

https://theses.gla.ac.uk/

Theses Digitisation:

https://www.gla.ac.uk/myglasgow/research/enlighten/theses/digitisation/

This is a digitised version of the original print thesis.

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk

STUDIES ON THE INHIBITION OF PLANT PATHOGENIC FUNGI BY CYCLOHEXYLAMINE AND SPERMIDINE ANALOGUES

Caroline Anne Mackintosh

ProQuest Number: 10391367

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10391367

Published by ProQuest LLC (2017). Copyright of the Dissertation is held by the Author.

All rights reserved.

This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346



STUDIES ON THE INHIBITION OF PLANT PATHOGENIC FUNGI BY CYCLOHEXYLAMINE AND SPERMIDINE ANALOGUES

A thesis submitted to the University of Glasgow

for the degree of Doctor of Philosophy in respect of research carried out in the Plant Science Department of the Scottish Agricultural College, Auchincruive.

by

Caroline Anne Mackintosh B.Sc. (Hons.) M.Sc.

October 1997

© Caroline A. Mackintosh 1997

|--|

1997 (N.S. &

Acknowledgements	i
Dedication	v
Publications	vi
List of Tables	vii
List of Figures	x
List of Plates	xi
Summary	xii
Chapter 1 INTRODUCTION	
1.1 Review of literature	2
1.1.1 Polyamines: an introduction	2
1.1.2 Functions of polyamines	2
1.1.2.1 Cell proliferation	4
1.1.2.2 Cell differentiation and organism growth	5
1.1.2.3 Regulation of stress	6
1.1.2.4 Regulation of senescence	7
1.1.2.5 DNA methylation	8
1.1.2.6 Protein synthesis and phosphorylation	9
1.1.3 Biosynthesis and regulation of intracellular polyamines	10
1.1.3.1 Polyamine biosynthesis	10
1.1.3.2 Regulation of polyamine biosynthetic enzymes	12
1.1.3.3 Other means of controlling intracellular polyamine levels	14
1.1.4 Inhibition of polyamine biosynthesis	17
1.1.4.1 Inhibitors of ODC	17
1.1.4.2 Inhibitors of AdoMetDC	18
1.1.4.3 Inhibitors of spermidme synthase	19
1.1.4.4 Inhibitors of sperintic synthese	20
1.1.4.5 Inhibitors of ADC	20
1.1.4.0 Polyalining analogues	20
1.1.5 Calleet research	25
1.1.6.1. Protozoal infections	20
1162 Bacteria	20
1163 Animal-infecting fungi	30
1164 Insecticides	32
1 1 7 Europal research	33
1.1.7.1 Inhibitors of ODC	33
1.1.7.2 Inhibitors of AdoMetDC	39
1.1.7.3 Inhibitors of spermidine synthase	41
1.1.7.4 Inhibitors of spermine synthase	41
1.1.7.5 Inhibitors of ADC	41
1.1.7.6 Polyamine analogues	43
1.1.8 Fungal research - further considerations	48
1.1.8.1 Effects of polyamine biosynthetic inhibitors on plants	48
1.1.8.2 Effects of polyamine biosynthetic inhibitors on mycorrhizal fungi	48
1.1.8.3 Control of plant pathogenic fungi - an overview	49
1.2 Aim of study	51

Chapter 2 EFFECT OF CYCLOHEXYLAMINE AND NORSPERMIDINE ON FUNGAL INFECTION OF HOST PLANTS

•

2.1 Introduction and Objectives	53
2.1.1 Broad bean rust	53
2.1.2 Broad bean chocolate spot	53
2.1.3 Barley powdery mildew	55
2.1.4 Apple powdery mildew	55
2.1.5 Potato late blight	56
2.2 Materials and Methods	57
2.2.1 Effect of CHA and norspermidine against rust on broad bean	57
2.2.1.1 Growth of plant material	57
2.2.1.2 Maintenance of pathogen	57
2.2.1.3 Inoculation of plant material and treatment with CHA and	
norspermidine	57
2.2.2 Effect of CHA and norspermidine against chocolate spot on broad bean	58
2.2.2.1 Growth of plant material	58
2.2.2.2 Maintenance of pathogen	58
2.2.2.3 Inoculation of plant material and treatment with CHA and	
norspermidine	58
2.2.3 Effect of CHA and norspermidine against powdery mildew on barley	58
2.2.3.1 Growth of plant material	58
2.2.3.2 Maintenance of pathogen	58
2.2.3.3 Inoculation of plant material and treatment with CHA and	
norspermidine	59
2.2.4 Effect of CHA and norspermidine against powdery mildew on apple	59
2.2.4.1 Growth of plant material	5 9
2.2.4.2 Maintenance of pathogen	59
2.2.4.3 Inoculation of plant material and treatment with CHA and	
norspermidine	59
2.2.5 Effect of CHA and norspermidine against late blight on potato	60
2.2.5.1 Growth of plant material	60
2.2.5.2 Maintenance of pathogen	60
2.2.5.3 Inoculation of plant material and treatment with CHA and	
norspermidine	60
2.3 Results	. 62
2.3.1 Effect of CHA and norsperimidine against rust on broad bean	62
2.3.2 Effect of CHA and norspermidine against chocolate spot on broad bean	62
2.3.3 Effect of CHA and norsperinidine against powdery mildew on barley	62
2.3.4 Effect of CHA and norspermidine against powdery mildew on apple	63
2.3.5 Effect of CHA and norspermidine against late blight on potato	63
2.4 Effects of CHA and norspermidine on germination and development of rust	
uredospores on broad bean leaves	68
2.4.1 Introduction and Objectives	68
2.4.2 Materials and Methods	68
2,4.3 Results	68
2.5 Discussion	72

Chapter 3 EFFECT OF CYCLOHEXYLAMINE AND SPERMIDINE ANALOGUES ON THE GROWTH OF PLANT PATHOGENIC FUNGI *IN VITRO*

.

3.1 Effect of CHA and spermidine analogues on mycelial growth of fungi on	
solid media	76
3.1.1 Introduction and Objectives	76
3.1.2 Materials and Methods	76
3.1.2.1 Maintenance of pathogens	76
3.1.2.2 Growth of fungi on inhibitor amended solid media	76
3.1.3 Results	78
3.1.3.1 Effects of CHA and spermidine analogues on mycelial growth of	
Pyrenophora avenae	78
3.1.3.2 Effects of CHA and norspermidine on mycelial growth of <i>Pyricularia</i>	
oryzae	79
3.2 Effect of CHA and spermidine analogues on growth of <i>Pyrenophora avenae</i>	
in liquid culture	85
3.2.1 Introduction and Objectives	85
3.2.2 Materials and Methods	85
3.2.2.1 Maintenance of pathogen	85
3.2.2.2 Growth of Pyrenophora avenae in inhibitor amended liquid media	85
3.2.3 Results	86
3.3 Effect of CHA and spermidine analogues on cell lengths of Pyrenophora	
avenae grown on solid media	91
3.3.1 Introduction and Objectives	91
3.3.2 Materials and Methods	9 1
3.3.3 Results	91
3.4 Discussion	93
Chapter 4 EFFECT OF CYCLOHEXYLAMINE AND SPERMIDINE ANALOGUES	
ON BIOSYNTHESIS AND CATABOLISM OF POLYAMINES IN PYRENOPHORA	
AVENAE	
4.1 Initial determination of the effects of CHA and spermidine analogues on	
polyamine biosynthesis in Pyrenophora avenae	96
4.1.1 Introduction and Objectives	96
4.1.2 Materials and Methods	96
4.1.2.1 Growth of fungal material	96
4.1.2.2 ODC assay	96
4.1.2.3 AdoMetDC assay	97
4.1.2.4 Incorporation of radio-labelled ornithine into polyamines	97
4.1.2.5 Analysis of free polymines in fungal fissue	98
4.1.3 Kesults	99
4.1.3.1 ODC and AdoMetDC activities	99
4.1.3.2 Incorporation of radio-labelled ornithing into polyamines	99 00
4.1.5.5 Analysis of tree polyamines in lungal tissue	99
4.2 Further determination of the effects of CHA and norspermidine on	100
orosynthesis and catadonsm of polyamines in <i>Pyrenophora avenae</i>	103
4.2.1 Introduction and Objectives	103
4.2.2 INTROTATS AND INCODES	104

4.2.2.1 Growth of fungal material	104
4.2.2.2 PAO assay	104
4.2.2.3 DAO assay	104
4.2.2.4 Analysis of acetylated polyamines in funga	1 tissue 105
4.2.3 Results	107
4.2.3.1 PAO and DAO activities	107
4.2.3.2 Analysis of acetylated polyamines in funga	I tissue 107
4.3 Determination of the effects of higher concentr	ations of CHA and
norspermidine on polyamine biosynthesis in Pyren	ophora avenae 109
4.3.1 Introduction and Objectives	109
4.3.2 Materials and Methods	109
4.3.3 Results	110
4.3.3.1 Effect of 4 days exposure to higher concent	trations of CHA and
norspermidine on growth of Pyrenophora avenae	110
4.3.3.2 Effect of 1 day exposure to higher concentre	ations of CHA and
norspermidine on growth of Pyrenophora avenae	110
4.3.3.3 Incorporation of radio-labelled ornithine in	to polyamines 110
4.3.3.4 Analysis of free polyamines in fungal tissu	e 110
4.4 Further determination of the effects of higher of	oncentrations of CHA and
norspermidine on biosynthesis and catabolism of p	olyamines in Pyrenophora
avenae	114
4.4.1 Introduction and Objectives	114
4.4.2 Materials and Methods	115
4.4.2.1 PAO activity	115
4.4.2.2 Analysis of acetylated polyamines in funga	1 tissue 115
4.4.2.3 Formation of acetylated polyamines	115
4.4.2.4 Determination of amine excretion	115
4.4.3 Results	116
4.4.3.1 PAO activity	116
4.4.3.2 Analysis of acetylated polyamines in tunga	I tissue 116
4.4.3.3 Formation of acceptated polyamines	110
4.4.3.4 Determination of annue excretion	110
4.5 Discussion	118
Chapter 5 REFECT OF NOVEL SPERMINNE ANAL	OCHES ON IN VITEO
GROWTH AND POLYAMINE BIOSYNTHESIS IN PYRI	ENOPHORA AVENAE
5.1 Introduction and Objectives	124
5.2 Effect of novel spermidine analogues on myce	lial growth of Pyrenophora
avenae on solid media	124
5.2.1 Materials and Methods	124
5.2.2 Results	124
5.3 Effect of novel spermidine analogues on growt	h of <i>Pvrenophora avenae</i> in
liquid culture	130
5.3.1 Materials and Methods	130
5.3.2 Results	<u>a.</u>
	Constant and a second second

Ż

NAME NAMES OF A

5.4 Effect of novel spermidine analogues on the incorporation of radio-labelled	
ornithine into polyamines in <i>Pyrenophora ayenae</i> grown in liquid culture	131
5.4.1 Materials and Methods	131
5.4.2 Results	131
5.5 Discussion	133
Chapter 6 FIELD STUDIES	
6.1 Introduction and Objectives	136
6.2 Materials and Methods	136
6.2.1 1995 Trial	136
6.2.2 1996 Trial	137
6.2.3 1996 Mini plots	138
6.3 Results	139
6.3.1 1995 Trial	139
6.3.2 1996 Trial	142
6.3.3 1996 Mini plots	145
6.4 Discussion	147
Chapter 7 ERRECT OF CYCLOHEXYLAMINE AND NORSPERMIDINE ON	
UPTAKE OF AMINO ACIDS. POLYAMINES AND SUCARS BY <i>PYRENOPHORA</i>	
AVENAE	
7 5 Tutus Augustian	1.61
7.1 1. Deview of literature	151
7.1.1 1 Transmentering on interduction	151
7.1.1.2 Transport mechanisms: an introduction	151
7.1.1.2 Transport mechanisms in tungi 7.1.1.2.1 Transport of environ	152
7.1.1.2.2. Transport of sugars	154
7.1.1.2.2. Transport of nalvernines	154
7.1.1.5 Transport of other substances	155
7.1.1.4 Transport of other substances	150
7.1.1.6 Factors affecting transport processes	159
7.1.2 Aim of studies	158
7.2 Materials and Methods	160
7.2.1 Experiment 1: Effect of CHA and norspermidine on the uptake of	1.00
compounds by <i>Pyrenophora avenae</i> during a four hour period	160
7.2.2 Experiment 2: Effect of prolonged exposure to CHA and norspermidine	4.60
on the uptake of compounds by <i>Pyrenophora avenae</i> during a four hour period	160
7.2.3 Experiment 3: Effect of higher concentrations of CHA and horsperimidine	1 1 1
on the uptake of compounds by <i>Pyrenophora avenae</i> during a 15 minute period	101
7.2.4 Experiment 4: Determination of uptake kinetics	101
7.3 Results	162
7.3.1 Experiment 1: Effect of CHA and norspermidine on the uptake of	
compounds by Pyrenophora avenae during a four hour period	162
7.3.2 Experiment 2: Effect of prolonged exposure to CHA and norspermidine	
on the uptake of compounds by Pyrenophora avenae during a four hour period	162
7.3.3 Experiment 3: Effect of higher concentrations of CHA and norspermidine	
on the uptake of compounds by <i>Pyrenophora avenae</i> during a 15 minute period	162
7.3.4 Experiment 4: Determination of uptake kinetics	165

.

.

1.1

. . . .

and the second sec

1. 1.60

. 4

7.4 Discussion

2.2

•

.

. . .

. . . .

Chapter 8 HYPUSINE BIOSYNTHESIS

8.1 Introduction	179
8.1.1 Review of literature	179
8.1.1.1 Hypusine: an introduction	179
8.1.1.2 Biosynthesis of hypusine	179
8.1.1.3 Biological significance of eIF-5A and hypusine	179
8.1.1.4 Polyamine biosynthesis and hypusine	181
8.1.1.5 Inhibition of hypusine biosynthetic enzymes	183
8.1.1.5.1 Deoxyhypusine synthase	183
8.1.1.5.2 Deoxyhypusine hydroxylase	185
8.1.2 Aim of studies	185
8.2 Effect of various deoxyhypusine synthase inhibitors on mycelial growth of	
Pyrenophora avenae on solid media	187
8.2.1 Introduction and Objectives	187
8.2.2 Materials and Methods	187
8.2.3 Results	188
8.3 Effect of various compounds on the growth of Pyrenophora avenae in	
liquid culture	193
8.3.1 Introduction and Objectives	193
8.3.2 Materials and Methods	193
8.3.3 Results	194
8.4 Effect of various compounds on fungal hypusine and deoxyhypusine	
content	196
8.4.1 Introduction and Objectives	196
8.4.2 Materials and Methods	197
8.4.3 Results	199
8.5 Discussion	203
Chapter 9 GENERAL DISCUSSION	207
Chapter 10 BIBLIOGRAPHY	214
Chapter 11 APPENDICES	257

ACKNOWLEDGEMENTS

This study would not have been at all possible without the support of both individuals and institutions. Additionally, many more people contributed to both the technical content of my work and the encapsulation of many fond memories of the past three years. I make no apologies for utilising this opportunity to thank them all.

First and foremost, I am indebted to Dr Dale Walters, my supervisor, who had the courage to present me with an opportunity to achieve a Ph.D., even though I had a mere inkling for research three years ago. Introducing me to polyamines, his enthusiasm for research was at once inspirational. Not only does that enthusiasm continue to inspire me, but his depth of knowledge and extraordinary wisdom, I aspire to. His guidance throughout the project was unfailing and his support phenomenal. He appeared to have the utmost faith in me, although in darkest times, I lacked confidence in myself. For that faith and his encouragement, I am most grateful. Not only did Dale provide an avenue for me to pursue a Ph.D., but his excellent supervisory role ensured that I have completed my studies with relative ease.

. V. S.

and the second of the second of the

I am also indebted to Dr George Marshall, Head of the Plant Science Department. He demonstrated continued interest throughout the duration of my studies, but above all, his faith in Dale and myself ensured the viability of this Ph.D. Recent reductions in funding have led to enormous restrictions on research resources. I am, therefore, extremely grateful for the opportunity to study for a Ph.D. I can only hope that what work I achieved can contribute to the research portfolio of the Department and aid its bid for further funding. On such a note, I am grateful to the Scottish Office Agriculture, Environment and Fisheries Department from whom SAC receive financial support.

I should also like to thank SAC and the University of Glasgow for the provision of a Dr. William Stewart Scholarship.

Parts of this study were carried out in collaboration with Professor David Robins of the Chemistry Department, Glasgow University. I would like to take this opportunity to thank him and his research team for the provision of the novel spermidine analogues. I am most grateful to Dr J. E. Folk of the National Institutes of Health, USA, who kindly provided me with samples of deoxyhypusine and inhibitors of deoxyhypusine synthase. His generosity and willingness to assist with my studies were most refreshing. Likewise, many thanks are due to Dr. T. Shiba of the Protein Research Foundation, Japan, for the provision of a sample of hypusine. He was also most generous and extremely helpful with regards to my hypusine work. I would like to acknowledge Marion Merrell Dow Pharmaceuticals for the generous

gift of DFMO and Professor A. Khomutov of the Russian Academy of Science, Moscow, for the provision of APA.

I am most grateful to Dr Neil Havis for his early contribution to this study when he effectively demonstrated many aspects of the practical work involved. His continued interest in my work and helpful comments throughout were welcomed. Above all, his friendship is valued enormously.

I should like to thank both Graeme Ligertwood and Donald Kiltic for their assistance regarding the field trials; planning, executing and above all, spraying. 'Ideal spraying days' and 'Ayrshire' are not synonymous and their comments as to the suitability or not of the weather for such an operation were highly regarded. Frustration gained a new meaning!

My thanks go to our resident Plant Pathologist who assumes the mantle of statistician now and again, Dr. Neil McRoberts. His helpful comments regarding my microscopy work were also gratefully received. Above all, I am in indescribable awe of anybody who appears to possess both a love for statistics and microscopy!

Thanks must also go to Kenny MacIsaac of the Biochemistry and Nutrition Department who provided his expertise with the hypusine related HPLC analysis. His was only too willing to assist me and although the analysis was unsuccessful, it was not from a lack of trying.

The photographic genius of Fiona Sutherland must be acknowledged. Her talent and flair resulted in some great visual records of my work.

Grateful thanks must go to Dr Ruairidh Bain and Lynda Raynor for their advice regarding the potato blight experiments. If only growers had as much success keeping plants disease free as I did!

The Hannah Research Institute must be acknowledged for the use of their fluorimeter, required for free polyamine analysis.

I would like to thank most sincerely Carol Steel and Martha Richmond for the use of their computer facilities out of working hours. Only such a provision allowed me to combine my Ph.D. with the wardening jobs of Oswald and Wilson Hall.

My time at Auchineruive would not have been nearly as enjoyable had it not been for those members of the Plant Science Department, past and present, and indeed, from all College

ii

Departments, who welcomed me in the beginning, showed genuine interest in my work, provided me with morale boosting encouragement and above all, have been my friends. A list I am not going to contemplate. There are, however, some who are worthy of a mention and will remain in my memories for years to come.

Which brings me to the unluckiest lottery syndicate ever. Together we've shared the thrills of winning.....a tenner, and the frustration of the all too frequent realisation that we were to be back at work on Monday morning and not meeting each other at Glasgow airport. The afore mentioned Neil Havis, who along with Neal Evans and Aileen Timmons, have moved onto pastures new. Their past support and continued friendship is valued. Fiona Sutherland - my caffeine drinking, cigarette puffing fellow Ph.D. student who, along with Laura Heggie, provided me with much empathy regarding the Ph.D. process, helping me enormously along the way. Tracy Cowley and Arlene McPherson, their comradeship and support in the lab made light of those long assay-filled days. Arlene is, of course, Michael Jackson's greatest fan. Did you know that plastic complexion is a 'skin condition'? Audrey Meikle, the only person I know who could give Imelda Marcos a run for her money in the shoe stakes and still have time for her other fetishes, fruit and cheese to name but two. Above all, her advice and friendship along the way is treasured. Richard Hetherington, to whom I am most grateful for giving up his time to proof read this thesis. His comments were invaluable. More than anything he kept the humour flowing in the lab, albeit at his own expense. Those ageism jokes will catch up with us all at some point but at least we've learned a thing or two about boxes - and onions! And so to the newest recruits, Celtic's number one fan, Billy Sinclair and Margaret Greenwood, with whom I have discovered a common bond in these last few weeks of writing up - you know who! Which leaves only Jacqueline Gemmell, who deserves the highest accolade. A true friend throughout, through laughter and tears, for whom I have the utmost admiration. Not only does she combine a hectic home life with a far reaching work schedule but she has endless time for all. Her friendship is treasured.

I should also like to thank Lynda Raynor, who along with Jacqueline, was my swimming partner and 'counsellor' - much to Graeme's amusement! And Dr. Mark McQuilken, for his continued interest in my work and helpful comments throughout. I doubt I'll meet anybody who can moan like him again! My eternal gratitude also goes to Carol Steel, a tremendous friend who kept me going through the darkest days of my second year. I'll never give out washing powder to strangers again!

Finally, but by no means least, I am, as always, indebted to my family. In particular, my parents, for their endless and unconditional support - both emotional and financial. And to my little brother, perhaps you will now realise that this thesis is not merely a wee pamphlet!

The 1996 World Food Summitt, organised by The Food and Agriculture Organisation of the United Nations reported the following facts:

800 million people in the world are undernourished 200 million of those are children under the age of five By 2030, the world population will have risen by 3 billion

With rising populations and reductions in availability of agricultural land, increased productivity is paramount.

The problem of hunger is, however, both a problem of production and distribution.

This thesis is dedicated to those farmers and scientists who strive to provide food for all, and to the politicians who should.

v

PUBLICATIONS

Parts of the work presented in this thesis appear in the following publications:

MACKINTOSII, C. A. & WALTERS, D. R. (1996). Spermidine analogues as novel fungicides. Proceedings Crop Protection in Northern Britain 1996, 195-200.

MACKINTOSH, C. A. & WALTERS, D. R. (1996). Fungicidal activity of two spermidine analogues. Journal of Phytopathology 144, 43-45.

WALTERS, D. R. & MACKINTOSH, C. A. (1997). Control of plant disease by perturbation of fungal polyamine metabolism. Physiologia Plantarum 100, 689-695.

MACKINTOSH, C. A., SLATER, L. A., McCLINTOCK, C. A., WALTERS, D. R., HAVIS, N. D. & ROBINS, D. J. (1997). Synthesis and antifungal activity of two novel spermidine analogues. FEMS Microbiology Letters 148, 21-25.

MACKINTOSH, C. A., SLATER, L. A., WALTERS, D. R. & ROBINS, D. J. (1997). Synthesis of six novel spermidine analogues and effects on growth of the fungal plant pathogen <u>Pyrenophora avenae</u>. Amino Acids, (in press).

MACKINTOSH, C. A. & WALTERS, D. R. (1997). Growth and polyamine metabolism in <u>Pyrenophora avenae</u> exposed to cyclohexylamine and norspermidine. Amino Acids, (in press).

vi

LIST OF TABLES

1. Effect of CHA and norspermidine against powdery mildew on apple	67
2. Effect of CIIA and norspermidine against late blight on potato	67
3. Effects of 2.0 mM CHA and 2.0 mM norspermidine on bean rust germling development on leaf surfaces	70
4. Effect of N^1 - and N^8 -acetylspermidine on mycelial growth of <i>Pyrenophora</i> avenae	80
5. Effect of varying concentrations of CHA and norspermidine on growth of <i>Pyrenophora avenae</i> in liquid culture for 4 days	81
6. Effect of 2.0 mM N ¹ - and N ⁸ -acetylspermidine on growth of <i>Pyrenophora</i> avenae in liquid culture for 4 days	81
7. Effect of 1.0 mM CHA and 0.01 mM norspermidine on growth of <i>Pyrenophora</i> avenae in liquid culture for 2, 3 or 4 days	88
8. Effect of 0.5 mM N ¹ - and N ⁸ -acetylspermidine on growth of <i>Pyrenophora</i> avenae in liquid culture for 2, 3 or 4 days	88
9. Effect of varying concentrations of CHA and spermidine analogues on cell lengths in <i>Pyrenophora avenae</i> grown on solid media for 8 days	92
10. Effect of CHA and spermidine analogues on ODC activity in <i>Pyrenophora avenae</i> grown for 2, 3 or 4 days	100
11. Effect of CHA and spermidine analogues on AdoMetDC activity in <i>Pyrenophora avenae</i> grown for 2, 3 or 4 days	100
12. Effect of CHA and spermidiue analogues on the incorporation of radio-labelled omithine into polyamines in <i>Pyrenophora avenae</i> grown for 2, 3 or 4 days	101
13. Effect of CHA and spermidine analogues on free polyamine levels in <i>Pyrenophora avenae</i> grown for 2, 3 or 4 days	102
14. Rf values of various compounds, developed on the plates using chloroform : isopropyl alcohol (95:5) (v/v)	106
15. Effect of 1.0 mM CHA on PAO and DAO activities in <i>Pyrenophora avenae</i> grown for 2, 3 or 4 days	108
16. Effect of 1.0 mM CHA and 0.01 mM norspermidine on acetylated polyamine levels in <i>Pyrenophora avenae</i> grown for 2, 3 or 4 days	108
17. Effect of 1 day exposure to higher concentrations of CHA and norspermidine on growth of <i>Pyrenophora avenae</i>	112

18. Effect of 1 or 4 day exposure to higher concentrations of CHA and norspermidine on the incorporation of radio-labelled ornithine into polyamines in <i>Pyrenophora avenae</i>	112
19. Effect of 1 day exposure to higher concentrations of CHA and norspermidine on free polyamine levels in <i>Pyrenophora avenae</i>	113
20. Effect of 4 day exposure to higher concentrations of CHA and norspermidine on free polyaminc levels in <i>Pyrenophora avenae</i>	113
21. Effect of 1 or 4 day exposure to 2.75 mM CHA on PAO activity in <i>Pyrenophora avenae</i>	117
22. Effect of 1 or 4 day exposure to higher concentrations of CHA and norspermidine on acetylated polyamine levels in <i>Pyrenophora avenae</i>	117
23. Effect of 1 or 4 day exposure to 2.75 mM CHA on polyamine excretion from <i>Pyrenophora avenae</i>	117
24. Structures of novel spermidine analogues	125
25. Effect of novel spermidine analogues on growth of <i>Pyrenophora avenae</i> in liquid culture for 4 days	132
26. Effect of novel spermidine analogues on the incorporation of radio-labelled ornithine into polyamines in <i>Pyrenophora avenae</i> grown for 4 days	13 2
27. Effect of CHA, norspermidine, E-TED and a flutriafol and carbendazim mixture on powdery mildew infection of spring barley: 1995 Trial	140
28. Effect of CHA, norspermidine, E-TED and a flutriafol and carbendazim mixture on plant height, plant dry weight and grain dry weight of spring barley: 1995 Trial	141
29. Effect of CHA, norspermidine, a flutriafol and carbendazim mixture, and tebuconazole on powdery mildew infection of spring barley: 1996 Trial	143
30. Effect of CHA, norspermidine, a flutriafol and carbendazim mixture, and tebuconazole on plant height, plant dry weight and grain dry weight of spring barley: 1996 Trial	144
31. Effect of CHA, norspermidine and DFMO on powdery mildew infection of spring barley: 1996 Mini plots	146
32. Uptake of methionine, ornithine, polyamines and sugars by <i>Pyrenophora avenae</i> exposed to 1.0 mM CHA or 0.01 mM norspermidine for 2 days	163
33. Uptake of methionine, ornithine, polyamines and sugars by <i>Pyrenophora avenae</i> exposed to 1.0 mM CHA or 0.01 mM norspermidine for 4 hours	163
34. Uptake of polyamines by <i>Pyrenophora avenae</i> exposed to 1.0 mM CHA or 0.01 mM norspermidine for 4 days	164

4

Ę

ς,

ĝ

1

, å

ź

÷

35. Uptake of methionine and spermidine by <i>Pyrenophora avenae</i> exposed to 2.75 mM CHA or 0.2 mM norspermidine for 4 days or 15 minutes	164
36. Uptake of methionine by <i>Pyrenophora avenae</i> exposed to 2.75 mM CHA or 0.2 mM norspermidine for 4 days	167
37. Structures of deoxyhypusine synthase inhibitors	189
38. Effect of deoxyhypusine synthase inhibitors on mycelial growth of <i>Pyrenophora avenae</i>	190
39. Effect of various compounds on the growth of <i>Pyrenophora avenae</i> in liquid culture for 4 days	195

3.a.

 \dot{z}_i

LIST OF FIGURES

• • •

1. Structures of some major polyamines	3
2. Biosynthetic pathways of the major plant polyamines	11
3. Polyamine interconversion pathways	16
4. Structures of spermidine, CHA and norspermidine	54
5. Effect of cyclohexylamine and norspermidine against rust on broad bean	64
6. Effect of cyclohexylamine and norspermidine against chocolate spot on broad bean	65
7. Effect of cyclohexylamine and norspermidine against powdery mildew on barley	66
8. Structures of spermidine, N^1 -acetylspermidine and N^8 -acetylspermidine	77
9. Effect of cyclohexylamine and norspermidine on mycelial growth of <i>Pyrenophora</i> avenae	81
10. Effect of cyclohexylamine and norspermidine on mycelial growth of <i>Pyricularia</i> oryzae	82
11. Effect of C59 and C73 on mycelial growth of Pyrenophora avenae	126
12. Effect of LAS 2/11 and LAS 2/12 on mycelial growth of Pyrenophora avenae	127
13. Effect of LAS 2/26 and LAS 2/27 on mycelial growth of Pyrenophora avenae	128
14. Effect of LAS 2/28 and LAS 3/8 on mycelial growth of Pyrenophora avenae	129
15. Eadie-Hofstee plot: Vo against Vo/[S] (Control)	168
16. Eadie-Hofstee plot: Vo against Vo/[S] (CHA treatment - expanded scale)	169
17. Eadie-Hofstee plot: Vo against Vo/[S] (CHA treatment - original scale)	170
 Eadie-Hofstee plot: Vo against Vo/[S] (norspermidine treatment - expanded scale) 	171
19. Eadie-Hofstee plot: Vo against Vo/[S] (norspermidine treatment - original scale)	172
20. Biosynthesis of hypusine	180
21. HPLC Chromatograms of authentic hypusine and extract from <i>Pyrenophora</i> avenae	202

LIST OF PLATES

83). 197

1. Effect of 2.0 mM CHA and 2.0 mM norspermidine on mycelial growth of	
Pyrenophora avenae, grown on solid media for eight days	83
2. Effect of 2.0 mM CHA and 2.0 mM norspermidine on mycelial growth of	
Pyricularia oryzae, grown on solid media for eight days	84
3. Effect of 1.0 mM CHA on growth of Pyrenophora avenae in liquid culture,	
grown for four days	89
4. Effect of 0.01 mM norspermidine on growth of Pyrenophora avenae in liquid	
culture, grown for four days	90
5. Effect of 1.0 mM concentrations of various deoxyhypusine synthase inhibitors	
on mycelial growth of Pyrenophora avenae, grown on solid media for six days.	191
6. Effect of varying concentrations of various deoxyhypusine synthase inhibitors	
on mycelial growth of Pyrenophora avenae, grown on solid media for six days.	192

SUMMARY

The potential inhibitory activity of several compounds against plant pathogenic fungal growth was studied. Prevention of growth was hypothesised to occur by the inhibition of polyamine biosynthesis and/or an effect on cellular functions associated with spermidine, the predominant polyamine in fungi.

The spermidine synthese inhibitor, cyclohexylamine (CHA), and the spermidine analogue, norspermidine, demonstrated protective, eradicant and systemic activity against a range of plant pathogenic infections of host plants: broad bean rust and chocolate spot, barley and apple powdery mildew, and potato late blight. For example, both 2.0 mM CHA and 2.0 mM norspermidine, applied pre inoculation, reduced powdery mildew infection of barley by 88 %. CHA and norspermidine were equally effective although apple powdery mildew and potato late blight infections were slightly more responsive towards norspermidine than CHA treatment. Against bean rust and barley powdery mildew, little increased efficacy was afforded by increasing inhibitor concentrations from 0.5 mM to 2.0 mM. Higher concentrations were, however, more effective against the other pathogenic infections examined. Pre inoculation treatments tended to be more effective than post inoculation applications of the inhibitors, although the reason for this is unclear as microscopic examination of bean rust spores and developing germlings on bean leaves found little difference between the treated leaves and the untreated control, suggesting the compounds acted on later stages of fungal development.

CHA and norspermidine were also effective against powdery mildew infection of spring barley in the field. The compounds were as effective as a commercially available fungicide in reducing disease. Neither the commercial fungicide nor the inhibitors caused any increase in yield over the untreated control, although this may have been the result of additional environmental pressures.

The compounds controlled mycelial growth of two plant pathogens, *Pyrenophora avenae* and *Pyricularia oryzae*, on artificial media and reduced fresh weight of *P. avenae* grown in liquid culture. Exposure of *P. avenae* for 4 days to 3.0 mM CHA reduced fungal growth by 93 % while 0.5 mM norspermidine completely inhibited growth. In vitro growth of the fungus was thus more responsive to norspermidine than CHA. Additional commercially available spermidine analogues, N¹- and N⁸-acetylspermidine, had limited efficacy against mycelial growth of *P. avenae in vitro*, although antifungal activity was apparent when the fungus was grown in liquid culture in the presence of the analogues. Varying antifungal

activities were found for eight novel spermidine analogues although four compounds did indeed successfully inhibit growth of *P. avenae in vitro*.

Neither 1.0 mM CHA, 0.01 mM norspermidine nor 0.5 mM N¹- and N⁸-acetylspermidine affected ODC, AdoMetDC or spermidine synthase activities in *P. avenae*. Free polyamine levels were unaffected, with the exception of treatment with CHA, which increased putrescine content. For example, exposure of fungus to 1.0 mM CHA for 4 days increased putrescine by 75 %. Higher concentrations of the inhibitors, 2.75 mM CHA and 0.2 mM norspermidine, caused similar effects on polyamine biosynthesis in *P. avenae*. Thus, even the most dramatic effects on growth were unlikely to have been caused by aberrant polyamine biosynthesis.

The cause of the CHA induced increase in putrescine concentration remains unclear. Increased catabolism of higher polyamines was unlikely to be responsible since PAO activity was unaffected. Reduced putrescine catabolism was similarly not responsible, since DAO activity was unaltered. Breakdown of conjugated N-acetylputrescine was also not likely to be the cause of the increased putrescine concentration. However the increase was caused, it is unlikely to wholly account for the antifungal activity of CHA.

ų,

CHA and norspermidine did cause effects on uptake of amino acids, polyamines and sugars into P. avenae. Indeed, exposure of fungus to the inhibitors for 4 days completely destroyed the methionine uptake system. Disruption of membrane function and permeability may have been responsible for the dramatic effects on the kinetics of methionine uptake by P. avenae caused by this pre exposure of fungus to CHA or norspermidine.

Known inhibitors of deoxyhypusine synthase successfully inhibited mycelial growth of *P. avenae in vitro*. Attempts to determine fungal hypusine content were, however, unsuccessful and it remains to be seen whether CHA and spermidine analogues exert their antifungal and fungicidal modes of action via perturbation of hypusine synthesis.

The study therefore showed that CHA and spermidine analogues possessed both antifungal and fungicidal activity against a range of plant pathogens. Although differences in *in vivo* and *in vitro* studies exist, perturbation of polyamine biosynthesis was unlikely to account for such activity of the compounds. The compounds were most likely exerting their effects via some other cellular function related to polyamines.

Chapter 1

Introduction

1.1 REVIEW OF LITERATURE 1.1.1 POLYAMINES: AN INTRODUCTION

Antoni van Leeuwenhoek is credited with the discovery of polyamines when, in 1678, he observed crystals whilst studying human semen. It was not until 1878, after many unsuccessful attempts, that these crystals were identified as the phosphate salt of an organic base and ten years later, named as spermine. An additional 38 years then passed before the chemical structure of spermine was finally elucidated, simultaneously in England and Germany (Smith 1982).

Since then, a number of polyamines have been described. The term 'polyamine' is, however, a misnomer as the compounds are in actual fact aliphatic diamines, triamines and tetraamines. Such organic compounds have low molecular weights, are soluble in water and are cationic at physiological pH, (around neutrality). Some, for example, putrescine (1, 4-diaminobutane), spermidine (4 azaoctane-1, 8-diamine) and spermine (4, 9 diazadodecane-1, 12-diamine) are widely distributed (Stevens & Winther 1979). Indeed, no eukaryotic or prokaryotic cell has been found in which polyamines are not present. Putrescine and spermidine are most common with spermine being less well distributed, some bacteria and filamentous fungi being devoid of the compound (Stevens & Winther 1979). The diamine cadaverine has been found in organisms such as plants (Smith 1985), mammalian cells (Pegg & McGill 1979) and *Neurospora crassa* (Paulus *et al.* 1982), but its occurrence in nature is generally restricted (Figure 1).

1.1.2 FUNCTIONS OF POLYAMINES

Polyamines have been implicated in many physiological functions, although their exact role in many of these is still not well understood (Tabor & Tabor 1985). Nevertheless, some of these functions with which polyamines are associated shall be reviewed including cellular proliferation and differentiation, and organism growth and development. Much work has concluded that polyamines are involved in regulating stress and senescence of plants and two additional specific cellular functions relating to polyamines, DNA methylation and protein phosphorylation, shall also be reviewed. Some of the work in this area has utilised polyamine biosynthesis inhibitors and such compounds will be discussed in later parts of this review. Other work used mutants of organisms, incapable of polyamine biosynthesis, to prove the essentiality of polyamines for growth (Tabor & Tabor 1985). Many studies have, therefore, concluded that polyamines are vital for a range of different functions associated with cellular growth and development.



÷

Figure 1. Structures of some major polyamines

1.1.2.1 Cell proliferation

Polyamines were initially associated with cellular proliferation in 1949 when Herbst and Snell observed that a factor in orange juice, later shown to be putrescine, was essential for growth of the bacterium *Hemophilus parainfluenzae* (Smith 1982). Since then, polyamines have been shown to stimulate cell division and growth in a number of bacteria, fungi, the beetle *Oryzaephilus*, tissue cultures of Chinese hamsters and tuber explants of Jerusalem artichoke (Tabor & Tabor 1964; Smith 1971). Also, an inhibitor of polyamine biosynthesis caused cessation of cell division in the tomato (Egea-Cortines *et al.* 1993), while Pfosser *et al.* (1992) demonstrated that polyamines decreased in cell suspension cultures of the alfalfa *Medicago varia* when cell division arrested.

The correlation between polyamines and cell division is thought to be largely due to their association with nucleic acids. Polyamines are the most cationic small molecules of the cell and therefore, bind readily to polyanionic macromolecules like DNA and RNA (Igarashi *et al.* 1982). Since the amines have a distributed charge, unlike multivalent ions, for example Mg^{2+} , they are able to interact more flexibly with the acidic phosphate groups of nucleic acids (Marton & Morris 1987). Thus, polyamines act by stabilising DNA and RNA molecules and indeed, several possible configurations for complexes formed between DNA and polyamines have been suggested through the use of X-ray crystallography (Smith 1982). The structural re-inforcement of DNA by polyamines is illustrated well by exposing the DNA double helix to increasing temperatures. High temperatures lead to an unwinding of the molecule to give a 'random coil form'. However, a considerably higher temperature is required to achieve the same degree of disorganisation when polyamines are present (Smith 1982).

Polyamines stimulate the DNA replicating enzyme although relatively little is known as to how this is achieved (Stevens 1970). Many reports provide evidence of the association between polyamines and DNA synthesis. Rapid accumulation of polyamines and in particular, putrescine, was a pre requisite for the enhancement of DNA synthesis in rat liver, induced by food intake (Kameji *et al.* 1979). Putrescine is known to increase in relation to the rate of DNA synthesis in the woody plant *Catharanthus* and other plant tissues (Serafini-Fracassini *et al.* 1980; Smith 1985; Minocha *et al.* 1991). Minocha *et al.* (1991) also observed that putrescine reversed inhibition of DNA synthesis caused by an inhibitor of polyamine biosynthesis in *C. roseus*, providing evidence of the absolute requirement for the polyamine in DNA synthesis.

Polyamines are also associated with RNA synthesis. Since the double stranded DNA separates for RNA synthesis, polyamines are suggested to enhance the rate of restoration of

the double helix, thus stimulating synthesis of RNA (Smith 1982). Polyamines may also stimulate RNA synthesis by facilitating the removal of transcripted RNA from the DNA single strand (Smith 1982). Gallardo *et al.* (1992) reported that the accumulation of polyamines during germination of chick-pea, *Cicer arietinum*, seeds coincided with DNA and RNA synthesis.

1.1.2.2 Cell differentiation and organism growth

Somatic embryogenesis in carrot cell cultures was accompanied by an increase in polyamines (Mengoli *et al.* 1989), while polyamine biosynthesis inhibitors reduced somatic embryo formation of the ligneous plant *Hevea brasiliensis* (El Hadrami & D'Auzac 1992). Polyamines were implicated in embryogenesis of celery where increased amine levels were necessary for embryo growth and plantlet organisation (Danin *et al.* 1993). Altamura *et al.* (1991) observed that rhizogenesis in tobacco explants was reduced by various inhibitors of polyamine biosynthesis. Polyamines were also associated with organogenesis in tissue culture systems derived from melon cotyledons (Leshem *et al.* 1991), while Cohen *et al.* (1982) demonstrated that the use of a polyamine biosynthesis inhibitor prevented tomato ovary development.

2

Polyamines are also implicated in higher developmental processes in plants. For example, pollen germination and pollen tube length in lily flowers were reduced following exposure to polyamine biosynthesis inhibitors (Rajam 1988). Harkess *et al.* (1992) observed that a correlation existed between polyamines and flowering in the long day plant *Rudbeckia hirta*. Polyamines were also reported to be involved in fruit growth during the initial stages of fruit development in apple trees (Biasi *et al.* 1991), while Matejko & Dahlhelm (1991) found that polyamine levels related to the physiological state of onion bulbs during dormancy.

Polyamines have been implicated in the development of fungi. Polyamine biosynthesis increased during germination of *Blastocladiella emersonii* (Mennucci *et al.* 1975), *Aspergillus nidulans* (Stevens *et al.* 1976) and morphogenesis of *Mucor racemosus* (Inderlied *et al.* 1980). In *Sclerotium rolfsii*, mycelial growth and sclerotium formation were associated with increasing levels of polyamines (Shapira *et al.* 1989). Polyamines were implicated in the dimorphic transition of *Mucor* species when biosynthesis increased during the yeast to mycelium conversion of *M. rouxii* and *M. bacilliformis* (Calvo-Mendez *et al.* 1987). Further evidence of the essentiality of polyamines in this process was provided when a mutant of *M. bacilliformis*, unable to grow mycelially, possessed very low levels of a polyamine biosynthetic enzyme (Ruiz-Herrera *et al.* 1983).

Exactly how polyamines affect differentiation and growth of various organisms is unclear. They are known to be associated with cell division and Evans & Malmberg (1989) suggested that this influence may extend to affecting patterns of cell division and thus, organism form. Other workers, however, have suggested that polyamines act as types of growth regulator or hormonal secondary messenger (Galston 1983; Davies 1987; Galston & Kaur-Sawhney 1987). Ruiz-Herrera (1994) suggested that polyamine levels may affect fungal differentiation via DNA methylation and this subject will be discussed in more detail shortly.

1.1.2.3 Regulation of stress

A great deal of work has implicated polyamines in the regulation of plant stress. Richards & Coleman (1952) first reported elevated levels of putrescine in potassium deficient barley. Klein *et al.* (1979) and Adams *et al.* (1990, 1992) reported similar results for peas and grapes while increased putrescine was observed under conditions of magnesium deficiency in barley (Smith 1973). Likewise, low pH in barley (Smith & Sinclair 1967), excess ammonium in peas (Klein *et al.* 1979) and atmospheric pollutants such as sulfur dioxide in peas (Priebe *et al.* 1978) caused accumulation of putrescine. Mung bean plants under salt stress (Friedman *et al.* 1989) and barley plants under water stress (Turner & Stewart 1986) also demonstrated an accumulation of putrescine. Wheat plants exposed to osmotic stress accumulated polyamines (Foster & Walters 1991), while elevated levels of polyamines were found in barley leaves infected with powdery mildew (Walters *et al.* 1985) and in wheat leaves inoculated with black stem rust (Foster & Walters 1992b).

The exact mechanism by which polyamines regulate each stress is unknown. However, there are many protective functions that are associated with polyamines which may assist plants to withstand stress. Srivastava & Smith (1982) suggested that polyamines interact with anionic groups of cell membranes, preventing leakage and causing stabilisation under conditions of stress. Evidence of the stabilising effect of polyamines on membranes was obtained with apple cells grown in culture, storage tissue from swede and spinach leaf discs. Induced ion efflux from these tissues could be reversed by the addition of spermine (Srivastava & Smith 1982). Naik & Srivastava (1978) suggested that polyamines may partially replace calcium and maintain membrane integrity by binding to phospholipid components of the numbrane. Marton & Morris (1987) similarly suggested that binding of polyamines to phospholipid groups of membranes would alter membrane stability and permeability. It should be noted that DiTomaso *et al.* (1989) suggested that increased levels of putrescine may have caused membrane damage in maize roots by producing hydrogen peroxide and free radicals via the action of diamine oxidase. However, Dumbroff (1991) suggested that polyamines possessed the potential to act as free radical scavengers.

Smith & Sinclair (1967) suggested that increased cytoplasmic putrescine could help maintain cytoplasmic pH at a constant value, and Altman & Levin (1993) did indeed find that polyamines acted as a metabolic buffer, maintaining cellular pH under conditions of ammonium accumulation in tobacco cell suspension cultures. The accumulation of putrescine has also been associated with enhanced protein synthesis (Kaur-Sawhney *et al.* 1980) and increased synthesis of protective alkaloids (Hartmann *et al.* 1988).

Such cases of polyamine association with plant stress have been observed in other species with Agrawal *et al.* (1992) suggesting that polyamines were involved in de-toxifying mercury from the green alga *Chlorogonium elongatum*.

1.1.2.4 Regulation of senescence

Polyamines are associated with the regulation of plant senescence, although the mechanisms by which this is achieved are not clear. Exogenous polyamines are known to retard plant senescence, as demonstrated by the application of putrescine or spermine to carnation buds (Upfold & van Staden 1991). Ethylene is known to induce senescence and since biosynthesis of polyamines and ethylene require a common pre cursor (Sadenosylmethionine), elevation of polyamines should reduce ethylene production and thus retard senescence (Evans & Malmberg 1989). Conversely, depletion of polyamines should elevate ethylene production and trigger senescence (Evans & Malmberg 1989).

Indeed, polyamines are known to inhibit ethylene formation in a number of plant tissues including apple fruits, bean and tobacco leaf explants (Apelbaum *et al.* 1981). However, elevated levels of polyamines are not always associated with reduced ethylene biosynthesis. Chen *et al.* (1991) showed that polyamines stimulated the production of ethylene in detached rice leaves, while Park & Lee (1994) reported that an accumulation of polyamines was ethylene induced in suspension cultures of tobacco cells. Lee & Chu (1992) showed that ethylene accumulation led to increased putrescine biosynthesis which was associated with the elongation of rice coleoptiles. Such findings suggest that reduction of ethylene synthesis may not be wholly responsible for polyamine associated retardation of plant senescence.

Company of the start of the

Botha & Whitehead (1992) and Chien *et al.* (1991) both dismissed ethylene biosynthesis inhibition as the main mechanism for regulation of senescence in *Petunia hybrida* and rice plants while putrescine application extended the effective pollination period in 'Comice' pears but was not associated with altered ethylene levels (Crisosto *et al.* 1992).

The prevention of softening of apples and tomatoes by polyamines was suggested to be caused by polyamines either rigidifying cell walls or altering the activity of degradation enzymes of such walls (Kramer *et al.* 1991; Law *et al.* 1991). Smith (1985) proposed that polyamines acted as antisenescence agents by binding to nucleic acids or membranes. Indeed, Cohen *et al.* (1979) implicated a polyamine induced stabilisation of chloroplast thylakoid membranes in the retardation of chlorophyll degradation in senescing barley tissue.

The accumulation of polyamines in diseased plants has been implicated in the formation of 'green islands' and a subsequent retardation of senescence in these regions of the infected leaf, although the exact mechanism for the cause of this effect is unknown (Walters & Wylie 1986; Coghlan & Walters 1990).

1.1.2.5 DNA methylation

Methylation of DNA bases, a reaction which occurs in both prokaryotes and eukaryotes, is known to be important in gene expression at different developmental stages of many organisms (Razin *et al.* 1984). Generally, methylation is indicative of gene inactivation (Jupe *et al.* 1986). Cytosine residues, the most common DNA base to be methylated in cukaryotes, are methylated in reactions involving methyl groups from S-adenosylmethionine, catalysed by methyl transferase enzymes (Magill & Magill 1989).

Cano *et al.* (1988) implicated an association between polyamines, DNA methylation and growth of dimorphic *Mucor rouxii* with different levels of methylation leading to either mycelial growth or yeast-like growth. Although polyamines were implicated in regulating the amount of methylation, the studies were unable to determine how exactly the polyamines acted (Cano *et al.* 1988).

Ruiz-Herrera (1994) also suggested that the association between polyamines and differentiation in Mucorales may be connected to DNA methylation and went on to suggest that increased polyamine levels may inhibit DNA methylation and thus allow expression of specific genes necessary for differentiation (Ruiz-Herrera 1994).

Demethylation has been suggested to occur not by direct removal of the methyl group from DNA, but by inhibition of cytosine methylation during DNA replication (Ruiz-Herrera 1994). Indeed, Ruiz-Herrera *et al.* (1995) showed how activities of cytosine DNA methyltransferases were sensitive to polyamine concentrations, in particular, high concentrations of spermidine reduced the number of methyl groups incorporated into *Escherichia coli* DNA. Interestingly, Cano and co-workers (1988) and Cano-Canchola *et*

8

. چېرو al. (1992) found that an inhibitor of polyamine biosynthesis prevented DNA methylation although the exact mechanism by which the inhibitor operated is unknown.

1.1.2.6 Protein synthesis and phosphorylation

The cationic nature of polyamines allows them to interact with ribosomes, the site of protein synthesis. This association has been implicated in the ability of polyamines to speed up the transcription-translation process of protein manufacture (Stevens 1970). Spermidine has been reported to induce the initiation of protein synthesis in isolated chloroplasts of the unicellular green alga *Chlamydomonas reinhardtii* (Blättler *et al.* 1992).

As well as synthesis of proteins, polyamines have been associated with the regulation of protein activity. Protein phosphorylation is the most common covalent modification of proteins and is used as a means of activating or deactivating enzymes. For the various phosphorylase kinase enzymes to phosphorylate proteins, the kinases must also be phosphorylated and this is achieved by various protein kinases. Conversely, protein dephosphorylation is achieved through the action of phosphorylase phosphatases and such enzymes are phosphorylated by protein phosphatases (Palmer 1991).

The activities of several important phosphorylation enzymes are modulated by abberant polyamine levels including proteins from pea plumules (Datta *et al.* 1987), maize coleoptiles (Velutthambi & Poovaiah 1984) and wheat (Polya & Micucci 1985).

Varying effects are, however, reported. Enzymes from maize coleoptiles (Velutthambi & Poovaiah 1984) and pea nuclei (Datta *et al.* 1986) were stimulated by polyamines while polyamines inhibited activity of enzymes from wheat embryos (Polya & Micucci 1985), wheat anthers (Bothma & Dubery 1991) and soybean hypocotyls (Lin & Key 1980). Nevertheless, polyamines are clearly associated with protein phosphorylation although their exact role is unclear. Polyamines may exert their effects on organism growth and development through regulation of various enzyme activities.

1.1.3 BIOSYNTHESIS AND REGULATION OF INTRACELLULAR POLYAMINES

1.1.3.1 Polyamine biosynthesis

The initial step of polyamine biosynthesis in mammalian cells and protozoa is the decarboxylation of ornithine to form putrescine, catalysed by the enzyme ornithine decarboxylase (ODC: EC 4. 1. 1. 17) (Figure 2). In plants and bacteria, putrescine can also be formed from a series of intermediates following decarboxylation of arginine by arginine decarboxylase (ADC: EC 4. 1. 1. 19) (Walters 1995).

Although putrescine formation in most fungi occurs solely through the operation of ODC, there is some evidence for the operation of ADC in certain fungi. Examples include the phytopathogenic fungi, *Ophiostoma ulmi* [Khan & Minocha 1989a (as *Certatocystis ulmi*); Biondi *et al.* 1993], *Verticillium dahliae*, *Certatocystis minor* (Khan & Minocha 1989b) and the mycorrhizal fungus Laccaria proxima (Zarb & Walters 1994c).

The polyamines spermidine and spermine are synthesised from putrescine by subsequent additions of aminopropyl groups donated by decarboxylated S-adenosylmethionine, which is formed from S-adenosylmethionine (AdoMet) in a reaction catalysed by the enzyme S-adenosylmethionine decarboxylase (AdoMetDC: EC 4. 1. 1. 50). The aminopropyl group additions are catalysed by the aminopropyltransferases spermidine synthase (EC 2. 5. 1. 16) and spermine synthase (EC 2. 5. 1. 22) respectively (Walters 1995).

Cadaverine is formed by the decarboxylation of the amino acid lysine. In plants, cadaverine is formed via the enzyme lysine decarboxylase (LDC: EC 4. 1. 1. 18) (Smith 1985). However, in animal cells and *Neurospora crassa*, cadaverine formation is catalysed by ODC, albeit inefficiently (Pegg & McGill 1979; Paulus *et al.* 1982). In cultured animal cells, bacteria and *N. crassa*, starved of polyamines, cadaverine is synthesised along with its higher derivative aminopropylcadaverine (APC) (Dion & Cohen 1972; Alhonen-Hongisto & Jänne 1980; Paulus *et al.* 1982). Zarb & Walters (1994b) showed that APC and another higher derivative of cadaverine, N,N-bis(3-aminopropylcadaverine) (3APC) were routinely formed in a range of mycorrhizal fungi.



Figure 2. Biosynthetic pathways of the major plant polyamines (Reproduced from Walters (1995))

1. arginine decarboxylase (ADC) 2. ornithine decarboxylase (ODC) 3. arginase 4. agmatine iminohydrolase 5. N-carbamoylputrescine amidohydrolase 6. S-adenosylmethionine decarboxylase (AdoMetDC) 7. spermidine synthase 8. spermine synthase

Note: Decarboxylation of arginine is common to plants and bacteria, however the intermediates preceeding putrescine differ; the plant biosynthetic route is shown.

1.1.3.2 Regulation of polyamine biosynthetic enzymes

The mammalian polyamine biosynthesis pathway consists of four irreversible steps, catalysed by the enzymes ODC, AdoMetDC, spermidine and spermine synthase. Putrescine synthesis appears to be regulated by the availability of ornithine and by the activity of ODC. The conversion of putrescine to spermidine and spermine is regulated by the activity of AdoMetDC, the supply of decarboxylated AdoMet determining the rate of synthesis of the higher polyamines. Whether decarboxylated AdoMet is utilised for the synthesis of spermidine or spermine depends on the amount of spermidine and spermine synthases available (Pegg 1988).

ODC activity is highly regulated and various mechanisms exist to facilitate enzyme response to fluctuating cellular polyamine levels. ODC also responds to a variety of growth stimulatory substances including hormones. In such responses, it has been shown that ODC activity is increased via an increased amount of ODC enzyme (Pegg 1986; Pegg 1987). This itself is mainly due to an elevated rate of protein synthesis although enzyme stabilisation may also occur (Pegg 1988).

In the presence of elevated levels of polyamines, ODC levels are reduced. Conversely, depletion of polyamines leads to an increase in cellular ODC content (Pegg 1988). Such changes in ODC concentration appear to be the result of changes in the rate of translation of the protein and in the rate of enzyme turnover (Pegg 1986; Pegg 1987). Indeed, all three key regulatory enzymes in the polyamine biosynthetic pathway, ODC, AdoMetDC and spermidine/spermine N^1 -acetyltransferase (whose role will be discussed shortly), have very short half lives, enabling rapid turnover in response to cellular polyamine fluctuations (Pegg 1988).

. .

Turnover of ODC protein may also be enhanced by its association with an inhibitory protein, 'antizyme'. This protein is induced by polyamines and inhibits ODC activity by forming a complex with the enzyme (Fong *et al.* 1976; Canellakis *et al.* 1979). Degradation of ODC in hepatoma tissue culture (HTC) cells and enzyme from Chinese hamster ovary (CHO) cells was indeed accelerated by 'antizyme' (Murakami *et al.* 1992a,b).

AdoMetDC activity is regulated by the availability of methionine which is the starting material for S-adenosylmethionine. Like ODC, AdoMetDC activity is highly regulated and is increased in response to various growth stimulatory substances (Pegg & McCann 1982; Pegg 1986). Such an increase has been reported to be the result of increased protein accumulation (Shirahata & Pegg 1985; Pegg *et al.* 1988b). This itself is probably due to
changes in rates of protein translation. Increased enzyme synthesis is less likely to be responsible (Pegg *et al.* 1988b).

Putrescine activates AdoMetDC directly, thus, a rise in putrescine levels will lead to an increase in decarboxylated AdoMet content so as to form spermidine and spermine (Pegg & McCann 1982; Pegg 1986). Elevated levels of spermidine and spermine act as products and repress AdoMetDC activity (Pegg 1984; Pegg 1986). Thus, as with ODC, depletion of polyamines results in increased AdoMetDC content while elevated polyamine levels lead to a reduction in AdoMetDC protein. These changes are mainly due to alterations in the rate of synthesis of the enzyme although degradation rates are also involved (Shirahata & Pegg 1985; Pegg *et al.* 1988b).

The third highly regulated enzyme involved in polyamine biosynthesis is the catabolic enzyme spermidine/spermine N^1 -acetyltransferase (SSAT). Enzyme activity can be induced by spermidine or spermine but is unaffected by putrescine. The increased activity of the enzyme results from both an increase in synthesis and a reduction in enzyme degradation (Pegg 1986).

The aminopropyltransferases are not as highly regulated as those enzymes discussed already. They do not turn over rapidly and fluctuations in intracellular polyamine levels do not generally lead to alterations in their activity (Pegg 1988).

Cells generally control their steady-state amount of biosynthetic end products through feedback inhibition (Davis 1990). Indeed, alterations in enzyme activity in Ehrlich ascites tumour cells treated with ODC inhibitors was attributed to the depleted polyamine concentrations exerting feedback control (Persson *et al.* 1989; Holm *et al.* 1989). However, control mechanisms involving the likes of enzyme turnover and protein translation rates are more elaborate and reflect the difficulty intracellular polyamines have in carrying out feedback inhibition (Davis 1990). Polyamines bind readily to cellular anions, thus Davis (1990) argued that feedback inhibition would be unworkable. Davis *et al.* (1985) suggested that only a small proportion of intracellular polyamines were involved in biosynthesis regulation while Paulus *et al.* (1983) demonstrated that the majority of polyamines in *Neurospora crassa* are bound to cell constituents. Bound polyamine can be released to overcome shortfalls in free polyamines available to the cell (Davis *et al.* 1992). Adjustments to biosynthetic enzyme activity on the basis of the free polyamine pool, by feedback inhibition, would be largely inaccurate in terms of the whole cellular requirements (Davis

1990). Therefore, the elaborate mechanisms of enzyme regulation have evolved to respond to intracellular levels of polyamines (Davis *et al.* 1992).

1.1.3.3 Other means of controlling intracellular polyamine levels

Although polyamines are necessary for normal cell growth and development, polyamines are also toxic to organisms when present in excess (Davis 1990). Indeed, a mutant of *Neurospora crassa* which accumulated putrescine grew at a substantially reduced rate (Davis & Ristow 1991), while high levels of putrescine were reported to be toxic to the cyanobacterium *Anabaena* (Guarino & Cohen 1979).

As well as highly regulated biosynthetic enzymes, a number of mechanisms exist enabling cells to deal with excess polyamines. Mammalian cells have an elaborate polyamine catabolism system in which polyamines are interconverted and degraded back to putrescine (Pegg 1988). Spermine and spermidine are acetylated by the enzyme spermidine/spermine N¹-acetyltransferase (SSAT) to form N¹-acetyl derivatives of spermine and spermidine. Such derivatives are then substrates for polyamine oxidase (PAO: EC 1.5.3.3), and spermidine and putrescine are formed respectively (Pegg 1988). The spermidine formed from spermine can likewise form putrescine and the diamine either be excreted or degraded by diamine oxidase (DAO: EC 1.4.3.6) to form hydrogen peroxide (Pegg 1988). Thus, eatabolism may prevent intracellular polyamines reaching toxic levels (Pegg 1986) (Figure 3).

Many plants and microbes also possess the ability to catabolise polyamines (Davis 1990). Indeed, catabolism in *Escherichia coli* is well characterised (Large 1992). In other microorganisms, however, the pathways taken are not well defined. Two major routes are available, one via N-acetyl derivatives, as described for mammalian cells, and the other via Δ^1 -pyrroline (Large 1992). The N-acetyl route has been described for yeasts such as *Candida boidinii*, however, it is likely that other catabolic routes exist for such organisms (Large 1992). Polyamine conjugates such as glutathionyl-spermidine exist yet their function is unknown. Large (1992) suggests that they may be part of an as yet uncharacterised catabolic pathway. Little is known of polyamine catabolism in plant pathogenic fungi (Walters 1995).

Excess polyamines can be removed from cells by conjugation. Acetyl derivatives and glutathionyl-spermidine have already been discussed with respect to degradation. Such derivatisation may also act as a means of removing polyamines from the free pool (Large 1992) (Figure 3). Glutathionyl-spermidine conjugates have been reported for *E. coli* and trypanosomes (Tabor & Tabor 1985). Conjugation as hydroxycinnamic acid amide

derivatives has been observed in plants (Flores *et al.* 1989). However, Flores *et al.* (1989) suggested that conjugated forms of polyamines may have more significance as reserves than as detoxified forms of polyamines. Indeed, some evidence exists that conjugated forms of putrescine may be reversibly hydrolysed to maintain free putrescine levels in tissues treated with putrescine biosynthesis inhibitors (Slocum & Galston 1985).

Sequestration of polyamines from the cytosol into cellular organelles is known to control excess polyamines. Organisms such as *N. crassa*, other fungi and plants are capable of moving polyamines into vacuoles (Davis 1990).

Excretion of polyamines can also enable cells to control intracellular levels of amines. Cancer cells frequently release excessive polyamines into biological fluids, as do many normal mammalian tissues, probably by simple diffusion once intracellular levels of polyamines become too high (Davis 1990). *N. crassa* continues to synthesise spermidine whilst in the stationary phase and controls its intracellular polyamine levels by excretion (Davis & Ristow 1989). Control of extracellular uptake of polyamines can be used to regulate intracellular polyamine levels (Davis 1990), and this will be discussed later in the thesis.





spermidine/spermine N¹-acetyltransferase (SSAT);
polyamine oxidase (PAO);
spermine synthase;
spermidine N⁸-acetyltransferase;
N⁸-acetylspermidine deacetylase;
spermine synthase

Note: N⁸-acetylspermidine and N-acetylputrescine are conjugated forms of polyamines

1.1.4 INITIBITION OF POLYAMINE BIOSYNTHESIS

Although the physiological functions of polyamines are not fully understood, polyamines are undoubtedly vitally important to cellular processes such as proliferation. Indeed, elevated levels of polyamines and their biosynthetic enzymes are known to occur in rapidly proliferating tissues (Sunkara *et al.* 1987). This knowledge prompted workers, attempting to develop novel chemotherapeutic approaches for cancer treatment, to target the polyamine biosynthesis pathway since its inhibition was hypothesised to afford control of rapidly proliferating cells. Instigation of this work soon led to studies examining perturbation of polyamine biosynthesis as a means of controlling growth of many different types of organism (Porter & Sufrin 1986; Schechter *et al.* 1987; Pegg 1988). The following section is designed to provide a brief insight into the various inhibitors of polyamine biosynthetic enzymes although more details will be discussed in the subsequent sections relating to growth inhibition of specific organisms.

1.1.4.1 Inhibitors of ODC

Research into the inhibition of ODC has occurred via three different routes (Bey *et al.* 1987). The first of these involved the synthesis of substrate (ornithine) and product (putrescine) analogues as potential competitive reversible inhibitors of the enzyme. Compounds included the substrate analogues α -fluoromethylornithine (Bitonti *et al.* 1982), dehydroornithine (Relyea & Rando 1975) and the product analogue dehydroputrescine (Relyea & Rando 1975). The second route involved the synthesis of molecules capable of interacting or combining with the enzyme co-factor, pyridoxal phosphate. Such molecules were also competitive reversible inhibitors and examples included α -hydrazinoornithine (Inoue *et al.* 1975) and 1-aminooxy-3-aminopropane (APA) (Khomutov *et al.* 1985). Although α -hydrazinoornithine inhibited ODC from rat liver with an affinity constant of 0.5 μ M (Inoue *et al.* 1975) and APA inhibited mouse kidney enzyme with a K₁ of 0.0032 μ M (Khomutov *et al.* 1985), these compounds lacked specificity, affecting other pyridoxal phosphate dependent enzymes (Khomutov *et al.* 1985).

The final route taken, in attempts to inhibit ODC, involved the design and synthesis of enzyme-activated or suicide irreversible inhibitors. Such compounds are chemically inert pseudosubstrates of the target enzyme. The inhibitors have a latent group incorporated into their structure which is transformed by catalytic turnover into a species that eventually inactivates the enzyme (Bey *et al.* 1987). Inhibitors belonging to this group include the ornithine analogues α -difluoromethylornithine (DFMO) and dehydromonofluoromethylornithine (Δ MFMO) which have affinity constants of 39 μ M and 2.7 μ M respectively for enzyme from rat liver (Metealf *et al.* 1978; Bey *et al.* 1983). Since Δ MFMO penetrates cell membranes poorly, *in vivo* use of the inhibitor utilises its methyl ester derivative,

AMFMO.CH₃ (Mamont *et al.* 1986). Putrescine analogues include (2R,5R)-6-heptyne-2,5-diamine (RR-MAP) which is a potent inhibitor of growth of rat hepatoma cells (Mamont *et al.* 1984).

Section Section 1

-2

i.

1 4434

Much exploration of ODC inhibition has been carried out with enzyme-activated irreversible inhibitors being particularly successful. It should be noted, however, that the relative potency of these inhibitors depends upon the source of the enzyme (Bey *et al.* 1987).

1.1.4.2 Inhibitors of AdoMetDC

Methylglyoxal bis(guanylhydrazone) (MGBG) is a powerful competitive inhibitor of mammalian AdoMetDC with an inhibition constant of less than 1 μ M (Williams-Ashman & Pegg 1981). The inhibitor is less active towards AdoMetDC from other sources such as *Escherichia coli*, Chinese cabbage and the slime mold *Physarum polycephalum* (Pegg & Jacobs 1983; Pegg 1984; Yamanoha & Cohen 1985). A number of analogues of MGBG have been synthesised and ethylmethylglyoxal bis(guanylhydrazone) (EMGBG), was found to be a more potent inhibitor of AdoMetDC than MGBG (Pegg & Jacobs 1983).

Much work has been carried out using MGBG as a potential inhibitor of polyamine biosynthesis. Indeed, the compound has been reported to cause changes in the polyamine content of cells and spermidine has been shown to prevent the inhibitory action of MGBG (Pegg & Williams-Ashman 1987). In spite of this, it is unlikely that the compound exerts its effects wholly via inhibition of AdoMetDC activity (Pegg & Williams-Ashman 1987). Transport of the compound into cells utilises the same carrier system as polyamines. Prevention of growth by spermidine depletion may be due to competition between MGBG and spermidine for transport into cells (Pegg & Williams-Ashman 1987). MGBG inhibits cellular respiration and this has been implicated in severe mitochondrial damage observed in mammalian cells (Jänne *et al.* 1985). Nikula *et al.* (1984) reported that MGBG inhibited mitochondrial fatty acid oxidation and suggested this to be the cause of mitochondrial damage. EMGBG does not appear to affect cellular respiration, but although it is a more potent inhibitor of AdoMetDC than MGBG, it failed to demonstrate antiproliferative activity against mouse L1210 leukemia cells *in vitro* (Elo *et al.* 1986).

Many other inhibitors of AdoMetDC have been described including nucleoside analogues of AdoMet, such as the irreversible inhibitor S-(5'-deoxy-5'adenosyl)-methylthioethylhydroxylamine (AMA) (Artamonova *et al.* 1986). Secrist (1987) synthesised a number of irreversible inhibitors including 5'-deoxy-5'-{N-methyl-N-[2-(aminooxy)ethyl]}aminoadenosine (MAOEA) and 5'-deoxy-5'-{N-methyl-N-(3-hydrazinopropyl)]aminoadenosine (MHZPA). These three compounds were reported to inhibit polyamine biosynthesis in L1210 cells (Pegg et al. 1988a; Kramer et al. 1989).

Recently, Regeness *et al.* (1994) described a cyclic analogue of MGBG, CGP 48 664 and reported how the compound inhibited AdoMetDC and exerted powerful antiproliferative effects against a variety of tumours (Regeness *et al.* 1994).

1.1.4.3 Inhibitors of spermidine synthase

Cyclohexylamine (CHA) has been sold as dicyclohexylammonium sulphate and confusion in the literature exists with many researchers incorrectly naming cyclohexylamine as dicyclohexylamine (Batchelor *et al.* 1986). The compound was initially reported as a strong inhibitor of mammalian spermidine synthase (Hibasami *et al.* 1980). Reports then followed of enzyme from trypanosomes, plants and various bacteria being inhibited by CIIA (Pegg *et al.* 1983; Pösö *et al.* 1983; Paulin *et al.* 1983; Mattila *et al.* 1984; Sindhu & Cohen 1984). Enzyme from some bacteria were not, however, affected by the compound (Pösö *et al.* 1983; Mattila *et al.* 1984).

CHA is readily taken up into mammalian cells, inhibiting growth and leading to a depletion in spermidine levels (Ito et al. 1982; Mitchell et al. 1985; Feuerstein et al. 1985). Growth inhibitory effects on tumours in rodents were also observed (Ito et al. 1982). In extracts of *Helianthus tuberosus*, CHA decreased spermidine levels and increased putrescine concentration (Torrigiani et al. 1987). Such changes in polyamines were also observed in carrot embryos treated with cyclohexylammonium sulphate (Khan & Minocha 1991) and in pine cotyledons treated with CHA (Biondi et al. 1986). No effect was observed on the growth of pine cotyledons, however, an increase in spermine was suggested to be responsible for continued growth (Biondi et al. 1986). Such an increase in spermine and reduction in spermidine was also observed in protoplasts from Chinese cabbage leaves where spermidine synthase was inhibited by CHA *in vivo* (Greenberg & Cohen 1985). Feirer et al. (1985) also reported how CHA reduced growth and embryogenesis in carrots and decreased spermidine levels.

However, when studying *in vitro* propagation of potatoes on inhibitor amended medium, CHA reduced growth yet did not affect spermidine levels, suggesting that growth inhibiton was occurring by some other mechanism (Masse *et al.* 1988). Tiburcio *et al.* (1987), studying organogenesis in tobacco callus culture systems, reported how the effect of CHA was unclear and unlikely to be spermidine synthase inhibition. Indeed, Pegg & Williams-Ashman (1987) noted how information on the specificity of CHA was lacking and that the compound may affect other aspects of cellular physiology.

Various other inhibitors of spermidine synthase exist, including S-adenosyl-1,8-diamino-3thiooctane (AdoDato) which successfully and specifically inhibited activity of spermidine synthase from various mammalian sources, bacteria, trypanosomes and plants (Pegg *et al.* 1983; Bitonti *et al.* 1984; Sindhu & Cohen 1984; Yamanoha *et al.* 1984). The compound did indeed deplete spermidine from mammalian cells and increased spermine levels (Pegg *et al.* 1982b). Hamaguchi and co-workers (1987) discovered an active compound juglorin in the culture filtrate of *Streptomyces malachiticus* which inhibited spermidine synthase but with weaker activity than that of CHA. Trans-4-methylcyclohexylamine was reported as inhibiting mammalian spermidine synthase with similar activity to CHA (Shirahata *et al.* 1988). Analogues of decarboxylated AdoMet had significant effects on polyamine levels in cultured L1210 cells, although they were suggested to affect both spermidine synthase and spermine synthase (Douglas *et al.* 1991).

1.1.4.4 Inhibitors of spermine synthase

Inhibitors of spermine synthase include S-methyl-5'-methylthioadenosine $(AdoS^+(CH_3)_2)$ (Pegg *et al.* 1986) and S-adenosyl-1,12-diamino-3-thio-9-azadodecane (AdoDatad) (Coward & Pegg 1987). These compounds deplete cells of spermine yet have no effect on cell growth (Pegg *et al.* 1986). However, spermidine levels are increased following treatment of cells with the inhibitors and such clevated levels may account for the continuation of growth (Pegg *et al.* 1986). Combination of such inhibitors with one of AdoMetDC has therefore been suggested as an approach to effective depletion of spermidine and spermine (Pegg 1988).

1.1.4.5 Inhibitors of ADC

A range of α -fluoromethyl analogues of arginine were synthesised and found to be potent inhibitors of bacterial ADC, with the specific, enzyme-activated, irreversible α -difluoromethylarginine (DFMA) being one such inhibitor (Kallio *et al.* 1981). Analogues of arginine and agmatine, the enzymatic product, were also synthesised and found to inhibit ADC from *Escherichia coli* and oats with greater activity than DFMA (Bitonti *et al.* 1987a).

1.1.4.6 Polyamine analogues

The use of polyamine analogues as inhibitors of polyamine biosynthetic enzymes has already been discussed briefly in relation to each specific enzyme. Indeed, MGBG is an analogue of spermidine (Dave & Caballeri 1973). In work which will be discussed later, the irreversible enzyme-activated inhibitor DFMO was shown to have several short-comings with regards to inhibition of cell growth. Continuation of cell growth in the presence of DFMO was attributed to the drug being unable to totally deplete polyamines from cells (Pegg *et al.* 1982a; Mamont *et al.* 1982). Thus, Porter & Sufrin (1986) speculated that

polyamine analogues, differing only slightly in structure from naturally occurring polyamines, might be more capable of reducing cell growth than specific enzyme inhibitors.

Polyamines are taken up into most cells utilising a carrier system (Porter *et al.* 1982; Porter *et al.* 1984), which is specific for polyamine-like molecules (Seppänen 1981; Bartholeyns *et al.* 1984). Polyamine analogues may be able to utilise this system by binding to the carrier proteins at sites specific for polyamines. Analogues would be able not only to participate in binding but also to compete for sites with the polyamines. As an added advantage in cancer chemotherapeutic strategies, transport of polyamines into malignant tissues is elevated compared to normal tissues (Porter & Sufrin 1986). Once in the cell, analogues were hypothesised to exert antiproliferative effects by one or more of a number of mechanisms. Inhibition of polyamine biosynthetic enzymes could occur with analogues acting as either substrate or product analogues. Regulation of polyamines for binding sites specific to cellular proliferation. Disruption of macromolecular structure and/or function could occur if analogues were to bind at associated polyamine sites and finally, analogues could act as vector molecules to deliver antiproliferative agents to cancerous tissues (Porter & Sufrin 1986).

2

ŝ

je N

Previous initial attempts using polyamine analogues as potential anticancer agents were limited since structural similarity to polyamines was minimal (Porter & Sufrin 1986). Analogues of spermidine and spermine were synthesised by various workers (Israel *et al.* 1964; Jorstad *et al.* 1980). However, only norspermidine inhibited cell growth *in vitro* (Porter & Bergeron 1983; Casero *et al.* 1984). Later work has shown this compound to possess antitumour activity against experimental animal tumours (Prakash *et al.* 1988). Norspermidine also inhibited HeLa cell growth in culture (Sunkara *et al.* 1988). Mitotic activity in such cells ceased while polyamines were depleted and ODC activity reduced. Interestingly, the growth inhibition could not be reversed by polyamines and norspermidine may thus have affected growth via an effect on mitosis of cells and/or perturbation of polyamine metabolism (Sunkara *et al.* 1988). Inhibition of spermidine biosynthesis was, however, associated with a retardation of growth of potatoes (Masse *et al.* 1988) and maize roots (Masse *et al.* 1985) on norspermidine amended media.

A number of spermidine analogues were designed, synthesised and evaluated (Porter *et al.* 1982; Porter *et al.* 1985). Spermidine analogues were chosen since studies with DFMO had indicated that it is the primary polyamine implicated in cell growth (Mamont *et al.* 1978). The analogues were evaluated against L1210 cell growth and three possessed significant inhibitory activity (Porter *et al.* 1985). Conclusions were made that two of the analogues,

 N^{1} , N^{8} -bis(ethyl)spermidine (BES) and N^{1} , N^{8} -bis(propyl)spermidine (BPS), did indeed inhibit cell growth via polyamine depletion by regulating the biosynthetic enzymes as polyamines do (Porter *et al.* 1985). The analogues were incapable of promoting cellular growth (Porter *et al.* 1985). Thus, the potential of polyamine analogues as antiproliferative agents was indeed apparent although why some of the analogues do not inhibit growth and how the other exerts its mode(s) of action is unclear.

Many other polyamine analogues exist but their uses will be discussed in relation to the growth inhibition of various organisms.

1.1.5 CANCER RESEARCH

It is thus apparent that many inhibitors of polyamine biosynthetic enzymes exist. Their use in cancer research shall now be discussed before considering applications in the control of growth of other organisms and a detailed examination of their uses against growth of plant pathogenic fungi.

As discussed, perturbation of polyamine biosynthesis was suggested as a potential cancer chemotherapeutic strategy. Targetting of cellular proliferation is not wholly selective towards cancerous tissues, and as with most chemotherapeutic strategies, side effects were envisaged to occur. However, a number of cancers were known to be curable by chemotherapy and the development of a strategy involving inhibition of polyamine biosynthesis was worthy of consideration (Porter & Sufrin 1986). Over the past twenty years, much work has been carried out in this area and only a brief review of this work shall be given.

The initial polyamine biosynthetic enzyme to be targetted was ODC, the rate limiting enzyme (Sunkara et al. 1987). Relative success with various reversible inhibitors quickly instigated the design and synthesis of specific irreversible enzyme inhibitors by Merrell Dow scientists, the best known of these being DFMO (Metcalf et al. 1978). DFMO treatment typically reduced ODC activity and depleted mammalian cells of putrescine and spermidine (Mamont et al. 1978). Spermine levels were not reduced, probably because ODC activity would have continued at low levels, permitting small amounts of putrescine to be formed and converted through spermidine to spermine (Pegg et al. 1982a; Mamont et al. 1982). Alternatively, spermine pools may have been unaffected since growth is largely dependent upon spermidine. As ODC is inhibited, spermidine levels decrease by two means; by dilution since the cells continue to divide, and by conversion to spermine. Thus, when spermidine reaches a critical level and cell division ceases, the spermine pool is maintained (Porter & Sufrin 1986). It was this phenomenon which was to prove problematic with respect to use of DFMO. Cytostasis occurs when cells cease to divide in order to conserve their intracellular polyamine pools. However, upon removal of the drug, the residual polyamines allow resumption of cellular growth. Thus, DFMO inhibitory action was often temporary (Porter & Sufrin 1986).

Nevertheless, growth of various cultured cells including HTC and L1210 leukemia cells was inhibited by DFMO (Mamont *et al.* 1978). The inhibitory effects were reversed by the addition of putrescine and spermidine, indicating depletion of polyamines to be critical (Mamont *et al.* 1978). Inhibition of growth of many other types of cell was reported, including HeLa cells (Sunkara *et al.* 1980) and Ehrlich ascites carcinoma cells (Oredsson *et al.*

al. 1980). Effects of DFMO were generally cytostatic, however, cytotoxicity was apparent in some cell lines including mouse melanoma cells (Sunkara *et al.* 1983a). DFMO appeared to be cytostatic or cytotoxic, depending on the cell line employed (Sunkara *et al.* 1987).

In spite of the short-comings of the drug upon cells in culture, the compound was used in attempts to inhibit the growth of experimental tumours. Significant inhibitory effects were obtained against a variety of tumours including L1210 leukemia (Prakash *et al.* 1978), and melanoma in mice (Sunkara *et al.* 1983b). DFMO also inhibited growth of human small-cell lung carcinoma cells in experimental mice, with cytotoxicity effects reported. The cells did eventually die under sustained polyamine deprivation (Luk *et al.* 1983).

Given the antiproliferative effects of DFMO against cultured cells and turnours in rodent models, clinical studies of DFMO in patients with cancer commenced in 1979. Patients suffering from far-advanced disease for whom no other therapy could be offered were used in the trials. The advanced stage of their illnesses may have contributed to the relatively poor results observed. DFMO did not affect turnour growth or disease progression significantly. Occasionally, stabilisation of disease occurred but no general pattern of response to DFMO was observed from the patients (Schechter *et al.* 1987). In a later trial using patients suffering from malignant melanoma, more disease stabilisation was observed (Meyskens *et al.* 1986). Side effects such as gastrointestinal effects and hearing loss were reversible upon dose reduction or discontinuation of treatment (Schechter *et al.* 1987).

Although DFMO was not sufficiently effective alone, it was observed to have low toxicity and was thus hypothesised to be more useful in combination with other drugs, perhaps increasing the cytotoxicity of other antitumour agents (Porter & Sufrin 1986). Indeed, uptake of MGBG into Ehrlich ascites cells was increased after pre treatment with DFMO with an associated increase in cytotoxicity of MGBG (Alhonen-Hongisto *et al.* 1980; Scppänen *et al.* 1981). A similar phenomenon was observed with HcLa cells (Sunkara *et al.* 1983b). Marton *et al.* (1981) showed how DFMO increased the cytotoxicity of the DNA reactive drug, 1,3-bis(2-chloroethyl)-1-nitrosourea (BCNU) in rat gliosarcoma cells in culture. However, Oredsson *et al.* (1982) failed to increase cytotoxicity of cis-platinum by pre treating cells with DFMO, demonstrating that results vary according to cell line and/or cytotoxic agent.

Such combinations of DFMO and other agents were observed to be effective against turnours in experimental model systems. The combination of DFMO with interferon suppressed growth of subcutaneous melanomas in mice (Sunkara *et al.* 1983a). Pre treatment with DFMO was hypothesised to deplete cellular polyamines, destabilising DNA

and thus making it more vulnerable to those drugs such as interferon which act upon DNA (Pegg 1988). Sequential therapy using DFMO followed by MGBG was also shown to be effective against L1210 cells in mice (Sunkara *et al.* 1993a). Similarly, DFMO treatment was envisaged to deplete cells of spermidine and since MGBG can utilise the spermidine uptake system, increased levels of MGBG could accumulate in the cell (Sunkara *et al.* 1987).

Initial clinical trials using combination therapy, for example utilising DFMO followed by MGBG against childhood leukemia were successful (Siimes *et al.* 1981). DFMO and interferon, together, were studied against malignant melanoma, however, limited responses were observed (Talpaz *et al.* 1986).

It was apparent that DFMO was not as effective as was originally hypothesised for the treatment of cancer. The cytostatic response of the drug has already been discussed. Other limitations include the rapid clearing of DFMO from the body (Grove *et al.* 1981) and its slow uptake into cells, via a passive diffusion mechanism (Erwin & Pegg 1982). However, the minimal host toxicity of DFMO was indeed promising and led to further work designing specific enzyme-activated irreversible inhibitors and studying other enzyme inhibitors and polyamine analogues, as discussed elsewhere in this review (Porter & Sufrin 1986).

Clearly, perturbation of polyamine biosynthesis and/or functions represents a promising approach to cancer treatment and much work is on-going in this area. Some of the most recent reported work includes an increase in cytotoxicity of cisplatin against L1210 leukemia cells by a polyamine analogue (Hawthorne & Austin 1996), lung tumour cell death caused by polyamine analogues (McCloskey *et al.* 1996) and an investigation into the use of a spermine conjugate as an inhibitor of polyamine transport into cells (Aziz *et al.* 1996).

25

1.1.6 PROTOZOAL, BACTERIA, ANIMAL-INFECTING FUNGI AND INSECTICIDE RESEARCH

1.1.6.1 Protozoal infections

Perturbation of polyamine biosynthesis was initially hypothesised by Cohen (1979) as a target for chemotherapeutic control of parasitic diseases. As a consequence of the relative success achieved with DFMO in inhibiting ODC and growth of mammalian cells, coupled with the low toxicity of the compound, work was instigated and showed DFMO to cure infection of mice with the protozoan cause of bovine trypanosomiasis, *Trypanosoma brucei brucei* (Bacchi *et al.* 1980). DFMO was also shown to be effective against mice infected with the human sleeping sickness pathogens *T. b. rhodesiense* and *T. b. gambiense* (McCann *et al.* 1986), while the compound reduced infection and prolonged survival time of mice infected with *T. evansi*, an important blood protozoan parasite of cattle in India (Damayanthi 1990).

Putrescine synthesis in *T. b. brucei* cultured *in vitro* was inhibited by DFMO, suggesting that the compound did indeed affect ODC (Bacchi *et al.* 1980). Inclusion of exogenous polyamines with the DFMO treatment reversed the curative action, confirming that DFMO caused an effect on polyamine levels (Nathan *et al.* 1981). Later work demonstrated that the compound inhibited cellular DNA and RNA synthesis (Bacchi *et al.* 1983). Interestingly, DFMO causes a morphological change in trypanosome populations from replicating parasitic bloodstream trypanosomes to non replicating short stumpy forms (Giffin *et al.* 1986)

Whatever the exact mechanism of action of DFMO against such trypanosomes, the parasites are much more sensitive to the compound than mammalian cells (Mamont *et al.* 1978). The reason for this is not clear as trypanosomal ODC from *T. b. brucei* has the same degree of sensitivity towards DFMO as ODC from mammalian cells (Bitonti *et al.* 1986a). DFMO is also known to enter the parasites by passive diffusion, as with mammalian cells (Bitonti *et al.* 1986a). Several suggestions have, however, been made to account for the difference in sensitivity observed between trypanosomes and mammalian cells with respect to DFMO. Enzyme from trypanosomes may turn over more slowly than that from mammalian cells (Phillips *et al.* 1987). ODC inhibition and subsequent depletion of spermidine could affect amounts of the trypanosome spermidine-containing enzyme co-factor, trypanothione, obligatory for glutathione reductase activity (Fairlamb *et al.* 1985). Also, the bigh multiplication rate of trypanosomes could place a greater dependance on spermidine for trypanosomes than mammalian cells (Byers *et al.* 1991).

1

In addition to DFMO, a number of ornithine and putrescine analogues, RR-MAP, Δ MFMO, Δ MFMO.CH₃ and Δ MFMO.C₂H₅ inhibited ODC from *T. b. brucei* as well as enzyme activity in trypanosomes (Bitonti *et al.* 1985). However, only Δ MFMO.CH₃ was as effective as DFMO against the infection in experimental mice (Bitonti *et al.* 1985). Indeed, Bacchi *et al.* (1987) found that Δ MFMO.CH₃ was effective at concentrations lower than those of DFMO in *T. b. brucei* infections of mice. Here, the compound was taken up by the host animals much more rapidly than DFMO (Bacchi *et al.* 1987). Evaluation of DFMO in combination with other drugs such as the antibiotic bleomycin (Clarkson *et al.* 1983) and the trypanocide suramin (Clarkson *et al.* 1984) against acute *T. b. brucei* infection in mice demonstrated synergistic action.

Interestingly, DFMO has no effect on T. cruzi, the cause of American trypanosomiasis (Kierszenbaum et al. 1987). However, Yakubu and co-workers (1992) demonstrated effective inhibition of T. cruzi growth by DFMA, the organism being unusual in synthesising putrescine via ADC.

Further studies using alternative polyamine biosynthesis inhibitors against trypanosomal parasites have been reported. MGBG inhibited AdoMetDC from *T. b. brucei* (Bitonti *et al.* 1986b). Another inhibitor of AdoMetDC, 5'-{[Z]-4-amino-2-butenyl]methylamino}-5'- deoxyadenosine (AbeAdo) cured infections in mice caused by *T. b. brucei* (Bitonti *et al.* 1990). Indeed, the drug caused the parasites to be cleared from the bloodstream extremely rapidly, before significant depletion of spermidine had occurred, suggesting a mechanism of action other than polyamine depletion (Bitonti *et al.* 1990). Substantial increases in AdoMet levels were observed in trypanosomes exposed to AbeAdo (Byers *et al.* 1991). Since only smaller increases in AdoMet were observed in similarly treated mammalian cells, the phenomenon was suggested to be specific to parasites (Byers *et al.* 1991).

Similarly, reduced AdoMetDC activity and elevated AdoMet levels were observed in trypanosomes taken from rats which had been infected with T. b. brucei and treated with DFMO (Byers *et al.* 1991). Thus, inhibitors may act against trypanosomes via the induction of elevated AdoMet levels (Byers *et al.* 1991). Since AdoMet donates a methyl group for methylation, the mechanism by which elevated AdoMet may lead to control of growth of trypanosomes may be associated with aberrant methylation (Byers *et al.* 1991).

AdoMetDC activity of T. b. brucei was also inhibited by the trypanocidal agents Berenil (4,4'-diamidinodiazoaminobenzene) and pentamidine (p, p'-(pentamethylenedioxy)-dibenzamidine (Bitonti et al. 1986b). Two compounds, AdoDato and dicyclohexylamine,

shown to inhibit spermidine synthase from *T. b. brucei*, failed however, to prolong the life span of infected mice (Bitonti *et al.* 1984).

Given the promising results against trypanosomal infections in experimental models, clinical evaluations of DFMO against human sleeping sickness were initially reported by Sjoerdsma & Schechter (1984). Effective control of the disease was observed in twenty patients with parasites being eradicated from the blood, usually within days of commencing the treatment. One patient relapsed but most others remained symptom free for several months to two years. Taelman et al. (1987) reported the remarkable response of a comatose patient, suffering from late stage sleeping sickness, to DFMO administration. Rapid recovery was observed and thereafter, the patient remained parasite free. In spite of the efficacy demonstrated, DFMO did cause side effects although these were no more than those expected with chemotherapeutic strategics. Indeed, the side effects, such as nausea and diarrhoea were reversible (Schechter et al. 1987). Many other investigations have been reported for DFMO against human and animal trypanosomiasis and successful results led to DFMO now having been marketed for several years by Merrell-Dow Pharmaceuticals as Ornidyl, for treatment against human West African (Gambian) sleeping sickness. It should, however, be noted that drug resistant populations may emerge since resistance to DFMO was observed in a mutant strain of T. b. brucei (Phillips & Wang 1987).

Research into inhibition of polyamine biosynthesis as a chemotherapeutic strategy against other protozoan infections has been carried out. Malaria remains one of the most important infectious diseases in the world and since resistance to antimalarial drugs is common-place, the development of new drugs is paramount (Peters 1985). Initial studies demonstrated that DFMO inhibited proliferation of *Plasmodium falciparum in vitro* (McCann *et al.* 1981). The compound was later shown to inhibit proliferation of *P. berghei* in mice although it did not cure the infection (Bitonti *et al.* 1987b).

A number of bis(benzyl)polyamine analogues were found to be potent inhibitors of growth of chloroquine-resistant and chloroquine-sensitive strains of *P. falciparum in vitro* (Bitonti *et al.* 1989). Analogues also inhibited growth of *P. berghei in vivo* and combination of the analogues with DFMO cured mice with established infections of *P. berghei* (Bitonti *et al.* 1989). Inhibition appeared to be cytotoxic, in contrast to effects of DFMO which tend to be cytostatic (Pegg & McCann 1982). Thus, prevention of parasitic growth by the analogues was suggested to be due to means other than the repression of polyamine biosynthesis (Bitonti *et al.* 1989). Indeed, binding of the analogues to DNA and subsequent disruption of macromolecular synthesis was concluded to be the major cytotoxic effect of the analogues against parasitic infections (Bitonti *et al.* 1989).

Clinical trials using DFMO against the protozoa *Pneumocystis carinii*, which causes pneumonia in immunocompromised patients, and is known as the most common lethal infection of patients with Aquired Immune Deficiency Syndrome (AIDS) have also been carried out (Schechter *et al.* 1987). While drugs are available for treatment of the infection, patients often suffer severe side effects or even fail to respond to these drugs (Golden *et al.* 1984). These patients were involved in the first clinical trials of DFMO against *P. carinii* with beneficial results obtained which warranted further study (Golden *et al.* 1984).

DFMO has also demonstrated a high degree of efficacy towards *Eimeria tenella*, the cause of coccidiosis in chickens (McCann *et al.* 1981) as well as other protozoan species; *Trichomonas* (Yarlett & Bacchi 1994) and *Leishmania* (Kaur *et al.* 1986). Clearly, perturbation of polyamine biosynthesis is useful against protzoan infections.

Much work is on-going with regards to the perturbation of polyamine biosynthesis and/or function in the sphere of protozoal research. Most recently, selective toxicity of DFMO for *P. carinii* has been reported (Merali & Clarkson 1996), MGBG derivatives have been shown to be effective against trypanosomes (Bacchi *et al.* 1996) and an inhibitor of AdoMetDC demonstrated inhibitory activity against growth of *Leishmania donovani* (Mukhopadhyay *et al.* 1996).

1.1.6.2 Bacteria

Putrescine biosynthesis in most bacteria occurs via both the ODC and ADC route, therefore, any desired inhibition of growth through polyamine biosynthesis inhibition must consider both pathways (Kallio & McCann 1981). Nevertheless, studies have been carried out to evaluate various inhibitors as potential bacterial growth retardants.

ODC activity in the bacteria *Escherichia coli* and *Pseudomonas aeruginosa* exposed to DFMO was reduced although, since ADC activity increased as a compensatory measure, no growth reduction was observed (Kallio & McCann 1981). Interestingly, ODC extracted from *E. coli* was unaffected by DFMO, yet the enzyme from *P. aeruginosa* was inhibited. Enzyme from another bacteria, *Klebsiella pneumoniae*, was unaffected by DFMO (Kallio & McCann 1981). Another ODC inhibitor, DL- α -monofluoromethylputrescine, inhibited enzyme from *E. coli* but failed to affect the enzyme from *P. aeruginosa* (Kallio *et al.* 1982). This is in direct contrast to those effects of DFMO (Kallio & McCann 1981). The studies showed differences in specificity for enzyme inhibitors by the bacterial enzymes as well as *in vitro* and *in vivo* differences.

Kallio and co-workers (1982) demonstrated that a combination of DL- α -monofluoromethylputrescine with DFMA reduced ODC and ADC activity in *E. coli*, agreeing with the suggestion that a two-pronged attack on bacterial putrescine synthesis was necessary to prevent compensatory increases in enzyme activity (Kallio & McCann 1981). However, putrescine levels were only reduced to one third of control values (Kallio *et al.* 1982).

Growth of *E. coli* and *P. aeruginosa* was inhibited using a combination of DL- α monofluoromethylomithine (MFMO), DFMA and dicyclohexylammonium sulphate. Putrescine and spermidine levels were reduced and the growth inhibition reversed by the inclusion of exogenous polyamines in the growth medium, indicating that the growth reduction was indeed due to polyamine depletion (Bitonti *et al.* 1982).

In contrast, growth of *Agrobacterium tumefaciens* was inhibited through the use of two single inhibitors, DFMO and CHA. DFMO reduced putrescine and spermidine levels while CHA had no effect on intracellular polyamines. Growth inhibition by DFMO was reversed by putrescine but retardation of growth by CHA was not affected by exogenous spermidine, suggesting that CHA did not specifically inhibit polyamine biosynthesis (Ponappa *et al.* 1992). DFMA did not inhibit growth of the bacterium and indeed, ODC was observed to be the primary enzyme for the synthesis of putrescine in *A. tunefaciens* (Ponappa *et al.* 1992). However, the converse was observed by Speranza and Bagni (1977), with these authors identifying ADC as the primary enzyme. Such a difference may be attributed to the specific strain of bacterium (Speranza & Bagni 1977).

The ADC inhibitor DFMA reduced enzyme activity in *E. coli* and *P. aeruginosa* as well as enzyme extracted from these bacteria. DFMA also inhibited ADC from *K. pneumoniae* (Kallio *et al.* 1981). Arginine and agmatine analogues successfully inhibited ADC from *E. coli*, being more potent than DFMA (Bitonti *et al.* 1987a)

Inhibition of polyamine biosynthesis in *E. coli* has been further complicated by the reported finding of a probable new pathway for putrescine, not involving ODC or ADC (Cataldi & Algranati 1986). Nevertheless, perturbation of polyamine biosynthesis and/or function may prove useful in the control of bacterial growth.

1.1.6.3 Animal-infecting fungi

Inhibition of polyamine biosynthesis as a means of chemotherapeutic control of fungal infections of animals has also been studied. *In vitro* growth of *Candida albicans*, *C. tropicalis* and *C. parapsilosis* was inhibited by DFMO although *C. albicans* and

C. parapsilosis were less susceptible to the inhibitor than *C. tropicalis* (Pfaller *et al.* 1987). In later work, DFMO inhibited ODC activity in *C. albicans* and *C. tropicalis*, however, enzyme inhibition was greater for *C. albicans* compared to *C. tropicalis*. Such a result does not correlate with the given effects on yeast growth where *C. tropicalis* growth was more sensitive to DFMO than *C. albicans* (Pfaller *et al.* 1988). Growth of *Cryptococcus neoformans*, a pathogenic yeast associated with meningitis infections which occur in patients with AIDS (Patterson & Andriole 1989), was also inhibited by DFMO (Pfaller *et al.* 1990). In all of the fungal species, fluctuations in polyamine levels were consistent with inhibition of ODC and growth inhibition was reversed by exogenous polyamines, indicating that DFMO was indeed inhibiting ODC activity and not exerting non specific toxic effects (Pfaller *et al.* 1987, 1988, 1990).

Growth of *C. albicans*, *C. tropicalis* and *C. neoformans* was also inhibited by CHA. Fluctuations in polyamine levels, increases in putrescine and decreases in spermidine, and the ability of exogenous polyamines to reverse the growth effects indicated that CHA caused specific enzyme inhibition and not non specific toxic effects (Pfaller *et al.* 1988, 1990). A sequential treatment of DFMO followed by CHA in *C. albicans* proved to inhibit growth to a greater extent than CHA alone (Pfaller *et al.* 1988).

As well as pathogenic yeasts, inhibitors of polyamine biosynthesis have been studied against other animal-infecting fungal species. Growth of six species each of *Microsporum* and *Trichophyton* dermatophytic fungi were inhibited by DFMO and DFMA. *Trichophyton* species were generally more sensitive than *Microsporum* species and indeed, lower concentrations of DFMA compared to DFMO were required to inhibit growth (Boyle *et al.* 1988). The inhibitory activity of DFMA was thought to be the result of conversion of the compound to DFMO by arginase and the increased efficacy of DFMA over DFMO, due to increased uptake of DFMA (Boyle *et al.* 1988). Thus, Δ MFMO.CH₃, known to be taken up faster than DFMO by parasites (Bacchi *et al.* 1979) and mammalian cells (Mamont *et al.* 1986), was found to be more effective at inhibiting growth of the fungal species than DFMO (Boyle *et al.* 1988). The inhibition of putrescine synthesis in the fungi species was shown to be specific, being reversed by exogenous putrescine (Boyle *et al.* 1988).

Mycelial morphology was dramatically altered in the presence of DFMO or DFMA. Abnormal hyphae and swollen cells were characteristics of the fungal species observed after treatment with the inhibitors (Boyle *et al.* 1988).

In later work, DFMO and DFMA inhibited ten further dermatophytic fungal species and while DFMO-induced growth reductions were associated with decreased ODC activity and

reductions in polyamine levels, the mechanism of DFMA growth inhibition was unclear. ADC activity was not detected, nor arginase-mediated conversion of DFMA to DFMO (Gruhn & Boyle 1991).

1.1.6.4 Insecticides

Polyamine biosynthesis inhibitors have been suggested to act as novel insecticides. Insects synthesise putrescine solely via the ODC route while plants possess both the ODC and the ADC routes for polyamine formation, thus specific control of insects on crop plants should be possible (Rajam 1991). DFMO, MGBG and bis(cyclohexylammonium) sulphate all exhibited insecticidal activity against the tobacco caterpillar (*Spodoptera litura*) when larvae were allowed to feed on treated leaf discs. MGBG was the most effective compound in terms of mortality and reducing body weight of surviving larvae (Rajam 1991).

1.1.7 FUNGAL RESEARCH

As discussed in section 1.1.3.1, page 10, most fungi form putrescine from ornithine, in a reaction catalysed by ODC. Plants, however, can synthesise putrescine both via ODC and from arginine in a series of reaction steps, initially catalysed by ADC. Thus, specific control of fungal growth on plants should be possible by inhibiting ODC. Since plants have an alternative pathway to putrescine, they should not be adversely affected by ODC enzyme inhibition. It was this theory that instigated research into inhibition of polyamine biosynthesis in fungi. Such research will now be reviewed, begining with inhibition of ODC.

1.1.8.1 Inhibitors of ODC

Rajam & Galston (1985) initially studied the potential of polyamine biosynthesis inhibitors to control plant pathogenic fungi by examining the effects of DFMO on mycelial growth of *Botrytis sp, Botrytis cinerea, Rhizoctonia solani* and *Monilinia fructicola*. DFMO strongly inhibited growth of all the fungal species examined, with the highest dose of 1.0 mM almost completely depressing growth. *Botrytis sp* was the most sensitive species with regards to the effects of DFMO, differences being apparent between the responses of the different species to the inhibitor (Rajam & Galston 1985).

Later work by Birecka and co-workers (1986) not only demonstrated that DFMO inhibited mycelial growth of *Helminthosporium maydis*, but also noted that differences were apparent between sensitivity of different fungal species towards DFMO. These authors speculated that the differences may be due to variations in inhibitor uptake and distribution within the cell, enzyme sensitivity towards the inhibitor and fungal polyamine requirements (Birecka *et al.* 1986).

Fungal growth of *Botrytis sp* and *M. fructicola* on media amended with DFMO plus polyamines was greater than that growth on medium amended with only DFMO, suggesting that DFMO was indeed affecting growth through the depletion of polyamines (Rajam & Galston 1985). These findings were also observed for *H. maydis* when the inclusion of putrescine into the media prevented DFMO-induced inhibition of growth (Birecka *et al.* 1986).

Since then, there have been many reports of DFMO inhibiting mycelial growth of various phytopathogenic fungi on solid media. West & Walters (1989) demonstrated that 4.0 mM DFMO reduced growth of 'grey' *Pyrenophora teres* and *Gaeumannomyces graminis*. Culturing of 'grey' *P. teres* produced stable 'orange' variants and thus, both such variants were studied by these researchers. Smaller growth reductions of 'orange' *P. teres* and

Fusarium culmorum were obtained with DFMO while no inhibitory effect was observed against *Septoria nodorum*, indeed a 2.0 mM concentration of DFMO increased growth of that fungus (West & Walters 1989). Additional ODC inhibitors were examined; Δ MFMO, Δ MFMO.CH₃ and RR-MAP were less effective than DFMO in inhibiting growth of the various fungi. RR-MAP actually increased growth of 'grey' *P. teres* (West & Walters 1989). The addition of polyamines to DFMO amended media resulted in increased mycelial growth compared to DFMO treatment alone. Such a restoration of growth indicated that DFMO did indeed affect growth by reducing polyamine levels in the fungal species (West & Walters 1989).

Thus, the work of these authors agreed with the findings of Birecka *et al.* (1986) whereby differences in sensitivity of fungal species towards inhibitors was demonstrated. Indeed, West & Walters (1989) took these findings further with mutant variants of the same fungus, *P. teres*, displaying varying effects towards inhibitors. Interestingly, reduced efficacy of DFMO towards *F. culmorum* was later observed not to be a consequence of reduced inhibitor uptake (West & Walters 1991a).

Various growth increases observed by enzyme inhibitors may have been the result of compensatory increases in non-target enzyme activities (West & Walters 1989). As discussed in section 1.1.3.2, page 12, organisms possess a variety of means enabling them to overcome fluctuations in their intracellular polyamine concentrations, including enzyme activity increases (Jäme *et al.* 1985).

Mycelial growth of various other pathogens has been inhibited by DFMO. Mussell and coworkers (1987) reported the inhibition of *Verticillium dahliae* at concentrations as low as 5μ M. The wheat pathogen *Neovossia indica* (Singh *et al.* 1989), *Botrytis cinerea* (Smith *et al.* 1990b), *Pyrenophora avenae* (Foster & Walters 1990) and *Ophiostoma ulmi* (Biondi *et al.* 1993) were all inhibited on solid media by DFMO. Khan & Minocha (1989a) observed DFMO-induced inhibition of *Helminthosporium carbonum* and *Fusarium oxysporum*. Inhibition was reversed in all these cases by the inclusion of polyamines into the medium.

More recently, Bharti & Rajam (1996) demonstrated inhibition by DFMO of mycelial growth of four plant pathogenic fungal species; *Helminthosporium oryzae* (causing leaf spot of rice), *Curvularia lunata* (black rot of chilli), *Pythium aphanidermatum* (damping off disease in chilli and pea) and *Colletotrichum capsici* (fruit rot of chilli). *H. oryzae* and *C. lunata* were more sensitive towards DFMO than the other fungi and all instances of growth inhibition were prevented by the addition of putrescine to inhibitor amended media (Bharti & Rajam 1996).

Other ODC inhibitors have also demonstrated inhibition of mycelial growth of fungi; growth of *B. cinerea* was reduced by Δ MFMO (Smith *et al.* 1990a) while Δ MFMO and RR-MAP reduced growth of *Phytophthora infestans* (Barker *et al.* 1993).

Mycelial morphology and cell size were dramatically altered by DFMO in *Botrytis sp* and *M. fructicola* where fewer hyphae resulted and fungal cell lengths were reduced (Rajam & Galston 1985). West & Walters (1989) observed varying effects on cell length caused by ODC inhibitors when exposed to fungi growing on solid media. DFMO reduced cell lengths of *P. teres* and *G. graminis* but increased those of *S. nodorum* and *F. culmorum*. The other ODC inhibitors examined, Δ MFMO, Δ MFMO.CH₃ and RR-MAP increased fungal cell lengths of *S. nodorum* and *F. culmorum* and reduced those of *P. teres*. Interestingly, mycelial growth of *G. graminis* was relatively sensitive towards the inhibitors yet only Δ MFMO affected its cell lengths, increasing them (West & Walters 1989). DFMO reduced cell lengths of *C. ulmi* (Garcia *et al.* 1991) while, in contrast, the inhibitor failed to alter cell lengths in *P. avenae* (Foster & Walters 1990).

It has already been noted that growth of some fungal species is unaffected by DFMO and other ODC inhibitors (for example, West & Walters (1989)). Other such instances of insensitivity towards inhibitors have been reported. DFMO failed to affect mycelial growth of Pythium ultimum at a concentration of 5.0 mM (Walters et al. 1995b). This result was interesting given that growth of the related fungus, Phytophthora infestans, was well controlled by the inhibitor (Walters et al. 1995b). DFMO reduced polyamine concentrations and ODC activity in P. infestans but failed to affect such enzyme activity and polyamine levels in P. ultimum when each fungus was exposed to the inhibitor in liquid culture. In vitro sensitivity of ODC from the two fungi to DFMO differed, with enzyme from P. infestans being more sensitive than that from P. ultimum. Inhibitor uptake was also greater in P. infestans compared to P. ultimum (Walters et al. 1995b). Barker et al. (1993) also reported differences in responses to DFMO by related fungi; growth of P. infestans was reduced while that of Phytophthora cactorum was unaffected. These authors speculated that the difference in growth inhibition was due to the lower amounts of polyamines found in P. infestans. Enzyme activity in the fungus was therefore thought to be lower and thus more sensitive towards inhibitors (Barker et al. 1993). Similarly, growth of both Septoria tritici and Ustilago maydis was insensitive towards DFMO and other ODC inhibitors. These findings, however, were unusual since exposure of the fungi to DFMO reduced ODC activity and polyamine levels (Smith et al. 1992). One possible explanation for continued growth in the presence of inhibition of polyamine biosynthesis would be the availability of an alternative pathway for putrescine synthesis (Smith et al. 1992).

Havis & Walters (1992b) reported how activity of ODC from *Pyrenophora avenae*, *Pyricularia oryzae* and *Uromyces viciae-fabae* was unaffected by 1.0 mM DFMO. Activity of ODC from *P. avenae* could be reduced by 5.0 mM DFMO and was also inhibited by MFMO. RR-MAP, however, did not affect activity of the fungal enzyme (Havis & Walters 1992b). Such work demonstrated that enzymes from different organisms vary in their sensitivity towards inhibitors (Havis & Walters 1992b).

Nevertheless, in addition to controlling plant pathogenic fungi grown using artificial media, ODC inhibitors have been shown to inhibit fungal infections on various plants. Rajam *et al.* (1985) initiated such studies by examining the effect of DFMO on rust infection (*Uromyces phaseoli*) of french beans (*Phaseolus vulgaris*). Complete protection was afforded by concentrations of DFMO of 0.5 mM and higher. At lower concentrations, post inoculation treatments were more effective than pre inoculation sprays (Rajam *et al.* 1985). DFMO also conferred protection to unsprayed parts of plants, demonstrating an ability to be translocated (Rajam *et al.* 1985). In further studies, complete inhibition of rust was achieved by 0.5 mM DFMO when it was applied up to three days post inoculation, while treatment 4 or 5 days after inoculation only resulted in partial pathogen control (Rajam *et al.* 1986).

Similarly, Walters (1986) found that 0.4 mM DFMO gave good control of rust (*Uromyces viciae-fabae*) infection on broad bean (*Vicia faba*) with post inoculation sprays being more effective than pre inoculation treatments. The translocational ability of DFMO was demonstrated when untreated leaf halves were protected from infection (Walters 1986).

Such translocational ability of DFMO was later observed against powdery mildew on barley seedlings when inhibitor applied to plant roots and other leaves reduced disease infection (Walters & Kingham 1990).

DFMO demonstrated good control of *Verticillium dahliae*, the cause of Verticillium wilt, a vascular disease of tomato, by post inoculation applications of the inhibitor (Mussell *et al.* 1987). Good control of *Botrytis cinerea* on various plants, for example, tomato and pepper, was afforded by concentrations of DFMO as low as 0.01 mM (Elad 1991).

Three biotrophic fungal pathogens, *Puccinia recondita* (leaf rust), *P. graminis* (stem rust) and *Erysiphe graminis* (powdery mildew) on wheat were controlled by DFMO at relatively low concentrations (Weinstein *et al.* 1987). In general, inhibitor application after inoculation with the pathogen was more effective than a pre inoculation treatment (Weinstein *et al.* 1987). Foster & Walters (1992a) also reported how DFMO reduced stem

rust infection of wheat. However, pre inoculation treatments were more effective than post inoculation sprays in this instance (Foster & Walters 1992a).

Powdery mildew infection on barley was reduced by a range of ODC inhibitors; DFMO, Δ MFMO, Δ MFMO.CH₃, RR-MAP and α -hydrazinoornithine in glasshouse experiments, with post inoculation treatments generally being more effective than pre inoculation sprays (West & Walters 1988). Control of powdery mildew on barley in the field by DFMO was demonstrated by Havis & Walters (1992a). Here, DFMO controlled early season powdery mildew infection as well as a commercial fungicide and although late season control was not as good, grain weights were increased compared to untreated controls (Havis & Walters 1992a).

While agreement exists as to the effectiveness of DFMO and ODC inhibitors against plant pathogens, disagreements are apparent with regards to the optimum timing of application. Foster & Walters (1992a) indicated that pre inoculation treatments were more effective while various other workers suggested a preference for post inoculation treatments (Rajam *et al.* 1985; Walters 1986; Weinstein *et al.* 1987). Weinstein *et al.* (1987) suggested that DFMO applied pre inoculation could enter the leaf and react with plant ODC enzyme, rendering the inhibitor unavailable to act upon fungal ODC. After observing DFMO to possess systemic activity, Walters (1986) suggested that the inhibitor may require to be applied prior to inoculation to allow time for translocation of the compound to untreated parts of the plant. Indeed, it may be translocation of DFMO, a metabolite of the inhibitor or an induced antifungal compound which is required for fungicidal activity (Rajam *et al.* 1985). Alternatively, post inoculation treatments may be more effective were DFMO to act upon early germling development of the fungus on the leaf surface, as suggested by Walters (1995).

Indeed, DFMO inhibited the germination of rust uredospores in a culture medium. In addition, spore germination was delayed and germ tube growth restricted (Rajam *et al.* 1989). Spore germination on plants was completely inhibited by 0.5 mM DFMO applied 1 day before inoculation, while application 1 day after inoculation caused greater damage to the uredosporelings (Rajam *et al.* 1989). DFMO inhibited the germination of *C. ulmi* spores *in vitro* (García *et al.* 1991) and inhibited sporidial formation of *Neovossia indica* in culture (Singh *et al.* 1989). Spore germination and hyphal growth of the mycorrhizal fungus *Glomus mosseae* were both reduced by DFMO (El Ghachtouli *et al.* 1996).

In contrast, DFMO had little effect on germination and germ tube growth of wheat stem rust fungal spores on plants although production of spores was reduced (Machatschke *et al.* 1990). These authors speculated that enough endogenous polyamines were present in

37

ALL STORES

162469

 $\mathcal{Z}_{\mathcal{O}}^{(i)}$

spores to support germination and growth for some time, thus accounting for the lack of inhibitor effects initially observed (Machatschke *et al.* 1990). Germination of wheat bunt fungal spores *in vitro* was unaffected by the inhibitor, however, DFMO did inhibit hyphal growth (Trione *et al.* 1988). Similarly, Reitz *et al.* (1995a) found that DFMO did not affect germination of rust uredospores on artificial membranes, however, the inhibitor did affect germling development by reducing appressorium formation almost completely when used at a concentration of 2.0 mM.

Anomalies exist between fungal growth on plants and that in culture; fungi on plants are exposed to inhibitors early on in their development while in culture, growing mycelium is used. Biotrophic fungi cannot be grown easily in culture and inhibitor effects on such fungi can only be suggested through studies using necrotrophic fungi grown on artificial media. Differences between *in vivo* and *in vitro* studies may exist and the exact mechanism(s) by which DFMO and other ODC inhibitors act against pathogenic fungi on plants may be unclear.

ODC inhibition has been demonstrated for some fungal species grown *in vitro*. DFMO reduced ODC and increased AdoMetDC activities of *P. avenae* whilst causing reductions in putrescine and spermidine concentrations (Foster & Walters 1990). The inhibitor reduced ODC activity in four phytopathogenic fungi, accompanied by decreases in putrescine and spermidine and an increase in spermine levels (Khan & Minocha 1989a). ODC activity in *P. infestans* was reduced by DFMO and accompanied by reductions in putrescine, spermidine and spermine (Walters *et al.* 1995b). Interestingly, cadaverine concentration decreased in *P. avenae* exposed to DFMO (Foster & Walters 1990). The exact mechanism for this reduction is unknown and it is in contrast to previous work where cadaverine actually increased in *Neurospora crassa* when polyamine depletion occurred (Paulus *et al.* 1982).

In vitro enzyme studies have also been carried out, although these may not always reflect in vivo effects on enzyme activities. Smith and co-workers (1990a) extracted ODC from *B. cinerea* and observed that DFMO did indeed inhibit the enzyme. Interestingly, RR-MAP and Δ MFMO also inhibited the enzyme well whilst they were less effective at inhibiting growth of *B. cinerea* on solid media (Smith *et al.* 1990a).

Nevertheless, given the many instances of inhibition reversal by polyamines, polyamine biosynthesis and/or function would appear to be involved in the antifungal and fungicidal effects of DFMO and other ODC inhibitors.

Only ODC inhibition can specifically inhibit polyamine biosynthesis in most species of plant pathogenic fungi since AdoMetDC, spermidine and spermine synthase are vital to both fungi and plants. However, research into inhibition of the other enzymes has been

38

1.11.20

carried out given the success of ODC inhibitors in controlling fungal growth and the possibility that such inhibitors may not wholly act through inhibition of polyamine biosynthesis but rather by disrupting function(s) of the amines. While ADC is primarily a plant enzyme, some fungi are known to possess both the ODC and ADC pathways to putrescine. Thus, research into inhibition of ADC activity has also been carried out.

1.1.8.2 Inhibitors of AdoMetDC

West & Walters (1989) observed that MGBG inhibited the growth of four necrotrophic fungi species, *P. teres, G. graminis, S. nodorum* and *F. culmorum* Indeed, MGBG was the most effective inhibitor examined, being more effective than DFMO and CHA, and causing substantial reductions in fungal growth when used at a concentration of 2.0 mM. Combination with DFMO was effective in reducing growth of all pathogens but no more so than MGBG alone (West & Walters 1989). Such combinations of inhibitors, targetting different enzymes, was proposed to be more effective in inhibiting polyamine biosynthesis than a single inhibitor, providing greater reductions of intracellular polyamines (Jänne *et al.* 1985). Indeed, mycelial growth of *B. cinerea* was reduced by MGBG and this reduction increased when the inhibitor was used along with DFMO (Smith *et al.* 1990b). Mycelial growth of *O. ulmi* was controlled successfully by a mixture of MGBG and CHA (Biondi *et al.* 1993).

Foster & Walters (1990) demonstrated that growth of *P. avenae* was inhibited by both MGBG and EMGBG. Indeed, mycelial growth was almost completely depressed by 2.0 mM MGBG. When combined with DFMO, MGBG successfully inhibited mycelial growth of *P. avenae*, although no more effectively than MGBG alone (Foster & Walters 1990). Interestingly, inhibitory effects of MGBG, EMGBG and MGBG/DFMO were not reversed by the inclusion of putrescine or spermidine in the media. However, the respiration rate of the fungus was reduced by MGBG and the mixture of MGBG and DFMO, thus, the inhibitory effects of these treatments may not have been solely due to effects on AdoMetDC (Foster & Walters 1990).

Recently, Bharti & Rajam (1996) observed that MGBG successfully inhibited mycelial growth of four pathogenic fungi. Indeed, MGBG was more effective than DFMO and CHA in controlling fungal growth of the species examined (Bharti & Rajam 1996).

Various effects on fungal cell lengths were caused by exposure to MGBG (West & Walters 1989). Cell lengths of *P. teres* and *G. graminis* were reduced while those of *F. culmorum* were increased (West & Walters 1989). Similar findings were reported for MGBG and DFMO combined (West & Walters 1989). Foster & Walters (1990) found little effect on

4.

cell size in *P. avenae* following exposure to MGBG or EMGBG. A mixture of MGBG and DFMO did, however, reduce cell lengths (Foster & Walters 1990).

Powdery mildew on barley was reduced by MGBG applied both as pre and post inoculation treatments, although post inoculation sprays were generally more effective (West & Walters 1988). A concentration of 2 mM almost completely prevented mildew infection. Mixtures of MGBG and DFMO were also effective in reducing disease and were generally more effective than either compound alone (West & Walters 1988).

ý.

£?

Foster & Walters (1992a) reported that MGBG and MGBG/DFMO controlled infection of wheat by the stem rust fungus, although they found that pre-inoculation treatments were more effective than post inoculation sprays. Mixtures of MGBG and DFMO were observed to be more consistent in reducing infection on the plants (Foster & Walters 1992a).

Thus variation exists with regard to the efficacy of pre and post applications of MGBG. As Weinstein *et al.* (1987) proposed for DFMO and ODC, pre inoculation treatments may be less effective if MGBG were to bind to plant AdoMetDC and thus render it unavailable to act upon fungal AdoMetDC. However, inhibitors may be acting upon development of the fungus on the leaf surface as suggested by Walters (1995). Indeed, Reitz *et al.* (1995a) observed that MGBG reduced germination and prevented appressorium formation of rust urcdospores on artificial membranes when used at a concentration as low as 0.025 mM.

Various effects of MGBG and EMGBG on intracellular enzyme activities and polyamine concentrations in *P. avenae* grown in liquid culture were observed by Foster & Walters (1990). MGBG reduced ODC activity while EMGBG had no effect on this enzyme. AdoMetDC activity was substantially reduced by EMGBG but only slightly by MGBG. Both EMGBG and MGBG reduced spermidine concentrations while MGBG also reduced spermine. A mixture of MGBG/DFMO reduced ODC activity while no effect on AdoMetDC was observed. Putrescine and spermidine concentrations were reduced and spermine remained unaltered (Foster & Walters 1990). Interestingly, cadaverine was also reduced by MGBG, EMGBG and the mixture of MGBG/DFMO (Foster & Walters 1990).

Further work on inhibitors of AdoMetDC was carried out by Walters and co-workers (1995a). The trypanocidal agents Berenil and pentamidine, shown to irreversibly inhibit AdoMetDC from rat liver, yeast, *Escherichia coli* and *Trypanosoma brucei brucei* (Bitonti *et al.* 1986b), reduced mycelial growth of *Pyricularia oryzae* and infection of barley powdery mildew. However, activity of AdoMetDC from fungus was inhibited by the compounds yet the enzyme in fungus exposed to the compounds was unaffected (Walters *et*

40

al. 1995a). Thus, in vivo and in vitro enzyme studies may not be comparable. Putrescine levels increased while spermidine was unaltered and spermine levels reduced. The authors, however, suggested that such altered polyamine concentrations were unlikely to account for the antifungal and fungicidal activity of the compounds observed (Walters *et al.* 1995a). Interestingly, cadaverine concentration of *P. oryzae* was greatly increased upon exposure of the fungus to both Berenil and pentamidine, however, the reason for such an increase without considerable depletion of polyamines in unknown (Walters *et al.* 1995a). Instances of cadaverine and derivative accumulation in response to polyamine depletion have recently been reported by Zarb & Walters (1994b).

1.1.8.3 Inhibitors of spermidine synthase

West & Walters (1989) reported how the spermidine synthase inhibitor, CHA, inhibited mycelial growth of *G. graminis*, with a concentration of 4.0 mM reducing growth by 80%. Little effect was however observed for *P. teres*, *S. nodorum* and *F. culmorum* (West & Walters 1989). Cell lengths of *P. teres* were reduced in the presence of CHA while those of *S. nodorum* were unaltered (West & Walters 1989). Recently, Bharti & Rajam (1996) reported inhibition of four pathogenic fungal species by bis-(cyclohexylammonium) sulphate (BCHA). CHA failed to affect mycelial growth of *B. cinerea* when used at a concentration of 1.0 mM (Smith *et al.* 1990b). Interestingly though, the compound acted as a synergist for DFMO mediated inhibition, increasing control of growth afforded by DFMO alone (Smith *et al.* 1990b).

Powdery mildew on barley was inhibited by CHA in pre and post inoculation treatments, although post inoculation treatments were generally more effective (West & Walters 1988). The mechanisms by which CHA affects fungal growth on plants has not been studied in any great detail, but Reitz *et al.* (1995a) found that rust uredospore germination was not affected although appressorium formation was reduced on artificial membranes by a concentration of 3.0 mM CHA.

1.1.8.4 Inhibitors of spermine synthase

No specific inhibitors of spermine synthase have been examined as potential inhibitors of plant pathogenic fungi. Some fungi are indeed capable of synthesising spermine, however, in other species, spermine is either absent, or present in only very low quantities (Stevens & Winther 1979).

1.1.8.5 Inhibitors of ADC

Rajam & Galston (1985) studied the effects of the ADC inhibitor DFMA on mycelial growth of four plant pathogenic fungal species. DFMA strongly inhibited growth of

Botrytis sp, B. cinerea, Rhizoctonia solani and Monilinia fructicola and was indeed, observed to be more effective than DFMO. Inclusion of polyamines with the inhibitor reversed the inhibition, suggesting that DFMA affected growth through an effect on polyamine levels.

Inhibition of mycelial growth of various other fungal species has been reported to have been caused by DFMA including *Helminthosporium carbonum*, *Fusarium oxysporum* and *Ceratocystis ulmi* (Khan & Minocha 1989a). In general, inhibition of growth of these fungal species was less pronounced with DFMA compared to DFMO. The inclusion of polyamines into media amended with DFMA reversed the growth inhibition and indicated that DFMA was acting via the depletion of polyamines (Khan & Minocha 1989a). Growth of *Ophiostoma ulmi* was inhibited by DFMA, DFMA being more effective than DFMO in this instance (Biondi *et al.* 1993). DFMA produced varying results on mycelial growth of four species of pathogenic fungi examined by Bharti & Rajam (1996). *Helminthosporium oryzae* and *Curvularia lunata* were inhibited while growth of *Pythium aphanidermatum* and *Colletotrichum capsici* was stimulated (Bharti & Rajam 1996). The inhibitor was ineffective against mycelial growth of *Neovossia indica* (Singh *et al.* 1989) and *Helminthosporium maydis* (Khan & Minocha 1989a).

Mycelial growth of *Botrytis cinerea* was reduced by DFMA, however, genuine ADC activity could not be detected in *B. cinerea* (Smith *et al.* 1990a,b). Since the inhibitor had no effect on activity of ODC extracted from the fungus, the growth inhibitory effect of DFMA was thought to be due to its conversion to DFMO by fungal arginase (Smith 1990a,b), as suggested by Rajam & Galston (1985). Nevertheless, genuine ADC activity has been reported to be predominant in *O. ulmi* (Biondi *et al.* 1993) and *C. ulmi* (Khan & Minocha 1989a,b). Generic differences between fungi may be responsible for the varying effects on growth observed for DFMA and DFMO, as suggested by Birecka *et al.* (1986). Indeed, instances of increased efficacy of DFMA compared to DFMO have been implicated with increased uptake of the compound (Boyle *et al.* 1988).

Mycelial morphology and cell size were dramatically altered by DFMA in *Botrytis sp* and *M. fructicola* with numbers of hyphae reduced and fungal cell lengths decreased (Rajam & Galston 1985).

DFMA did not offer protection against rust on french beans even at a concentration of 5.0 mM (Rajam *et al.* 1985). The inhibitor did, however, control stem rust and leaf rust on wheat plants albeit not as effectively as DFMO (Weinstein *et al.* 1987). Powdery mildew on barley was reduced by DFMA both pre and post inoculation, with pre inoculation

treatments being more effective (West & Walters 1988). The mechanisms by which DFMA affects fungi on leaf surfaces have not been studied in any detail. However, DFMA inhibited the germination of rust uredospores in a culture medium, delayed spore germination and restricted germ tube growth (Rajam *et al.* 1989). Also, DFMA had no effect on the germination of wheat bunt fungal spores *in vitro* but inhibited hyphal growth (Trione *et al.* 1988).

1.1.8.6 Polyamine analogues

Recent studies on the control of plant pathogenic fungi through the perturbation of polyamine biosynthesis have examined the potential of polyamine analogues as antifungal and fungicidal agents. As discussed previously (section 1.1.4.6, page 20), several reasons exist to suggest that polyamine analogues may be more effective in inhibiting polyamine biosynthetic enzymes than specific enzyme inhibitors.

ļ

Foster & Walters (1993) reported how the putrescine analogue, keto-putrescine, provided control of a number of plant pathogens. Powdery mildew on barley, rust and chocolate spot on broad bean, brown rust on barley, powdery mildew on apple and late blight on potato were all effectively controlled by the analogue. Pre and post inoculation sprays were both effective, however, post inoculation treatments tended to afford greater disease control. Keto-putrescine was most effective against chocolate spot infection of broad bean (Foster & Walters 1993).

The analogue reduced *in vitro* growth of *P. infestans, P. avenae* and *P. oryzae*, although it was less effective against *in vitro* growth than fungal infections on plants (Foster & Walters 1993). Indeed, keto-putrescine was relatively poor at inhibiting mycelial growth of *B. cinerea* (Smith *et al.* 1990b). Although such phenomena cannot be directly correlated, germination of *Aspergillus nidulans* was delayed by keto-putrescine in previous work (Stevens *et al.* 1977), suggesting that the analogue might indeed affect fungal development on the leaf surface, and be less effective against growing mycelium (Foster & Walters 1993). Alternatively, greater uptake and accumulation of the compound by a developing fungus could lead to greater effects *in vivo* as oppose to effects on mycelium (Foster & Walters 1993)

Interestingly, another putrescine analogue, N-acetylputrescine, did not affect barley powdery mildew infection of plants, nor *in vitro* growth of pathogens examined (Foster & Walters 1993). Similarly, no effect of the analogue on growth of *B. cinerea* was observed by Smith and co-workers (1990b). However, the lack of fungicidal and antifungal activity

reported may have been due to a deacetylase, found in some fungi, which can metabolise N-acetylputrescine (Haywood & Large 1986).

Both ODC and AdoMetDC activities in *P. avenae* were reduced when exposed to 1.0 mM keto-putrescine. Spermidine levels were decreased, spermine levels increased and fungal putrescine concentration remained unaffected (Foster & Walters 1993). Interestingly, Stevens and co-workers (1977) exposed *A. nidulans* to keto-putrescine and observed ODC activity to increase. This may have been due to enzyme stabilisation. An increase in putrescine concentration was also observed along with reductions in spermidine and spermine levels (Stevens *et al.* 1977). Clearly, the analogue has differing effects on polyamine biosynthesis in different fungal species. Keto-putrescine also caused a substantial increase in the concentration of cadaverine in *P. avenae* (Foster & Walters 1993). As discussed earlier, increased cadaverine concentrations have been observed to occur in response to polyamine depletion (Paulus *et al.* 1982; Zarb & Walters 1994b). Foster & Walters (1993) noted how the increased cadaverine levels could support fungal growth in the absence of spermidine.

Keto-putrescine was relatively successful as a fungicidal and antifungal agent. Varying effects on fungal polyamine biosynthesis have been reported and therefore, no conclusions with regards to the mode(s) of action of the analogue can be made. Nevertheless, novel putrescine analogues were synthesised in an attempt to control fungi via a perturbation of polyamine biosynthesis. Good control of a number of plant pathogen interactions; powdery mildew on barley, rust and chocolate spot on broad bean, powdery mildew on apple and late blight on potato, were afforded by (E)-1,4-diaminobut-2-ene (E-BED) and the cis isomer, Z-BED (Havis et al. 1994a). E-BED was generally more effective than Z-BED and while pre and post inoculation sprays both gave good control of infections, post inoculation treatments were more effective (Havis et al. 1994a). Best control of plant infections was observed against chocolate spot on broad bean with 1.0 mM E-BED (Havis et al. 1994a). E-BED also demonstrated good early season control of powdery mildew on barley in the field, being as effective as a commercial fungicide, and although later season control was not as effective, an equivalent increase in grain yield compared to untreated controls was observed (Havis et al. 1994b). (E)-N,N,N',N'-tetracthyl-1,4-diaminobut-2-ene (E-TED) and derivatives similarly controlled plant pathogenic fungal infections well (Havis et al. 1994c). As with the other analogues, post inoculation sprays were most effective and best control was achieved by E-TED against barley powdery mildew (Havis et al. 1994c).

Reitz et al. (1995b) observed that E-BED and E-TED reduced germination and appressorium formation of rust uredospores on artificial membranes. Such a result offers

44

evidence that, as suggested for keto-putrescine, the putrescine analogues affect fungal development on the leaf surface. Similarly, a range of polyamine biosynthesis inhibitors have been shown to inhibit conidial germination, germ tube growth and haustorial development of barley powdery mildew (S. Hannif & D. R. Walters unpublished results).

E-BED and Z-BED reduced mycelial growth of *B. cinerea* and *P. oryzae* and growth of *P. avenae* in liquid culture (Havis *et al.* 1994a). Both compounds displayed similar efficacy towards *P. oryzae*, however, E-BED was the more effective of the compounds against *B. cinerea* and *P. avenae* (Havis *et al.* 1994a). Indeed, *B. cinerea* and *P. avenae* were more sensitive towards the analogues than *P. oryzae* (Havis *et al.* 1994a).

 $\sum_{i=1}^{n-1} \frac{1}{i} = \frac{1}{i}$

E-TED completely inhibited growth of *P. avenae* in liquid culture, at a concentration of 1.0 mM. That same concentration, however, failed to affect mycelial growth of the fungus on solid media (Havis *et al.* 1994c).

ODC and AdoMetDC activities of *P. avenae* exposed to E-BED were substantially reduced. Putrescine and spermine concentrations were increased while spermidine levels decreased (Havis *et al.* 1994a). In contrast, Z-BED reduced ODC activity slightly while AdoMetDC activity was increased. Spermidine concentrations were reduced but putrescine and spermine levels remained unchanged (Havis *et al.* 1994a).

E-TED increased ODC activity in *P. avenue* while AdoMctDC activity was reduced. Putrescine and spermidine concentrations decreased while spermine levels were unaltered (Havis *et al.* 1994c).

A common effect in fungus exposed to all analogues described above was the reduction in spermidine levels. Since spermidine is the predominant polyamine in fungi (Stevens & Winther 1979), this may have been responsible for the observed antifungal effects. However, as substantial pools of spermidine generally remained, perturbation of polyamine biosynthesis was suggested by the authors not to be the major inhibitory mechanism of these putrescine analogues (Havis *et al.* 1994a,c). Often, an increase in putrescine was observed. Whatever the cause of this increase, it may have been responsible for the antifungal effects observed (Havis *et al.* 1994a).

More recent work has shown that E-BED inhibits DNA methylation (Walters 1997), and that it also causes a very substantial decrease in uptake of amino acids, polyamines and carbohydrates in *P. avenae* (Walters & McPherson 1997). Havis and co-workers (1996b) also reported how E-TED increased spermine concentration in *P. infestans* and suggested

the spermine was displaced from intracellular binding sites. Exactly how these effects of E-BED and E-TED relate to their antifungal activity is unknown at present.

Following studies on aliphatic diamines as analogues of putrescine, cyclic diamines were synthesised and tested for fungicidal and antifungal effects on plant pathogenic fungi. 1,2bis(aminomethyl)-4,5-dimethylcyclohexa-1,4-diene (BAD) and various derivatives controlled powdery mildew on barley with post inoculation treatments providing greater control over pre inoculation treatments (Havis *et al.* 1996a). As with the aliphatic diamines, differences in efficacy were apparent between cis and trans isomers, trans-BAD being more effective against powdery mildew of barley than cis-BAD (Havis *et al.* 1996a). Unlike the aliphatic diamines, BAD and derivatives did not affect growth or polyamine biosynthesis in *P. avenae* to any great extent (Havis *et al.* 1996a).

Additional cyclic diamines were synthesised and their fungicidal activity tested against powdery mildew on barley. Trans-1,2-bis(diethylaminomethyl)cyclopentane (compound 1), trans-5,6-bis(aminomethyl)bicyclo[2.2,1]hept-2-ene (compound 2) and 1,2-bis(dimethylaminomethyl)-4,5-dimethylcyclohexa-1,4-diene (compound 3) afforded good control of powdery mildew infection with post inoculation treatments being more effective than pre inoculation treatments (Havis et al. 1997). Only limited activity was observed against chocolate spot and rust of broad bean (Havis et al. 1997). All three compounds reduced growth of P, avenae in vitro. However, differing effects were observed on polyamine levels in fungus exposed to the compounds. Compound 1 reduced levels of putrescine, spermidine and spermine, decreased ODC activity and slightly increased AdoMetDC activity. Compound 2 increased spermidine levels and reduced putrescine and spermine concentrations. ODC activity was increased while AdoMetDC activity was unaltered. Compound 3 reduced putrescine levels while spermidine and spermine concentrations were unaltered. ODC activity was reduced only slightly while AdoMetDC activity increased substantially (Havis et al. 1997). Such changes in polyamine levels could not be explained by the corresponding alterations in biosynthetic enzyme activities. Thus, as with the aliphatic diamines, the fungicidal and antifungal effects of these cyclic diamines is unlikely to have been wholly caused by an effect on polyamine metabolism (Havis et al. 1997).

Fungicidal activity of spermidine analogues has been briefly studied. Powdery mildew on barley was controlled in glasshouse experiments by 1.0 mM concentrations of N¹- and N⁸- acetylspermidine. Mycelial growth of *P. avenae* and *P. oryzae*, however, was largely unaffected by 1.0 mM concentrations of the compounds (S. A. Foster unpublished results). Two novel spermidine analogues, C59 and C73, also gave good control of barley powdery

mildew, rust on broad bean and powdery mildew on apple when applied as post inoculation treatments in glasshouse experiments (N. D. Havis unpublished results).

. . . .

Ŕ

Ĵ

۰,

1.1.8 FUNGAL RESEARCH - FURTHER CONSIDERATIONS

1.1.8.1 Effects of polyamine biosynthetic inhibitors on plants

Specific control of fungal growth via inhibition of polyamine biosynthesis should, in theory, not affect plants. Nevertheless, research into the effects of polyamine biosynthesis inhibitors on plants has been carried out, being necessary if inhibitors are to be used against fungi growing on plants.

In attempting to control rust on broad beans, Walters (1986), observed that no effect on plant growth nor polyamine levels of beans was caused by DFMO application. However, Rajam and co-workers (1986) demonstrated that putrescine and spermidine levels increased in french bean plants treated with DFMO. This increase was suggested to be the result of stimulated plant ADC activity in response to ODC inhibition (Rajam et al. 1986). Later, Rajam et al. (1991) examined the effects of DFMO and DFMA on endogenous polyamine levels in bean plants and although polyamine concentrations were slightly affected, growth continued unabated (Rajam et al. 1991). Application of DFMO, DFMA, MGBG and CHA, alone and in combinations did not significantly alter polyamine levels or affect growth of tomato seedlings (Felix & Harr 1989). Interestingly, the rate of photosynthesis of detached barley leaves decreased after only a 30 minute exposure to 1.0 mM DFMO. However, the return to near normal rates after 12 hours of exposure to the chemical indicates that the effect of DFMO on plant growth should be minimal (Coghlan & Walters 1992). Recently, MGBG and BCHA were shown to be non-phytotoxic towards wheat seedlings (Bharti & Rajam 1995). Thus, the potential of inhibitors to control plant pathogenic fungi is recognised as being viable.

1.1.8.2 Effects of polyamine biosynthetic inhibitors on mycorrhizal fungi

Mycorrhizal fungi are intimately associated with most plants and if polyamine biosynthesis inhibitors are to be used to control pathogenic infections, consideration must be given to their effects on such fungi.

Research has shown that different mycorrhizal fungi exhibit quite different responses when exposed to polyamine biosynthetic inhibitors *in vitro*. Zarb & Walters (1994c) found that mycelial growth of *Laccaria proxima* was only slightly reduced by DFMO yet slightly increased by DFMA. However, growth of *Paxillus involutus* was substantially reduced by DFMO (Zarb & Walters 1994a). Mycelial growth of *Crinipellis perniciosa* was reduced by both DFMO and DFMA as well as the ODC inhibitors MFMO.CH₃ and RR-MAP (Zarb & Walters 1993). DFMO also reduced mycelial growth of *Thelephora terrestris*, yet had no effect on *Hebeloma mesophaeum* (Zarb 1995). Varying effects on polyamine biosynthetic enzyme activities and polyamine concentrations were observed in the different fungal
species (Zarb & Walters 1993, 1994a,c; Zarb, 1995). Interestingly, in *P. involutus* and *L. proxima*, DFMO was suggested to affect cellular processes other than polyamine biosynthesis, for example, transmethylation reactions and amino acid synthesis (Zarb & Walters 1994a,c). Although different mycorrhizal fungi differ in their responses to polyamine biosynthetic inhibitors *in vitro*, the effect of such inhibitors on mycorrhizal infection and establishment on plant roots is not known and would have to be examined in detail should an effective polyamine biosynthetic inhibitor suggest itself for commercial use.

1.1.8.3 Control of plant pathogenic fungi - an overview

Even with effective disease control, around 12 % of crop yield potential is lost to diseases (Agrios 1988). Many methods are used to control plant disease including cultural measures and the use of resistant varieties. Fungicide use is, however, paramount.

In the UK, cereal diseases are the most important plant diseases which require consideration. These have been well controlled in the past by morpholines and triazoles, compounds which inhibit ergosterol biosynthesis. Ergosterol is a vital component of fungal membranes and its complicated biosynthetic pathway presents many potential enzymes for targetted inhibition. Morpholines inhibit biosynthesis at two particular sites, namely prevention of $\Delta^{7-}\Delta^{8}$ isomerisation and Δ^{14} reduction while triazoles inhibit the C14 demethylation step. Such specific inhibition has, however, allowed resistant pathogens to evolve and consequently, the fungicides tend to be used in mixtures (Hassall 1990).

Recent developments in fungicide research include the use of new compounds known as strobilurins which are synthetic analogues of naturally occuring antifungal compounds produced by the basidiomycete fungus *Strobilurus tenacellus*. Such compounds act by inhibiting the respiratory process of fungi. Two molecules are currently marketed, kresoxim-methyl is added to fenpropimorph (Ensign, BASF), and azoxystrobin (Amistar, Zeneca) is to be commercially launched fully in 1998. These new compounds offer a wide spectrum of disease control and as natural compounds, are environmentally acceptable (Anon 1997).

Other developments include cyprodinil (Unix, Novartis), as yet unavailable in the UK but offering control of eyespot, *Septoria nodorum* and mildew on wheat. Quinoxyfen (Fortress, DowElanco) is a quinolene and offers protectant action against mildew. Finally, Novartis are awaiting approval of their Plant Activator SAR which is not in actual fact a fungicide, but acts by activating the plants own resistance response (Anon 1997).

In spite of continuing developments, the number of available active ingredients for fungicide usage is decreasing. This is a result of increasing legislation with regards to toxicological and environmental considerations. Coupled with the threat of resistance, the need for novel fungicides with alternative modes of action is paramount (DeWaard *et al.* 1993).

Ì

1.2 AIM OF STUDY

The inhibition of polyamine biosynthesis has been recognised as a potential target for specific control of plant pathogenic fungi for over 10 years. During this time research has progressed and the most recent work has studied putrescine analogues in an attempt to inhibit ODC and/or affect cellular functions for which putrescine is vital.

Despite affording good control of a number of plant pathogens *in vitro* and *in vivo*, putrescine analogues failed to deplete spermidine pools completely. Since spermidine is known to be the predominant polyamine in fungi and in some species, is absolutely essential for growth, residual spermidine has been implicated in the limited efficacy of putrescine analogues. Thus, it was suggested that spermidine analogues might be more efficient at perturbing polyamine biosynthesis and/or cellular function(s) related to polyamines in fungi.

Although inhibition of spermidine synthase is not specific with regards to control of growth of pathogenic fungi on plants, putrescine analogues are now thought perhaps to afford control of fungal pathogens by mechanisms unconnected to polyamine biosynthesis. Thus, exploration of spermidine analogues as potential inhibitors of polyamine biosynthesis and/or functions in fungi on plants was warranted.

Analogues are proposed to be more effective than specific enzyme inhibitors since they may affect any function for which polyamines are essential. Spermidine synthase is not as highly regulated as ODC and AdoMetDC, thus perturbation of its activity is not likely to lead to elaborate compensatory measures within the cell.

This study was thus designed to explore fully the possibility of controlling growth of plant pathogenic fungi through the inhibition of fungal polyamine biosynthesis and/or an effect on cellular functions associated with spermidine by the use of spermidine analogues.

Chapter 2

Effect of cyclohexylamine and norspermidine on fungal infection of host plants

2. EFFECT OF CYCLOHEXYLAMINE AND NORSPERMIDINE ON FUNGAL INFECTION OF HOST PLANTS

2.1 INTRODUCTION AND OBJECTIVES

Preliminary experiments were designed to determine the fungicidal properties of two commercially available compounds: cyclohexylamine (CHA), an inhibitor of spermidine synthase and norspermidine, a spermidine analogue (Figure 4). *In vivo* effects of these compounds upon several plant pathogens were examined, in most cases investigating the protectant action of the compounds as well as their eradicant properties. Systemic activity of the compounds against one pathogen was also examined. Before detailing the experiments performed, the biology and control of each of the pathogens will be discussed briefly, along with an indication of their agricultural importance.

2.1.1 Broad bean rust

Broad beans are an important crop worldwide, providing protein for livestock feedstuffs. Production in the UK alone accounts for almost half of that in the European Union (Outsiders Guide 1995). The pathogen responsible for rust of broad bean (*Vicia faba* L.) is a basidiomycete, *Uromyces viciae-fabae* (Pers.) Schroet. In general, the family of rusts are among the most destructive of plant diseases (Agrios 1988). Rust on broad beans is usually seen late in the growing season with leaves becoming randomly covered in reddish-brown pustules, often surrounded by chlorotic tissue. Early infections can lead to defoliation of plants. The pathogen survives only on living plant material, such as volunteer beans or crop debris, primarily as uredospores. Infection is spread by air-borne spores. Disease control measures include cultural methods such as the removal of volunteer plants and crop debris. Chemical control is also available with, for example, a fungicide like fenpropimorph (Corbel, BASF) (Parry 1990).

2.1.2 Broad bean chocolate spot

Chocolate spot disease of broad bean is caused by two organisms: the deuteromycetes *Botrytis fabae* Sardina and *Botrytis cinerea* Pers. ex Fr. Together, these species are the most important pathogens of field beans. Symptoms of the disease are classified into two phases: the non-aggressive disease which can be caused by either pathogen, resulting in discrete brown lesions on leaves, stems and pods and the more aggressive phase, attributable mainly to *B. fabae* and leading to sizeable lesions and the destruction of large areas of tissue. Aggressive chocolate spot may also result in infected pods aborting, and if stems are affected, the occurrence of lodging. This phase of the disease is favoured by high humidity or prolonged rainfall. Both pathogens survive as sclerotia as well as on volunteers and green debris. Wind-borne conidia spread infection throughout crops. Chemical control



Figure 4. Structures of spermidine, CHA and norspermidine

is widely practised with foliar-applied fungicides being used at regular intervals throughout the season. Fungicides used include vinclozolin (Ronilan FL, BASF) and carbendazim + chlorothalonil (Bravocarb, ISK Biosciences) (Parry 1990).

Ĵ,

19/11/2011

2.1.3 Barley powdery mildew

Powdery mildew (Erysiphe graminis DC f.sp. hordei Marchal) of barley (Hordeum vulgare L.) is the most important cereal disease in the UK (Carlile 1995). The ascomycete is an obligately biotrophic parasite, surviving only on living plant material. Volunteers are thus sources of infection. Another great source of infection arises from autumn sown crops, spreading disease to spring sown crops. Wind-borne conidia spread infection, a process which is favoured by warm, dry conditions. Leaves are most commonly affected although symptoms of white fluffy pustules can be found on all aerial parts of the plant. Control measures include cultural methods such as the removal of volunteers and the use of resistant varieties. Resistance, however, can be short lived and growers are recommended to make use of different varieties with varying forms of genetic resistance. Market demands, however, often make such choices impractical and fungicides remain the main method of disease control. Systemic seed treatments will protect against early attacks, for example, ethirimol + flutriafol + thiabendazole (Ferrax, Bayer), while foliar triazole and morpholine fungicides are used throughout the growing season, multiple applications often being required. Such fungicides have begun to encounter resistance problems and are therefore used together in mixtures to provide maximum disease control. Examples include fenpropimorph + propiconazole (Glint 500 EC, Novartis) and fenpropimorph + prochloraz (Sprint, AgrEvo) (Parry 1990). Recent additions to the mildewicide armoury include the quinolene, quinoxyfen (Fortress, DowElanco) and the strobilurin, azoxystrobin (Amistar, Zeneca) (Anon 1997).

2.1.4 Apple powdery mildew

Worldwide, apples are economically the most important fruit grown (de Rougemont 1989). Powdery mildew is an important disease of apples and is caused by the pathogen *Podosphaera leucotricha* (Ell. & Ev.) Salm. An ascomycete, the pathogen survives as mycelium in vegetative buds. Air-borne conidia spread infection to all types of tissue. Symptoms of fine white conidia and spreading mycelium are similar in appearance to barley powdery mildew. Severe infection may result in defoliation of plants, and fruits may fail to develop (Carlile 1995). Apple varieties vary in their susceptibility to the disease, but chemical control is available, with fungicides which provide control of apple seab also affording control of powdery mildew (Toms & Dahl 1976).

2.1.5 Potato late blight

Potatoes are the most important non-cereal crop in the world. The most historically important crop disease is still the most important disease of potatoes: potato late blight, caused by the oomycete pathogen *Phytophthora infestans* (Mont) De Bary. The disease epidemics of the 1840s led to one quarter of the 8 million population of Ireland dying and another one million emigrating, mostly to Northern America, thereby altering the course of world history (Parry 1990).

The pathogen survives as mycelium in tubers, such as seed stock, groundkeepers or dumped potatoes. Diseased tubers give rise to diseased haulms and the sporangiospores produced are dispersed by rain-splash or wind currents. Haulm symptoms of brown necrotic patches are often seen at the edges or tips of leaves. Infected tubers have inward progressing regions of brown rotted tissue. Zoospores are also produced and their contact with tubers can lead to infection. Such spores can be washed downwards through the soil or can contaminate tubers during harvesting. Disease progression is favoured by warm humid conditions, however, risk periods can be successfully predicted. Cultural control measures include the elimination of groundkeepers and infected haulms at dumps and the planting of certified seed. Resistant varieties are also available. Fungicide use is, however, routine and several sprays are used throughout the season. These include protectants such as fentin hydroxide (Du-Ter 50, AgrEvo) and mixtures of a protectant and a systemic fungicide, for example, mancozeb + metalaxyl (Fubol 58 WP, Novartis) (Parry 1990).

2.2 MATERIALS AND METHODS

2.2.1 Effect of CHA and norspermidine against rust on broad bean

2.2.1.1 Growth of plant material

Broad bean seeds (cv. Threefold White) were sown individually in 9 cm pots using Fison's Levington compost. Plants were grown in a ventilated glasshouse under natural daylight supplemented to a 16 hour photoperiod with 400W mercury vapour lamps. The average maximum temperature attained during the day was 24°C with an average minimum of 9°C at night.

2.2.1.2 Maintenance of pathogen

Bean seedlings were inoculated when they were approximately 20 days old by painting the leaves with a rust spore suspension (25 mg per 100 ml distilled water), using a small haired brush. The spore suspension was prepared with 0.01 % Tween 20 (v/v) to act as a surfactant. Plants were loosely covered with clear plastic bags for 48 hours to ensure the high relative humidity necessary for spore germination. New stock plants were inoculated at approximately 4 week intervals.

2.2.1.3 Inoculation of plant material and treatment with CHA and norspermidine

Previous work demonstrated that barley powdery mildew was reduced by post inoculation applications of 0.5 mM and 1.0 mM solutions of CHA (West & Walters 1988). Thus, to examine the effects of CHA and norspermidine on the various pathogens, 0.5, 1.0 and 2.0 mM solutions of the compounds were prepared with 0.01 % Tween 20 (v/v). Pre inoculation treatments were applied three hours before inoculation while post inoculation treatments were applied 3 days after inoculation. Timings were chosen on the basis of previous work in this laboratory which found them to be satisfactory for the study of fungicidal activity of putrescine analogues (Havis *et al.* 1994a, c).

Bean seedlings were used for experimentation when they were approximately 20 days old. Plants were sprayed to run off with either CHA or norspermidine solutions using hand held compressed air spray units, either pre or post inoculation. Seedlings were inoculated by painting a suspension of rust spores (25 mg per 100 ml distilled water) onto leaves using a small haired brush before covering the plants with clear plastic bags for 48 hours. Spore suspensions were prepared with 0.01 % Tween 20 (v/v). Disease infection was assessed by estimating the percentage area of each of the third pair of leaves affected, using a standard area diagram 19 days after inoculation. Controls consisted of plants which were inoculated only and eight replicates per treatment were used. As with other experiments performed to

examine the effect of CHA and norspermidine against plant pathogens, significance was assessed using Student's *t*-test. All experiments were repeated and similar results obtained.

2.2.2 Effect of CHA and norspermidine against chocolate spot on broad bean 2.2.2.1 Growth of plant material

Beans were grown as described in section 2.2,1.1.

2.2.2.2 Maintenance of pathogen

Stock cultures of *Botrytis fabae* were maintained on sterile medium X as described by Last and Hamley (1956)¹. Culture plates were kept in a darkened incubator at a temperature of 24°C and fresh plates initiated approximately every ten days.

2.2.2.3 Inoculation of plant material and treatment with CHA and norspermidine

With a few exceptions, experiments were carried out as detailed in section 2.2.1.3. The suspension of *B. fabae* spores contained approximately $4 \ge 10^5$ conidia per ml of distilled water. Infection intensity was assessed by estimating the percentage leaf area affected using a standard area diagram 7 days after inoculation.

2.2.3 Effect of CHA and norspermidine against powdery mildew on barley

2.2.3.1 Growth of plant material

Seeds of barley (cv. Delibes) were sown in 20 cm x 15 cm trays using Fison's Levington compost and grown under the glasshouse conditions described in section 2.2.1.1.

2.2.3.2 Maintenance of pathogen

Barley seedlings were inoculated when they reached GS 13 (Zadoks growth stage: three leaves unfolded), by simply dusting with powdery mildew conidia. Fresh stock plants were initiated approximately every 4 weeks.

⁴ Composition of media, made up in 11 of distilled water, was as follows: KH_2 .PO₄ (1.52 g), MgSO₄.7H₂O (0.52 g), NaNO₃ (6 g), KCI (0.52 g), Dextrose (10 g), Peptone (2 g), Casein hydrolysate (3 g), Yeast nucleic acids (0.5 g) and Agar (20 g).

2.2.3.3 Inoculation of plant material and treatment with CHA and norspermidine

Seedlings were used for experimentation when they reached GS 13 (Zadoks growth stage: three leaves unfolded). Plants were sprayed to run off with solutions of CHA or norspermidine using hand held compressed air sprayers either pre or post inoculation. Seedlings were inoculated using the method of Nair and Ellingboe (1962). Stock plants were shaken 6 hours prior to experimental inoculation, allowing the production of fresh conidia. Such fresh conidia were shaken onto microscope slides and cotton buds used to lift the conidia and place them onto the barley leaves, in a rolling fashion. This method ensured a more controlled inoculation rather than simply dusting the plants with the powdery mildew fungus. Infection intensity was assessed 14 days later by estimating the percentage area of the third leaf affected using a standard area diagram. Ten replicates were used per treatment with controls consisting of plants which were inoculated only.

2.2.4 Effect of CHA and norspermidine against powdery mildew on apple 2.2.4.1 Growth of plant material

Seeds of apple (*Malus communis* L.) were sown in 20 cm x 15 cm trays using a 3:1 mixture of Fison's Levington compost and horticultural sand. Seed trays were covered with a black plastic bag and seeds stratified by placing in warm storage (20-25°C) for two weeks before a 10-14 week period in cold storage (0-5°C). Seed trays were then uncovered and placed in the glasshouse to germinate under the conditions described in section 2.2.1.1. Before experimentation, seedlings were transplanted to individual 9 cm pots using the same compost/sand mixture.

2.2.4.2 Maintenance of pathogen

Apple seedlings were inoculated after they had been in the glasshouse for 4-5 weeks, having developed 3-4 leaves, by simply dusting with conidia of the powdery mildew fungus. Fresh stock plants were initiated approximately every 4 weeks.

2.2.4.3 Inoculation of plant material and treatment with CHA and norspermidine

Seedlings were used for experimentation when they had developed three unfolded leaves. Poor seed germination resulted in the production of lower numbers of seedlings and thus, for experiments using apples, plants were only treated post inoculation. Post inoculation was chosen as the treatment to be used, as opposed to pre inoculation, since previous work in this laboratory had used post inoculation treatment of apples to study the fungicidal activity of putrescine analogues against powdery mildew (Havis *et al.* 1994a, c). Seedlings

were sprayed to run off with solutions of CHA or norspermidine three days after inoculation using hand held compressed air sprayers. Inoculation was carried out by brushing conidia onto the leaves using a small haired brush. Infection intensity was assessed 21 days after inoculation using an infection key, assessing the third leaf. Six replicates were used for each treatment with controls consisting of plants which were inoculated only.

2.2.5 Effect of CHA and norspermidine against late blight on potato 2.2.5.1 Growth of plant material

Potato tubers (*Solanum tuberosum* L. cv. King Edward) were sown in 15 cm pots using Fison's Levington compost. Plants were grown in the glasshouse under the conditions described in section 2.2.1.1.

2.2.5.2 Maintenance of pathogen

Stock cultures of *Phytophthora infestans* were maintained on sterile potato dextrose agar (PDA). Plates were kept in a darkened incubator at a temperature of 24°C and fresh culture plates initiated approximately every 10 days. Prior to experimentation, the fungus had to be grown on whole potato leaves, since attempts to inoculate experimental leaf discs directly using spore suspensions prepared from artificially grown cultures were unsuccessful. This may have been due to a loss of vigour by the pathogen, having been grown on artificial media for some time. Leaves were removed from potato plants when they were approximately 4-5 weeks old and placed, upper surface downwards, on dampened filter paper in 90 mm plastic Petri dishes. A spore suspension (approximately 5 x 10^4 spores per ml of sterile distilled water) was prepared using fungus maintained on PDA. Leaves were inoculated by pipetting droplets of this suspension onto their surface. Petri dishes were scaled and placed in an incubator at 15° C with a photoperiod of 16 hours, for 10 days.

2.2.5.3 Inoculation of plant material and treament with CHA and norspermidine

Potato leaf discs (15 mm diameter) were cut from stock leaves when plants were approximately 4-5 weeks old. Discs were floated, upper surface downwards, on solutions of CHA or norsperimidine in 90 mm plastic Petri dishes. Controls used distilled water only. Discs were used as opposed to whole plants as it is an established method of testing fungicidal activity against blight in the department. The method reduces the space required in incubators which are necessary for disease progression. Such a method examines the systemic activity of CHA and norspermidine against blight. Leaf discs were inoculated by pipetting spore suspension (20 μ l) onto the centre point of each disc surface. The spore suspension (approximately 5 x 10⁴ spores per ml of sterile distilled water), was prepared

using fungus that had been grown on whole plant leaves. Petri dishes were covered and placed in incubators as described in section 2.2.5.2. Infection intensity on each disc was assessed 8 days later using an infection key. Six replicates per treatment were used.

2.3 RESULTS

2.3.1 Effect of CHA and norspermidine against rust on broad bean

Disease effects aside, none of the plants in any of the experiments appeared to have suffered detrimentally as a result of CHA or norspermidine solution application. The compounds did not appear to have caused phytotoxic effects on any of the plants studied. Therefore, 'fungicidal' effects observed can indeed be attributed to such activity rather than non specific toxic effects of CHA and norspermidine.

All treatments reduced rust infection of broad bean although the reduction caused by the 1.0 mM CHA post inoculation treatment was not significant (Figure 5; Appendix 1). Little difference was observed between the treatments, with higher concentrations giving no better control than the lower concentrations of inhibitors. The efficacies of CHA and norspermidine at comparable concentrations were similar. Pre inoculation sprays appeared to be more effective than post inoculation treatments, although rarely significantly so. Best control was achieved using 1.0 mM CHA, applied pre inoculation, which reduced infection by 71.1 % ($p \le 0.001$) (Figure 5; Appendix 1).

2.3.2 Effect of CHA and norspermidine against chocolate spot on broad bean

All treatments reduced chocolate spot infection of broad bean although several of the treatments did not do so significantly (Figure 6; Appendix 2). More effective control was achieved through the use of pre inoculation sprays with only the highest concentrations of CHA and norspermidine causing significant reductions in infection when applied post inoculation. Little difference was observed between the efficacies of CHA and norspermidine. Best disease control was achieved with the pre inoculation application of 1.0 mM CHA, which reduced infection by 40.0 % ($p \le 0.001$) (Figure 6; Appendix 2).

2.3.3 Effect of CHA and norspermidine against powdery mildew on barley

All treatments reduced powdery mildew infection of barley, with 1.0 mM CHA providing best control; applied pre inoculation it reduced infection by 89.7 % ($p \le 0.01$) (Figure 7; Appendix 3). Little differences were observed between the treatments, with higher concentrations providing no greater disease control. In addition, no differences in efficacy between CHA and norspermidine were detected. Pre inoculation treatments appeared to provide superior control to those treatments applied post inoculation, although this difference was rarely significant (Figure 7; Appendix 3).

2.3.4 Effect of CHA and norspermidine against powdery mildew on apple

All treatments afforded good control of powdery mildew infection on apple (Table 1). Higher concentrations of each of the compounds provided superior disease control compared to lower concentrations. Norspermidine appeared to provide better control than CHA at comparable concentrations (Table 1).

2.3.5 Effect of CHA and norspermidine against late blight on potato

All treatments reduced blight infection of potato leaf discs (Table 2). Increasing concentrations of each compound provided superior disease control when compared to the lower concentrations. Infection was completely prevented by 2.0 mM norspermidine and very substantially reduced by 2.0 mM CHA. Norspermidine thus appeared to provide superior disease control to CHA at comparable concentrations (Table 2).















Treatment	Mean disease score	
Control	3	
0.5 mM CHA	2	
1.0 mM CHA	2	
2.0 mM CHA	1.5	
0.5 mM norspermidine	1.5	
1.0 mM norspermidine	1	
2.0 mM norspermidine	1	

Table 1: Effect of CHA and porspermidine against powdery mildew on apple

Note: Values are shown as the mean disease score of six replicates.

Infection key used for assessment:	3: \geq 50 % infection
	2: < 50 % infection
	1: few isolated spores
	0: no infection

Table 2: Effect of CHA and norspermidine against late blight on potato

Treatment	Mean disease score
Control	3
0.5 mM CIIA	2.5
1.0 mM CHA	2.5
2.0 mM CHA	1
0.5 mM norspermidine	2.5
1.0 mM norspermidine	2.5
2.0 mM norspermidine	0

Note: Values are shown as the mean disease score of six replicates.

Infection key used for assessment:

3: ≥ 50 % infection
2: < 50 % infection
1: few isolated spores

0: no infection

2.4 EFFECTS OF CHA AND NORSPERMIDINE ON GERMINATION AND DEVELOPMENT OF RUST UREDOSPORES ON BROAD BEAN LEAVES

2.4.1 Introduction and Objectives

The effects of CHA and norspermidine against rust and chocolate spot on broad bean and powdery mildew on barley indicated that pre inoculation treatments were more effective than post inoculation sprays. Although such differences were not always significant, further work was carried out in an attempt to extend these findings, examining the effects of the compounds on the development of rust germlings on the leaf surface. Bean rust was chosen since it was used previously in this laboratory to examine the effects of the putrescine analogues E-BED and E-TED, DFMO, MGBG and indeed, CHA, on germling development on artificial membranes (Reitz *et al.* 1995a, b). Only the highest concentrations of CHA and norspermidine (2.0 mM) were studied, using the specific treatment timings of either 3 hours before or 3 days after inoculaton.

2.4.2 Materials and Methods

Broad beans were grown as described in section 2.2.1.1 and rust maintained as detailed in section 2.2.1.2. When the beans were approximately 20 days old, they were treated with 2.0 mM solutions of either CHA or norspermidine and inoculated as described in section 2.2.1.3. At 12 hourly intervals, leaves were harvested from plants and placed into 100 % absolute alcohol, killing the growing fungus as well as removing chlorophyll from the leaves. Harvesting continued up to 120 hours after inoculation. Six replicates per treatment were used with controls consisting of plants that were inoculated only.

Upon completion of harvesting, leaves were re-hydrated by placing them in 80 % absolute alcohol for 3-4 hours, followed by similar periods in 50 % absolute alcohol and finally 20 % glycerol. Random sections of leaf were then cut, placed onto microscope slides and stained using a dilute solution of lacto-fuchsin. Leaves were examined using an Olympus CH2 microscope at a magnification of 10 x (eye lens) and 10 x (objective lens). Thirty spores were counted from each slide. The number of germinated spores was recorded, germ tube lengths measured using a calibrated micrometer, and the number of germinated spores forming appressoria noted. Significance was assessed using Student's t-test.

2.4.3 Results

Great variability within each replicate of each treatment was observed, a phenomenon reflected in the standard errors of the means. Little differences were apparent between any of the treatments at any of the timings examined (Table 3). CHA and norspermidine, applied pre inoculation, did appear to reduce germination at the earliest timings studied.

However, the effect was never significant. Likewise, germ tubes of those germinated spores appeared to be reduced in length and the percentage of them forming appressoria also reduced when beans were treated with CHA and norspermidine pre inoculation. However, again, these differences were rarely significant. Post inoculation treatments of CHA and norspermidine appeared to have little effect on the developing germling. Indeed, rust development had already considerably progressed when the post inoculation treatment of CHA and norspermidine was applied.

Time after inoculation (hrs)	Treatment	% spores germinated	average length of germ tubes	% of germ tubes forming
			(µm)	appressoria
12	Control	6.1 ± 1.02	73 ± 15.9	0
	pre CHA	3.3 ± 1.71	$21 \pm 10.1 *$	0
	pre norspermidine	4.4 ± 2.68	32 ± 18.4	0
24	Control	11.1 ± 5.07	61 ± 17.1	0
	pre CHA	5.0 ± 2.39	48 ± 23.6	0
	pre norspermidine	5.6 ± 2.67	47 ± 23.7	0
36	Control	30.6 ± 10.90	60 ± 22.5	5.2 ± 3.28
	pre CHA	17.8 ± 5.35	31 ± 9.2	4.0 ± 2.62
i	pre norspermidine	21.1 ± 6.97	23 ± 11.1	8.3 ± 8.33
48	Control	34.5 ± 10.70	61 ± 17.3	15.5 ± 8.68
	pre CHA	25.6 ± 6.75	46 ± 21.2	13.8 ± 10.30
	pre norspermidine	33.9 ± 5.74	40 ± 8.9	6.9 ± 4.52
60	Control	43.9 ± 11.20	76 ± 14.4	20.4 ± 8.03
	pre CHA	23.9 ± 4.08	44 ± 18.9	15.2 ± 8.24
	prc norspermidine	26.7 ± 7.10	56 ± 11.8	6,7 ± 4.94
72	Control	46.6 ± 9.51	76 ± 16.1	20.4 ± 8.03
	pre CHA	35.1 ± 3.86	51 ± 16.7	14.9 ± 8.17
	pre norspermidine	36.2±5.87	56 ± 11.8	8.9 ± 4.30

Table 3: Effects of 2.0 mM CHA and 2.0 mM norspermidine on bean rust germling development on leaf surfaces

... Table 3 continued overleaf.

.

				120					108					96					84
post norspermidine	post CHA	pre norspermidine	pre CHA	Control	post norspermidine	post CHA	pre norspermidinc	pre CHA	Control	post norspermidine	post CIIA	pre norspermidine	pre CHA	Control	post norspermidine	post CHA	pre norspermidine	pre CHA	Control
45.0 ± 5.07	42.8 ± 5.27	33.3 ± 5.08	33.3 ± 4.63	43.3 ± 7.40	46.1 ± 7.47	53.3 ± 6.62	75.6 ± 6.62	35.0 ± 2.81	40.6 ± 7.76	41.7 ± 7.59	$\textbf{45.0} \pm \textbf{8.55}$	32.8 ± 4.66	37.8 ± 4.92	42.2 ± 6.93	43.3 ± 6.44	43.9 ± 4.75	32.8 ± 5.27	33.9 ± 5.05	41.7 ± 5.95
47 ± 7.7	57 ± 8.7	72 ± 20.0	70 ± 16.6	70 ± 12.2	47 ± 7.6	44 ± 9.0	53 ± 9.0	45 ± 9.1	72 ± 14.5	72 ± 16.2	66 ± 15.4	52 ± 8.2	56 ± 17.0	82 ± 8.0	34 ± 5.6	66 ± 11.1	54 ± 4.8	43 ± 6.6	64 ± 11.9
9.6 ± 3.79	10.3 ± 3.87	14.6±7.27	9.9 ± 3.84	16.6 ± 7.72	16.0 ± 7.57	14.9 ± 8.17	14.0 ± 7.57	15.7 ± 7.12	19.3 ± 6.99	19.9 ± 6.97	16.6 ± 7.72	6.4 ± 2.95	14.9 ± 8.17	17.3 ± 7.63	10.4 ± 8.18	14.2 ± 8.21	18.8 ± 7.74	16.6 ± 7.72	24.2 ± 15.50

- 21

A STATE OF STATE

... Table 3 continued from previous page:

Note: Values are shown as the means \pm SE of thirty measurements from each of six replicates.

Significant difference from control is shown as: * $p \leq 0.05$

. .

. .

· · · · · ·

. .

2.5 DISCUSSION

These studies have shown that CHA and norspermidine possess fungicidal activity against a number of plant pathogens. Such findings are in agreement with those of West and Walters (1988) who found that CHA reduced powdery mildew infection of barley in glasshouse studies. Little information exists as to the effect of norspermidine upon plant pathogenic fungi, although antiproliferative activity was observed in experimental animal turnours (Prakash *et al.* 1988). The present findings are in agreement with these studies and demonstrate the growth inhibitory properties of norspermidine.

The compounds demonstrated protective, eradicant and systemic activity against a broad range of plant pathogenic fungi. Such findings are in agreement with the broad fungicidal activity found for putrescine analogues (Havis *et al.* 1994a, c). As suggested by Rajam *et al.* (1985), the systemic action observed against late blight on potato leaf discs could be due to the compounds themselves, metabolites of the compounds or induced fungicidal compounds. Further work would be necessary to determine which of those possibilities is responsible for this systemic activity.

Various differences were apparent between the responses of the different pathogens towards the compounds. Against rust on broad bean and powdery mildew on barley, no differences were apparent between the higher and lower concentrations of inhibitors. The other pathogens responded more effectively to increasing concentrations of CHA and norspermidine. Bean rust and barley powdery mildew may be more sensitive to the compounds and thus, lower concentrations may be optimal for control of such plant infections.

Such differences in responses of fungi to inhibitors is well documented. Birecka *et al.* (1986) suggested that differences may be due to variations in uptake of inhibitor, enzyme sensitivity towards the inhibitor and fungal polyamine requirements. Similar fungal species can possess markedly different sensitivities towards a single inhibitor, for example, Walters *et al.* (1995b) found that *Phytophthora infestans* was well controlled by DFMO while *Pythium ultimum* was unaffected by the inhibitor. These authors found that differences in inhibitor uptake and enzyme sensitivity towards the inhibitor accounted for the differences observed against fungal growth. Further work would be necessary to clucidate the mechanism(s) by which bean rust and barley powdery mildew afforded increased sensitivity towards CHA and norspermidine.

Additional response differences were observed since bean rust and chocolate spot, and barley powdery mildew did not appear to be affected differently by CHA or norspermidine.

Apple mildew and potato blight, however, were more responsive to norspermidine than to CHA. Similarly, this effect could be due to differences in fungal uptake of inhibitor or enzyme sensitivity towards the inhibitor. Further work would be required to elucidate these findings.

The study indicated that pre inoculation sprays of the compounds were more effective than post inoculation treatments in all the plant/pathogen interactions examined, i.e. broad bean rust, chocolate spot on beans and powdery mildew of barley. This is in agreement with Foster and Walters (1992a) who concluded that a pre inoculation application of DFMO was more effective than a post inoculation treatment against the stem rust fungus on wheat, However, the findings are in contrast to much of the previous work carried out to examine the effects of various inhibitors on plant pathogenic fungi, where post inoculation treatments were invariably superior to pre inoculation sprays (Rajam et al. 1985; Walters 1986; Weinstein et al. 1987; Havis et al. 1994a, c). Walters (1995) suggested that inhibitors may effect fungal germination and/or development on the leaf surface. In recent work, Reitz et al. (1995a) found that CHA did not affect germination but reduced appressoria formation of rust uredospores on artificial membranes. While this would suggest that CHA should be more effective when applied post inoculation, the chosen timings of inhibitor application were 3 hours pre inoculation and 3 days post inoculation. Thus, the post inoculation treatments of CHA and norspermidine may have been too late to critically affect early fungal development on the leaf surface.

Thus, microscopy studies were carried out but little differences were found between pre inoculation applied CHA, norspermidine and the untreated control. A slight tendancy towards reduced germination, length of germ tube and appressoria formation was indicated on treated leaves, although this was rarely significant. No differences were apparent between the inhibitors applied post inoculation and the control. However, while microscopic examination of rust development on leaves found little differences between the treated leaves and the untreated control, it should be noted that the compounds had indeed effectively demonstrated an ability to reduce infection on plants. The inhibitors perhaps exert a much greater effect upon the growing fungus at a later stage of development than on spore germination or early germling development.

Great variability was observed even between the replicates of each treatment and perhaps a less variable fungus would have proved a more suitable subject for such microscopy studies. Powdery mildew development is well documented (Jenkyn & Bainbridge 1979), and future work in this area may be better utilising such a fungus.

However the compounds act, CHA and norspermidine undoubtedly possess fungicidal activity against a variety of plant pathogens and no phytotoxicity was observed. Previous work has indicated that inhibitors of polyamine biosynthesis do not adversely affect polyamine levels in plants (Walters 1986). However, future work would be necessary to confirm this for plants treated with CHA and norspermidine.

Chapter 3

Effect of cyclohexylamine and spermidine analogues on the growth of plant pathogenic fungi *in vitro*

3. EFFECT OF CYCLOHEXYLAMINE AND SPERMIDINE ANALOGUES ON THE GROWTH OF PLANT PATHOGENIC FUNGI *IN VITRO*

3.1 EFFECT OF CHA AND SPERMIDINE ANALOGUES ON MYCELIAL GROWTH OF FUNGION SOLID MEDIA

3.1.1 Introduction and Objectives

Preliminary laboratory experiments were carried out to determine the antifungal activity of CHA, norsperimidine and two additional commercially available spermidine analogues: N^{1} - and N^{8} -acetylspermidine (Figure 8). In vitro effects of these compounds on mycelial growth on solid media were studied in two plant pathogenic fungi: the ascomycete *Pyrenophora avenae* Ito & Kuribay, causing leaf-stripe in oats and the deuteromycete *Pyricularia oryzae* Br. & Cav., responsible for rice blast.

As a plant pathogen *P. avenae* has declined in importance through the advent of routine effective seed treatments (Jones & Clifford 1978). Rice blast remains an important disease in those parts of the world, for example, the Far East, which rely heavily on rice production (Stapley & Gayner 1969). Such pathogens were, however, primarily chosen for experimentation as representative of those fungi able to grow on artificial media, and which had been shown in previous work to be moderately affected by polyamine biosynthesis inhibitors (Foster & Walters 1990; Havis *et al.* 1994a).

3.1.2 Materials and Methods

3.1.2.1 Maintenance of pathogens

Stock cultures of *P. avenae* and *P. oryzae* were maintained on sterile PDA in a darkened incubator at a temperature of 24°C. Fresh culture plates were initiated approximately every ten days for *P. avenae* and every 20 days for *P. oryzae*.

3.1.2.2 Growth of fungi on inhibitor amended solid media

Previous work by West and Walters (1989) demonstrated that CHA reduced mycelial growth of *Pyrenophora teres* and *Gaeumannomyces graminis* at concentrations ranging from 0.5 mM to 4.0 mM. However, when Foster and Walters (1990) examined the effect of ODC and AdoMetDC inhibitors on the mycelial growth of *P. avenae*, they found concentrations ranging from 0.1 mM to 2.0 mM inhibited growth of the fungus. Havis and co-workers (1994a) demonstrated that E-BED and its derivatives, used at 1.0 mM, successfully inhibited mycelial growth of *P. oryzae* and *Botrytis cinerea*. For experiments



N⁸-acetylspermidine



with *P. avenae*, the concentrations of CIIA and norspermidine thus chosen for study ranged from 0.05 mM to 2.0 mM. N¹- and N⁸-acetylspermidine were only examined at a concentration of 2.0 mM due to the limited amount of those chemicals available. The higher concentration was chosen since work had already shown mycelial growth of *P. avenae* to be unaffected by 1.0 mM concentrations of these compounds (S. A. Foster unpublished results).

Higher inhibitor concentrations (1.0 mM - 5.0 mM) were chosen for experiments with *P. oryzae* since an initial investigation had shown that concentrations ranging from 0.05 mM - 2.0 mM had little effect on mycelial growth of the fungus (data not shown). Effects of N¹- and N⁸-acetylspermidine upon growth of *P. oryzae* were not examined due to the limited amounts of chemicals available. Higher concentrations of these compounds, compared to CHA and norspermidine, were required to affect growth of *P. oryzae*, excessively high concentrations of N¹- and N⁸-acetylspermidine would have been required to affect growth of *P. oryzae*.

Filter-sterilised inhibitor solutions (10 ml) were added to 70 ml of sterile PDA at 45-47°C to obtain the final desired concentration of each inhibitor. Sterile medium (20 ml) was added aseptically to a 90 mm sterile plastic Petri dish. An 8 mm cork borer was used to remove plugs of stock mycelium which were inverted and placed in the centre of each Petri dish. Inoculated plates were incubated in the dark at 24°C. Radial mycelial growth, excluding the mycelial plug, was measured 3, 6 and 8 days after inoculation with *P. avenae* and 7, 14 and 21 days after inoculation with the slower growing *P. oryzae*. Three measurements were taken from each plate. Control plates contained culture medium only. Four replicates were used for each treatment with significance assessed using Student's *t*-test. All experiments were repeated with similar results obtained.

3.1.3 Results

3.1.3.1 Effects of CHA and spermidine analogues on mycelial growth of *Pyrenophora avenae*

Mycelial growth of *P. avenae* on solid media was reduced in the presence of 2.0 mM CHA by 55.1 %, eight days after inoculation ($p \le 0.001$) (Figure 9; Appendix 4) (Plate 1). Lower concentrations of CHA had little effect on fungal growth. Norspermidine produced a slightly greater effect on fungal growth, with 1.0 mM reducing growth by 41.0 % and 2.0 mM reducing growth by 63.1 % eight days after inoculation ($p \le 0.001$) (Figure 9; Appendix 4) (Plate 1). N¹- and N⁸-acetylspermidine had little effect on fungal growth at a concentration of 2.0 mM (Table 4).

3.1.3.2 Effects of CHA and norspermidine on mycelial growth of *Pyricularia* oryzae

Mycelial growth of *P. oryzac* on solid media was reduced by all of the concentrations of CHA examined although the reduction caused by 1.0 mM CHA after 21 days growth was not significant (Figure 10; Appendix 5). Fungal growth was reduced by 5.0 mM CHA, by 35.9 %, 21 days after inoculation ($p \le 0.001$) (Plate 2). Norspermidine had a greater effect on fungal growth with a concentration of 5.0 mM reducing growth by 50.6 % after 21 days growth ($p \le 0.001$) (Figure 10; Appendix 5) (Plate 2).

	Mycelial growth, measured as mean colony radius (mm)									
Treatment	3 days	6 days	8 days							
Control	12.0 ± 0.26	28.7 ± 0.42	$\textbf{36.2} \pm \textbf{0.79}$							
N ¹ -acetylspermidine	12.1 ± 0.42	26.7 ± 0.80 *	$\textbf{36.0} \pm \textbf{0.33}$							
N ⁸ -acetylspermidine	12.2 ± 0.48	29.2 ± 0.48	36.8 ± 0.31							

190

Sec. B.

<u>Table 4. Effect of N^1 - and N^8 -acetylspermidine on mycelial growth of *Pyrenophora avenae*</u>

Note: Values are shown as the means \pm SE of three measurements from each of four replicates.

Significant difference from control is shown as: * $p \le 0.5$

ţ





Mean radial mycelial growth (mm) 10 5 8 8 8 8 8 0 ch CONTROL H Effect of cyclohexylamine and norspermidine on mycelial growth of Pyricularia oryzae 1 CHA concentration (mM) 2 H 3 H 4 н 5 H 1 H Norspermidine concentration (mM) 2 H 3 H 4 H 5 FH □21days 14 days T days

standard errors of means indicated. Significant differences from control are indicated in Appendix 5. mycelial growth was measured 7, 14 and 21 days after inoculation of plates. Values are the means of 3 measurements from each of 4 replicates with Figure 10. Effect of varying concentrations (1 - 5 mM) of CHA and norspermidine on mycelial growth of Pyricularia oryzae on solid media. Radial



Plate 1. Effect of 2.0 mM CHA and 2.0 mM norspermidine on mycelial growth of Pyrenophora avenae, grown on solid media for eight days.



Plate 2. Effect of 2.0 mM CHA and 2.0 mM norspermidine on mycelial growth of Pyricularia oryzae, grown on solid media for eight days.
3.2 EFFECT OF CHA AND SPERMIDINE ANALOGUES ON GROWTH OF PYRENOPHORA AVENAE IN LIQUID CULTURE

3.2.1 Introduction and Objectives

Because fungal growth on solid media is not always a reliable indicator of the effects of inhibitors added to the growth medium (Isaac & Jennings 1995), studies were carried out to determine the effect of CHA and the spermidine analogues on the growth of *P. avenae* in liquid culture. This would give a more realistic indication of the effects of the inhibitors on fungal growth, and would be a useful foundation for subsequent work on metabolic effects of the compounds. *P. avenae* was chosen as it had demonstrated an intermediate response to polyamine biosynthesis inhibitors, allowing collection of sufficient fungal material for biochemical analysis (Foster & Walters 1990; Havis *et al.* 1994a, c).

ý.

3.2.2 Materials and Methods

3.2.2.1 Maintenance of pathogen

P. avenae was maintained as described in section 3.1.2.1.

3.2.2.2 Growth of Pyrenophora avenae in inhibitor amended liquid media

Since mycelial growth of *P. avenae* had been reduced in the presence of 2.0 mM CHA, a preliminary experiment to examine the effect of CHA on growth of *P. avenae* in liquid culture was instigated, studying a range of concentrations (1.0 mM - 5.0 mM: data not shown). Fungal growth was markedly affected by 3.0 mM CHA and thus, concentrations ranging from 2.0 mM - 3.0 mM CHA were used to determine the effect of CHA upon growth of *P. avenae* in liquid culture.

Lower preliminary concentrations were chosen for norspermidine since greater fungal sensitivity had been exhibited towards norspermidine compared to CHA. Fungal growth was dramatically reduced in the presence of 0.5 mM norspermidine when concentrations ranging from 0.01 mM - 1.0 mM were used (data not shown). Thus, the effects of 0.1 mM - 0.5 mM concentrations of norspermidine on growth of *P. avenae* in liquid culture were examined.

The effect of 0.5, 1.0 and 2.0 mM concentrations of N¹- and N⁸-acetylspermidine on *P. avenae* growth in liquid culture were also studied. Although such concentrations did not affect mycelial growth of *P. avenae*, it is well established that fungal growth in liquid culture is more sensitive towards inhibitors. Indeed, Havis *et al.* (1994c) found that mycelial growth of *P. avenae* on solid medium was unaffected by 1.0 mM E-TED, yet the same concentration of inhibitor completely inhibited growth of the fungus in liquid culture.

85

Filter-sterilised inhibitor solutions (10 ml) were added to 140 ml of sterile liquid Potato Dextrose Broth (PDB) in 250 ml conical flasks to obtain the desired final concentration of each inhibitor. Flasks were inoculated with a 10 mm dise of *P. avenae* mycelium and placed in a Gallenkamp orbital shaker (140 rev.min⁻¹) at 24°C. After 4 days growth, the fungus was washed with distilled water through a fine mesh seive and centrifuged at 16000 g at 0°C for 10 minutes before weighing. Control flasks contained 150 ml of sterile PDB only. Four replicates per treatment were used. Significance was assessed using Student's *t*-test and experiments were repeated with similar results obtained.

•

In order to carry out biochemical analyses, sufficient fungal material had to be amassed. It was desirable to study fungus after it had been growing for only a short time in the presence of the compounds, i.e. 2 days. Therefore, *P. avenae* was grown in liquid culture in the presence of lower concentrations of the compounds: 1.0 mM CHA, 0.01 mM norspermidine and 0.5 mM N¹- and N⁸-acetylspermidine. Although such concentrations of inhibitors did not always cause a significant reduction in fungal fresh weight (Tables 7-8), growth changes were always apparent (Plates 3-4). Fungus grown in the presence of the compounds was darker than the control. Control flasks contained distinct balls of fungal mycelium, with the growing fungus giving those balls a spikey exterior look, growing outwards like needles from the fungal mass. Fungus grown in the presence of CHA and the spermidine analogues, however, was much less distinct and appeared to be more of a broken up mass. Given the growth changes, biochemical analyses were warranted. Havis *et al.* (1994c), similarly, used lower concentrations of E-TED to measure enzyme activity and polyamine content of *P. avenae* exposed to the inhibitor.

3.2.3 Results

ł

Fungal growth was significantly inhibited by CHA over the range of concentrations, 2.25 mM - 3.00 mM (Table 5). A concentration of 3.0 mM CHA inhibited growth in liquid culture by 92.9 %. Fungal growth was also significantly inhibited by concentrations of norspermidine over the range 0.1 mM - 0.5 mM with the latter concentration completely inhibiting growth (Table 5). The fresh weight of fungus grown in liquid culture in the presence of N¹- and N⁸-acetylspermidine was unaffected by concentrations as high as 2.0 mM (Table 6).

Growt	h in liquid culture, mea	sured as mean fresh weight	(g)
Concentration (mM)	CHA	Concentration (mM)	norspermidine
0	4.2 ± 0.37	0	5.2 ± 0.30
2.00	4.3 ± 0.16	0.1	1.8 ± 0.11 **
2.25	2.6 ± 0.06 *	0.2	1.3 ± 0.11 **
2.50	1.7 ± 0.19 **	0.3	0.7 ± 0.08 ***
2.75	0.7 ± 0.37 ***	0.4	0.3 ± 0.02 ***
3.00	0.3 ± 0.02 **	0.5	NG ^a

Table 5. Effect of varying concentrations of CHA and norspermidine on growth of *Pvrenophora avenae* in liquid culture for 4 days

Note: Values are shown as the means \pm SE of four replicates. Significant differences from control are shown as: * p \leq 0.05; ** p \leq 0.01; *** p \leq 0.001 ^a NG: no growth

Table 6. Effect of 2.0 mM N¹- and N⁸-acetylspermidine on growth of Pyrenophora avenuein liquid culture for 4 days

Treatment	Growth in liquid culture, measured as mean
	fresh weight (g)
Control	3.5 ± 0.01
N ¹ -acctylspermidine	3.3 ± 0.02
Control	3.4 ± 0.01
N ⁸ -acetylspermidine	3.8 ± 0.10

Note: Values arc shown as the means \pm SE of four replicates.

	Growth in liquid	culture, measured as me	an fresh weight (g)
Days of growth	Control	CHA	norspermidine
2	0.6 ± 0.01	0.6 ± 0.01	0.6 ± 0.01
3	1.1 ± 0.02	1.2 ± 0.07	1.1 ± 0.04
4	3.4 ± 0.02	3.3 ± 0.41	2.2 ± 0.09

Table 7. Effect of 1.0 mM CHA and 0.01 mM norspermidine on growth of Pyrenophoraavenae in liquid culture for 2, 3 or 4 days

Note: Values are shown as the means \pm SE of four replicates.

Table 8. Effect of 0.5 mM N¹- and N⁸-acetylspermidine on growth of Pyrenophora avenaein liquid culture for 2, 3 or 4 days

	Growth in liqu	id culture, measured as mea	n fresh weight (g)
Days of growth	Control	N ¹ -acetylspermidine	N ⁸ -acetylspermidine
2	0.8 ± 0.05	1.0 ± 0.08	0.9 ± 0.09
3	1.5 ± 0.05	1.6 ± 0.06	1.2 ± 0.07
4	4.0 ± 0.07	4.1 ± 0.53	3.8 ± 0.23

Note: Values are shown as the means \pm SE of four replicates.



Plate 3. Effect of 1.0 mM CHA on growth of Pyrenophora avenae in liquid culture, grown for four days.





3.3 EFFECT OF CHA AND SPERMIDINE ANALOGUES ON CELL LENGTHS OF *Pyrenophora avenae* grown on solid media

3.3.1 Introduction and Objectives

Given the obvious changes in growth of P. avenae in liquid culture in the presence of 1.0 mM CHA, 0.01 mM norspermidine and 0.5 mM N¹- and N⁸-acetylspermidine, although not always accompanied by a reduction in fungal fresh weight, microscopic examination of the fungus grown on solid media in the presence of the compounds was undertaken, in an attempt to detect morphological changes induced by the inhibitors. Mycelial growth on solid media was chosen for examination as it is inherently easier and more accurate than examination of fungus grown in liquid culture.

3.3.2 Materials and Methods

Fungus was grown on inhibitor amended media as described in section 3.1.2.2. Treatments examined were 0.05 mM - 2.0 mM CHA and norspermidine, and 2.0 mM N¹- and N⁸- acetylspermidine. After 8 days growth, mycelium was scraped from the Petri dishes using a scalpel, spread onto microscope slides and stained with lactophenol cotton blue. Cell lengths were measured using a calibrated micrometer in an Olympus CH2 microscope at a magnification of 10 x (eye lens) and 100 x (objective oil lens). Ten measurements were taken from each of four replicates of each treatment, and significance was assessed using Student's *t*-test. The experiment was repeated and similar results obtained.

3.3.3 Results

Fungal cell lengths were quite variable within each treatment and so little significance was observed between the treatments, although all treatments did cause a reduction in fungal cell length (Table 9). The addition of 0.1 mM and 2.0 mM CHA, and 1.0 and 2.0 mM norspermidine to the growth medium resulted in a significant reduction in the lengths of *P. avenae* cells (Table 9). Neither 2.0 mM N¹- nor N⁸-acetylspermidine had any significant effect on the fungal cell length (Table 9).

Treatment	Mean cell length (µm)
Control	60.98 ± 7.15
0.05 mM CHA	51.40 ± 4.69
0.1 mM CHA	43.60 ± 3.38 *
0.5 mM CHA	45.07 ± 4.53
1.0 mM CHA	52.29 ± 7.29
2.0 mM CHA	41.68 ± 2.85 *
0.05 mM norspermidine	44.92 ± 5.61
0.1 mM norspermidine	51.11 ± 6.36
0.5 mM norspermidine	52.58 ± 3.62
1.0 mM norspermidine	31.96 ± 4.40 **
2.0 mM norspermidine	38.88 ± 4.30 *
2.0 mM N ¹ -acetylspermidine	46.84 ± 3.17
2.0 mM N ⁸ -acetylspermidinc	47.42 ± 3.89

 Table 9. Effect of varying concentrations of CHA and spermidine analogues on cell lengths

 in Pyrenophora avenae grown on solid media for 8 days

Note: Values are shown as the means \pm SE of ten measurements from each of four replicates.

Significant differences from control are shown as: * $p \le 0.5$; ** $p \le 0.01$

-

3.4 DISCUSSION

This study found CHA and norspermidine to be effective in reducing the mycelial growth of two pathogenic fungi on solid media. Such findings are in agreement with those of West and Walters (1989) who found that CHA inhibited mycelial growth of *Gaeumannomyces* graminis. Little information exists on the fungicidal activity of norspermidine, although, antiproliferative activity has been demonstrated against experimental animal tumours (Prakash et al. 1988). This study has also shown, therefore, that norspermidine possesses growth inhibitory activity in fungal systems. Similarly, both compounds exhibited antifungal properties when *P. avenae* was grown in liquid culture.

No effect on mycelial growth was observed for N¹- or N⁸-acetylspermidine at the concentrations examined, although growth changes were apparent when *P. avenae* was grown in the presence of the compounds in liquid culture. These findings are in contrast to the reported reduction of powdery mildew on barley in glasshouse trials by the spermidine analogues (S. A. Foster unpublished results). However, insensitivity of fungal mycelium towards inhibitors is not unusual and as discussed previously, Havis *et al.* (1994c) found that E-TED did not affect mycelial growth of *P. avenae* at a concentration of 1.0 mM, yet at the same concentration, it completely inhibited growth of the fungus in liquid culture. Mycelial growth is often regarded as less reliable as fungus can 'grow away' from inhibitor in the medium (Isaac & Jennings 1995).

Higher concentrations of CHA and norspermidine were required to achieve inhibition of *P. oryzae* on solid media compared to *P. avenae*. Birecka *et al.* (1986) suggested that differences between fungi in their response to a single inhibitor may be the result of differences between genera. Such a phenomenon has been discussed previously in relation to differences between fungi growing on plants and their response to CHA and norspermidine (section 2.5, page 72). Birecka *et al.* (1986) suggested that such differences may be due to variations in inhibitor uptake by the fungus or fungal enzyme insensitivity towards the inhibitor. Further work would, however, be necessary to elucidate exactly why *P. oryzae* is less sensitive than *P. avenae* to CHA and norspermidine.

Similarly, both species of fungus were more sensitive towards norspermidine than CHA. Norspermidine and CHA were more effective than N¹- and N⁸-acetylspermidine against growth of *P. avenae* on solid media. Further study would be necessary to determine the cause of the differing sensitivity of the fungal species to the various inhibitors.

Growth changes were apparent when fungus in liquid culture was exposed to concentrations of CHA, norspermidine and N¹- and N⁸-acetyispermidine that did not always affect fungal

fresh weight. Inhibitors of polyamine biosynthesis have been reported to cause morphological changes in fungus growing on solid media, for example, Boyle *et al.* (1988) found that abnormal hyphae and swollen cells were caused in species of *Microsporum* and *Trichophyton* exposed to DFMO. Clearly, CHA and the spermidine analogues affected growth of *P. avenae* although further microscopic analysis would be required to detect the morphological effects caused.

This study found that cell lengths of *P. avenae* grown on solid media were reduced by higher concentrations of CHA and norspermidine. Such findings are in agreement with those of West & Walters (1989) who found that CHA reduced cell lengths in *Pyrenophora teres*. However, these authors also observed fungal cells of *Septoria nodorum* to be unaffected by CHA (West & Walters 1989), and is therefore similar to the observed lack of effect on cell lengths of *P. avenae* exposed to 2.0 mM N¹- and N⁸-acetylspermidine. Thus, *P. avenae* has been shown to respond quite differently to a number of inhibitors, resulting in differing effects on cell lengths following exposure to different inhibitors. This agrees with previous studies which report varying effects to be caused by different inhibitors on different fungal species.

Thus, CHA and norsperimidine were shown to possess antifungal activity against two pathogenic fungal species on solid media and one in liquid culture. N^{1} - and N^{8} - acetylspermidine were less effective in reducing fungal growth but a degree of antifungal activity was observed.

Chapter 4

Effect of cyclohexylamine and spermidine analogues on biosynthesis and catabolism of polyamines in *Pyrenophora avenae*

4. EFFECT OF CYCLOHEXYLAMINE AND SPERMIDINE ANALOGUES ON BIOSYNTHESIS AND CATABOLISM OF POLYAMINES IN *PYRENOPHORA AVENAE* GROWN *IN VITRO*

4.1 INITIAL DETERMINATION OF THE EFFECTS OF CHA AND SPERMIDINE ANALOGUES ON POLYAMINE BIOSYNTHESIS IN *PYRENOPHORA AVENAE*

4.1.1 Introduction and Objectives

Given the observed fungicidal and antifungal effects of CHA and norspermidine *in vivo* and *in vitro*, studies were undertaken to determine the effect of those compounds on polyamine biosynthesis in *P. avenae* in an attempt to determine their fungicidal and antifungal mode(s) of action. Studies were also undertaken with N¹- and N⁸-acetylspermidine, even though the present study had failed to demonstrate antifungal properties attributable to either of these compounds: the compounds had, however, been observed to reduce powdery mildew infection of barley in previous preliminary work (S. A. Foster unpublished results).

4.1.2 Materials and Methods

4.1.2.1 Growth of fungal material

As described in section 3.2.2.2 (page 85), *P. avenae* was grown in liquid culture amended with either 1.0 mM CHA, 0.01 mM norspermidine, 0.5 mM N¹-acetylspermidine or 0.5 mM N⁸-acetylspermidine for 2, 3 or 4 days. The fungal pellet obtained after centrifugation was frozen at -20°C and used for enzyme and polyamine analyses as required. All results are the means of four replicates and statistical significance was assessed using Student's *t*-test. All analyses were repeated and similar results obtained.

4.1.2.2 ODC assay

Fungus (0.5 g) was ground using a pre chilled mortar and pestle with 1.75 ml of a buffer containing 10 mM potassium phosphate, 2 mM 1,4-dithiothreitol, 1 mM magnesium chloride, 0.1 mM ethylenediaminetetra-acetic acid (EDTA) and 0.1 mM pyridoxal-5-phosphate, adjusted to a pH of 7.6, as described by Stevens *et al.* (1976). The crude enzyme extract was sonicated on ice using a Soniprep 150 for 10 cycles of 10 seconds on/20 seconds off before being centrifuged at 24000 g for 15 minutes at 0°C. The supernatant (cytosolic fraction) was dialysed against 30 times its volume of buffer, overnight in the dark at 4°C, using dialysis tubing with a molecular weight cut off of 12000. Protein assays were carried out using the method of Lowry *et al.* (1951) with bovine serum albumin (BSA) as the standard².

² Protein calibration curve used was as described by Havis, N. D. (1993). Control of fungal plant pathogens using novel putrescine analogues. Ph.D. Thesis, Glasgow University.

ODC activity was assayed by measuring the ${}^{14}\text{CO}_2$ released after incubation with [1- ${}^{14}\text{C}$] ornithine. Enzyme extract (0.1 ml) and 0.125 µCi of L-[1- ${}^{14}\text{C}$] ornithine hydrochloride (specific activity: 57 mCi.mmol⁻¹, Amersham International, UK) were added to 0.3 ml of a buffer containing 50 mM Tris/HCl, 0.05 mM L-ornithine monohydrochloride and 0.031 mM pyridoxal-5-phosphate, adjusted to a pH of 8.0, as described by Stevens *et al.* (1976). The assay was carried out in 100 mm glass test tubes fitted with silicone rubber stoppers and 35 mm long, 22 guage needles. A 25 mm² filter paper impregnated with 10µl of 2 M KOH was fitted to each needle to trap ${}^{14}\text{CO}_2$ released during the reaction. The test tubes were placed in a water bath at 37°C for 30 minutes, after which 0.2 ml of 6% perchloric acid (v/v) was added to each tube to stop the reaction before incubation for a further 30 minutes. The filter paper was removed and placed in 12 ml Emulsifier-Safe scintillant (Packard) and left overnight before counting the radioactivity using a Packard 1900 TR liquid scintillation counter. Values obtained as disintegrations per minute were converted and expressed as pmol ${}^{14}\text{CO}_2$ (mg protein)⁻¹ hour⁻¹.

4.1.2.3 AdoMetDC assay

The crude enzyme extract was prepared, sonicated and centrifuged as described in section 4.1.2.2. Ammonium sulphate (754 mg) was dissolved in the supernatant (cytosolic fraction) and the suspension centrifuged at 24000 g for 20 minutes at 0°C. The pellet obtained was redissolved in 1.75 ml buffer and dialysed as described in section 4.1.2.2. AdoMetDC activity was assayed by measuring the $^{14}CO_2$ released after incubation with S-adenosyl-[1- ^{14}C] methionine. Enzyme extract (0.1 ml) and 0.025 µCi of S-adenosyl-L-[carboxyl- ^{14}C] methionine (specific activity: 55 mCi.mmol⁻¹, Amersham International, UK) were added to 0.3 ml of a buffer containing 0.1 M sodium phosphate, 0.2 mM S-adenosyl-L-methionine and 1.0 mM putrescine, adjusted to a pH of 7.4, as described by Stevens *et al.* (1976). The remainder of the assay was carried out as described in section 4.1.2.2.

4.1.2.4 Incorporation of radio-labelled ornithine into polyamines

Several complications arise in attempting to assay spermidine synthase activity. The assay is known to be complex and time consuming, and a necessary substrate, decarboxylated AdoMet, is not commercially available (Stevens & Winther 1979). Thus, to examine spermidine synthase activity, radio-labelled ornithine is allowed to form polyamines and by determining the radioactivity in the end products, an indirect examination of spermidine synthase activity can be made.

An ODC assay was carried out as described in section 4.1.2.2, except that 0.125 μ Ci of L-[U-¹⁴C] ornithine hydrochloride (specific activity: 242 mCi.mmol⁻¹, Amersham

International, UK) was used in the reaction mixture. After completion of the incubation process in the water bath, an aliquot (100 µl) of the reaction mixture was removed, 200 µl of saturated sodium carbonate solution and 400 µl dansyl chloride in acctone solution (30 mg per ml acetone) added, and the reaction mixtures incubated in darkness at 60°C for 25 minutes. L-proline solution (100 µl of a 100 mg per ml solution in distilled water) was added and the samples incubated in darkness for a further 10 minutes at room temperature to convert excess dansyl chloride to dansyl proline. Toluene (500 µl) was added and the reaction mixtures shaken for 20 seconds to extract the dansylated polyamines. Aliquots of extract (25 µl) were spotted onto silica-gel thin layer chromatography plates which had been activated in an oven at 110°C for 90 minutes. The plates were developed using a chloroform:triethylamine (12:1) (v/v) solvent mixture and the polyamines visualised under ultraviolet light and identified against standards. Spots were scraped off the plate into 10 ml of Emusifier-Safe scintillant (Packard) and the radioactivity counted using a Packard 1900 TR liquid scintillation counter. Values obtained as disintegrations per minute were converted and expressed as dpm g⁻¹ fresh weight.

4.1.2.5 Analysis of free polyamines in fungal tissue

Fungus (0.5 g) was ground in a pre chilled mortar and pestle using 1 ml of 10 % perchloric acid. The suspension was centrifuged at 16000 g for 25 minutes at 0°C. 100 μ l of the supernatant was used for polyamine analysis as described in section 4.1.2.4. The spots were scraped off the plate into 4 ml of ethyl acetate and fluorescence measured in a Perkin-Elmer LS5 luminescence spectrometer at excitation 350 nm, emission 540 nm. Values obtained as units of fluorescence were converted and expressed as μ mol g⁻¹ fresh weight³.

³ Polyamine calibration curves used were as described by Havis, N. D. (1993). *Control of fungal* plant pathogens using novel putrescine analogues. Ph.D. Thesis, Glasgow University.

4.1.3 Results

4.1.3.1 ODC and AdoMetDC activities

Neither 1.0 mM CHA, 0.01 mM norspermidine, 0.5 mM N¹-acctylspermidine nor 0.5 mM N⁸-acctylspermidine had any effect on the activities of ODC and AdoMetDC enzymes in *P. avenae* exposed to the inhibitors for 2, 3, or 4 days (Tables 10-11).

4.1.3.2 Incorporation of radio-labelled ornithine into polyamines

None of the compounds examined had any effect on the flux of labelled ornithine through to polyamines in fungus grown in the presence of the inhibitors for 2, 3, or 4 days (Table 12). No radio-label was found in putrescine (data not shown) with all of the label found in spermidine and spermine (Table 12). This indirect determination of spermidine synthase activity showed the enzyme to be unaffected by the treatments examined.

と、1997年1月1日 - 日く

4.1.3.3 Analysis of free polyamines in fungal tissue

Exposure of fungal tissue to 1.0 mM CHA for 2, 3 or 4 days resulted in an increase in the amount of putrescine found in *P. avenae* (Table 13), e.g. a 74.8 % increase was found in fungus exposed to the inhibitor for 4 days. No other compound examined had any effect on the amounts of free polyamines in fungal tissue (Table 13).

	ODC activ	ity [pmol CO2 (mg pro	otein) ⁻¹ hr ⁻¹]
Treatment	2 days	3 days	4 days
Control	42.6 ± 8.91	33.0 ± 3.47	48.2 ± 7.61
1.0 mM CHA	31.5 ± 7.55	39.2 ± 5.77	50.5 ± 7.40
Control	55.3 ± 8.61	68.2 ± 4.14	54.2 ± 8.13
0.01mM norspermidine	46.2 ± 5.57	50.6 ± 6.51	61.2 ± 6.69
Control	334 ± 27.3	312 ± 32.9	255 ± 12.5
0.5 mM N ¹ -acetylspermidine	330 ± 33.9	293 ± 23.8	233 ± 28.2
0.5 mM N ⁸ -acetylspermidine	316 ± 25.8	271 ± 35.8	227 ± 24.2

 Table 10. Effect of CHA and spermidine analogues on ODC activity in Pyrenophora

 avenae grown for 2, 3 or 4 days

Note: Values are shown as the means \pm SE of four replicates.

Variability in control values of ODC activity may be the result of different growth rates of fungus measured during the different experiments (data not shown). Such an observation has also been noted by Havis and Walters (1992b).

	AdoMetDC ac	ctivity [pmol CO ₂ (mg	protein) ⁻¹ hr ⁻¹]
Treatment	2 days	3 days	4 days
Control	61.5 ± 7.10	69.6 ± 4.32	92.5 ± 11.01
1.0 mM CHA	70.3 ± 7.32	57.3 ± 4.91	99.4 ± 10.32
Control	51.9 ± 5.03	58.9 ± 6.93	83.0 ± 2.06
0.01mM norspermidine	59.8 ± 2.10	49.7 ± 7.37	75.2 ± 6.66
Control	32.2 ± 8.64	28.8 ± 2.88	28.7 ± 3.51
0.5 mM N ¹ -acetylspermidine	33.3 ± 6.21	22.5 ± 5.49	39.6 ± 4.71
0.5 mM N ⁸ -acetylspermidine	39.7 ± 6.9 1	39.2 ± 4.82	29.0 ± 6.32

 Table 11. Effect of CHA and spermidine analogues on AdoMetDC activity in *Pyrenophora*

 avenae grown for 2, 3 or 4 days

Note: Values are shown as the means \pm SE of four replicates.

Variability in control values of AdoMetDC activity may be the result of different growth rates of fungus measured during the different experiments (data not shown). Such an observation has also been noted by Havis and Walters (1992b).

Treatment Control 1.0 mM CHA	2 days 43.9 ± 6.54 50.3 ± 4.46	Spermidine 3 days 40.1 ± 7.31 35.3 ± 3.85	Radioactivity in poly 4 days 78.0 ± 9.54 56.6 ± 8.39	7amine (dpm g- ¹ FW 2 days 5.1 ± 1.39 9.8 ± 2.88) Spermine 3 days 3.1 ± 0.89 7.1 ± 3.41	4 da 12.0 ± 15.4 ±
			Radioactivity in poly	′amine (dpm g-1 FW		
		Spermidine			Spermine	
Treatment	2 days	3 days	4 days	2 days	3 days	4 d:
Control	43.9 ± 6.54	40.1 ± 7.31	78.0 ± 9.54	5.1 ± 1.39	3.1 ± 0.89	12.0 ±
1.0 mM CHA	50.3 ± 4.46	35.3 ± 3.85	56.6 ± 8.39	9.8 ± 2.88	7.1 ± 3.41	15.4 ±
0.01 mM norspermidine	47.5 ± 6.96	38.7 ± 3.77	60.3 ± 9.26	4.7 ± 1.93	2.7 ± 0.51	16.9±
Control	54.2 ± 0.01	$58.4 \pm 6,40$	107.2 ± 3.76	4.5 ± 1.01	9.0 ± 0.74	8.8 ±
0.5 mM N ¹ -acetylspermidine	52.2 ± 3.13	39.7 ± 5.85	109.0 ± 5.10	8.2 ± 1.92	6.3 ± 1.46	$10.8 \pm$

Table 12. Effect of CHA and spermidine analogues on the incorporation of radio-labelled omithine into polyamines in Pyrenophora avenae grown for 2. 3 or 4 days

、おからした。1997年にあるようないができた。そのでは、1997年に、1997年には、1997年には、1997年に、1997年には、1997年に

Note: Values are shown as the means \pm SE of four replicates.

•

•

· `,` · ;

0.5 mM N⁸-acetylspermidine

 52.7 ± 6.57

 39.0 ± 4.97

 103.0 ± 7.04

 6.6 ± 1.75

 12.8 ± 1.26

 12.4 ± 1.22

(data not shown). Such an observation has also been noted by Havis and Walters (1992b). Variability in control values of radioactivity in polyamines may be the result of different growth rates of fungus measured during the different experiments

					Polyamine	concentation	ı (jumol g- ¹)	FW)				
		Putrescine			Cadaverin	ë		Spermidine			Spermine	
Treatment	2 days	3 days	4 days	2 days	3 days	4 days	2 days	3 days	4 days	2 days	3 days	4 days
Control	200 ± 20.4	273 ± 14.0	369 ± 22.4	50 ± 7.0	115 ± 9.5	103 ± 8.3	143 ± 11.8	206 ± 16.6	270±9.7	42±3.7	75 ± 6.2	86 ± 4.1
1.0 mM CHA	467 ± 34.5 **	608 ± 26.7 ***	645 ± 29.1 **	59 ± 6.7	131 ± 12.3	126 ± 12.2	123±8.5	179 ± 13.6	266 ± 10.7	51 ± 5.1	87±5.3	97 ± 13.8
Control	200 ± 20.4	273 ± 14.0	523 ± 35.2	300 ± 18.0	258±13.8	222 ± 15.1	171 ± 9.7	$\textbf{224}\pm\textbf{31.2}$	419±33.4	42±3.7	80 ± 12.9	150 ± 18.6
0.01 mM norspermidine	220 ± 16.4	$\textbf{230} \pm \textbf{29.2}$	587±42.8	315 ± 27.4	234 ± 12.8	201 ± 16.6	174 ± 9.0	253 ± 11.1	457 ± 19.9	46 ± 5.2	89±9.2	167 ± 14.6
Control	140 ± 8.3	185 ± 26.9	72 ± 3.4	70 ± 4.6	54 ± 5.0	45±3.2	215±17.9	334 ± 22.6	44 ± 1.8	86 ± 13.6	93 ± 15.5	32±3.5
0.5 mM N ¹ -acctylspermidine	127 ± 12.0	197 主 24.6	86 ± 4,3	69±8.1	71 ± 10.1	49 ± 2.7	245 ± 14.3	327 ± 11.1	42 ± 2.0	84 ± 11.5	92 ± 3.5	39 ± 3.0
0.5 mM N ⁸ -acetylspermidine	176±11.0	197 ± 24-5	82 ± 3.1	65 ± 6.4	62 ± 8.4	45±3.9	182 ± 19.8	358 ± 24.7	58±3.7	89±13.4	99±8.7	34 ± 2.3
Note: Values are shown as	s the means	± SE of four re	plicates.									
Significant differences fro	m control ar	c shown as: *	* p≤0.01; *	** p ≤ 0.0	01							

Table 13. Effect of CIIA and spermidine analogues on free polyamine levels in *Pyrenophora avenae* grown for 2, 3 or 4 days

(data not shown). Such an observation has also been noted by Havis and Walters (1992b). Variability in control values of polyamine concentrations may be the result of different growth rates of fungus measured during the different experiments

.

.

.

4.2 FURTHER DETERMINATION OF THE EFFECTS OF CHA AND NORSPERMIDINE ON BIOSYNTHESIS AND CATABOLISM OF POLYAMINES IN *PYRENOPHORA AVENAE*

4.2.1 Introduction and Objectives

Given the increase in putrescine observed in fungal tissue exposed to 1.0 mM CHA for 2, 3 or 4 days, without an effect on ODC, AdoMetDC or the flux of labelled ornithine through to polyamines, further study was necessary to clucidate the mechanism(s) responsible for such an increase.

As reviewed in section 1.1.3.3 (page 14), organisms possess a variety of mechanisms to enable them to control intracellular polyamine levels. Polyamine catabolism, whereby polyamines are interconverted and degraded back to putrescine, is well characterised for mammalian cells, and many plants and microbes (Pegg 1988; Davis 1990). Generally, spermine can be converted to spermidine in a two step reaction, initially catalysed by the enzyme spermidine/spermine N¹-acetyltransferase (SSAT), to form an N¹-acetyl derivative of spermine before forming spermidine via the action of polyamine oxidase (PAO). Spermidine can similarly be converted to an N¹-acetyl derivative by SSAT before forming putrescine via PAO (Figure 3, page 16). Putrescine can then be degraded by diamine oxidase (DAO) (Pegg 1988, Large 1992). Further investigations carried out to study the cause of the observed increase in putrescine in CHA treated *P. avenae*, therefore, involved the determination of PAO and DAO activities, along with levels of N¹-acetylspermidine. It was decided to determine acetylated polyamine levels as a means of studying catabolism since the SSAT enzyme assay is inherently unreliable. N¹-acetylspermine was not determined as its commercial availability was restricted.

Levels of N^8 -acetylspermidine and N-acetylputrescine were also determined in case either conjugation of polyamines, or breakdown of the conjugates was occurring, mechanisms which are present in mammalian cells and which may cause fluctuations in the free polyamine pool (Figure 3, page 16).

Acetylated polyamine levels were also determined for fungus exposed to 0.01 mM norspermidine. Although 0.01 mM norspermidine failed to affect levels of free polyamines in *P. avenae*, intracellular polyamine levels are often regulated by conjugation, selective uptake and excretion of polyamines (Davis 1990). An effect on free polyamine levels may have been masked by such processes and therefore, acetylated polyamine levels were studied. Since no effect on free polyamine levels had been observed, a comprehensive examination of the catabolic enzymes, (i.e. including determination of PAO and DAO activities), was not thought to be necessary and the determination of acetylated polyamine

103

levels would indicate if further study of those enzymes was desirable for norspermidine treated fungus.

Due to limited amounts of chemicals available, no further study of the effects of the spermidine analogues, N^{1} - and N^{8} -acetylspermidine, on fungal polyamine biosynthesis was carried out.

4.2.2 Materials and Methods

4.2.2.1 Growth of fungal material

As described in section 3.2.2.2 (page 85), *P. avenae* was grown in the presence of 1.0 mM CHA or 0.01 mM norspermidine for 2, 3 or 4 days. The pellet obtained was used to assay the activities of PAO and DAO, and to determine acctylated polyamine levels. All results are the means of four replicates and statistical significance was assessed using Student's *t*-test. All analyses were repeated and similar results obtained.

4.2.2.2 PAO assay

Fungus (0.6 g) was ground using a mortar and pestle with 4 ml of a buffer containing 100 mM potassium phosphate and 2 mM 1,4-dithiothreitol, adjusted to a pH of 8.0, as described by Okuyama and Kobayashi (1961). The crude enzyme extract was centifuged at 20000 g for 20 minutes at 4°C. Supernatant (0.5 ml) and 0.15 μ Ci of $|^{14}C|$ spermidine trichloride (specific activity; 108 mCi.mmol⁻¹, Amersham International, UK) were added to 0.5 ml of a buffer containing 100 mM potassium phosphate, 1 mM spermidine and 30 µg catalase, adjusted to a pH of 8.0, as described by Okuyama and Kobayashi (1961). The assay was carried out in 100 mm glass test tubes fitted with silicone rubber stoppers. Test tubes were incubated in a water bath at 37°C for 30 minutes before the reaction was stopped by adding 4 M sodium hydroxide solution (1 ml) and shaking. Enzymic products were extracted into toluene by adding toluene (2 ml) and vortexing the mixture for 10 seconds before leaving to stand for 30 minutes. An aliquot (1 ml) of the upper tolucne phase was added to 10 ml Emusifier-Safe scintillant (Packard) before counting the radioactivity using a Packard 1900 TR liquid scintillation counter. Values obtained as disintegrations per minute were converted and expressed as pmol product (mg protein)⁻¹hour⁻¹. Protein concentrations were determined using the method of Lowry et al. (1951) with BSA as the standard.

4.2.2.3 DAO assay

DAO activity was determined as described for PAO activity in section 4.2.2.2. However, the reaction mixture consisted of 0.5 ml enzyme extract, 0.15 μ Ci [1,4-¹⁴C] putrescine dihydrochloride (specific activity: 108 mCi.mmol⁻¹, Amersham International, UK) and

0.5 ml of a buffer containing 100 mM potassium phosphate, 1 mM putrescine and 30 μ g catalase, adjusted to a pH of 8.0, as described by Okuyama and Kobuyashi (1961).

4.2.2.4 Analysis of acetylated polyamines in fungal tissue

Acetylated polyamines were determined as described in section 4.1.3.3 for free polyamines. The tle plates, however, were developed using a chloroform : isopropyl alcohol (propan-2-ol) (95:5) (v/v) solvent mixture as described by Libby (1978). The different developing solvent was required since the use of chloroform : triethylamine (12:1) (v/v) resulted in incomplete separation of the acetylated amines and putrescine. Rf values for the common mixture constituents are detailed in Table 14. Calibration curves were calculated using standards with N-acetylputrescine (100 μ g/ml) used to calibrate the fluorimeter on every occasion (Appendices 6-7). During recovery of the amines from the tle plates, N¹-acetylspermidine and N⁸-acetylspermidine were pooled together as their separation was not totally distinct.

Compound	Rf value	
Putrescine	0.76	
Cadaverine	0.79	
Spermidine	0.85	
Spermine	0.88	
N ¹ -acetylspermidine	0.41	
N ⁸ -acetylspermidine	0.37	
N-acetylputrescine	0.15	
CHA	0.84	
Norspermidine	0.66	

Table 14: Rf values of various compounds, developed on tic plates usingchloroform : isopropyl alcohol (95:5) (v/v)

A Star New Low

Ĵ.

(2013) N (20

4.2.3 Results

4.2.3.1 PAO and DAO activites

Neither PAO nor DAO activity was affected when *P. avenae* was exposed to 1.0 mM CHA for 2, 3, or 4 days (Table 15).

4.2.3.2 Analysis of acetylated polyamines in fungal tissue

Little changes in levels of acetylated polyamines were detected in fungus exposed to 1.0 mM CHA and 0.01 mM norspermidine for 2, 3 or 4 days (Table 16).

	Enz	yme activity [pmol pr	oduct (mg protein)	⁻¹ lm ⁻¹]
	Polyamir	ie Oxidase	Diamine	e Oxidase
Days	Control	1.0 mM CHA	Control	1.0 mM CHA
2	58.1 ± 5.04	77.6 ± 11.2	144 ± 15.6	167 ± 5.3
3	94.1 ± 11.9	82.7 ± 9.33	164 ± 10.2	143 ± 4.3
4	94.8 ± 10.9	75.3 ± 8.35	176 ± 8.7	162 ± 12.4

Table 15. Effect of 1.0 mM CHA on PAO and DAO activities in Pyrenophora avenaegrown for 2, 3. or 4 days

Note: Values are shown as the means \pm SE of four replicates.

Table 16: Effect of 1.0 mM CHA and 0.01 mM norspermidine on acetylated polyaminelevels in Pyrenophora avenae grown for 2, 3, or 4 days

		Polyamine concent	ration (µmol g ⁻¹ FW)
	Treatment	N-acetylputrescine	N-acetylspermidines
2 day exposure	Control	45.5 ± 5.14	22.5 ± 5.34
	CHA	64.2 ± 3.85	26.4 ± 1.99
	norspermidine	43.0 ± 4.82	29 .3 ± 1,82
3 day exposure	Control	54.5 ± 3.24	56.1 ± 7.58
	CHA	61.8 ± 3.86	59.0 ± 6.51
	norspermidine	55.1 ± 4.87	55.6 ± 6.80
4 day exposure	Control	46.4 ± 6.81	55.8 ± 2.85
	CHA	57.9 ± 1.53	59.6 ± 6.19
	norspermidine	46.8 ± 4.05	50.9 ± 7.68

Note: Values are shown as the means \pm SE of four replicates.

4.3 DETERMINATION OF THE EFFECTS OF HIGHER CONCENTRATIONS OF CHA AND NORSPERMIDINE ON POLYAMINE BIOSYNTHESIS IN *PYRENOPHORA AVENAE*

4.3.1 Introduction and Objectives

The study has so far shown that 0.01 mM norspermidine, 0.5 mM N¹- and 0.5 mM N⁸- acetylspermidine have no effect on ODC and AdoMetDC activities, the flux of labelled ornithine to polyamines or free polyamine levels in *P. avenae* exposed to the compounds for 2, 3 or 4 days. Exposure of fungus to 1.0 mM CHA caused an increase in free putrescine, a phenomenon which does not appear to be the result of catabolic mechanisms.

Given that fungal growth was more markedly affected by increased concentrations of the above compounds, experiments were carried out to eliminate the possibility of an effect on fungal polyamine biosynthesis occurring when fungus was exposed to higher concentrations of CHA and norspermidine. For such studies, fungus had to be exposed to the inhibitors for 4 days in order to amass sufficient material for analysis. Another possibility is that an early effect on polyamine biosynthesis could occur and yet be nullified later on, preventing detection. Experiments were therefore designed to allow the study of fungus which had only been exposed to the inhibitors for one day.

4.3.2 Materials and Methods

As described in section 3.2.2.2 (page 85), *P. avenae* was grown in liquid culture amended with higher concentrations of compounds: either 2.75 mM CHA or 0.2 mM norspermidine, for 4 days. Concentrations chosen were the maximum which permitted growth to occur.

Fungus was also exposed to these higher concentrations of inhibitors for 1 day. Fungus was grown in culture medium only for two days before being harvested and transferred to inhibitor amended medium for one further day. The pellet obtained thus contained sufficient material for analysis yet had only been exposed to the inhibitor for one day.

Fungus obtained was used to study the incorporation of $[U-^{14}C]$ ornithine into polyamines and to determine free polyamine levels. ODC and AdoMetDC assays were not thought to be necessary, given that an examination of the flux from labelled ornithine through to polyamines would also indicate if ODC or AdoMetDC activities had been affected. Analyses were carried out as described previously in sections 4.1.2.4 and 4.1.2.5. Results are the means of four replicates with all analyses repeated and similar results obtained. Significance was assessed using Student's *t*-test.

4.3.3 Results

4.3.3.1 Effect of 4 days exposure to higher concentrations of CHA and norspermidine on growth of *Pyrenophora avenae*

Exposure of fungus to 2.75 mM CHA and 0.2 mM norspermidine for 4 days resulted in substantial reductions in fungal fresh weight, similar to those results presented in Table 5 (section 3.2.3, page 87). A reduction in fungal fresh weight of 83 % was caused by 2.75 mM CHA while 0.2 mM norspermidine reduced growth by 75 % (Table 5).

4.3.3.2 Effect of 1 day exposure to higher concentrations of CHA and norspermidine on growth of *Pyrenophora avenae*

When fungus was exposed to higher concentrations of CHA or norspermidine for 1 day only, by transferring 2 day old mycelium into fresh medium, the fresh weight of the fungus decreased (Table 17). Fresh weight of control fungus increased by 118 % while fungus exposed to 2.75 mM CHA decreased in weight by 10 % and that exposed to 0.2 mM norspermidine decreased by 44 % (Table 17). The exact mechanism by which the compounds affected the fungus, causing a reduction in fresh weight as opposed to simply preventing further fungal growth, is unclear. It is possible that CHA and norspermidine affected the composition of the fungal cells so that, for example, the treated cells contained less water.

4.3.3.3 Incorporation of radio-labelled ornithine into polyamines

No effect was observed for the incorporation of radio-labelled ornithine into polyamines in fungus exposed to higher concentrations of CHA and norspermidine for 1 or 4 days (Table 18). Little radioactivity was found in putrescine (data not shown) with most being detected in spermidine and spermine (Table 18).

4.3.3.4 Analysis of free polyamines in fungal tissue

When fungus was exposed to higher concentrations of the inhibitors for 1 day, spermidine content in tissue exposed to 2.75 mM CHA decreased by 56.6 % (Table 19). Spermine content increased, although this result should be discounted as CHA co-chromatographed with spermine. No effect on putrescine was observed. Exposure to 0.2 mM norspermidine led to an increase in spermidine, although this should also be discounted since norspermidine co-chromatographs with spermidine (Table 19).

Exposure of fungus to higher concentrations of CHA and norspermidine for 4 days caused quite different effects. Thus, 2.75 mM CHA led to a 364 % increase in putrescine and a 20 % reduction in spermidine (Table 20). Again, the observed increase in spermine levels should be discounted. Little effect was caused by exposure of fungus to 0.2 mM norspermidine since the increase in spermidine should be discounted as indicated above (Table 20). The increase in spermine levels was probably the result of norspermidine co-chromatographing with spermidine, the high level of spermidine/norspermidine causing some to be detected along with spermine which is found very close to spermidine on the tlc plates.

Such instances of co-chromatography were indeed apparent in initial investigations into the effects of 1.0 mM CHA and 0.01 mM norspermidine on fungal free polyamine levels, albeit, infrequently. However, these later studies have used increased concentrations of CHA and norspermidine and it is possible that these compounds and not the polyamines, account for the increases in fluorescence observed at the 'spermine' and 'spermidine' spots respectively.

Treatment	Mean fresh weight after	Mean fresh weight after	Percentage change wrt 2
	2 days growth (g)	1 further days growth (g)	day old fungus
Control	0.86 ± 0.012	$\textbf{1.88} \pm \textbf{0.072}$	+118
2.75 mM CHA	0.91 ± 0.034	0.82 ± 0.152	- 10
0.2 mM norspermidine	0.93 ± 0.017	0.52 ± 0.011	- 44

М

1000

Table 17: Effect of 1 day exposure to higher concentrations of CHA and norspermidine on growth of *Pyrenophora avenae*

Note. Values are shown as the means \pm SE of four replicates.

<u>Table 18.</u> Effect of 1 or 4 day exposure to higher concentrations of CHA and norspermidine on the incorporation of radio-labelled ornithine into polyamines in *Pyrenophora avenae*

Treatment	Radioactivity in poly	nine (dpm g ⁻¹ FW)		
Exposure for 4 days	Spermidine	Spermine		
Control	83.8 ± 9.73	22.1 ± 3.6		
2.75 mM CHA	89.1 ± 9.28	22.9 ± 1.55		
0.2 mM norspermidine	98.4 ± 10.92	25.5 ± 5.53		
Exposure for 1 day	Spermidine	Spermine		
Control	184 ± 18.2	32.0 ± 1.77		
2.75 mM CHA	177 ± 10.4	36.1 ± 1.90		
0.2 mM norspermidine	197 ± 12.1	45.0 ± 5.07		

Note: Values are shown as the means \pm SE of four replicates.

	Polyamine concentration (μ mol g ⁻¹ FW)		
	Control	CHA	norspermidine
Putreseine	273 ± 12.5	280 ± 25.9	256 ± 17.9
Cadaverine	75.2 ± 5.65	84.3 ± 9.59	85.9 ± 13.40
Spermidine	318 ± 3.8	138 ± 13.7 **	594 ± 41.2 **
Spermine	134 ± 7.7	192 ± 12.2 **	188 ± 27.9

Table 19. Effect of 1 day exposure to higher concentrations of CHA and norspermidine on free polyamine levels in *Pyrenophora avenae*

<u>,</u>

Note: Values are shown as the means \pm SE of four replicates. Significant differences from control are shown as: ** $p \le 0.01$

Table 20. Effect of 4 day exposure to higher concentrations_of CHA and norspermidine on free polyamine levels in *Pyrenophora avenae*

	Polyamine concentration (µmol g ⁻¹ FW))
	Control	CHA	Control	norspermidine
Putrescine	77 ± 11.2	357 ± 19.7 ***	137 ± 5.0	152 ± 6.8
Cadaverine	26.8 ± 1.88	25.8 ± 2.54	32.5 ± 2.07	42.3 ± 4.14
Spermidine	118 ± 5.7	94 ± 5.2 *	150 ± 2.8	667 ± 15.9 ***
Spermine	37.1 ± 5.00	67.6 ± 6.61 *	49.1 ± 4.22	64.6±3.76*

Note: Values are shown as the means \pm SE of four replicates.

Significant differences from control are shown as: * $p \le 0.05$; ** $p \le 0.01$; *** $p \le 0.001$

4.4 FURTHER DETERMINATION OF THE EFFECTS OF HIGHER CONCENTRATIONS OF CHA AND NORSPERMIDINE ON BIOSYNTHESIS AND CATABOLISM OF POLYAMINES IN *PYRENOPHORA AVENAE*

4.4.1 Introduction and Objectives

Polyamine biosynthetic enzymes appeared to be unaffected in fungus exposed to higher concentrations of CHA and norspermidine as measured by examining the flux of radio labelled ornithine through to polyamines. Exposure to 2.75 mM CHA for 4 days did, however, cause an increase in fungal free putrescine levels. A slight decrease in spermidine levels was also detected. Interestingly, exposure of fungus to 2.75 mM CHA for one day failed to produce an increase in putrescine but caused a substantial decrease in spermidine, suggesting that the fluctuations in spermidine and putrescine levels were connected. Further work was therefore carried out to clarify these results.

The changes in free polyamine levels observed may have been caused by one or more of the numerous possible mechanisms. As reviewed earlier, (section 1.1.3.3, page 14), catabolism processes known for mammalian cells include the degradation of spermidine to an acetylated derivative which is in turn acted upon by PAO to yield putrescine. This process may account for the increase in putrescine detected and the reduction in spermidine levels observed. Thus, PAO activity was determined.

PAO may have acted upon conjugated N^{1} -acetyl derivatives of spermine and spermidine; i.e. derivatives that exist in their own right and not as products of SSAT. This would have led to the formation of spermidine and putrescine, and if catabolism of that spermidine to putrescine occurred, would account for the highly elevated levels of putrescine detected. N^{8} -acetylspermidine and N-acetylputrescine may have been hydrolysed to spermidine and putrescine to add to the free polyamine pool, or indeed, spermidine and putrescine removed by conjugation. Detection of acetylated amines was therefore carried out with the exception of N^{1} -acetylspermine whose availability was restricted.

Such acetyl derivatisation of spermidine, removing free spermidine from the polyamine pool and leading to reductions in spermidine levels warranted further investigation. Formation of acetylated amines from labelled ornithine was therefore studied.

Fluctuations in polyamine levels may result from excretion of amines, thus, this process was also studied. Similarly, uptake of amines can often account for fluctuations in the free polyamine pool, and this process will be considered in chapter 7.

Although the higher concentration of norspermidine did not cause any effects on free polyamine levels in fungus exposed for 1 or 4 days, acetylated amine levels were determined in case excretion or uptake of amines had masked a perturbation of polyamine content. As previously described, (section 4.2.1), this would indicate whether further enzymatic studies were necessary.

4.4.2 Materials and Methods

4.4.2.1 PAO activity

PAO activity was determined as described in section 4.2.2.2,

4.4.2.2 Analysis of acetylated polyamines in fungal tissue

Levels of acetylated polyamines were determined as described in section 4.2.2.4.

4.4.2.3 Formation of acetylated polyamines

Study of the formation of acetylated amines was attempted as follows. Fungus was grown in control medium for 2 days before being transferred into medium (50 ml), either unamended or amended to give a concentration of 2.75 mM CHA. 4 µl of $\{U^{-14}C\}$ ornithine was added to each flask and the flasks incubated at 24°C for 24 hours in a Gallenkamp orbital shaker (140 rpm). Fungus was harvested and analysed for free and acetylated polyamines as described previously except that amines were scraped from the tle plates and placed into scintillant and radioactivity counted.

4.4.2.4 Determination of amine excretion

As described previously, fungus was exposed to 2.75 mM CHA for 1 or 4 days. After harvesting, fungal mycelium was placed into 10 ml of an amine free buffer. The buffer used was a Citric acid - Na₂HPO₄ (MeIlvaine) buffer at pH 4^4 . This buffer was chosen since fresh PDB has a pH of 4.85 and it is known to acidify in the presence of growing fungus. Fungal mycelium was mixed gently in the buffer using a rotamixer and left to stand at room temperature for 6 hours before harvesting the fungus by centrifugation. The buffer was retained and aliquots (100 µl) were analysed for free polyamine presence as described in section 4.1.2.5.

⁴ 100 ml of buffer contained 61.45 ml of 0.1 M citric acid monohydrate and 38.55 ml of 0.2 M Na₂HPO₄, as described in "Data for Biochemical Research", (1986), (eds., Dawson, R. M. C., Elliott, D. C., Elliot, W. C. & Jones, K. M.), 3rd edition, Claredon Press, Oxford.

4.4.3 Results

4.4.3.1 PAO activity

Little difference was observed in PAO activity in fungus exposed to 2.75 mM CHA for 1 or 4 days (Table 21).

4.4.3.2 Analysis of acctylated polyamines in fungal tissue

Pooled acetylated spermidine levels were reduced upon exposure of fungus to 2.75 mM CHA, for 1 or 4 days, although the reduction for 1 day exposure was not always significant (Table 22). N-acetylputrescine levels were unaltered in fungus exposed to 2.75 mM CHA for either 1 or 4 days. Exposure of fungus to 0.2 mM norspermidine did not affect acetylated polyamine levels at either of the timings examined (Table 22).

4.4.3.3 Formation of acetylated polyamines

Determination of acetylated polyamines from labelled ornithine was largely unsuccessful. Results showed that despite using a large amount of radio-labelled ornithine, radioactivity was found largely in free spermidine and spermine with little or no radioactivity being detected in acetylated amines (data not shown). It is conceivable that acetylated amines may be formed from a spermidine pool quite distinct from the spermidine pool formed from ornithine. Different pools of free polyamines may exist in fungal cells, as they do in plants (Davis 1990).

4.4.3.4 Determination of amine excretion

Little differences were observed in amines excreted from fungus exposed to 2.75 mM CHA for 1 or 4 days compared to the control fungus (Table 23). Spermine was not detected (data not shown).

	Enzyme activity [pmol product (mg protein) ⁻¹ hr ⁻¹]		
Exposure	Control	CHA	
l day	17.6 ± 1.61	23.3 ± 2.21	
4 days	31.1 ± 5.60	33.0 ± 1.72	

Table 21: Effect of 1 or 4 day exposure to 2.75 mM CHA on PAO activity in Pyrenophora

<u>avenae</u>

Note: Values are shown as the means \pm SE of four replicates.

<u>Table 22:</u> Effect of 1 or 4 day exposure to higher concentrations of CHA and norspermidine on acetylated polyamine levels in *Pyrenophora avenae*.

	_	Polyamine concentration (μ mol g ⁻¹ FW)		
	Treatment	N-acetylputrescine	N-acetylspermidines	
1 day exposure	Control	109 ± 7.4	116 ± 13.4	
	2.75 mM CHA	125 ± 19.7	61 ± 2.5	
	0.2 mM norspermidine	116 ± 18.9	110 ± 23.0	
4 day exposure	Control	67.1 ± 1.23	65.3 ± 8.28	
	2.75 mM CHA	58.1 ± 5.12	20.6 ± 1.23 *	
	Control	75.4 ± 2.88	59.9 ± 8.14	
	0.2 mM norspermidine	68.0 ± 4.36	46.6 ± 4.38	

Note: Values are shown as the means \pm SE of four replicates. Significant differences from control are shown as: * $p \le 0.05$

Table 23: Effect of 1 or 4 day exposure to 2.75 mM CHA on polyamine excretion from Pyrenophora avenae.

	Polyamine concentration (µg / ml buffer)				
	Control	CHA	Control	CHA	
	1 day exposure		4 day exposure		
spermidine	11.45 ± 0.846	15.30 ± 1.780	12.36 ± 0.466	16.82 ± 1.580	
putrescine	9.01 ± 0.530	11.56 ± 1.140	9.07 ± 0.460	8.62±1.340	

Note: Values are shown as the means \pm SE of four replicates.

4.5 DISCUSSION

The present study found that 1.0 mM CHA increased the level of free putrescine in fungal tissue exposed to the inhibitor. Increased putrescine and decreased spermidine levels are the widely reported results of spermidine synthase inhibition. Such CHA-induced changes in polyamine levels have been observed in *Helianthus tuberosus* extracts (Torrigiani *et al.* 1987), pine cotyledons (Biondi *et al.* 1986) and cultured animal cells (Mitchell *et al.* 1985). However, in the absence of an effect on ODC and AdoMetDC activities or the flux of radio labelled ornithine into polyamines, indicating that spermidine synthase was unaffected, the increase in putrescine cannot be attributed to inhibition of spermidine synthase.

In mammalian cells, putrescine can be formed by a 'back-conversion' pathway involving the action of spermidine/spermine N^1 -acetyltransferase and polyamine oxidase on spermine and spermidine (Pegg 1988). However, levels of N^1 -acetylspermidine and PAO activity were unaltered, thus, increased catabolism of amines was unlikely to be the cause of the putrescine increase. Such non correlation between the elevated putrescine level and SSAT activity is in agreement with the reports that SSAT activity is not induced by putrescine (Pegg 1988).

The increase in putrescine does not appear to be the result of reduced diamine oxidase activity, the enzyme degrading putrescine. No increase in N^8 -acetylated spermidine or N-acetylated putrescine levels was detected, and thus, the increase in free putrescine cannot be accounted for by the breakdown of acetylated polyamine derivatives. Although such conjugation and breakdown pathways have been characterised for mammalian cells only (Seiler 1988), deacetylase enzymes have been reported in fungi (Haywood & Large 1986), therefore suggesting that breakdown of conjugated derivatives is indeed possible in fungi.

Neither 0.01 mM norspermidine, 0.5 mM N¹- nor 0.5 mM N⁸-acetylspermidine had any effect on the polyamine biosynthetic enzymes examined or the levels of free polyamines in *P. avenae*. Little information exists on the effects of such analogues on spermidine synthase activity. Masse and co - workers (1985, 1988) did indicate that norspermidine-induced growth effects on maize roots and potatoes were due to the perturbation of spermidine biosynthesis. Similarly, the spermidine analogues N¹, N⁸-bis-(ethyl)spermidine (BES) and N¹, N⁸-bis(propyl)spermidine (BPS), not only inhibited growth of cultured L1210 cclls, but altered polyamine biosynthesis (Porter *et al.* 1985). The results of the present study are, therefore, in contrast to previous findings.

However, a norspermidine-induced effect on cultured HeLa cells was attributed both to effects on polyamine biosynthesis and mitotic activity (Sunkara *et al.* 1988), while Porter *et al.* (1985) found an additional spermidine analogue inhibited L1210 cell growth, but were unable to determine its mode(s) of action. The results of the present study, therefore, agree with those findings, although exactly why differences exist in the effects of the analogues in different organisms is unclear.

Ņ

The higher (0.2 mM) concentrations of norspermidine had similar effects on *P. avenae:* no effects on the flux of radio labelled ornithine through to polyamines was observed. Free and acetylated polyamine levels were also unaltered. Thus, norspermidine appeared to be exerting its antiproliferative effects without an effect on polyamine biosynthesis.

The higher concentration of CHA (2.75 mM) failed to produce an effect on the flux of radio labelled ornithine through to polyamines in *P. avenae*, suggesting that spermidine synthase activity was unaffected by the compound. However, after 1 days exposure to the compound, spermidine levels were depleted while after 4 days exposure to CHA, spermidine levels were still depleted although, less so, and putrescine levels were elevated substantially. The absence of an initial effect on putrescine and the later smaller depletion of spermidine, suggests that the two phenomena are related. Had catabolism occurred to deplete spermidine and increase putrescine, this would have accounted for the observed results. However, no effect on PAO activity was detected. No effect on acetylated putrescine levels was found, thus breakdown of conjugated polyamines does not account for the increase in the free putrescine pool.

Thus, with both 1.0 mM and 2.75 mM CHA, an increase in putrescine levels in fungus exposed to the inhibitors remains unaccounted for. Exactly why spermidine levels decrease is also unclear. Conjugation would account for a reduction in the free spermidine pool, although, levels of acetylated spermidines also decreased. It should be noted that such reductions in acetylated spermidines may have accounted for the increase in putrescine but without an effect on PAO activity, this is unlikely.

Nevertheless, the above discussion of catabolism is based largely upon mechanisms characterised for mammalian cells. Knowledge of catabolism in many microorganisms is fragmentary (Large 1992), and indeed, little is known of such pathways in plant pathogenic fungi (Walters 1995). Large (1992) reports how two main catabolic pathways exist for many microorganisms, the N-acetylated route and an alternative route, involving Δ^1 -

pyrroline as an intermediate. Thus, it may be possible that an as yet uncharacterised catabolic pathway exists in *P. avenae.*

Regulation of intracellular polyamines may also involve selective uptake of polyamines, and this is considered later in this thesis. Excretion of polyamines may also be associated with abberant intracellular polyamine levels, although no evidence of such a mechanism was detected. Release of bound polyamines from, for example, macromolecular binding sites, may account for observed increases in putrescine content while stimulated binding of polyamines would account for the reduction in free spermidine. Indeed, Paulus *et al.* (1993) noted how the majority of polyamines in *Neurospora crassa* were bound to cell constituents. Further work would be necessary to determine bound polyamine levels in *P. avenae*.

It should be noted that an accumulation of putrescine is a widely reported response to stress in an organism, particularly plants (for example, Foster & Walters 1991). Bound putrescine could, therefore, have been released in an attempt to off-set the stress caused by exposure of fungus to CHA. A similar phenomenon was observed by Havis *et al.* (1996b), with spermine being displaced from binding sites following the exposure of *Phytophthora infestans* to E-TED. Exactly how elevated putrescine regulates stress is unknown but it has been associated with increasing membrane stability (Marton & Morris 1987), maintenance of cellular pH (Smith & Sinclair 1967), increased protein synthesis (Kaur-Sawhney *et al.* 1980) and synthesis of protective alkaloids (Hartmann *et al.* 1988).

In contrast to the hypothesis that elevated putrescine occurs to enable the fungus to withstand the effects of CHA, the antifungal activity of the compound may also be associated with the increase in putrescine concentration. An accumulation of putrescine was suggested by Davis & Ristow (1991) to be responsible for the reduction in growth of a mutant of *N. crassa*. Excess putrescine has been shown to be toxic in the cyanobacterium *Anabaena* (Guarino & Cohen 1979). Also, in spite of Dumbroff (1991) suggesting that polyamines may act as free radical scavengers, DiTomaso *et al.* (1989) hypothesised that catalytic breakdown of excess putrescine by DAO to hydrogen peroxide and free radicals, caused membrane damage of maize roots.

The findings for CHA are thus in contrast to many which indicate that CHA inhibits spermidine synthase from a variety of sources. Examples include spermidine synthase from mammalian cells, trypanosomes, certain bacteria (Pegg & Williams-Ashman 1987) and from Chinese cabbage leaves (Sindhu & Cohen 1984). However, work on other polyamine biosynthetic enzymes has shown that enzymes will react differently upon exposure to an

2
inhibitor *in vitro* and *in vivo* (Walters *et al.* 1995a). Such a phenomenon could, therefore, explain the lack of effect by CHA on spermidine synthase in fungi. Nevertheless, CHA is known to inhibit spermidine synthase in protoplasts extracted from Chinese cabbage leaves (Greenberg & Cohen 1985). Other authors, however, have observed CHA to cause an inhibitory growth effect on potatoes (Masse *et al.* 1988) and in tobacco callus culture systems (Tiburcio *et al.* 1987), but have been unable to fully account for this effect by perturbation of polyamine biosynthesis, and indeed, suggest that spermidine synthase inhibition is unlikely.

and the second water and the

1, 2 + 1, 2 + 2, 2

and the second second

Previous work with CHA, norspermidine and spermidine analogues in other organisms aside, the findings of the present study with fungi were not entirely unexpected. Indeed, work using putrescine analogues had concluded that observed effects on polyamine biosynthesis in fungi grown *in vitro* were unlikely to have been wholly responsible for reductions in fungal growth (Havis *et al.* 1994a,c, 1997). Havis *et al.* (1996a) also found cyclic diamines possessed fungicidal activity against powdery mildew on barley yet failed to affect polyamine biosynthesis of *P. avenae* grown *in vitro*. Further work is therefore necessary to clucidate the mechanism(s) by which CHA and spermidine analogues exert their antifungal effects.

Future work will be discussed in the conclusion of this thesis. There are, however, some specific areas associated with the work reported in this chapter which may be worthy of future exploration.

CHA was found to co-chromatograph with spermine whilst norspermidine cochromatographed with spermidine. Such a phenomenon meant that fluctuations in these polyamines were generally ignored, especially when dealing with fungus exposed to the higher concentrations of CIIA and norspermidine. Development of a more accurate method of polyamine analysis would be prudent since 'discounting' such results may mean that fluctuations in free polyamine levels are undetected. It should also be noted that APC and 3APC, higher cadaverine homologues, also co-chromatograph with spermidine and spermine respectively (Zarb & Walters 1994b). Production of these homologues may also affect amine detection. Indeed, Alhonen-Hongisto & Jänne (1980) reported how tumour cells in which polyamine biosynthesis had been perturbed, produced APC and 3APC. APC and 3APC formation has been reported for *P. avenae* and mycorrhizal species (Zarb & Walters 1994b) as well as *N. crassa* starved of polyamines (Paulus *et al.* 1982). However, cadaverine levels were unaltered in *P. avenae* exposed to CHA and spermidine analogues, thus, biosynthesis of the higher homologues was unlikely to have occurred. Interestingly, CHA inhibited formation of the higher homologues in extracts of *P. avenae* (Zarb & Walters 1994b).

Determination of amine excretion was attempted by placing fungus, which had been grown for 1 or 4 days in the presence of CHA, into an amine free buffer for some hours before analysing the buffer for polyamine presence. However, this failed to account for those polyamines already excreted over the four day growing period. Excretion is likely to occur via a passive diffusion mechanism once intracellular polyamine levels become too high (Davis 1990). Thus, excretion may have occurred before its determination was attempted in this study. Polyamines are present in the growing medium of fungus, thus, taking aliquots of the growing medium for analysis throughout the growth period of the fungus is not desirable. However, given a sensitive analytical method for determining polyamines, this process would give a clearer impression of the effects of inhibitors on excretion of amines by a growing fungus.

As discussed, excess putrescine may regulate stress by stabilising membranes but may also cause inhibition of growth by damaging membranes. Further work examining ion efflux from fungus grown in the presence of inhibitors may indicate if cellular membranes are damaged.

Chapter 5

Effect of novel spermidine analogues on *in vitro* growth and polyamine biosynthesis in *Pyrenophora avenae*

5. EFFECT OF NOVEL SPERMIDINE ANALOGUES ON *IN VITRO* GROWTH AND POLYAMINE BIOSYNTHESIS OF *PYRENOPHORA AVENAE*

5.1 INTRODUCTION AND OBJECTIVES

Given the antifungal and fungicidal activity of commercially available spermidine analogues, albeit with their mode(s) of action as yet undetermined, various novel spermidine analogues were synthesised and their antifungal activity determined by examining their effect on the growth of *P. avenae* on solid media. This work was part of an on-going collaborative research programme with Professor David Robins of the Chemistry Department at Glasgow University, who synthesised the novel analogues. Solid media studies were chosen over a study of the effects in liquid culture since limited amounts of each chemical were available, even though effects on growth on solid media can be misleading. Analogues which possessed significant antifungal activity were then examined further, determining their effect on growth of *P. avenae* in liquid culture and on fungal polyamine biosynthesis by carrying out biochemical analyses.

5.2 EFFECT OF NOVEL SPERMIDINE ANALOGUES ON MYCELIAL GROWTH OF *Pyrenophora avenae* on solid media

5.2.1 Materials and Methods

P. avenae was maintained as described in section 3.1.2.1 (page 76). Fungus was grown on solid media amended individually with eight novel spermidine analogues (Table 24), to give final concentrations between 0.05 mM and 2.0 mM. The concentrations were chosen on the basis of previous work using CHA and norspermidine which utilised such concentrations (section 3.1.2.2, page 76). Four replicates were used for each treatment with significance being assessed using Student's *t*-test. All experiments were repeated and similar results obtained.

5.2.2 Results

Mycelial growth was successfully inhibited by two of the novel spermidine analogues, LAS 2/12 and LAS 2/28, at concentrations of 2.0 mM (Figures 11-14; Appendix 8). LAS 2/12 reduced fungal growth by 47.3 %, while LAS 2/28 reduced growth by 34.5 %, eight days after inoculation ($p \le 0.001$). None of the other compounds had any great effect on mycelial growth of *P. avenae* on solid media at the concentrations examined (Figures 11-14; Appendix 8).

Analogue	Structure
C59	H2N(CH2)3NII(CH2)4NHC2H5
C73	H2N(CH2)3NH(CH2)5NHC2H5
LAS 2/11	H2N(CH2)3NHCH2CH=CHCH2N(C2H5)C2H5
LAS 2/12	H2N(CH2)3NH(CH2)4N(C2H5)C2H5
LAS 2/26	H2N(CH2)3NH(CH2)3N(CH3)CH3
LAS 2/27	H2N(CH2)3NH(CH2)4N(CH3)CH3
LAS 2/28	H2N(CH2)3NH(CH2)3N(C2H5)C2H5
LAS 3/8	H2N(CH2)3NHCH2CH=CHCH2N(CH3)CH3

Table 24: Structures of novel spermidine analogues

24

solid media. Radial mycelial growth was measured 3, 6 and 8 days after inoculation of plates. Values are the means of 3 measurements from each of 4 replicates with standard errors of means indicated. Significant differences from control are indicated in Appendix 8a. Figure 11. Effect of varying concentrations (0.05 - 2 mM) of the novel spermidine analogues C59 and C73 on mycelial growth of Pyrenophora avenae on



measurements from each of 4 replicates with standard errors of means indicated. Significant differences from control are indicated in Appendix 8a. Pyrenophora avenae on solid media. Radial mycelial growth was measured 3, 6 and 8 days after inoculation of plates. Values are the means of 3 Figure 12. Effect of varying concentrations (0.05 - 2 mM) of the novel spermidine analogues LAS 2/11 and LAS 2/12 on mycelial growth of







6 days □8 days

■3 days

Mean radial mycelial growth (mm) 8

3

8

10

1

5

8

8

Effect of LAS 2/26 and LAS 2/27 on mycelial growth of Pyrenophora avenae

H

H

H





5.3 EFFECT OF NOVEL SPERMIDINE ANALOGUES ON GROWTH OF *Pyrenophora Avenae* in Liquid Culture

the second se

5.3.1 Materials and Methods

P. avenae was grown in liquid culture as described in section 3.2.2.2 (page 85), in the presence of either 1.0 mM LAS 2/28, C59, C73 or 0.5 mM LAS 2/12. A concentration of 1.0 mM was chosen for those compounds since previous work had shown that CHA affected growth of *P. avenae* in liquid culture at that concentration. 0.5 mM was chosen as the concentration of LAS 2/12 to be studied since a preliminary experiment found that 1.0 mM LAS 2/12 completely inhibited growth of *P. avenae* (data not shown).

LAS 2/12 and LAS 2/28 were chosen for further investigation given their successful inhibition of mycelial growth of *P. avenae* on solid media. Although C59 and C73 failed to affect mycelial growth of *P. avenae* on solid media, previous glasshouse experiments had shown that these compounds possessed fungicidal activity against powdery mildew on barley, rust on broad bean and powdery mildew on apple (N. D. Havis unpublished results). Therefore, further investigation of these compounds was also appropriate.

Since the amounts of these compounds was limited, fungus was only grown in liquid culture for 4 days. Previous work with the commercially available spermidine analogues had found little differences in *P. avenae* exposed to the compounds for 2, 3 or 4 days, thus examination of 4 day old fungus only was deemed appropriate. Limited amounts of chemical meant that only two replicates were used. Significance was assessed using Student's *t*-test. Experiments were repeated and similar results were obtained.

5.3.2 Results

All of the compounds examined reduced fungal growth of *P. avenae* in liquid culture 4 days after inoculation. 0.5 mM LAS 2/12 reduced growth by 92.4 %, while 1.0 mM concentrations of C59, C73 and LAS 2/28 reduced fungal growth by 56.0, 65.5 and 87.1 % respectively (Table 25). Only growth in LAS 2/28 was significantly different from the control since a small number of replicates were used.

5.4 EFFECT OF NOVEL SPERMIDINE ANALOGUES ON THE INCORPORATION OF RADIO-LABELLED ORNITHINE INTO POLYAMINES IN *Pyrenophora avenae* grown In Liquid culture

5.4.1 Materials and Methods

The incorporation of radio-labelled ornithine into polyamines was carried out as described in section 4.1.2.4 (page 97). Such an analysis was chosen to determine if spermidine formation was affected in *P. avenae* grown in liquid culture in the presence of the novel spermidine analogues. Due to the limited amount of fungal material available, this was the only analysis carried out, although, it would also provide an indication of the activities of ODC and AdoMetDC. All results are the means of four replicates with significance being assessed using Student's *t*-test. All experiments were repeated with similar results obtained.

5.4.2 Results

Neither 0.5 mM LAS 2/12, 1.0 mM LAS 2/28, 1.0 mM C59 or 1.0 mM C73 had any effect on the incorporation of radio-labelled ornithine into polyamines. All radioactivity was found in spermidine or spermine with none in putrescine (Table 26).

「おいたいで、 というない しょう

<u>Table 25.</u>	Effect of novel	spermidine	analogues	<u>on growth</u>	of Pyrenophor	<u>a avenae i</u>	<u>n liquid</u>
		į	culture for	4 days			

Treatment	Growth of fungus, measured as mean fresh
	weight (g)
Control	4.6 ± 0.19
1,0 mM LAS 2/28	0.6±0.02 *
Control	5.6 ± 0.87
1.0 mM C59	$\textbf{2.5}\pm\textbf{0.16}$
1.0 mM C73	1.9 ± 0.12
0.5 mM LAS 2/12	0.4 ± 0.01

Note: Values are shown as the means \pm SE of two replicates. Significant difference from control is as: * $p \le 0.05$

Table 26. Effect of novel spermidine analogues on the incorporation of radio-labelled

ornithine into polyamines in Pyrenophora avenae grown for 4 days

	Radioactivity in poly	amine (dpm g-' FW)
Treatment	Spermine	Spermidine
Control	8.6 ± 3.04	284 ± 11.0
1.0 mM LAS 2/28	17.4 ± 1.7	253 ± 18.2
Control	6.4±0.35	160 ± 13.4
0.5 mM LAS 2/12	11.8 ± 2.29	135 ± 17.7
1.0 mM C59	7.2 ± 1.87	194 ± 11.1
1.0 mM C73	11.4 ± 2.35	172 ± 26.2

Note: Values are shown as the means \pm SE of four replicates.

5.5 DISCUSSION

The observed inhibition of fungal growth by novel spermidine analogues agrees with previous work by Porter *et al.* (1985) who showed that analogues of spermidine inhibited growth of L1210 cultured cells. Also, the findings are in agreement with the previous work carried out in this study, when a commercially available analogue, norspermidine, was observed to possess antifungal properties against pathogenic fungi *in vitro*.

and the second se

However, growth of P. avenae was insensitive towards many of the inhibitors, although this may be indicative of the unreliability of studies of fungal growth on solid media (Isaac & Jennings 1995). As discussed previously, 1.0 mM E-TED was ineffective against P. avenae on solid media yet completely inhibited growth of the fungus in liquid culture (Havis *et al.* 1994c). Thus, compounds which did not affect mycelial growth may possess antifungal activity. Limited amounts of the compounds prevented further work, although this may an area for future interest.

On the other hand, the compounds may be truly insensitive towards *P. avenae*. Growth of some fungal species on solid media was reported to be unaffected by CHA (West & Walters 1989). There may be a number of reasons for this phenomenon, as already discussed in previous chapters. Briefly, Birecka *et al.* (1986) suggested that differences in fungal sensitivity to inhibitors may be due to varying uptake of inhibitor by the fungus or differences in fungal enzyme sensitivity to the inhibitors. Another possibility for fungal insensitivity towards the analogues may be rapid metabolism of the inhibitors, as suggested by Khan & Minocha (1989a). Further work would be required to establish reasons for the observed differences between the analogues in their effect on *P. avenae* growth. It should be noted that even stereo-isomers of the same compound, for example E-BED and Z-BED (Havis *et al.* 1994a) and trans BAD and cis BAD (Havis *et al.* 1996a) produce very different effects on growing fungus.

The four compounds examined did not appear to affect the flux of radio labelled ornithine through to polyamines. This indirect examination of spermidine synthase activity demonstrates that the analogues do not inhibit fungal growth *in vitro* through an effect on the enzyme. Given the results of the study so far, this finding is not unexpected. In other work, novel putrescine analogues did indeed perturb polyamine metabolism yet this phenomenon was not suggested as the main cause of the observed antifungal effects (Havis *et al.* 1994a,c). These workers also went onto demonstrate how some cyclic putrescine analogues perturbed polyamine metabolism while others had no effect and, as with the aliphatic analogues, any effect on polyamine biosynthesis was not thought to wholly account for the antifungal properties of the analogues (Havis *et al.* 1996a, 1997).

In their original hypothesis, Porter & Sufrin (1986) suggested inhibition of a polyamine biosynthetic enzyme to be only one potential mode of action by which analogues may exert antiproliferative activity. Alternative suggested mechanisms of effect include regulation of enzymes, binding to sites associated with cellular proliferation and binding to macromolecules and disrupting their structure and/or function. Thus, the novel spermidine analogues showed potential as antifungal agents although further work is necessary to elucidate their mode(s) of action.

in an

合います

Chapter 6

Field studies

6. FIELD STUDIES

6.1 Introduction and Objectives

It has already been demonstrated in this study that CHA and norspermidine possess fungicidal activity against a number of plant pathogens (Chapter 2). While glasshouse experiments are important for preliminary screening, such studies are not always true reflections of the behaviour of a compound in the field. For compounds to be of any use agriculturally, they must show activity against fungal pathogens in the field. Thus, despite the mode of action of the compounds being undetermined, small scale field trials were carried out to evaluate CHA and norspermidine as novel fungicides against powdery mildew on spring barley.

6.2 Materials and Methods

6.2.1 1995 Trial

Spring barley (cv. Prisma) was sown at a row spacing of 12 cm with a seed rate of 220 kg.ha⁻¹. Plots (4.0 m x 2.0 m) were arranged in a randomised block design with four replicates. With the exception of disease control, normal husbandry practices were employed throughout the growth of the erop. 500 kg.ha⁻¹ of a 20:10:10 compound fertiliser was applied at sowing (100:50:50 kg.ha⁻¹ NPK) and a mixture of two broad spectrum herbicides was applied at GS 15 (Zadoks growth stage: five leaves unfolded). Metsulfuronmethyl (Ally, Du Pont) and fluroxypyr + ioxynil + bromoxynil (Advance, Zeneca) control many broad leaved weeds including charlock, chickweed, cleavers, hemp-nettle, knotgrass, redshank and deadnettles.

To evaluate CHA and norspermidine as novel fungicides against powdery mildew, various treatments were applied at GS 32 (Zadoks growth stage: second node detectable), when mildew was initially observed. CHA and norspermidine were both applied at concentrations of 1.0 mM and 5.0 mM along with mixtures of 5.0 mM CHA + 5.0 mM E-TED⁵ and 5.0 mM norspermidine + 5.0 mM E-TED. Jänne *et al.* (1985) suggested that by using an inhibitor of AdoMetDC along with one of ODC, complete depletion of putrescine, spermidine and spermine in mammalian cells should be possible. DFMO often failed to deplete cells of all polyamines and those remaining amines were thought to be responsible for sustained cellular growth. The combination of two analogues, proposed to affect spermidine synthase and ODC, was thus thought to be a useful additional investigation in this field trial.

⁵ E-TED was provided by Professor D. J. Robins of Glasgow University. Synthesis of the compound is described by Havis *et al.* (1994a).

The treatments were applied using an AZO compressed air sprayer at a rate of 300 l.ha⁻¹. Treatments were made up in water with Agral (alkyl phenol ethylene oxide, Zeneca) at a rate of 0.5 l.ha⁻¹. A commercial fungicide, a mixture of flutriafol + carbendazim, (Early Impact, Zeneca; 118 g and 188 g a.i.ha⁻¹ respectively) was also evaluated. Ten plants per plot were assessed 7 and 14 days after treatment by estimating the percentage mildew infection on the third leaf using a standard area diagram.

and the second se

All plots except those treated with the mixture of 5.0 mM CHA + 5.0 mM E-TED were sprayed a second time at GS 59 (Zadoks growth stage: emergence of inflorescence) and the percentage mildew infection on the flag leaves assessed 7, 14 and 21 days later. Limited amounts of E-TED restricted use of this chemical. Since preliminary results had shown the initial CHA + E-TED spray to be less effective in controlling powdery mildew infection than the other treatments, it was omitted from the second spray programme.

Ten plants and 100 cars per plot were harvested and measurements made of plant height, plant dry weight and grain dry weight at 14.5 % moisture content.

Disease assessment results were analysed using an analysis of variance of square root transformed data. Data were transformed to acquire the normal distribution necessary for statistical analysis to be carried out. Plant height, plant dry weight and grain dry weight results were simply analysed using an analysis of variance.

6.2.2 1996 Trial

The 1996 field trial was carried out as described for the 1995 trial in section 6.2.1 with the exception of the herbicide application, where metsulfuron-methyl (Ally, Du Pont) was applied at GS 32. Fungicidal evaluation was carried out with sprays at GS 32 and GS 59 and two and three disease assessments after the respective sprays. CHA (1.0 mM and 5.0 mM) and norspermidine (1.0 mM and 5.0 mM), and the mixture of flutriafol + carbendazim (Early Impact, Zeneca), were re-evaluated. Additional compounds were also evaluated: tebuconazole (Folicur, Bayer; 250 a.i.ha⁻¹), a half dose of the fungicide and mixtures of a half dose of the fungicide ± 1.0 mM CHA or 1.0 mM norspermidine. Strategies for combatting the onset of fungicide resistance are vital to extend the life of the decreasing number of active ingredients available. Mixtures of fungicides with different modes of action appear to be one way of counteracting the development of resistance (Bolton and Smith 1988). Thus, it was decided to evaluate mixtures of CHA or norspermidine with a known fungicide. The lower 1.0 mM concentrations of CHA and norspermidine were chosen since the 1995 field trial failed to differentiate between the two concentrations (i.e. 1.0 mM and 5.0 mM) with respect to mildew control.

6.2.3 1996 Mini plots

Surplus areas sown with spring barley for the 1996 field trial were sub-divided to give a randomised block design with 14 plots (2.0 m x 1.0 m) (four replicates). Treatments in the large field trials were limited to 1.0 mM and 5.0 mM CHA and norspermidine: thus, 1.0, 2.0, 3.0, 4.0, and 5.0 mM concentrations of the compounds were evaluated using the mini-The combination of a putrescine analogue with CHA or norspermidine was plots. investigated in the 1995 field trial. Since plots were unavailable for such an evaluation in 1996, mini plots were used to examine the effect of an alternative ODC inhibitor, DFMO⁶, on spring barley alone, and in combination with 1.0 mM concentrations of CIIA or norspermidine. 1.0 mM concentrations were again chosen since the 1995 field trial had found no difference in fungicidal activity of 1.0 mM and 5.0 mM concentrations of the compounds. The mini-trial was carried out almost in the same manner as the 1996 field trial as described in section 6.2.2. However, treatments were made up in water with 0.01 % Tween 20 (v/v) to act as a surfactant and were applied to the plots using hand held compressed air sprayers. Since the treatments were applied with a reduced pressure compared to the AZO compressed air sprayer, only infection on the flag leaf was assessed. Measurements of plant height, plant dry weight and grain dry weight were not made for the mini plots.

⁶ DFMO was kindly provided by Marion Merrell Dow Pharmaceuticals.

6.3 Results

6.3.1 1995 Trial

All treatments significantly reduced mildew levels on 19 June except the mixture of 5.0 mM CHA and 5.0 mM E-TED (Table 27). All treatments significantly reduced infection on the second and fourth assessments. No significant differences were observed between any of the treatments at the third assessment, and was probably due to the low levels of mildew present. The reduction in mildew observed with 5.0 mM CHA and the mixture of 5.0 mM CHA and 5.0 mM E-TED at the fifth assessment (19 July) was not significant. All other treatments significantly reduced the infection level on this date.

Little differences were observed between the treatments. Although the commercial fungicide was significantly better than 5.0 mM CHA on both the first and final assessments, generally, the treatments were no less effective than the mixture of flutriafol and carbendazim in controlling mildew infection. No differences were observed between the lower and higher concentrations of either CHA or norspermidine. Similarly, the addition of a putrescine analogue to CHA and norspermidine did not enhance the disease control conferred (Table 27).

No significant differences were observed between any of the treatments in either plant height, plant dry weight or grain dry weight (Table 28).

An estimation of the grain yield per plot was made to give a more meaningful value for the field trial yield data. The number of stems (equivalent to the number of ears) in a 1 m row were counted at 20 random locations over the field trial area. Thus, the mean value was calculated and with 14 rows per plot, the number of ears per plot was estimated. The known grain weight of 100 ears in each plot was then converted to give the grain weight per plot. Statistical analysis was carried out on the unconverted data since the value of grain weight per hectare is only an estimation and assumes equal plant establishment over the trial area.

Treatment		Mean percen	tage powdery mild	ew infection	
	19 June	26 June	5 July	12 July	19 July
untreated	19.6	38.4	8.8	24.6	28.8
flutriafol and carbendazim	7.9	28.1	6.1	10.8	9.5
1.0 mM CHA	12.2	25.6	5.9	12.6	11.5
5.0 mM CHA	12.4	25.1	7.3	11.5	19.5
1.0 mM norspermidine	11,4	27.4	6,1	6.8	9.6
5.0 mM norspermidine	10.0	27.1	5.4	11.2	14.6
5.0 mM CHA + 5.0 mM E-TED	19.3	28.5	7.4	16.0	19.6
5.0 mM norspermidine + 5.0 mM E-TED	8.6	22.8	5.5	12,4	13.4
LSD ($p \le 0.05$)	4.28	4.81	ns	6.22	8.37
sed *	0.287	0.224	ns	0.355	0.530

. .

of transformed data

ж

.

.

;

č

,

freatment	Mean plant height	Mean nlant dry weight	Grain dry weight of	Estimated grain vield
	(cm)	(g)	100 ears (g)	рег plot (kg)
untreated	70.51	76.0	107.08	4.92
flutriafol and carbendazim	70.25	1.02	111.92	5.14
1.0 mM CHA	66.98	0.91	107.64	4,94
5.0 mM CHA	71.08	0.97	108.32	4.97
1.0 mM norspermidine	68.58	0.98	107.10	4.92
5.0 mM norspermidine	67.20	0.95	107.39	4.93
5.0 mM CHA + 5.0 mM E-TED	70.19	0.93	106.00	4.87
5.0 mM norspermidine + 5.0 mM E-TED	69.84	1.02	107.97	4,96

ANOVA f value probabilities were > 0.05 (0.403, 0.149 and 0.556 respectively for plant height, plant dry weight and grain dry weight of 100 ears)

-

5 Y 2 Y

.

. . . .

· • •

6.3.2 1996 Trial

In all of the assessments, all treatments significantly reduced powdery mildew infection of spring barley (Table 29). Little differences were observed between the mixture of flutriafol and carbendazim and the CHA and norspermidine treatments. Similarly, little differences were observed between the tebuconazole (full dose) treatment, tebuconazole (half dose) and the mixtures of half doses of tebuconazole and 1.0 mM CHA or 1.0 mM norspermidine.

Tebuconazole (full dose) was more effective in controlling disease than the mixture of flutriafol and carbendazim, and the CHA and norspermidine (1.0 mM and 5.0 mM) treatments, although only significantly so at the second assessment.

No differences were found between the heights, dry weights and grain weights of plants harvested from the plots (Table 30).

		Mean percer	ntage powdery mild	ew infection	
Treatment	12 June	19 June	15 July	22 July	29 July
untreated	27.8	38.8	6.6	16.7	24,9
flutria fol and carbendazim	16.2	17.5	2.6	4.5	8.7
1.0 mM CHA	18.1	24.5	2.8	2.9	5.8
5.0 mM CHA	18.8	22.2	2.4	3.0	7.0
1.0 mM norspermidine	17.5	22.9	2.2	4.2	12.0
5.0 mM norspermidine	18.2	22.8	4.5	2.6	9.1
tebuconazole	14.8	10.5	1.6	3.6	8.1
1/2 dose tebuconazole	14.4	13.8	1.6	2.2	7.4
1/2 dose tebuconazole plus 1.0 mM CHA	17.5	16.3	2.2	4.0	9.7
1/2 dose tebuconazole plus 1.0 mM norspermidine	14.0	10.7	1.6	2.3	9.7
LSD ($p \le 0.05$)	5.10	7.21	4.20	5.10	5.52
sed *	0.302	0.475	0,449	0.385	0.350

М.

Table 29. Effect of CHA, norspermidine, a flutriafol and carbendazim mixture and tebuconazole on powdery mildew infection of spring barley: 1996 Trial

11, MA

1000

of transformed data

¥

ж. . -~

•

.

Treatment	Mean plant height (cm)	Mean plant dry weight (g)	Grain dry weight of 100 cars (g)	Estimated grain yield per plot (kg)
untreated	70.26	0.77	93.4	4.29
flutriafol and carbendazim	72.27	0.87	83.2	3.82
1.0 mM CHA	72.11	0.81	74.1	3.40
5.0 mM CHA	67.45	0.70	86.9	3.99
1.0 mM norspermidine	70.19	0.80	83.7	3.84
5.0 mM norspermidine	69.64	0.79	87.7	4.03
tebuconazole	69.95	0.82	82.5	3.79
1/2 dose tebuconazole	69.40	0.83	88.1	4.05
1/2 dose tebuconazole plus 1.0 mM CHA	73.00	0.82	78.3	3.60
1/2 dose tebuconazole plus 1.0 mM norspermidine	70.35	0.88	90.3	4.15

ANOVA f value probabilities were > 0.05 (0.630, 0.117 and 0.460 respectively for plant height, plant dry weight and grain dry weight of 100 cars)

6.3.3 1996 Mini plots

As discussed in section 6.2.3, only the flag leaves on the mini plot plants were assessed for mildew infection. Consequently, only three assessments were made, 7, 14 and 21 days after the second spray. On the dates of the two assessments after the first spray, no disease was present on the flag leaves.

All of the treatments examined reduced powdery mildew infection significantly on all three assessment dates (Table 31). Little differences were apparent between the treatments and none were observed between the CHA treatments themselves, the norspermidine treatments or the DFMO \pm CHA / norspermidine treatments.

いたないです。

	Mean	percentage powdery mildew infi	ection
Treatment	19 July	26 July	2 August
untreated	10.1	14.3	26.7
1.0 mM CHA	4.0	5.6	12.6
2.0 mM CHA	3.9	4.7	10.6
3.0 mM CHA	2.3	3.C	11.9
4.0 mM CHA	2.0	3.7	10.3
5.0 mM CHA	2.6	4.9	10.2
1.0 mM norspermidine	2.8	4.5	12.4
2.0 mM norspermiduae	4.6	4.0	12.9
3.0 mM norspermidine	2.4	8,£	12.7
4.0 mM norspermidine	3.1	2.6	11,4
5.0 mM norspermidine	2.2	3.5	12.6
1.0 mM DFMO	3.1	3.6	13.6
1.0 mM DFMO plus 1.0 mM CHA	2.3	د ز در	10.1
.0 mM DFMO plus 1.0 mM norspermidine	2.6	3.2	10,9
LSD ($p \le 0.05$)	2.26	2.06	4.59
sed *	0.319	0.423	0.318

Table 31. Effect of CHA. norspermidine and DFMO on powdery mildew infection of spring barley: 1996 Mini plots

·· .

.*

2

2

of transformed data

-86

6.4 Discussion

Little difference was observed between the 1.0 mM and 5.0 mM CHA and norspermidine treatments, and the mixture of flutriafol and carbendazim in years 1995 and 1996, suggesting that the compounds controlled mildew infection as effectively as the commercial fungicide. For example, both the mixture of flutriafol and carbendazim and 1.0 mM norspermidine reduced disease by 67 % on the final assessment of the 1995 trial. Such results are in agreement with West and Walters (1988) who showed that CHA controlled powdery mildew on barley in the glasshouse. No comparable data exist for norspermidine but the compound has demonstrated antiproliferative activity against animal tumours (Prakash *et al.* 1988), and thus, the fungicidal properties of the compound are in agreement with previous findings. The results for CHA and norspermidine also compare favourably with DFMO and the novel putrescine analogue E-BED, which both provided early season control of powdery mildew in the field when compared with the same commercial fungicide (Havis & Walters 1992a; Havis *et al.* 1994b).

No additional control was conferred by increasing the concentration of the spermidine analogues. This suggested that the optimum dose of CHA and norspermidine was less than 5.0 mM. Such results were also observed in the mini plots where no difference was seen in mildew control afforded by the range of 1.0, 2.0, 3.0, 4.0 and 5.0 mM CHA and norspermidine. This further suggested that the optimum field dose of each analogue was 1.0 mM.

In the 1995 field trial, no additional control was observed through the combined use of a putrescine and spermidine analogue although the later control afforded by 5.0 mM CHA and 5.0 mM E-TED was interesting since the plots were only sprayed once. The combination of 1.0 mM CHA or norspermidine with 1.0 mM DFMO on the mini plots yielded a similar result with no additional control over DFMO alone observed. A synergistic effect was expected with the combined use of an ODC inhibitor and a spermidine synthase inhibitor. Jänne et al. (1985) suggested that the combined use of an AdoMetDC and an ODC inhibitor would lead to complete depletion of putrescine, spermidine and spermine in mammalian cells. Thus, if a plant pathogenic fungus possessed only one route to putrescine, the combined use of polyamine biosynthesis inhibitors should give more effective control. Indeed, West and Walters (1988) found various combinations of MGBG and DFMO reduced powdery mildew infection of barley more consistently than either compound alone in glasshouse experiments. CHA acted as a synergist for DFMO-mediated growth reduction of Botrytis cinerea on solid media (Smith et al. 1990b) and Pfaller et al. (1988) reported how a sequential treatment of DFMO followed by CHA provided increased control over CHA alone against growth of the yeast Candida albicans. However, spermidine content of carrot cell suspension cultures treated with a combination of cyclohexylammonium phosphate and DFMO was unchanged (Minocha & Khan 1991). These authors suggested that the two inhibitors had counteracted the effects of each other (Minocha & Khan 1991). This would, therefore, account for the lack of synergism observed in this study, although a number of other mechanisms may be possible.

ODC is a rate limiting enzyme of polyamine biosynthesis and inhibition of the enzyme often leads to compensatory mechanisms enabling the conservation of cellular polyamine levels (Pegg 1988). Spermidine synthase is not as well regulated (Pegg 1988), but ODC regulation may be responsible for the lack of increased response through the combined use of an ODC inhibitor and a spermidine synthase inhibitor. Additional environmental pressures may also have attributed to the lack of synergism observed, as discussed below.

Despite the substantial control of mildew in both field trials, no significant differences in the final yield were found between the treatments. This is in contrast to previous field studies with inhibitors of polyamine biosynthesis whereby DFMO and E-BED caused an increase in grain yield over untreated controls (Havis & Walters 1992a; Havis et al. 1994b). In the present study, even commercial fungicides did not increase grain yields although reductions in disease were apparent. It is well documented that net photosynthesis increases in healthy leaves of barley otherwise infected with powdery mildew (Ayres 1984), suggesting that partial control of infection will result in a yield benefit. However, the 1995 season was particularly dry and this may have influenced the results. Ayres (1984) suggests that the ability of a plant to cope with an infection may be seriously diminished by the simultaneous occurrence of an abiotic stress, for example drought, which could minimise the chances of compensatory growth and activity occurring in uninfected tissues. Indeed, Johnson (1995) reported how winter wheat had been affected in the 1994/5 season. She suggested that it was drought rather than disease which had limited canopy life and thus responses to fungicides were lower than would normally have been expected. On a similar note, the observed level of Rhynchosporium secalis infection was much higher in the 1996 field trial compared to the previous year. The incidence of this additional pathogen may have reduced the ability of the plants to compensate for the occurrence of powdery mildew, perhaps by reducing the amount of healthy tissue available for photosynthesis. This may have contributed to the lack of fungicide response in the 1996 season.

Tebuconazole treatments in 1996 controlled mildew more effectively than the CHA and norspermidine treatments and the mixture of flutriafol and carbendazim. The increase in efficacy over flutriafol and carbendazim is not surprising since tebuconazole has been developed more recently for mildew control. It should also be noted when comparing CHA and norspermidine to tebuconazole, that the experimental compounds are not formulated as commercial fungicides but only combined with a surfactant.

No additional disease control was observed through combining CHA or norspermidine with a reduced dose of tebuconazole in the 1996 field trial. Indeed, the reduced dose was as effective as the full dose of the fungicide. Wale *et al.* (1993) reported how powdery mildew of barley can be effectively controlled using low doses of fungicides under a range of optimum conditions which include the timing of fungicide application, disease pressure and resistance of the cultivar. However, rather than an effective low dose, it may be possible that the full dose did not achieve its potential as a result of additional environmental pressures on the crop, as already discussed.

CHA and norspermidine were thus shown to possess fungicidal activity in the field, an important phenomenon when attempting to determine potential compounds for crop disease control.

A DAY AND A

Chapter 7

Effect of cyclohexylamine and norspermidine on uptake of amino acids, polyamines and sugars by *Pyrenophora avenae*

7. EFFECT OF CYCLOHEXYLAMINE AND NORSPERMIDINE ON UPTAKE OF AMINO ACIDS, POLYAMINES AND SUGARS BY *PYRENOPHORA AVENAE*

7.1 INTRODUCTION

7.1.1 Review of literature

7.1.1.1 Transport mechanisms: an introduction

Nutrients which are essential for cells to survive generally enter those cells by either passive diffusion or carrier mediated transport. Passive diffusion is dependent upon the concentration of the nutrient on either side of the cell membrane. Thus, if the difference in intracellular and extracellular concentrations increases, the rate of diffusion increases and vice versa. Passive diffusion is non-specific and nutrients taken up by this mechanism usually do so such that their concentrations on either side of the membrane become equal (Jenkins 1992).

Nutrients taken up by carrier mediated transport do so by binding to proteins of the cell membrane, and indeed, most nutrients are transported by such systems. In order to be transported across the cell membrane, each substrate reversibly binds to a specific site of a carrier protein. With a limited number of binding sites on a protein, saturability of the transport system can occur, and the rate of transport across the membrane will reach a maximum limit. Such processes can therefore, like enzymes, be described by Michaelis-Menten kinetics, the equation of which is shown below (Jenkins 1992).

$$V = \frac{V_{max} [S]}{K_m + [S]}$$

V is the transport rate; V_{max} , the maximum transport rate; [S], the concentration of substrate to be transported and K_m , the Michaelis-Menten constant, which is equal to the substrate concentration taken up at half the maximum rate of transport. K_m is a measure of the affinity of the carrier site for the substrate. Thus, low K_m values reflect a high affinity while high K_m values correspond to lower affinity or binding strength between the carrier site and the substrate molecule (Jenkins 1992).

The specificity of such transport processes allows inhibition to occur, which may be competitive or non-competitive. Competitive inhibition occurs with substances closely related to the substrate. The amount of inhibition depends on the ratio of substrate and inhibitor present and the affinity of each of these substances for the carrier sites. Thus, the

effect of a competitive inhibitor decreases as its concentration is reduced. Kinetic effects of competitive inhibition are therefore observed as an increase in K_m with no change in V_{max} (Jenkins 1992).

Non-competitive inhibition can also occur with substances unrelated to the substrates being transported. Such inhibition is not related to the concentration of the inhibitor. Kinetic effects of non-competitive inhibition are therefore observed as a reduction in V_{max} with no change in K_m (Jenkins 1992).

Different types of carrier mediated transport occur and further distinction can be made, for example, with active and passive transport. Active transport requires energy while passive transport, also known as facilitated diffusion, does not. The energy for active transport can be provided by the proton motive force, generated as a result of cellular metabolism where H^+ ion gradients arise across membranes. H^+ ions are thus known as coupling ions and combined movement of those ions and substrate molecules is used to force movement of the substrate molecules across membranes. This situation where both H^+ ions and the substrate molecule are moved in the same direction is known as a symport transport system. Antiport transport systems occur when the two entities are being moved in opposite directions while uniport transport refers to movement of **a** single ion in response to the particular electrostatic gradient between the exterior and interior of the cell (Jenkins 1992).

Many more uptake mechanisms exist, reflecting the diversity of organisms and individual substrates. For example, proteins can form channels in cellular membranes, allowing movement of small molecules and ions. No binding to the proteins occurs and movement is regarded as a faster form of passive diffusion, with Michaelis-Menton kinetics not applicable (Jenkins 1992).

7.1.1.2 Transport mechanisms in fungi

Much work has studied transport of cations, anions, sugars and amino acids into fungal cells. However, only uptake of the later two shall be reviewed. Much more work on the uptake of polyamines has been carried out in other organisms and thus, polyamine uptake by fungi will be examined along with that associated work.

7.1.1.2.1 Transport of sugars

Sugars are transported into fungal cells via carrier mediated transport mechanisms, either passively or actively, depending upon the fungus, substrate and growth conditions. Comparisons between different organisms is thus difficult, as are comparisons between the same organism grown under different conditions (Griffen 1994).

Furthermore, tremendous diversity exists in fungal transport systems. Symport transport with hydrogen ions was implicated in active uptake of glucose by *Neurospora crassa* and many different yeast species (Deak 1978). *N. crassa* was, however, reported to possess the ability to take up glucose via passive transport when grown in high glucose concentrations (Scarborough 1970). Interestingly, H^+ symport transport was associated with uptake of maltose by *Saccharoinyces cerevisiae* via passive transport (Okada & Halvorson 1964). This phenomenon was indeed unusual and symport transport with hydrogen ions has been widely accepted as the main active transport system for fungal uptake of sugars (Griffen 1994). Active transport has also been reported for sugar uptake into *Rhodotorula gracilis* and *Aspergillus nidulans* (Griffen 1994).

A CONTRACTOR

Glucose transport into *S. cerevisiae* occurred via passive transport (Heredia *et al.* 1968; Novak *et al.* 1990). Two systems were apparent, a low affinity system and a high affinity system (Heredia *et al.* 1968; Novak *et al.* 1990). Such uptake processes are known as biphasic. At low concentrations of substrate, high affinity between the substrate and the carrier site is necessary to move the substrate across the membrane. However, when the concentration of substrate increases, less affinity is required and the low affinity process predominates. Such biphasic uptake of sugars by fungus is well documented (McDermott & Jennings 1976; Sheard & Farrar 1987; Walters *et al.* 1996). It has also been suggested that the low affinity system does not operate over the concentrations at which the high affinity system operates (Reinhold & Kaplan 1984).

Biphasic sugar uptake in *Pyrenopeziza brassicae* appeared to occur wholly via passive transport (Walters *et al.* 1996). However, work on *Phytophthora palmivora* suggested that active uptake of sugars took place for the high affinity system while the low affinity system operated via passive transport (Sheard & Farrar 1987).

In studying sugar uptake into *P. brassicae*, Walters and co-workers (1996) found that the kinetics for glucose and fructose uptake were similar. K_{ms} for the high affinity system for glucose and fructose were 3.5 μ M and 4.6 μ M respectively and 69 μ M and 75 μ M for the low affinity system. Such similar values and studies whereby glucose and fructose were taken up together, suggested that glucose and fructose shared a common carrier (Walters *et al.* 1996). This phenomenon was also found in *P. palmivora* for glucose and fructose uptake (Sheard & Farrar 1987). Kinetic parameters for sucrose uptake in *P. brassicae* were lower and competition studies where uptake of sucrose was unaffected by extracellular concentrations of glucose and fructose suggested a separate uptake carrier system existed for sucrose. Interestingly though, high invertase activity was detected (Walters *et al.* 1996).

Such an enzyme is known to hydrolyse sucrose to glucose prior to uptake, eliminating the need for a separate carrier in *P. palmivora* (Sheard & Farrar 1987).

7.1.1.2.2 Transport of amino acids

As with sugar transport into fungus, amino acid transport is extremely complex not least because of the large variety of different substrates as well as diversity of fungal organisms. Also, the stage of development of the fungus, the age of the culture and growth medium contents are all known to greatly influence the properties of the fungal transport system, and thus, different studies are not generally comparable (Griffen 1994).

Amino acid uptake is carrier mediated, however, differences are apparent in the specific mechanisms of such transport. Hunter & Segel (1971) reported that uptake of amino acids into *Penicillium chrysogenum* was sensitive towards inhibitors of fungal respiration, suggesting that transport was energy dependent. Similarly, Jones (1963) suggested active transport was responsible for amino acid uptake into *Botrytis fabae*. Active transport of amino acids was reported to occur via H^+ symport transport in yeast (Seaston *et al.* 1973; Eddy 1980; Cooper 1982).

Walters and co-workers (1997) however, suggested that uptake into *Pyrenopeziza brassicae* of glutamine and glutamic acid was independent of energy and likely to occur by passive transport. Interestingly, these authors found uptake of amino acids into the fungus to be a biphasic process. Kinetic parameters for the two amino acids were similar: K_{ms} for the high affinity system were 4.0 μ M and 4.4 μ M for glutamic acid and glutamine respectively and 580 μ M and 560 μ M for the low affinity system. Such similar parameters and inhibition of uptake of glutamic acid by glutamine and vice versa, suggested a single carrier protein to operate for both compounds. Studies using other amino acids; ornithine, arginine, lysine and asparagine suggested further that glutamic acid and glutamine were not transported using carrier systems associated with these other amino acids (Walters *et al.* 1997).

- オモンド・シンド

Similarly, in experiments with *Neurospora crassa*, *Aspergillus nidulans* and *P. chrysogenum*, competitive inhibition was observed between certain amino acids while little or no inhibition was observed among others (Griffen 1994). Thus, the existence of multiple systems for fungal amino acid uptake was suggested and studies in *N. crassa* found that five systems existed for such uptake. Aromatic and aliphatic amino acids were proposed to share a carrier system as were aromatic, aliphatic and basic amino acids. Three more systems were proposed to exist for basic amino acids, acidic amino acids and methionine

(Whitaker 1976). It can thus be seen that such multiple systems have overlapping specificities (Griffen 1994).

An interesting phenomenon of amino acid uptake is transinhibition, whereby large intracellular concentrations of an amino acid prevent further uptake of that compound. Inhibition is primarily non competitive, thus the maximum rate of transport is affected but K_m values unaltered. Transinhibition is system specific and only amino acids which compete for carrier binding sites are capable of transinhibition of each other. It has been suggested that the internal amino acid concentration may reduce the mobility of the carrier thus reducing the number of binding sites available for the external amino acid (Pall & Kelly 1971).

7.1.1.3 Transport of polyamines

As mentioned in section 1.1.3.3 (page 14), cells possess a large number of mechanisms enabling them to control their intracellular polyamine pools. Uptake of polyamines is one such process available to cells. A wide range of mammalian cells contain uptake pathways for polyamines (Porter & Sufrin 1986; Pegg 1987). Knowledge of these pathways is fragmentary, however, their importance in regulating intracellular levels of polyamines is demonstrated by the increased uptake of polyamines into cells in which amine synthesis is prevented through treatment with, for example, DFMO (Alhonen-Hongisto *et al.* 1980; Porter & Jänne 1987). Further studies using CHO cells in which a mutation had caused the transport system to be inactivated, showed how exogenous application of putrescine and spermidine failed to stimulate cells to grow in the presence of DFMO, demonstrating the transport system to be vital should cellular polyamine biosynthesis be prevented (Pegg 1988). Thus, the polyamine uptake system has been implicated in the ineffective treatment of cancer patients with, for example, DFMO. Polyamines may have been available from the diet and their uptake induced to compensate for drug related losses of intracellular amines (Pegg 1988). いたがない。そのできるまたに、ここのいで、読んであた。また、このでは、たちまた、いたがないではないであった。そのでは、「たい」、「そう」でいたが、また、それ、これできた。また、これがないでは、たちまた、これがないでは、たちまた、いたがないでは、これでは、また、これがないです。 いたい いたい しゅうしょう しゅうしょう しょうしょう しょうしょう

Mammalian polyamine transport is thought to be specific for polyamines and distinct from amino acid uptake (Pegg 1988). Indeed, CHO cells were shown to possess a transport system specific for polyamines (Byers & Pegg 1989). This system was saturable and energy dependent, and thus suggested to be an active carrier-mediated transport system (Byers & Pegg 1989). Various mammalian cell types were used to study uptake of spermidine (Khan *et al.* 1990). Spermidine uptake was inhibited by putrescine and spermine, suggesting that the three polyamines shared the same carrier. Also, pre loading of cells with amino acids such as ornithine and lysine did not affect spermidine uptake, suggesting that different transport systems existed for the polyamines and amino acids. Polyamine uptake has also been studied in other organisms, for example, putrescine uptake by maize root cells operated via a single saturable component (DiTomaso *et al.* 1992), as did putrescine uptake into *Saintpaulia* petals (Bagni & Pistocchi 1985). In contrast, uptake of the diamine into carrot cell cultures was biphasic, with two saturable components existing (Pistocchi *et al.* 1987).

「「「「「「「「「「「「「「「「「」」」」」」「「「」」」」」」

Studies on the uptake of polyamines into Aspergillus nidulans revealed that a single component saturable uptake system was operating (Spathus et al. 1982). The authors concluded that putrescine and spermidine appeared to be transported by different carrier systems, the presence of one amine inhibiting uptake of the other (Spathus *et al.* 1982). In contrast, uptake into Neurospora crassa, revealed that putrescine, spermidine and spermine were likely to share a transport system (Davis & Ristow 1988). These authors also found polyamine uptake to be biphasic (Davis & Ristow 1988). Similarly, putrescine and spermidine uptake in the fungus Fusarium culmorum was biphasic (West & Walters 1991b). $K_{\rm m}$ values for the high affinity system were 0.17 mM and 0.34 mM for putrescine and spermidine respectively while values for the low affinity system were 32.73 mM and 42.17 mM. Interestingly, uptake of each of the polyamines was not inhibited by the presence of the other amine in the uptake medium, suggesting that different carrier systems existed for each compound. Uptake was inhibited by sodium azide, suggesting that active carrier mediated transport occurred (West & Walters 1991b). Growth of the fungus in the presence of DFMO resulted in a slight increase in uptake of the amines, suggesting that polyamine depletion caused a commitant increase in uptake from extracellular sources. Interestingly though, growth in the presence of MGBG decreased uptake of the polyamines. This was, however, suggested to be the result of other effects of MGBG such as mitochondrial damage affecting the polyamine uptake systems (West & Walters 1991b).

7.1.1.4 Transport of other substances

Many studies have examined uptake of polyamine biosynthetic inhibitors. Uptake of DFMO into the fungus *Fusarium culmorum* was linear and non saturable, suggesting a passive diffusion mechanism. However, uptake was inhibited by ornithine, putrescine and spermidine, suggesting that DFMO shared a carrier with the amino acid and polyamines. Also, uptake was inhibited in the presence of sodium azide, implicating the occurrence of active carrier mediated transport. The authors thus suggested that two processes were responsible for uptake of DFMO into *F. culmorum*, passive diffusion and carrier mediated transport (West & Walters 1991a).

Passive diffusion was reported to be responsible for DFMO uptake in mouse fibroblasts (Erwin & Pegg 1982) and *Trypanosoma brucei brucei* (Bitonti *et al.* 1986a) since
extracellular amino acids and polyamines did not affect the uptake processes (Erwin & Pegg 1982; Bitonti *et al.* 1986). In contrast, transport of DFMO into *T. b. brucei* was suggested by Phillips & Wang (1987) not to be entirely due to passive diffusion.

Salar Salar

A CALMER

DFMO uptake into barley seedlings was inhibited by ornithine, suggesting that DFMO may utilise an amino acid carrier (Walters & Kingham 1990). Active transport was also implicated by the energy requirement for such uptake which was biphasic (Walters & Kingham 1990). DFMA and DFMO were indeed suggested by Slocum & Galston (1987) to utilise the amino acid transport system in plants and animals.

Interestingly, the total herbicide paraquat is taken up into cells utilising the polyamine transport system (Byers *et al.* 1987).

7.1.1.5 Polyamine analogues

The advantages of using analogues of polyamines in chemotherapeutic strategies has already been discussed in the thesis (section 1.1.4.6, page 20). Porter & Sufrin (1986) suggested that polyamine analogues may be able to utilise the polyamine uptake system to be transported into cells and since rapidly proliferating cells take up polyamines faster than normal cells, a degree of selectivity towards cancerous tissues would be possible. Once inside the cell, the analogues could exert an antiproliferative effect.

Indeed, although the polyamine transport system appears to be specific for polyamines, polyamine analogues are known to utilise the system. MGBG, a spermidine analogue, is taken up into cells via the same carrier as spermidine (Porter & Sufrin 1986; Porter & Jänne 1987). Such uptake was implicated in the antiproliferative success of MGBG with prevention of spermidine uptake through competition for carrier binding sites occurring rather than an inhibition of spermidine formation (Porter & Sufrin 1986; Porter & Jänne 1987).

Byers & Pegg (1989) reported how various bis(ethyl) polyamine analogues were taken into CHO cells while other polyamine analogues, for example, N'-N⁴-diacetylputrescine and N¹,N¹²-diacetylspermine inhibited spermidine uptake, probably by competing for binding sites, in a range of mammalian cells (Khan *et al.* 1990).

Recently, Walters & McPherson (1997) reported how uptake of methionine, ornithine, putrescine, spermidine, spermine, fructose and glucose into *Pyrenophora avenae* was reduced when the fungus had been grown in the presence of the novel putrescine analogue, E-BED. Exposure of *P. avenae* to the related analogue, E-TED, also reduced uptake of

methionine into the fungus. Competition studies showed that uptake of the compounds into *P. avenae* was reduced in the presence of E-BED, suggesting E-BED shared the transport systems of the amino acids, polyamines and sugars (Walters & McPherson 1997).

Kinetic studies were carried out for uptake of two of the compounds, methionine and fructose. In both cases, uptake was biphasic. The authors found that exposure of fungus to E-BED did not affect the K_m values of methionine or fructose uptake but reduced V_{max} values, suggesting non competitive inhibition. E-BED was thus suggested to inhibit uptake of methionine and fructose by perhaps binding to carrier sites associated with methionine and fructose.

7.1.1.5 Factors affecting transport processes

Many factors affect the uptake of substances into cells including the structure of the compound and the individual organism into which uptake is to occur (Griffen 1994). The pH of the external medium also influences uptake by various mechanisms. It can determine the charge of the substance to be carried and the charge on the carrier protein, thus affecting binding. If active transport operates, the pH of the external medium can influence the proton gradient across the membranes. Thus, for a given external concentration of substrate, the rate of uptake will vary with the pH of the external medium (Whitaker 1976). Putrescine uptake in Saintpaulia petals varied with external pH and two different optima were determined for low and high external concentrations of putrescine (Bagni & Pistocchi 1985). Similarly, two optima were reported for glutamine and glutamic acid uptake by Pyrenopeziza brassicae (Walters et al. 1997). Temperature affects transport of substances into cells as it affects all chemical reactions. Thus, the rate of carrier mediated uptake transport will increase with temperature to a point at which protein denaturation will occur, and uptake fall off rapidly (Hunter & Segal 1971). Cellular energy production will also affect the uptake process, if active transport is involved (Seaston et al. 1973). Clearly, transport studies are complex, however, uptake of amino acids, polyamines and sugars into cells is essential for such cell viability.

7.1.2 Aim of studies

Porter & Sufrin (1986) speculated that polyamine analogues may be able to utilise the polyamine transport system to enter cells before exerting their antiproliferative effects by one or more of various mechanisms. Antifungal and fungicidal activity have been demonstrated by norspermidine although no specific mode of action has been determined. Khan *et al.* (1990) reported how various analogues inhibited uptake of spermidine into mammalian cells. Since spermidine is the major polyamine in fungi (Stevens & Winther 1979), it seemed prudent to examine the uptake of norspermidine by fungus in an attempt to

establish the cause of the antifungal properties of norspermidine in the absence of an effect on polyamine biosynthesis.

Also, the cause of the increase in fungal putrescine caused by exposure to CHA is unknown. Exposure of fungus to E-BED also increased putrescine levels (Havis *et al.* 1994a), and recent work by Walters and McPherson (1997) has reported that E-BED caused substantial reductions in uptake of amino acids, polyamines and sugars in *Pyrenophora avenae*. Whether the E-BED induced reductions in uptake were responsible for the antifungal effects of the compound is not known. However, given these findings, it seemed prudent to examine the effect of CHA on the uptake of various compounds into *P. avenae*.

Experiments thus sought to determine the effects of CHA and norspermidine on fungal uptake of various compounds in an attempt to establish the cause of the antifungal properties of the compounds in the absence of polyamine biosynthetic inhibition.

7.2 MATERIALS AND METHODS

7.2.1 Experiment 1: Effect of CHA and norspermidine on the uptake of compounds by *Pyrenophora avenae* during a four hour period

and the second of the second second

1.2.2

P. avenae was maintained and grown in liquid culture amended with inhibitors to give final concentrations of 1.0 mM CHA and 0.01 mM norspermidine as described previously in sections 3.1.2.1 and 3.2.2.2 (pages 76 & 85). Two day old fungus (100 mg) was placed in a vial with 5 ml PDB and 3.7 kBq of a radio-labelled substance added. The compounds studied were [³H] methionine (2608.5 GBq mmol⁻¹, NEN Research Products, UK), [1-¹⁴C] ornithine (2.11 GBg mmol⁻¹, Amersham International, UK), [¹⁴C] putrescine, ^{[14}C] spermidine (both 4.0 GBq mmol⁻¹, Amersham International, UK), ^{[14}C] spermine (4.1 GBq mmol⁻¹, Amersham International, UK), D-[U-¹⁴C] glucose, D-[U-¹⁴C] fructose (both 10.5 GBq mmol⁻¹, Amersham International, UK) and [U-¹⁴C] sucrose (23.3 GBq mmol⁻¹, Amersham International, UK). The vials were incubated for 4 h in a shaking water bath at 27°C after which the mycelium was removed by centrifugation at 16000 g for 10 min at 0°C. The pellet obtained was washed twice with PDB (8 ml) by resuspending the mycelium and centrifuging before depositing it in a vial with 2 ml Soluene-350 (Packard), followed by shaking on a rotamixer and incubating at 60°C for 3 h. Vials were then left at room temperature overnight before 10 ml of Hionic Fluor (Packard) was added and the radioactivity counted using a Packard 1900 TR scintillation spectrometer.

To examine the competitive effect of CIIA and norspermidine on uptake of compounds, fungus grown only in culture medium for 3 days was used. Experiments were carried out as described above with one exception; fungus was incubated for 4 h in PDB amended with either CHA or norspermidine to give concentrations of 1.0 mM and 0.01 mM respectively. Controls used unamended PDB.

As with all experiments detailed in this chapter, values obtained as disintegrations per minute were converted and expressed as nmol g^{-1} hour⁻¹. Results are the means of four replicates and statistical significance was assessed using Student's *t*-test. Experiments were repeated and similar results obtained.

7.2.2 Experiment 2: Effect of prolonged exposure to CHA and norspermidine on the uptake of compounds by *Pyrenophora avenae* during a four hour period

To examine the possibility of an effect on uptake of compounds by fungus exposed to inhibitors for longer than two days, uptake of putrescine and spermidine were examined by fungus exposed to 1.0 mM CHA or 0.01 mM norspermidine for 4 days. Experiments were carried out as described above. Putrescine and spermidine were chosen as representative compounds of those available.

7.2.3 Experiment 3: Effect of higher concentrations of CHA and norspermidine on the uptake of compounds by *Pyrenophora avenae* during a 15 minute period

Further studies were carried out to examine the possibility that fungal uptake may be affected by exposure of the fungus to higher concentrations of the inhibitors. Thus, the effect of 2.75 mM CHA and 0.2 mM norspermidine on uptake processes was investigated. Studies were also carried out to determine the effect of CHA and norspermidine on fungal uptake of compounds over a short period. Uptake effects may occur quickly yet be masked if detection is delayed.

Two compounds, methionine and spermidine, were chosen as representative compounds of the original eight. Fungus was exposed to 2.75 mM CHA or 0.2 mM norspermidine for 4 days so as to gain enough fungal material for analysis. Fungal uptake of the compounds was allowed to proceed for 15 minutes as described above. The competitive nature of CHA and norspermidine was also examined by exposing 3 day old fungal mycelium, grown in culture medium only, to the inhibitors for the 15 minute uptake period.

Experiments were repeated 3-4 times with similar results obtained.

7.2.4 Experiment 4: Determination of uptake kinetics

To examine effects on the kinetics of the uptake process, fungus was grown in the presence of 2.75 mM CHA or 0.2 mM norspermidine for 4 days in order to amass sufficient fungal material. *P. avenae* was then allowed to take up varying levels of methionine (0.01 mM -128 mM, containing different levels of radioactive methionine, 0.25 μ l - 8 μ l) for a 15 minute period as in the procedure described above. Since smaller amounts of radioactive methionine could not be added precisely, the lower concentrations of unlabelled substrate had proportionately more radioactivity than was desired. However, this was accounted for in later calculations. Controls used fungus that was grown in culture medium only. Data were analysed using regression analysis from Eadie-Hofstee plots.

7.3 RESULTS

7.3.1 Experiment 1: Effect of CHA and norspermidine on the uptake of compounds by *Pyrenophora avenae* during a four hour period

and the second se

Neither 1.0 mM CHA nor 0.01 mM norspermidine had any significant effect on the uptake of amino acids, polyamines or sugars by *P. avenae* when the fungus was exposed to the inhibitors for 2 days (Table 32). Exposure to CHA caused a decrease in uptake of spermine while exposure to norspermidine caused an increase in the uptake of fructose, when fungus was exposed to CHA or norspermidine for the four hour uptake period only (Table 33).

7.3.2 Experiment 2: Effect of prolonged exposure to CHA and norspermidine on the uptake of compounds by *Pyrenophora avenae* during a four hour period

Exposure of fungus to 1.0 mM CHA or 0.01 mM norspermidine for 4 days had no effect on uptake of putrescine or spermidine (Table 34).

7.3.3 Experiment 3: Effects of higher concentrations of CHA and norspermidine on the uptake of compounds by *Pyrenophora avenae* during a 15 minute period

Various effects on the uptake of methionine and spermidine by P. avenue were detected when fungus was exposed to the inhibitors for 4 days and the uptake process allowed to proceed for 15 minutes. Exposure to 2.75 mM CHA caused an increase in the uptake of methionine, while exposure to 0.2 mM norspermidine resulted in a reduction in uptake of the compound. Treatment of fungus with norspermidine also caused an increase in uptake of spermidine although this was not always significant (Table 35).

CHA, however, failed to affect uptake when placed in competition with the compounds, i.e. when *P. avenae* was only exposed to the inhibitor for 15 minutes. Nevertheless, the presence of norspermidine for 15 minutes along with methionine or spermidine, increased uptake of methionine and decreased that of spermidine (Table 35).

162

	Uptake (nmol g ⁻¹ h ⁻¹)		
Compound	Control	CHA	norspermidine
Methionine	1.0 ± 0.11	$\textbf{0.8} \pm \textbf{0.06}$	1.0 ± 0.03
Ornithine	46.4 ± 6.32	54.6 ± 2.79	57.9 ± 3.61
Putrescine	124.8 ± 8.15	103.5 ± 5.91	119.4 ± 13.67
Spermidine	1754 ± 117.4	1483 ± 156.2	2322 ± 210.0
Spermine	2158 ± 177.6	2415 ± 151.7	2655 ± 34.1
Glucose	18.4 ± 1.46	16.7 ± 1.42	19.3 ± 1.08
Fructose	71.1 ± 3.62	84.6 ± 10.26	81.0 ± 5.23
Sucrose	12.7 ± 1.66	13.8 ± 2.24	11.1 ± 0.87

Al D

Table 32. Uptake of methionine. ornithine, polyamines and sugars by Pyrenophora avenaeexposed to 1.0 mM CHA or 0.01 mM norspermidine for 2 days

Note: Values are shown as the means \pm SE of four replicates.

<u>Table 33.</u>	Uptake of methionine, ornithine, polyamines and sugars by Pyrenophora avenae
	exposed to 1.0 mM CTIA or 0.01 mM parspermidine for 4 hours
	exposed to 1.0 mm chive of 0.301 mm norsperimente tot -4 mours

	Uptake (nmol g ⁻¹ h ⁻¹)		
Compound	Control	СНА	norspermidine
Methionine	1.2 ± 0.06	1.4 ± 0.18	1.3 ± 0.10
Ornithine	89.1 ± 11.76	61.6 ± 12.35	103.1 ± 3.39
Putrescine	202.3 ± 7.63	182.3 ± 10.54	217.3 ± 35.60
Spermidine	$\textbf{2211} \pm \textbf{241.7}$	1556 ± 235.3	1293 ± 127.9
Spermine	3412 ± 186.3	2215 ± 101.6 **	3604 ± 181.0
Glucose	23.0 ± 0.89	20.4 ± 0.92	19.5 ± 1.82
Fructose	89.1 ± 8.89	149.9 ± 20.93	209.6 ± 9.75 **
Sucrose	29.2 ± 2.45	28.9 ± 1.74	30.7 ± 2.09

Note: Values are shown as the means \pm SE of four replicates.

Significant differences from control are shown as: ** $p \leq 0.01$

,	Uptake (nmol g ⁻¹ h ⁻¹)		
Compound	Control	СНА	norspermidine
Putrescine	91.3 ± 8.67	71.3 ± 5.99	81.5 ± 2.61
Spermidine	382 ± 57.3	202 ± 23.2	371 ± 18.8

Table 34. Uptake of polyamines by Pyrenophora avenue exposed to 1.0 mM CHA or 0.01mM norspermidine for 4 days

Note: Values are shown as the means \pm SE of four replicates.

Table 35. Uptake of methionine and spermidine by Pyrenophora avenue exposed to 2.75mM CHA or 0.2 mM norspermidine for 4 days or 15 minutes

Uptake (nmol $g^{-1}h^{-1}$) in fungus exposed to inhibitors for 4 c			
Compound	Control	СНА	norspermidine
Methionine	3.78 ± 0.296	12.26 ± 0.095 **	0.66 ± 0.087 **
Spermidine	$\textbf{8224} \pm \textbf{349}$	7243 ± 420	11350 ± 823
	Uptake (nmol g ⁻¹ h ⁻¹) in fungus exposed to in	hibitors for 15 minutes
Compound	Control	CHA	norspermidine
Methionine	0.87 ± 0.010	0.87 ± 0.087	1.32 ± 0.091 *
Spermidine	11671 ± 912	11792 ± 1052	4 507 ± 832 **

Note: Values are shown as the means \pm SE of four replicates.

Significant differences from control are shown as: * $p \le 0.05$; ** $p \le 0.01$

7.3.4 Experiment 4: Determination of uptake kinetics

The kinetics of uptake of methionine by P. avenae exposed to 2.75 mM CHA or 0.2 mM norspermidine for 4 days were calculated from the classical equation relating to enzymecatalysed reactions, the Michaelis-Menton equation, as described in section 7.1.1.1.

$$V_{o} = \frac{V_{max}[S]}{[S] + K_{m}}$$

The initial velocity (V_0) of carrier mediated uptake varies with the concentration of substrate supplied [S]. A plot of the initial velocity against the substrate concentration gives a rectangular hyperbolic curve, demonstrating that at low substrate concentrations, the initial velocity of the reaction is directly proportional to the substrate concentration but at higher concentrations, the initial velocity reaches a maximum. The initial velocity at that point is denoted as V_{max} .

Since the reciprocal of the equation of a rectangular hyperbola is the equation of a straight line, the equation can be re-arranged to take the form of an Eadic - Hofstee equation as shown below.

$$V_0 = -K_m \times \frac{V_0}{---} + V_{max}$$
[S]

In accordance with the equation, plots of V_0 against $V_0/[S]$ will produce straight lines of gradient equal to $-K_m$ and which intercept the y-axis at V_{max} . Thus, values of K_m and V_{max} for the uptake process were established.

Exposure of fungus to 2.75 mM CHA for 4 days caused a reduction in the uptake of methionine by *P. avenae* at all concentrations of methionine investigated. Exposure of fungus to 0.2 mM norspermidine reduced uptake of methionine only at the lowest concentrations examined. At concentrations of 16 - 128 mM, no effect on methionine uptake was observed (Table 36).

The plot of V_0 against $V_0/[S]$ for the control fungus was best fitted by two straight lines, revealing the presence of a biphasic uptake system (Figure 16). Linear regression analysis of system 1 (high affinity) found V_{max} and K_m to have values of 0.87 nmol g⁻¹ hr⁻¹ and

3.84 mM respectively. Values for system 2 (low affinity) were 27.3 nmol g^{-1} hr⁻¹ and 520 mM.

The Eadie-Hofstee plot for fungus exposed to 2.75 mM CHA for 4 days revealed that the two phases were less defined (Figure 17). Analysis of the two phases failed to give significant regressions and a plot of V_0 against $V_0/[S]$ using the same scale as the control demonstrated that uptake had indeed been dramatically altered by the presence of CHA in the growing medium of the fungus (Figure 18).

The effect of norspermidine on fungal uptake of methionine resulted in complete disruption of the two uptake phases (Figure 19). Examination of the Eadie-Hofstee plot using the same scale as the control fungus further demonstrated this result (Figure 20).

166

_	Uptake (11mol g ⁻¹ h ⁻¹)		
Methionine conc. (mM)	Control	CHA	norspermidine
0.01	0.0026 ± 0.00011	0.0003 ± 0.00006 ***	0.0006 ± 0.00013 ***
0.1	0.020 ± 0.0027	0.002 ± 0.0004 **	0.006 ± 0.0012 **
1	0.153 ± 0.0109	0.016 ± 0.0011 **	0.052 ± 0.0087 ***
2	0.29 ± 0.022	0.02 ± 0.003 **	0.09 ± 0.008 **
4	0.44 ± 0.017	0.06 ± 0.002 ***	0.20 ± 0.040 **
8	0.58 ± 0.042	0.08 ± 0.004 **	0.35 ± 0.049 *
16	0.80 ± 0.079	0.15 ± 0.016 **	1.08 ± 0.198
32	1.59 ± 0.200	0.35 ± 0.016 **	1.53 ± 0.037
64	3.05 ± 0.356	0.53 ± 0.033 **	3.88 ± 0.338
128	5.36 ± 0.434	0.96 ± 0.150 **	7.45 ± 1.590

Table 36. Uptake of methionine by Pyrenophora avenae exposed to 2.75 mM CHA or0.2 mM norspermidine for 4 days

Note: Values are shown as the means \pm SE of four replicates. Significant differences from control are shown as: * p \leq 0.05; ** p \leq 0.01; *** p \leq 0.001 analysis. was allowed to proceed for 15 mins. Results are the means of four replicates and values for Vmax and Km defined from the graph using regression Figure 15. Uptake of varying concentrations of methionine (0.01 - 128 mM) by Pyrenophora avenae, grown as an experimental control. Uptake



was allowed to proceed for 15 mins. Results are the means of four replicates and the data were analysed using regression analysis. Figure 16. Uptake of varying concentrations of methionine (0.01 - 128 mM) by Pyrenophora avenae, exposed to 2.75 mM CHA for 4 days. Uptake



allowed to proceed for 15 mins. Results are the means of four replicates and the data were analysed using regression analysis. Figure 17. Uptake of varying concentrations of methionine (0.01 -128 mM) by Pyrenophora avenae, exposed to 2.75 mM CHA for 4 days. Uptake was



Uptake was allowed to proceed for 15 mins. Results are the means of four replicates and the data were analysed using regression analysis. Figure 18. Uptake of varying concentrations of methionine (0.01 - 128 mM) by Pyrenophora avenae, exposed to 0.2 mM norspermidine for 4 days.



Uptake was allowed to proceed for 15 mins. Results are the means of four replicates and the data were analysed using regression analysis Figure 19. Uptake of varying concentrations of methionine (0.01 - 128 mM) by Pyrenophora avenae, exposed to 0.2 mM norspermidine for 4 days.



7.4 DISCUSSION

Exposure of fungus to 1.0 mM CHA or 0.01 mM norspermidine for 2 days had no effect on the uptake of amino acids, polyamines or sugars by *P. avenae*. This is in contrast to the findings of Walters & McPherson (1997), who showed that exposure of fungus to E-BED for two days reduced the uptake of numerous compounds by *P. avenae*. Similarly, longer exposure of *P. avenae* to CHA and norspermidine, for 4 days, had no effect on uptake of those compounds examined.

When the various compounds were taken up by the fungus in the presence of 1.0 mM CHA, only spermine uptake was reduced. Walters & McPherson (1997), however, found that uptake of numerous compounds was reduced when uptake proceeded in the presence of E-BED. CHA and spermine may share a common carrier for fungal uptake, although kinetic studies would be necessary to determine whether the observed inhibition was competitive or non-competitive. Conversely, CHA may not utilise the carrier systems of the other compounds examined, thus explaining why their uptake into *P. avenae* was unaffected.

Competitive studies with 0.01 mM norspermidine only affected uptake of fructose, whose uptake into *P. avenae* was actually increased. This increase was unexpected, but may have been caused by a number of mechanisms. Binding of norspermidine to the carrier protein may have affected binding site affinity for fructose and increased the affinity. Also, since sugar uptake is generally considered to operate via active carrier mediated transport, norspermidine may have affected the disappation of energy and effectively increased the rate of fructose uptake by the fungus, although why this should have only affected fructose is unclear. Kinetic studies would indicate if such effects had occurred: an affinity increase would be reflected by a reduction in K_m while an increase in the rate of substrate transport would lead to an increase in V_{max} .

The comparative lack of effect of CIIA and norspermidine is, as discussed, in contrast to the findings of Walters & McPherson (1997). The use of 1.0 mM CHA and 0.01 mM norspermidine as opposed to higher concentrations of the compounds has already been discussed, (section 3.2.2.2, page 85). Although reductions in fresh weight of fungus exposed to such concentrations do not always occur, growth effects are clearly visible. Nevertheless, the effect of higher concentrations of the compounds, 2.75 mM CHA and 0.2 mM norspermidine, on fungal uptake were also examined. Shorter uptake times were investigated in case uptake effects had been masked by the long four hour uptake period.

Thus, exposure of fungus to 2.75 mM CHA for four days resulted in an increase in methionine uptake and no change in spermidine uptake by *P. avenae*. No effect was

observed in the uptake of methionine or spermidine when fungus was exposed to CHA only during the uptake process. This suggests that CHA does not share a common uptake carrier with methionine or spermidine. The effect on methionine uptake differed from previous work in this study and may be the result of the higher CHA concentration or the shorter uptake time. Further work would be necessary to determine which condition led to the uptake effect and indeed, why the effect occurred. Exposure of fungus to CHA may have caused changes to the carrier binding proteins and resulted in an increased affinity for methionine. Alternatively, the energetics of the transport system may have been affected and the rate of uptake thus increased.

Exposure of fungus to 0.2 mM norspermidine for four days resulted in a slight increase in spermidine uptake and reduced methionine uptake. The converse effects were found when the compounds were allowed to be taken up in the presence of norspermidine: i.e. an increase in methionine and a reduction in spermidine uptake was caused. Whether these effects were due to the increased concentration of norspermidine or the reduced uptake time is not known.

Exposure of fungus to norspermidine may have affected the carrier proteins of the fungus or the membrane potential of the cells and therefore affected either the binding affinity of the carrier sites towards the substrates or the energy disappated for active transport. The causes of the differing effects for methionine and spermidine are unclear and future work would be required to clarify the findings.

In the competitive studies, norspermidine may have inhibited the uptake of spermidine competitively or non-competitively. Kinetic studies would be required to establish which mechanism was responsible. Since the two compounds are structurally similar, they may have utilised the same carrier and competed with each other for binding sites. The increase in methionine uptake in the presence of norspermidine may be the result of altered binding affinity of the carrier towards methionine or changes in the energy available for active transport. Further work would be required in this area.

The results so far have been discussed assuming that active carrier mediated transport was operating for fungal uptake of the compounds. Certainly, carrier mediated transport was suggested by saturation kinetics of methionine uptake into *P. avenae* by Walters & McPherson (1997). Further work would be required using, for example, sodium azide, to examine if uptake did indeed require energy.

The kinetics of uptake were examined in fungus exposed to 2.75 mM CHA or 0.2 mM norspermidine for 4 days. For all the concentrations of methionine examined, methionine uptake was reduced in fungus pre exposed to 2.75 CHA. This is, however, in direct contrast to the previous work in this study which showed that fungus exposed to the same concentration of CHA for the same time and allowed to take up methionine for the same period, caused increased amino acid uptake. Griffen (1994) stated that even the age of the culture used could influence the uptake process, therefore, such results, differing from experiment to experiment are not unexpected, even though all cultures were used at approximately the same age.

Exposure of fungus to norspermidine reduced methionine uptake from external concentrations of 0.01 mM - 8 mM, but at higher concentrations (16 mM - 128 mM), no differences in uptake between the norspermidine treated fungus and the control were observed. The reasons for this phenomenon are not clear since methionine uptake into the control fungus had not reached its maximum level at the external concentration of 8 mM. The unaltered methionine uptake into norspermidine treated fungus could not therefore be explained by saturation. The high concentrations of external methionine appeared able to overcome the norspermidine-induced effects causing reduced methionine uptake.

and the second of the second of

An Eadie-Hofstee plot of the control fungus was best fitted by two straight lines, suggesting the uptake to be biphasic (Figure 15). Such uptake of amino acids is in agreement with previous work on glutamic acid and glutamine into *Pyrenopeziza brassicae* (Walters *et al.* 1997) and methionine into *P. avenae* (Walters & McPherson 1997). K_m values for the high and low affinity systems were 3.84 mM and 520 mM respectively. Such values are in contrast for those reported by Walters & McPherson (1997) for *P. avenae*, (0.02 mM and 1.68 mM). However, as discussed previously, differences can exist between different experiments with respect to the age of the fungal culture used. V_{max} values were 0.87 nmol g⁻¹ h⁻¹ and 27.3 nmol g⁻¹ h⁻¹ respectively. This, again, is in contrast to those previous findings of 0.004 mmol g⁻¹ h⁻¹ and 0.39 mmol g⁻¹ h⁻¹ by Walters & McPherson (1997).

Eadie-Hofstee plots for fungus exposed to CHA or norspermidine clearly show that inhibition of methionine uptake in these two fungi was not caused by competitive or noncompetitive inhibition. The kinetics of biphasic uptake was greatly distorted, more so for norspermidine than CHA. If the hypothesis that such findings are true for uptake of any substance into treated fungus holds, previous attempts to explain variations in uptake in experiments on the basis of classical competitive or non-competitive inhibition were unfounded. A more likely explanation for the findings lies in the known association between polyamines and cellular membranes. CHA and norspermidine may be binding to those sites associated with the polyamine-induced stabilisation of membrane integrity. Membrane structure may thus be disrupted and cause abberant effects on amino acid, polyamine and sugar transport.

Exactly why the analogues do not act as the putrescine analogue E-BED does with regards to fungal uptake of various compounds, is unknown. E-BED had been shown to reduce uptake of various compounds into fungus and such a reduction in polyamine uptake may be associated with the induced putrescine accumulation found in *P. avenae* by Havis and co-workers (1994a). However, Walters & McPherson (1997) could not conclude whether the E-BED induced reductions in uptake were responsible for the antifungal effects of the compound or indeed, related to the increased putrescine concentration. The present study has indicated that CHA may interfere with membrane structure and/or function but the cause of the CHA induced putrescine accumulation is still unknown.

「「「「「「「「「「」」」」

Porter & Sufrin (1986) suggested that polyamine analogues would utilise the polyamine transport system. Thus, norspermidine was hypothesised to compete with spermidine for uptake into fungal cells. Competitive uptake studies between spermidine and 0.2 mM norspermidine suggested that this was indeed true with spermidine uptake into *P. avenae* inhibited in the presence of norspermidine. However, kinetic studies revealed the apparent destruction of cell uptake systems by the spermidine analogue, although, it should be noted that such studies examined fungus already exposed to the analogue and little is known of norspermidine uptake into fungal cells during growth.

It is interesting that the results failed to demonstrate existence of distinct amino acid, polyamine and sugar uptake systems in *P. avenae*. Norspermidine affected uptake of methionine, fructose and spermidine, three chemically distinct compounds. This is, however, in agreement with work using E-BED which found that uptake of amino acids, polyamines and sugars was inhibited by the putrescine analogue (Walters & McPherson 1997). In mammalian cells, polyamines were reported to utilise a different uptake system from several amino acids (Khan *et al.* 1990), while in the fungus *P. brassicae*, different uptake systems were demonstrated for different amino acids (Walters *et al.* 1997). However, given the destruction of the integrity of the uptake system, the lack of observation of specific uptake systems is not surprising.

Many factors influence uptake into cells including the substrate, organism, extracellular pH, temperature and cellular energy production (Hunter & Segal 1971; Seaston *et al.* 1973; Griffen 1994). Furthermore, fungal uptake can even be affected by the age of the culture used (Griffen 1994). Despite the variability in the results of this study, CHA and norspermidine clearly affected uptake of amino acids, polyamines and sugars into cells of *P. avenae*. Whether these effects were primarily responsible for the antifungal activity of CHA and norspermidine or the result of some other antiproliferative mechanism is unknown.

Chapter 8

Hypusine biosynthesis

8. HYPUSINE BIOSYNTHESIS

8.1 INTRODUCTION

8.1.1 Review of literature

8.1.1.1 Hypusine: an introduction

The unusual amino acid hypusine was initially discovered in extracts of bovine brain by Shiba and co-workers (Shiba *et al.* 1971). Shortly afterwards, these workers determined the chemical structure of the compound to be [N^e-(4-amino-2-hydroxybutyl)lysine] (Shiba *et al.* 1972). The amino acid was subsequently found both in its free form (Nakajima *et al.* 1971), and bound to protein (Imaoka & Nakajima 1973; Sano *et al.* 1984) in various animal tissues. Hypusine has since been detected in lower eukaryotes such as yeast (Gordon *et al.* 1987), *Drosophila melanogaster* (Gordon *et al.* 1987), *Dictyostelium discoideum* (Sandholzer *et al.* 1989), *Neurospora crassa* (Chen & Yang 1988; Yang *et al.* 1990) and plants (M. H. Park unpublished results). Substantial evidence for the ubiquity of hypusine in all eukaryotic cells would thus now appear to exist. Hypusine has also been detected in many archaebacterial species but not in eubacteria (Schümann & Klink 1989; Bartig *et al.* 1990).

8.1.1.2 Biosynthesis of hypusine

Hypusine is the product of the post-translational modification of one specific cellular protein which was identified as cukaryotic translation initiation factor 5A (eIF-5A) (Cooper *et al.* 1983). Biosynthesis of the amino acid occurs in two steps. Firstly, free specific lysine cleaved and the aminobutyl moiety transferred to the ε -amino group of a specific lysine residue in the precursor of eIF-5A, forming the intermediate deoxyhypusine. This enzymatic transfer is catalysed by deoxyhypusine synthase. The subsequent reaction step consists of hydroxylation of the second carbon of the added group, catalysed by deoxyhypusine hydroxylase, to complete hypusine synthesis and eIF-5A maturation (Figure 20) (Park *et al.* 1993a).

8.1.1.3 Biological significance of eIF-5A and hypusine

The translation initiation factor eIF-5A was initially identified as a stimulant for two *in vitro* protein synthesis systems (Kemper *et al.* 1976). Indeed, Hershey (1991) stated that the mammalian eIF-5A appeared to act in the initiation phase of protein synthesis by promoting formation of the first peptide bond. However, recent reports of continued protein synthesis in yeast after the apparent complete depletion of eIF-5A suggest that the *in vivo* activity of cIF-5A may lie outwith the realm of protein synthesis (Park *et al.* 1993b).

	eIF-5A precursor	Lysyl residue NH2 (CH2)4 (CH2)4 NH-CH-COO	
Figure 2		$H_{2}^{(CH)} - H_{2}^{CH} - H_{2}^{CH} - H_{2}^{CH} - H_{2}^{CH} - H_{2}^{CH}$	Spermidine
0: Biosynthesis of hypusin	ţ	Deoxyhypusine synthase →	
e (Reproduced from Park	cIF-5A intermediate	$\begin{array}{c} \mathrm{NH}_2 \\ \mathrm{CH}_2 \\ \mathrm{CH}_2 \\ \mathrm{CH}_2 \\ \mathrm{CH}_2 \\ \mathrm{H}_2 \\ \mathrm{H}_2 \\ \mathrm{NH} \\ \mathrm{NH} \\ \mathrm{CH}_2 \\ \mathrm{Orb}_2 \\ \mathrm{NH} \end{array}$	Deoxyhypusyl residue
<i>et al.</i> 1993b)	Ţ	Deoxyhypusine hydroxylase →	
	eIF-5A	$\begin{array}{c} {\rm NH}_2 \\ {\rm CH}_2 \\ {\rm H}_2 \\ {\rm $	Hypusyl residue

· · · ••

.

1.111.1

۰.

180

Hershey and co-workers studied the function of cIF-5A by genetically modifying yeast cell lines, having identified two genes responsible for eIF-5A in yeast. Expression of either of the two genes alone promoted growth of the cells while disruption of both genes was lethal (Schnier *et al.* 1991; Schwelberger *et al.* 1993). Similarly, incorporation of a plasmid containing either one of the genes into a double null mutant caused cells to grow at almost the normal rate (Schnier *et al.* 1991; Schwelberger *et al.* 1993). The necessity of eIF-5A for yeast cellular growth was demonstrated when cells depleted of eIF-5A ceased to divide (Park *et al.* 1993b).

Ambiguity thus exists with respect to the exact role of eIF-5A in cellular biology. It may participate in protein synthesis itself or some other function relating to cellular proliferation. However, studies such as those described above, as well as the highly conserved nature of the protein, do suggest that role to be critical (Park *et al.* 1993a). The activity of eIF-5A is undoubtedly conferred by the hypusine residue (Park *et al.* 1991), and further studies have thus attempted to elucidate the exact cellular role of hypusine.

An increase in hypusine synthesis coincided with an increase in protein synthesis when human lymphocyte cells were activated to grow (Cooper *et al.* 1982). Various groups of researchers have demonstrated that modification of the hypusine residue of eIF-5A results in the failure of eIF-5A to stimulate protein synthesis *in vitro* (Park 1989; Smit-McBride *et al.* 1989; Park *et al.* 1991).

Studies have also demonstrated an association between hypusine synthesis and cellular proliferation with general correlations between the two processes having been reported for various mammalian cell lines (Chen 1983; Torrelio *et al.* 1984; Gerner *et al.* 1986). Interestingly, the post-translational events leading to hypusine synthesis occur only after initiation of cell growth (Cooper *et al.* 1982; Chen 1983; Torrelio *et al.* 1984).

Thus, as with eIF-5A, the precise cellular role of hypusine would appear to be unknown at present. Yet, whether it be associated with protein synthesis or some other function relating to cellular proliferation, synthesis of the amino acid is critical for cell viability. The necessity of hypusine has been substantiated further through studies involving depletion of the spermidine substrate and inhibition of the two biosynthetic enzymes involved in the synthesis of hypusine.

8.1.1.4 Polyamine biosynthesis and hypusine

Although the significance of polyamines in cellular replication and growth has been known for almost 50 years (section 1.1.2.1, page 4), it was not until relatively recently that a

specific physiological role for spermidine was discovered: that of hypusine formation (Park *et al.* 1981). Early work, as reviewed in Chapter 1, found that cellular growth inhibition by the ODC inhibitor DFMO was accompanied by changes in intracellular polyamine levels, including a reduction in spermidine (Tabor & Tabor 1984). However, the cessation of cellular growth was observed to be delayed with respect to the spermidine depletion (Hölttä *et al.* 1979). Hölttä and co-workers (1979) thus speculated that the delay in growth inhibition was not due to specific spermidine loss but rather due to depletion of some spermidine derived component that was required for cellular proliferation and/or growth. Subsequent studies revealed that DFMO treatment of cells not only caused depletion of spermidine but resulted in a reduced rate of hypusine synthesis (Gerner *et al.* 1986; Park 1987).

Contraction of the second

In further studies, the effects of the ODC inhibitor, APA, were examined. As with DFMO, inhibitor treatment of HTC cells resulted in the arrest of cellular proliferation accompanied by reductions in both the spermidine level and hypusine content (A. Shirahata unpublished results). Similarly, treatment of HTC cells with APA resulted in reduced hypusine content and spermidine levels, however, treated cells appeared to grow as well as control cells in this instance (Beppu *et al.* 1996).

Thus, evidence relating the inhibition of polyamine biosynthesis and resultant effects on cellular proliferation and growth to hypusine synthesis was apparent. Further work with polyamine biosynthesis inhibitors revealed further the necessity of spermidine for hypusine formation and cell growth. Byers and co-workers exposed L1210 cells to AbeAdo, an inhibitor of AdoMetDC. As expected, spermidine and spermine levels were depleted while putrescine levels increased. The cellular growth rate declined and was followed by the onset of cytostasis whereby cells cease to grow in order to conserve intracellular levels of polyamines (Byers *et al.* 1992). The onset of cytostasis was, however, delayed relative to the depletion of spermidine and thus, AbeAdo-induced cytostasis was proposed to be associated with hypusine synthesis (Byers *et al.* 1992).

eIF-5A precursor levels did indeed become measurable upon the onset of AbcAdo-induced cytostasis, indicating a reduction in hypusine biosynthesis. Cells could be rescued from cytostasis by the addition of exogenous spermidine upon which the level of accumulated precursor decreased (Byers *et al.* 1992). Spermine also could rescue cells from cytostasis but was thought to undergo intracellular conversion to spermidine and the spermidine, not spermine rescue the cells (Byers *et al.* 1992). Many polyamine analogues were tested for their ability to reverse AbcAdo-induced cytostasis but only one of those was successful; the substrate for deoxyhypusine synthase, N-(3-aminopropyl)-1,4-diamino-cis-but-2-ene (Byers

et al. 1992). Other analogues tested failed to rescue cells and were themselves unable to act as substrates for deoxyhypusine synthase. Interestingly, the trans-isomer of N-(3aminopropyl)-1,4-diamino-cis-but-2-ene did act as a substrate for deoxyhypusine synthase yet failed to rescue AbeAdo-induced cytostatic cells (Byers *et al.* 1992). In later work, two synthetic analogues, 1-methylspermidine and 1,12-dimethylspermine were tested for their ability to rescue cells. 1-methylspermidine was successful but 1,12-dimethylspermine failed to reverse AbeAdo-induced cytostasis. Treatment by 1-methylspermidine, a substrate for deoxyhypusine synthase, caused a reduction in the levels of unmodified eIF-5A and this correlation between hypusine synthesis and AbeAdo-induced cytostasis was further suggested by the failure of 1,12-dimethylspermine, which was not a substrate of deoxyhypusine synthase, to affect levels of eIF-5A (Byers *et al.* 1994).

8.1.1.5 Inhibition of hypusine biosynthetic enzymes

8.1.1.5.1 Deoxyhypusine synthase

Inhibition of deoxyhypusine synthase, the first enzyme of hypusine biosynthesis, was achieved using several mono-, di- and polyamines (Park & Wolff 1988). These initial studies indicated some structural features that were possibly important with regards to enzyme inhibition and thus, led to more detailed work by Jakus and co-workers (1993). These researchers demonstrated effective inhibition of deoxyhypusine synthase by several diamines and polyamines. 1,8-diaminooctane and 1,7-diaminoheptane were strong inhibitors of the enzyme while caldine (norspermidine) and N-(3-aminopropyl)cadaverine provided moderate inhibition. Inhibition was observed to be greatest with compounds that resembled spermidine in carbon chain length. The exception to this was the effective inhibition afforded by 1,3-diaminopropane (Jakus *et al.* 1993). This inhibition was, however, explained by product inhibition with 1,3-diaminopropane being produced when spermidine is cleaved prior to its addition to the lysine residue of eIF-5A (Park & Wolff 1988).

Since the fungicide guazatine, a guanylated polyamine derivative, had been reported to successfully inhibit deoxyhypusine synthase (Murphey & Gerner 1987), guanyl derivatives of these inhibitory diamines and polyamines were also synthesised. Monoguanyl derivatives were found to be more effective than the parent compounds, and indeed, the bisguanyl derivatives in inhibiting the enzyme. N¹-guanyl-1,7-diaminoheptane was the most effective inhibitor examined, having an IC₅₀ of 0.017 μ M (Jakus *et al.* 1993).

Compounds were tested for their ability to inhibit hypusine formation *in vivo* in CHO cells. All of the compounds tested: mono-guanyl derivatives of 1,7-diaminoheptane and 1,8diaminooctane and bis-guanyl derivatives of 1,6-diaminohexane, 1,7-diaminoheptane and 1,8-diaminooctane successfully reduced hypusine formation. N¹-guanyl-1,7diaminoheptane caused almost complete cessation of hypusine production at a concentration of 10 μ M (Jakus *et al.* 1993). Similar recent studies using HTC cells exposed to the parent diamines revealed interesting results. None of the four diamines examined affected cell growth and, indeed, 1,3-diaminopropane failed to produce an effect on hypusine content (Beppu *et al.* 1996). 1,6-diaminohexane and 1,7-diaminoheptane did, however, reduce hypusine synthesis, as did 1,8 diaminooctane although to a lesser degree. The lack of effect on cellular growth aside, the reduction in hypusine formation *in vivo* by the diamines was unlikely to wholly be caused by inhibition of deoxyhypusine synthase. Spermidine content was reduced in cells exposed to the inhibitors and this may have contributed to the observed reduction in hypusine content (Beppu *et al.* 1996).

Guanylated diamine studies concluded that N¹-guanyl-1,7-diaminoheptane and N¹-guanyl-1,8-diaminooctanc inhibit deoxyhypusine synthase *in vitro*, hypusine formation *in vivo* and reduce cellular growth in a correlated manner. Thus, the growth inhibition would appear to be due to enzyme inhibition and not the result of non-specific toxic effects. Compounds showed little effect on cellular polyamine levels, implying that the inhibitors do not interfere with other enzymes for which spermidine is a substrate (Jakus *et al.* 1993, Park *et al.* 1994). However, the growth inhibition afforded by, for example, the mono-guanyl derivative of 1,6-diaminohexane and the bis-guanyl derivatives of 1,6-diaminohexane and 1,8-diaminooctane may have been due to causes other than inhibition of hypusine synthesis. *In vivo* studies revealed that these compounds reduced growth of CHO cells but failed to produce comparative reductions in hypusine content. Intervention in cellular processes other than hypusine synthesis is likely to account for the arrest of growth by these compounds (Park *et al.* 1994). Solver Screek and

Further studies revealed that N¹-guanyl-1,7-diaminoheptane was actively taken up into CHO cells utilising the polyamine transport system. Guanylated diamines generally inhibited growth of other cell lines although the degree of inhibition varied and appeared to depend on the efficiency of inhibitor transport in different cell lines (Park *et al.* 1994). N¹-guanyl-1,7-diaminoheptane further demonstrated its antiproliferative ability with the supression of growth of mouse neuroblastoma and murine crythroleukemia cells at micromolar concentrations (Chen *et al.* 1996), as well as several other tumorigenic cell lines (Shi *et al.* 1996).

Work progressed when diamine and triamine analogues and derivatives were synthesised and tested for their ability to inhibit deoxyhypusine synthase. Few of these compounds were comparable with N¹-guanyl-1,7-diaminoheptane *in vitro* and none were as efficient in reducing hypusine synthesis *in vivo*. The arrest of cellular growth by these compounds could not be wholly attributed to inhibition of deoxyhypusine synthase (Lee *et al.* 1995).

8.1.1.5.2 Deoxyhypusine hydroxylase

Other studies have focussed on inhibition of deoxyhypusine hydroxylase, the second enzyme of hypusine biosynthesis, although this work is somewhat limited. The enzyme was successfully inhibited by metal chelators such as α,α -dipyridyl (Park *et al.* 1982), and hydralazine (Paz *et al.* 1984) *in vitro*. Polyamines are also known to inhibit the enzyme (Abbruzzese *et al.* 1989). Inhibition of deoxyhypusine hydroxylase by L-mimosine *in vivo* resulted in the arrest of cell cycle progression (Watson *et al.* 1991). *In vivo* inhibition was also observed in CHO cells cultured in the presence α,α -dipyridyl. Deoxyhypusine was observed to accumulate in cellular protein (Park *et al.* 1982; Beppu *et al.* 1996).

8.1.2 Aim of studies

Knowledge of the critical nature of hypusine for cellular proliferation and/or growth led to the supposition that control of amino acid production could provide a means of regulating cell growth. Control of growth of cancerous and normal cells has been achieved through the inhibition of hypusine biosynthetic enzymes (Park *et al.* 1994; Hanauske-Abel *et al.* 1994). In addition, eIF-5A has been implicated in human immuno-deficiency virus type 1 (HIV-1) replication and disruption of eIF-5A function may provide a means of combatting the virus (Ruhl *et al.* 1993).

Spermidine analogues were originally designed to interfere with the synthesis of spermidine via disruption of the enzyme spermidine synthase. They were also anticipated to act through disruption of cellular function(s) of spermidine. In hypusine synthesis, the provision of spermidine is critical, although in specific work related to AbeAdo-induced cytostasis various spermidine analogues have been observed to act as suitable substitutes for spermidine. Importantly, various analogues have been unable to act as substrates for deoxyhypusine synthase. Indeed, in related work, N¹-acetylspermidine was found to be unable to compete with spermidine *in vivo* (Gerner *et al.* 1986). Perhaps the most interesting aspect of work carried out so far lies in the observation that N-(3-aminopropyl)-1,4-diamino-trans-but-2-ene acted as a substrate for deoxyhypusine synthase yet failed to support hypusine synthesis (Byers *et al.* 1992). Presumably, analogues such as this compound could 'disable' biologically active eIF-5A, prevent hypusine formation and thus prove invaluable in preventing cellular proliferation and/or growth.

However, as observed in the studies relating to inhibition of the hypusine biosynthetic enzymes, many compounds differ in their effect on hypusine synthesis *in vitro* and *in vivo*.

Hypusine synthesis may also be affected without biosynthetic enzyme inhibition. Much work in this area is required. Nevertheless, the proposed use of spermidine analogues to affect the enzymes responsible for the synthesis of hypusine presents a specific antiproliferative strategy. Clearly, a specific target in an antiproliferative approach is undoubtedly superior to a wide ranging attempt to affect any cellular function(s) for which spermidine is important.

8.2 EFFECT OF VARIOUS DEOXYHYPUSINE SYNTHASE INHIBITORS ON MYCELIAL GROWTH OF *Pyrenophora avenae* on solid media

8.2.1 Introduction and Objectives

Although evidence of the presence of hypusine in *Neurospora crassa* does exist (Chen & Yang 1988; Yang *et al.* 1990), perturbation of hypusine formation as an antifungal or fungicidal strategy has not yet been explored. Given that CHA and spermidine analogues have demonstrated antifungal and fungicidal activity but no mode(s) of action have as yet been determined, investigation of fungal hypusine biosynthesis was deemed appropriate. Before attempting to study the effect of CHA and norspermidine on hypusine biosynthesis, the effects of known inhibitors of deoxyhypusine synthase on mycelial growth of *Pyrenophora avenae* on solid media was studied. This would give a clear indicator if perturbation of hypusine biosynthesis did indeed affect growth of *P. avenae*. Many of the known deoxyhypusine synthase inhibitors are related to guazatine (Panoctine, Rhone-Poulenc), a cereal seed dressing, thus antifungal activity was expected. Nevertheless, various inhibitors of deoxyhypusine synthase were generously supplied by Dr. J. E. Folk of the Enzyme Chemistry Section, National Institutes of Health, USA (Table 37). Although effects on fungal growth on solid media arc not always reliable indicators of *in vivo* fungal growth, such a study was appropriate given the limited amounts of compounds available.

8.2.2 Materials and Methods

P. avenue was maintained as described in section 3.1.2.1 (page 76). Fungus was grown on solid media amended with the various deoxyhypusine synthase inhibitors using the procedure described in section 3.1.2.2 (page 76), but modified to allow for the very limited amounts of inhibitors available.

Filter-sterilised inhibitor solutions (10 ml) were added to 30 ml of sterile potato dextrose agar (PDA) at 45-47°C to obtain the final desired concentrations of each compound. Concentrations varied from 1.0 mM depending on the amount of material available. Although 1.0 mM and lower concentrations of CHA and novel spermidine analogues did not produce inhibition of mycelial growth on solid media, norspermidine had inhibited growth at that concentration. The limited amounts of deoxyhypusine synthase inhibitors did not permit higher concentrations to be used for these studies. Sterile medium (10 ml) was added aseptically to a 60 mm plastic Petri dish. A 6 mm cork borer was used to remove plugs of mycelial from stock plates which were inverted and placed in the centre of each Petri dish. Inoculated plates were incubated in the dark at 24°C. Radial mycelial growth, excluding the mycelial plug, was measured 6 days after inoculation with *P. avenae*. Three measurements were taken from each plate. Control plates contained culture medium only.

Four replicates were used for each treatment with significance assessed using Student's *t*-test. All experiments were repeated and similar results obtained.

8.2.3 Results

All of the inhibitors examined reduced radial mycelial growth of *P. avenae* at the concentrations tested (Table 38, Plates 5-6). Four of the compounds completely inhibited fungal growth, with 4a doing so at a concentration as low as 0.61 mM. Compounds 48, 49 and 4e affected growth more profoundly than is reflected in the radial measurement data: growth was less dense in the presence of these compounds (Plate 6).

Compound	Structure	
3a ¹	H2NC(=NH)NH(CH2)8NH(CH2)8NHC(=NH)NH2	
3b 1	H ₂ NC(=NH)NH(CH ₂)7NH(CH ₂)7NHC(=NH)NH ₂	
4a ¹	H2N(CH2)8NH(CH2)8NHC(=NH)NH2	
4c '	H2N(CH2)4NH(CH2)4NHC(=NH)NH2	
11 ¹	$H_2N(CH_2)_3O(CH_2)_3NHC(=NH)NH_2$	
12 '	$H_2N(CH_2)_3S(CH_2)_3NHC(=NH)NH_2$	
21 '	H2NC(=NH)NH(CH2)7NHCH2CH3	
42 ²	H2NC(=NH)NH(CH2)7NHC(=NH)NH2	
43 ²	H2NC(=NII)NH(CH2)8NHC(=NII)NH2	
48 ²	H2NC(=NH)NH(CH2)7NH2	
49 ²	H2NC(=NH)NH(CH2)8NH2	

Table 37. Structures of deoxyhypusine synthase inhibitors

. . .

Note. Compounds are referred to as numbers as detailed in ¹ Lee et al. 1995 and ² Jakus et al. 1993.

Treatment	Concentration (mM)	Mycelial growth, measured as mean colony radius (mm)	Percentage reduction in radial mycelial growth
Control		25.7 ± 0.37	· · · · · · · · · · · · · · · · · · ·
4a	0.61	NG	100
3Ъ	0.9 7	NG	100
4e	0.84	20.7 ± 0.62	19.5
21	0.90	11.7 ± 0.21	54.4
42	1.0	1.2 ± 0.13	95.3
43	0.83	NG	100
48	1.0	$\textbf{20.1} \pm \textbf{0.43}$	21.8
49	1.0	17.8 ± 0.41	30.7
Control		24.3 ± 0.37	
3a	1.0	NG	100
11	1.0	12.2 ± 0.31	49.8
12	1.0	7.4 ± 0.18	69.5

Table 38: Effect of dcoxyhypusine synthasc inhibitors on mycelial growth of Pyrenophora avenae

のないのであるのである

ではないときの

1.00

Ĵ

Note: Values are shown as the means \pm SE of three measurements from each of four replicates.

For all treatments, significant differences were observed from the controls ($p \le 0.001$).

NG no growth



Plate 5. Effect of 1.0 mM concentrations of deoxyhypusine synthase inhibitors on mycelial growth of *Pyrenophora avenae*, grown on solid media for six days.



Plate 6. Effect of varying concentrations * of various deoxyhypusine synthase inhibitors on mycelial growth of *Pyrenophora avenae*, grown on solid media for six days.

* For concentrations used, see Table 38, page 190
8.3 EFFECT OF VARIOUS COMPOUNDS ON THE GROWTH OF *Pyrenophora avenae* in Liquid culture

8.3.1 Introduction and Objectives

Given that known inhibitors of deoxyhypusine synthase do indeed inhibit mycelial fungal growth of *P. avenae* on solid media, it was decided to determine the amount of deoxyhypusine and hypusine in fungus treated with CHA and norspermidine and thus investigate the effect of these compounds on hypusine biosynthesis. Various other compounds have demonstrated an effect on hypusine and deoxyhypusine formation *in vivo*, albeit in animal cells. Exposure of fungus to such compounds was thought to be worthwhile as an aid to developing the deoxyhypusine/hypusine determination procedure. Thus, if compounds acted in fungus as they do in animal cells, a series of known effects would be available for which comparison of CHA and norspermidine effects could be made.

DFMO (Gerner *et al.* 1986; Park 1987) and APA (A. Shirahata unpublished results; Beppu *et al.* 1996) are known to reduce synthesis of hypusine in cultured cells. 1,3diaminopropane and 1,6-diaminohexane have been reported to inhibit deoxyhypusine synthase (Park & Wolff 1988; Jakus *et al.* 1993) while 2,2'-dipyridyl, (α,α -dipyridyl), is known to inhibit the hydroxylation process and thus allow deoxyhypusine to accumulate (Park *et al.* 1982). Deoxyhypusine is normally present in cells at very low levels and can only be detected through disruption of deoxyhypusine hydroxylase.

8.3.2 Materials and Methods

P. avenae was grown in liquid culture in the presence of the various compounds associated with the inhibition of hypusine synthesis as detailed in section 3.2.2.2 (page 85). DFMO and APA⁷ were both used at 0.1 mM, limited amounts of these compounds not permitting higher concentrations to be used. 2,2'-dipyridyl was used at a concentration of 0.01 mM. A preliminary experiment had shown that fungal growth was completely prevented by 1.0 mM 2,2'-dipyridyl, whereby the normally yellow coloured PDB became pink in colour, indicating extreme toxicity. 1,3-diaminopropane and 1,6-diaminohexane were used at concentrations of 1.0 mM, such a concentration chosen on the basis of previous work which had shown CHA to affect *P. avenae* growth when used at 1.0 mM. Fungus was grown for 4 days in order to amass the material required for further analysis. Four replicates were used for each experiment and experiments were repeated with similar results. Statistical significance was assessed using Student's *t*-test.

⁷ APA was kindly provided by Professor A. Khomutov of the Russian Academy of Science, Moscow.

8.3.3 Results

1,3-diaminopropane failed to reduce the fresh weight of *P. avenae* at the concentration examined, although, it did appear to affect fungal growth in liquid culture. Fungus exposed to the compound was darker than the control fungus. DFMO, APA, 1,6-diaminohexane and 2,2'-dipyridyl all reduced the fresh weights of fungal mycelium grown in liquid culture in their presence for 4 days (Table 39), albeit, not always significantly.

Treatment	Concentration (mM)	Growth of fungus, measured as mean fresh weight (g)	
Control	,,	3.7 ± 0.31	
DFMO	0.1	3.2 ± 0.20	
APA	0.1	2.8 ± 0.18	
Control		3.9 ± 0.28	
1,3 diaminopropane	1.0	4.0 ± 0.19	
1,6 diaminohexane	1.0	2.7 ± 0.19 *	
Control	· · · · · · · · · · · · · · · · · · ·	3.3 ± 0.12	
2,2'-dipyridyl	0.01	2.8 ± 0.19	

Table 39: Effect of various compounds on the growth of Pyrenophora avenae in liquid culture for 4 days

Note: Values are shown as the means \pm SE of four replicates. Significant difference from control is shown as: * p \leq 0.05

8.4 EFFECT OF VARIOUS COMPOUNDS ON FUNGAL HYPUSINE AND DEOXYHYPUSINE CONTENT

8.4.1 Introduction and Objectives

Numerous methods have been reported for determination of hypusine and dcoxyhypusine. The majority of these procedures involve, firstly, the collection of cellular protein and then acid hydrolysis of that protein to release the bound amino acids. Ion-exchange chromatography is used to 'clean up' the sample by removing the bulk of the amino acids from hypusine and deoxyhypusine. Methods then vary in their preference for pre or post HPLC column derivatisation before separation and detection of the derivatives fluorometrically. Such a method using o-phthalaldehyde (OPA) pre column derivatisation was used by Beninati and co-workers (1990) while Bartig & Klink (1992) used 4-dimethylaminoazobenzene-4'-sulphonyl chloride derivatisation. Post column OPA derivatisation was employed by Sano and co-workers (1984) and Beppu *et al.* (1996).

Alternatively, various groups of workers have used deoxyhypusine synthase enzyme assays to study hypusine biosynthesis. One such *in vitro* enzyme assay procedure involves examination of the formation of radio-labelled products from labelled spermidine or lysine. Ion exchange chromatography is necessary to separate the products before radioactive counting to assess the flux of starting materials through to products (for example, Park *et al.* 1991). Another assay procedure involves measurement of the radioactivity found formed in protein bound hypusine and deoxyhypusine from labelled spermidine. Bound products are separated by gel electrophoresis before radioactivity counting of each protein (Chen & Dou 1988; Dou & Chen 1990). Assays have also been reported for deoxyhypusine hydroxylase, (for example, Csonga *et al.* 1996), however, all enzyme assays require the use of purified enzyme and precursor eIF-5A protein and were thus deemed unsuitable for the purposes of the present study.

It was decided to determine hypusine and deoxyhypusine content in fungal tissue by adapting the method reported by Beppu *et al.* (1996). These authors argued that previously reported methods using such a procedure of protein collection, acid hydrolysis, ion exchange chromatography and HPLC analysis, were unsatisfactory with regards to separation and sensitivity. However, the procedure reported by Beninati *et al.* (1990), using pre column OPA derivatisation was deemed satisfactory. Therefore, both of these previously reported studies were consulted during the 'design' of the present method. These procedures were deemed the most suitable in view of the resources and equipment at the disposal of the present study.

8.4.2 Materials and Methods

Authentic hypusine was kindly provided for the study by Dr. T. Shiba of the Protein Research Foundation, Osaka, Japan. Deoxyhypusine was generously donated by Dr. J. E. Folk.

The method reported by Beppu *et al.* (1996) initially involved collection of cellular protein. Briefly, HTC cells were washed twice and suspended in phosphate buffered saline before being sonicated and the lysate centrifuged. Protein in the supernataut was then determined.

Thus, fungus (7 g) was ground in a pre chilled mortar and pestle with 7 mls of a phosphate buffer, pH 7.6. This pH was chosen since it was the pH of the grinding buffer used for ODC and AdoMetDC enzyme assays. Portions of the crude extract wcrc sonicated on ice using a Soniprep 150 for 10 cycles of 10 seconds on/20 seconds off before being pooled and centrifuged at 16000 g for 25 min at 0°C, as for free polyamine analysis. Protein assays were carried out using the method of Lowry *et al.* (1951) with BSA as the standard.

「「「「「「「「「「」」」」

To precipitate the protein, Beppu et al. (1996) added 10 % trichloroacetic acid (3 ml) to HTC cell supernatant (1 ml) which contained 0.5 - 10 mg of protein. The resulting precipitate was washed once with 10 % trichloroacetic acid and then powdered using acctonc and ether, aiming to remove low molecular weight compounds, including trichloroacetic acid.

Thus, 10 % trichloroacetic acid (30 ml) was added to supernatant (10 ml) and the precipitate allowed to settle. Fungus generally contained 0.2 - 0.3 mg of protein per ml, thus approximately 2 - 3 mg of protein was collected. The supernatant was collected after filtering the mixture under gravity, allowing complete recovery of protein. Filtration under vacuum resulted in impregnation of the filter paper with the protein sample. The precipitate was washed once using 10 % trichloroacetic acid. Acetone (20 ml) was added followed by diethyl ether (30 mls) and the solvents removed *in vacuo*.

To the IITC cell residue, Beppu and co-workers (1996) added 500 pmol of an internal standard, a synthesised analogue of deoxyhypusine synthase, before hydrolysing the mixture in 6 M hydrochloric acid for 13 hours at 120°C.

These authors had tested many compounds for use as an internal standard and thus, 1,6diaminohexane (500 pmol) was added to the fungal residue. Under the HPLC conditions described by the authors, 1,6-diaminohexane was retained for 12.4 mins with hypusine and deoxyhypusine retained for 14.4 and 17.2 mins respectively. Thus, adequate separation between the internal standard and the desired compounds was envisaged. 6 M hydrochloric acid (20 ml) was added and the mixture hydrolysed for 13 hours, using an oil bath.

Upon completion of hydrolysis, the fungal sample was treated as the authors had done for hydrolysed HTC cell protein. Solvents were removed *in vacuo* and the residue dissolved in distilled water (10 ml). The solution was passed through a 0.45 μ m syringe filter and 1 M pyridine (100 μ l) added.

Using the method originally described by Samejima and co-workers (1976), the filtrate was applied to a small column (10 x 130 mm) of CM23 carboxymethyl cellulose, a fibrous cation exchanger (Whatman), equilibrated with 0.025 M pyridine-acetic acid buffer, pH 5. The column was eluted sequentially with 0.025 M (10 ml), 0.05 M (5 ml), 0.075 M (5 ml) and 0.1 M (5 ml) pyridine-acetic acid buffers, pH 5. Hypusine, deoxyhypusine and the internal standard were predicted to be recovered in the last fraction and this was evaporated to dryness *in vacuo* before dissolving the residue in distilled water (1 ml) and freezing at -20°C until required for HPLC analysis. Beppu *et al.* (1996) dissolved residues in HPLC clution buffer before directly injecting aliquots into the HPLC apparatus. In the present study, only one sample could be generated at any one time and intensive HPLC apparatus use meant that samples had to be frozen until their generation was complete. Nothing was known of the stability of the samples in the HPLC elution buffer, thus water was chosen as the medium for freezing the samples. Indeed, Beninati *et al.* (1990) froze stock solutions of hypusine and deoxyhypusine at -20°C in distilled water.

Thus, fungus grown as a control was prepared for HPLC analysis along with *P. avenae* exposed for 4 days to those compounds described in section 8.3: DFMO, APA, 1,6-diaminohexane, 1,3-diaminopropane, 2,2'-dipyridyl, 1.0 mM CHA and 0.01 mM norspermidine. Such CHA and norspermidine concentrations were chosen on the basis of previous work. The lower concentrations, as opposed to 2.75 mM CHA and 0.2 mM norspermidine, were used since large amounts of fungal material were necessary to collect sufficient protein for analysis.

An additional sample was prepared which would act as a test for hypusine presence, as described by Beppu *et al.* (1996). Fungus grown as a control was prepared in the manner as described previously. After ion-exchange chromatography, solvents were removed *in vacuo* from the final fraction recovered and the residue dissolved in 0.01 M hydrochloric acid (200 μ l). 100 μ l of the sample was frozen to await HPLC analysis. To the other 100 μ l, 100 μ l of 40 mM sodium periodate solution was added and the mixture left to stand at room temperature for 12 hours before being frozen to await HPLC analysis. Oxidative reaction

with sodium periodate is envisaged to cleave the β -aminoethanol portion of hypusine. Thus, HPLC analysis of the treated residue (50 µl) should reveal the absence of a peak when compared to the untreated sample (25 µl), indicating therefore that the peak should be hypusine. Twice the amount of the sodium periodate treated sample has to be injected compared to the untreated sample to allow for the 50 % dilution by sodium periodate.

State and the second second

HPLC apparatus consisted of an eluant pump (Model LC-10AD, Shimadzu, Japan) and a reagent pump (Model constaMetric 3500, LDC Analytical, Florida) connected to an autoinjector (Model SIL-9A, Shimadzu, Japan), fluorometric detector (Model 821-FP, Jasco, Japan) and chromato recorder (Model C-R6A, Shimadzu, Japan).

HPLC conditions were as follows: a C18 column (Spherclone ODS-2, 150 x 4.6 mm i.d., Phenomenex, UK) connected with a guard cartridge (ODS-2, 30 x 4.6 mm i.d., Phenomenex, UK), was kept at ambient temperature (22°C). Eluant buffer contained 0.1 M NaH₂PO₄.H₂O, adjusted to pH 2 using phosphoric acid, with 10 mM sodium *n*hexylsulfate and 3.5 % acetonitrile added. Buffer was eluted at a flow rate of 1 ml/min and fresh eluant was prepared daily.

Amines were detected fluorometrically by the OPA post column method as described by Seiler & Knödgen (1985). Reagent was prepared by dissolving boric acid (25 g) and potassium hydroxide (22 g) in 500 ml distilled water. 1.5 ml Brij-35 solution, 1.5 ml of β -mercaptoethanol and a solution of 200 mg of o-phthalaldehyde, dissolved in 2.5 ml methanol, were added to the solution. The reagent was stored in a dark bottle and prepared fresh daily. The derivatising reagent was delivered in the HPLC system at a rate of 1 ml/min.

The column eluate was thus mixed with the derivatising reagent in a 1:1 ratio and passed through a mixing column (50 x 4.6 mm i.d., filled with glass beads, 75 μ m, Supelco, UK), before fluorescence was detected using an excitation wavelength of 334 nm and emission wavelength of 440 nm.

8.4.3 Results

Due to the extremely limited amount of hypusine available, initial workings of the HPLC apparatus were undertaken using samples of 1,6-diaminohexane, the internal standard. Problems were immediately apparent with respect to the reproducibility of peaks. Indeed, a number of repeated injections of the same 1,6-diaminohexane solution resulted in ever decreasing peak areas although retention times remained steady. A similar phenomenon was observed with authentic deoxyhypusine, a small amount of which was used in case the

problems lay with 1,6-diaminohexane. Several means of fine tuning were employed in an attempt to solve the problem. Fresh derivatising reagent was prepared during a single day, although it was not thought that the reagent would deteriorate so quickly. Indeed, this made little difference to the results obtained. Injection volumes were increased from 50 μ l to 150 μ l, to ensure that excess sample was present and the full 50 μ l was loaded onto the column. Gain and response settings on the detector were also adjusted.

A mechanical problem did develop in that a loop was blocked, preventing sample reaching the column. Partial blocking may have contributed to the problems detected prior to discovery and clearance of the blockage.

Subsequently, an attempt to detect vitamins A and E for which the HPLC apparatus was normally used, was successful, indicating that mechanical problems had ceased. Such problems did, however, account for some considerable time, during which attempts were being made to develop the HPLC methodology.

The occurrence of multiple peaks (so called 'split peaks') indicated a void in the system. Attempts were made to eliminate the void and were apparently successful. Collection of a chromatogram of authentic hypusine was attempted. At this point, it was difficult to determine whether the appearance of multiple peaks was a characteristic of the compound i.e. contaminants, or that a void was still present in the system. Elimination of any void was attempted and consultation with Dr. Shiba revealed that the hypusine was indeed likely to consist of two diastereoisomers.

Aliquots of the control sample were injected and revealed many peaks early on in collection of the chromatogram yet little in the region where hypusine and the internal standard were expected. Aliquots of one of the treatment samples (1.0 mM CHA) were also injected. Still variability occurred with respect to the peak areas.

The possibility of carry over of sample, a possible cause of peak area irregularity was investigated. Thus, three injections of a sample were made before running 3 aliquots of distilled water through the system. No peaks were observed in the water samples and thus, carry over of sample was excluded.

Ultimately, comparison of the control sample with authentic hypusine revealed that indeed, hypusine may be present in the fungus (Figure 21). However, given the inherent variability of the repeated injections, the work could not be pursued further within the time permitted. A matter of weeks of HPLC apparatus use were negotiated originally when a much longer

200

time frame is generally accepted to be required for fine-tuning of HPLC methods. Further consultation with Dr. Shiba also revealed that the two peaks in the hypusine sample were not likely to correspond to two diastereoisomers, but that a contaminant was present.

Nevertheless, the work did indicate areas which may improve the workings of the HPLC system. The second (reagent) pump may have been operating variably. Pressure fluctuations may have been the cause of the erratic peak areas observed. Little resistance is offered against the pump and the inclusion of an inert column between the pump and the T-piece where the reagent meets the eluate may provide such an obstacle for the pump to operate against, and thus improve efficiency of the pump.

Indeed, the use of two pumps was an unusual means of operating for those who frequently used the HPLC apparatus. Pre column derivatisation would exclude the need for the second pump and may increase efficiency of the system.

Peaks corresponding to 1,6-diaminohexane were undetected in the fungal samples examined by HPLC. The inclusion of a compound to be used as an internal standard which acted more like hypusine and deoxyhypusine in the preparative procedure is desirable. 1,6diaminohexane may have been lost from the samples in the ion-exchange chromatography step.

Clearly, continuation of this work, developing a satisfactory method for hypusine and deoxyhypusine determination would be prudent. Methods reported until now have been for amino acid determination in cultured animal cells and mammalian organs. Fungal samples may require alternative or additional 'clean up' steps to remove compounds which hinder HPLC detection of hypusine and deoxyhypusine.

START



CHROMA	TOPAC C-R	(6A			r l L C	С — — — — —
SAMPLE	HO 9				METHOD	9661
REPORT	NO 1183					
PKNO	TIME	AREA	ħK	IDNO	CONC	NAME
1	2.375	7734			1.305	55
2	2.55	4127	ų		0,696	5
3	2.785	3578	¥		0.604	ł
4	3.642	3154			0.532	:4
5	5.242	1691			0.285	54
6	12.275	282239			47.638	}5
ž	13.642	289936	Ų		48.932	77
						~ ~~
	TOTAL	592460			100	

1.

あるという 二日 大学 二日 ちょうしい

Ζ.



CHROMA Sample Report	TOPAC C-R NO 0 NO 1104	6A			FILE Hethod	7 0061
PKNO	TIME	AREA	ΜK	IDNO	CONC	NAME
1 2 3 4 5 	2.365 2.533 2.865 3.63 4.412 12.298 13.482	26518 6948 17817 3054 46908 25006 12723	Ų Ų		19.882 4.994 12.821 2.197 33.755 17.994 9.155	3 3 9 1 1 3
	TOTAL	138967		-	100	



Chromatogram 1 (authentic hypusine) produced peaks at 12.28 and 13.64 minutes. Chromatogram 2 (fungus grown as control) produced peaks at 12.30 and 13.48 minutes.

8.5 DISCUSSION

The antifungal activity demonstrated against P. avenae grown on solid media by the inhibitors of deoxyhypusine synthase is in agreement with those findings which show that the compounds affect *in vitro* enzyme activity and *in vivo* growth of CHO cells (Jakus *et al.* 1993; Lee *et al.* 1995). Compound 3a, guazatine, was extremely effective, affording complete control of fungal growth at a concentration of 1.0 mM. Given that guazatine is a commercial fungicide, this control is not surprising. Similarly, the closely related compounds 3b and 4a completely prevented fungal growth when used at concentrations which were in fact less than 1.0 mM.

Compound 4a is the mono-guanylated derivative of bis-guanylated 3a and had demonstrated reduced inhibitory activity against deoxyhypusine synthase *in vitro* compared to 3a in previous work (Lee *et al.* 1995). However, growth inhibition of CHO cells *in vivo* by 4a was greater than that provided by 3a, leading the authors to conclude that growth inhibition was not wholly caused by an effect on deoxyhypusine synthase (Lee *et al.* 1995). In the present study, 4a reduced fungal growth at a lower concentration (0.61 mM) compared to 3a (1.0 mM). However, since both compounds completely prevented fungal growth, further study would be necessary to conclude which of the compounds was more effective. Nevertheless, the antifungal activity demonstrated by the compounds agrees with previous work which showed the compounds to possess antiproliferative activity.

Lee and co-workers (1995) made comparisons of the effects of 3a and 3b with the guanylated derivatives of 1,8-diaminooctane and 1,7-diaminoheptane. Firstly, 3b was found to be more effective at inhibiting deoxyhypusine synthase *in vitro* than 3a and also more effective at inhibiting growth of CHO cells. In the present study, both compounds completely prevented fungal growth, thus confirming the antiproliferative activity already reported. Further study would be necessary to distinguish between efficacy of the compounds.

1. S. W.

3a provided greater inhibition of deoxyhypusine synthase *in vitro* compared to bisguanylated 1,8-diaminooctane (compound 43) but less inhibition compared to the monoguanylated derivative, compound 49. A similar relationship was observed between 3b and the bis-guanylated derivative of 1,7-diaminoheptane (compound 42) and its monoguanylated counterpart, compound 48. Thus, mono-guanylated derivatives were shown to provide increased inhibitory activity of deoxyhypusine synthase *in vitro* over bis-guanylated derivatives (Lee *et al.* 1995). However, the present study demonstrated that compounds 42 and 43, the bis-guanylated derivatives, afforded greater control of fungal growth than their mono-guanylated counterparts, compounds 48 and 49, disagreeing with the previous work. Jakus and co-workers (1993) had also found the mono-guanylated derivatives inhibited deoxyhypusine synthase *in vitro* more effectively than the bis-guanylated derivatives. This disagreement with the present study was further substantiated since Jakus *et al.* (1993) and Park *et al.* (1994) had indicated that the mono-guanylated compounds 48 and 49 provided better control than the bis-guanylated compounds 42 and 43 in reducing hypusine formation and cellular proliferation of CHO cells.

As with 3b and 3a, the 1,7-diaminoheptane derivatives were more effective than 1,8diaminooctane derivatives both *in vitro*, (Jakus *et al.* 1993; Lee *et al.* 1995) and *in vivo* (Jakus *et al.* 1993). However, the present study showed little differences between the 1,7diaminoheptane derivatives (3b, 42 and 48) compared to their 1,8-diaminooctane counterparts (3a, 43 and 49). Further study, involving a range of inhibitor concentrations would be necessary before any firm conclusions could be made about the respective efficacies of these compounds.

Previous work indicated that N¹-guanyl-1,7-diaminoheptane (compound 48) is the strongest known inhibitor of deoxyhypusine synthase both *in vitro* and *in vivo* (Jakus *et al.* 1993; Park *et al.* 1994). However, the present study found the inhibitor to only reduce mycelial growth of *P. avenae* by 21.8 %, when used at a concentration of 1.0 mM. Workers noted that the inhibitory activity exerted by N¹-guanyl-1,7-diaminoheptane in various cell lines depended on uptake of the compound into cells (Park *et al.* 1994). Transport into fungal cells may thus have affected the degree of efficacy afforded by the compound. Compounds 48 and 49 did affect fungal growth more so than is reflected in the radial mycelial growth measurements (Plate 6). Indeed, Mandels (1965) stated that no necessary correlation between the spread of mycelial growth on a solid surface and the total amount of fungus produced existed and that the use of linear measurements for quantifying growth was misleading. Clearly, compounds 48 and 49 were less effective than 42 and 43. Further work using fungus grown in liquid culture would be prudent.

The effects of the remaining compounds, 4e, 11, 12 and 21 agree with the work of Lee *et al.* (1995) who showed that each compound effectively reduced deoxyhypusine synthase activity *in vitro* and reduced growth of CHO cells *in vivo*. Perhaps the control afforded by 4c is less than would have been expected given the low IC₅₀ value of 0.16 μ M and good inhibitory activity shown against CHO cells, however, Plate 6 does indicate that growth was more markedly affected than suggested by the mycelial growth measurements.

Interestingly, compounds 42, 43, 48, 49 and perhaps 4e are the only compounds for which evidence exists that their *in vivo* inhibitory activity is indeed related to deoxyhypusine

synthase inhibition (Jakus *et al.* 1993; Lee *et al.* 1995), although Park *et al.* (1994) suggested that the growth inhibition of bis-guanylated 1,8-diaminooctane may not be wholly due to inhibition of hypusine biosynthesis. The remaining compounds are likely to affect hypusine formation by some means other than deoxyhypusine synthase inhibition. Thus, their antifungal activity may be the result of more than one mode of action.

The inhibitory effects of the ODC inhibitors DFMO and APA against *P. avenae* grown in liquid culture are in agreement with previous work carried out in such areas. The growth inhibitory activity of DFMO is well documented, with such effects indeed being observed against fungal growth (for example, West & Walters 1989). APA has been shown to affect HTC cellular growth (A. Shirahata unpublished results) although the compound had little effect on the growth of CHO cells (Beppu *et al.* 1996). Such DFMO and APA treatment of cultured cells has been associated with a reduction in spermidine content as well as a reduction in the rate of hypusine synthesis (Gerner *et al.* 1986; Park 1987; A. Shirahata, unpublished results; Beppu *et al.* 1996). It should thus be noted that DFMO and APA perhaps cause a reduction in hypusine formation as a result of the lack of spermidine in the cells and not in fact due to an effect on the hypusine biosynthetic enzymes (Park 1987; Beppu *et al.* 1996). Nevertheless, perturbation of hypusine formation is envisaged to occur in fungus exposed to the compounds, and such fungus will thus be important in aiding the measurement of fungal hypusine and deoxyhypusine.

Fungal growth inhibition by 2,2'-dipyridyl agrees with the previous work carried out by Beppu *et al.* (1996), who showed that CHO cellular growth was affected by this compound. Hypusine content of CHO cells decreased while deoxyhypusine content increased. As spermidine levels did not differ from the control cells, inhibition of deoxyhypusine hydroxylase was suggested to have occurred (Beppu *et al.* 1996). Deoxyhypusine hydroxylase inhibition in fungus would allow detection of deoxyhypusine, and thus fungus exposed to 2,2'-dipyridyl should aid the development of a method to determine deoxyhypusine content of fungus.

The diamines 1,3-diaminopropane and 1,6-diaminohexane had no effect on CHO cellular growth (Beppu *et al.* 1996), thereby disagreeing with the findings of the present study where the compounds inhibited fungal growth. In such previous work, 1,3-diaminopropane had little effect on hypusine content of CHO cells (Beppu *et al.* 1996), even though it is well documented to inhibit deoxyhypusine synthase *in vitro* (Park & Wolff 1988; Wolff *et al.* 1992; Jakus *et al.* 1993). 1,6-diaminohexane reduced hypusine content of CHO cells, however, since spermidine content was also reduced, the reduction in hypusine may have

been due to a lack of spermidine rather than inhibition of deoxyhypusine synthase (Beppu *et al.* 1996). 1,6-diaminohexane has also demonstrated inhibitory activity towards deoxyhypusine synthase *in vitro* (Jakus *et al.* 1993). Whatever effects these compounds have on fungal hypusine biosynthesis, hypusine content is likely to be altered, and thus, the use of these compounds could prove useful in determining a method for the study of hypusine formation in fungal cells.

In spite of this preliminary work, hypusine and deoxyhypusine levels in control and treated fungus are still undetermined as yet. The study has shown that known inhibitors of deoxyhypusine synthase can successfully inhibit mycelial growth of *P. avenae* on solid media. Whether these compounds do in fact act by inhibition of the enzyme remains to be seen. Some compounds may act by inhibition of deoxyhypusine synthase and others by disruption of hypusine formation via alternative mechanisms.

「日本」となるとないないのである

「「「「「「「「「」」」」」

Spermidine analogues have demonstrated inhibitory activity towards deoxyhypusine synthase *in vitro*. For example, caldine, otherwise known as norspermidine, affected the enzyme with an IC₅₀ value of 41.2 μ m (Jakus *et al.* 1993). *In vitro* inhibition of the enzyme by the compound was also observed by Wolff and co-workers (1992) and Park & Wolff (1988). However, N¹-acetylspermidine did not possess inhibitory properties against deoxyhypusine synthase *in vitro* (Jakus *et al.* 1993), nor did it compete with spermidine in hypusine formation (Gerner *et al.* 1986). Clearly, further work in this area would be prudent to determine whether an effect on hypusine formation and/or its biosynthetic enzymes is caused by CHA and norspermidine in *P. avenae*.

Chapter 9

General discussion

9. GENERAL DISCUSSION

This study has shown that CHA and norspermidine possess fungicidal and antifungal activity against a number of plant pathogens grown on host plants and on artificial media. Furthermore, the compounds controlled powdery mildew on barley in the field. Such findings for CHA are in agreement with reported inhibition of *Gaeumannomyces graminis* (West & Walters 1989) and powdery mildew on barley (West & Walters 1988). In addition, growth inhibitory activity has been demonstrated against other organisms such as tumours (Ito *et al.* 1982) and carrot cell cultures (Feirer *et al.* 1985). Norspermidine has been shown to inhibit growth of cultured cells (Sunkara *et al.* 1988), tumours (Prakash *et al.* 1988).

 N^{1} - and N^{8} -acetylspermidine possessed limited antifungal activity although fungicidal activity was observed against powdery mildew in previous work (S. A. Foster unpublished results). Such findings are in agreement with reports of inhibition of cultured cell growth by the spermidine analogues, N^{1} , N^{8} -bis(ethyl)spermidine (BES) and N^{1} , N^{8} -bis(propyl)spermidine (BPS) (Porter *et al.* 1985).

In addition, novel spermidine analogues possessed antifungal activity and although some analogues did not affect growth of *P. avenae*, previous work had shown that two of the analogues possessed fungicidal activity against a number of plant pathogens grown on host plants (N. D. Havis unpublished results). Thus, CHA, norspermidine, N¹- and N⁸- acetylspermidine and novel spermidine analogues have been shown to provide control of a wide range of plant pathogenic fungi on plants and when grown on artificial media.

Interestingly, there was generally little difference in the relative efficacies of CHA and norspermidine, at comparable concentrations, against plant pathogens on host plants. Norspermidine did appear to affect apple powdery mildew and potato late blight more effectively than CHA, but the differences were small. Against plant pathogenic fungi grown on artificial media however, norspermidine was much more effective than CHA, with lower concentrations required for growth inhibition. The reasons for such differences are not known, but it should be noted that differences do exist between fungi growing on plants and fungi growing on artificial media. On plants, the inhibitors act upon early development of the fungus while on solid media or in liquid culture, fungal mycelium is exposed to the inhibitors. *In vitro* growth may not therefore reflect *in vivo* sensitivity. Also, since biotrophic fungi cannot be grown easily on artificial media, other species of fungi must be used for *in vitro* analyses. Differences may therefore exist in the effects of the inhibitors on the biotrophic fungal species and those fungal species grown in culture.

ALCHARTRANS LEODARY Although the compounds demonstrated antifungal and fungicidal activity, they did not affect spermidine biosynthesis of *P. avenae* grown in liquid culture. CHA is a known inhibitor of spermidine synthase from a variety of sources including mammalian cells (Hibasami *et al.* 1980), trypanosomes, plants and bacteria (Pegg *et al.* 1983; Pösö *et al.* 1983; Paulin *et al.* 1983; Mattila *et al.* 1984; Sindhu & Cohen 1984). *In vivo* inhibition of spermidine synthase has also been demonstrated in protoplasts from Chinese cabbage leaves (Greenberg & Cohen 1985), while growth inhibition of pathogenic yeasts was associated with polyamine biosynthesis inhibition (Pfaller *et al.* 1988, 1990). The present study is therefore in disagreement with the widely regarded view that CHA is an inhibitor of spermidine synthase.

However, there are a number of reports which indicate that such effects of CHA are not true for every organism. The enzyme from some bacteria was unaffected by the compound (Pösö *et al.* 1983; Mattila *et al.* 1984). CHA growth inhibitory action has also been demonstrated against potatoes on inhibitor amended medium, and organogenesis in tobacco callus culture systems without an effect on spermidine biosynthesis (Tiburcio *et al.* 1987; Masse *et al.* 1988). Similarly, growth of *Agrobacterium tumefaciens* was reduced by CHA, yet was not associated with a reduction in polyamine biosynthesis (Ponappa *et al.* 1992). Pegg & Williams-Ashman (1987) noted how information on the specificity of CHA was lacking. AdoDato is known to be a more specific inhibitor of spermidine synthase (Pegg 1988), although, for the purposes of this study, CHA was used as it is commercially available and inexpensive.

Little is known of the effects on spermidine synthase of norspermidine, N^{1} - and N^{8} acetylspermidine. A norspermidine-induced effect on potato and maize root growth on inhibitor amended media was associated with spermidine biosynthesis inhibition (Masse *et al.* 1985, 1988). However, Sunkara *et al.* (1988) concluded that an effect by the compound on HeLa cell growth in culture was related to mitosis in cells. N^{1} , N^{8} -bis(ethyl)spermidine and N^{1} , N^{8} -bis(propyl)spermidine were associated with an inhibition of spermidine biosynthesis in cultured L1210 cells (Porter *et al.* 1985). However, one other analogue possessed inhibitory activity yet its mode of action was unclear (Porter *et al.* 1985). Thus it can be seen that while a lack of effect on spermidine synthase by CHA, norspermidine, N^{1} and N^{8} -acetylspermidine and the novel spermidine analogues in this study was somewhat unexpected, such effects are clearly not unusual in other organisms.

Consideration should be given at this point to the difficulties in comparing work in fungal systems to work in related areas. Many examples are available of different effects of the same inhibitor with regards to different organisms, testament to the diversity of organisms.

Examples include the ODC inhibitor RR-MAP which was extremely effective against growth of rat HTC cells (Mamont *et al.* 1984), yet only partially effective against growth of powdery mildew on barley (West & Walters 1988). Also the AdoMetDC inhibitor, EMGBG, failed to affect mouse leukemia cells in culture (Elo *et al.* 1986), yet was effective in reducing mycelial growth of *P. avenae* on solid media (Foster & Walters 1990).

The lack of effect on spermidine synthase, in spite of demonstrating antifungal and fungicidal activity should not be unexpected in view of previous work using putrescine analogues. Such analogues were indeed effective in reducing plant pathogens grown on host plants and in culture. However, varying effects were observed on polyamine biosynthetic enzyme activities and intracellular polyamines in *P. avenae*. The authors concluded that the effects could not wholly account for the observed antifungal and fungicidal activity (Havis *et al.* 1994a, c; Havis *et al.* 1997). Interestingly, some cyclic analogues demonstrated fungicidal activity yet failed to even affect growth of *P. avenae* in culture or polyamine biosynthesis in the fungus (Havis *et al.* 1996a).

Despite not causing an effect on spermidine synthase activity in *P. avenae*, CHA caused an accumulation of putrescine, the cause of which could not be determined. Such an increase in putrescine has also been found in *P. avenae* exposed to E-BED and has been implicated in the antifungal activity of the putrescine analogue (Havis *et al.* 1994a). Although an accumulation of putrescine has been associated with maintenance of membrane integrity (Srivastava & Smith 1982), it is likely that this increase may have lead to destruction of the membrane, as suggested by DiTomaso *et al.* (1989) for maize roots.

Although Walters & McPherson (1997) found that E-BED caused reductions in uptake of amino acids, polyamines and sugars in *P. avenae*, the findings of the present study indicated over-whelmingly that destruction of the biphasic uptake system was caused by exposure of the fungus to CHA or norspermidine. This was thought to be caused by the compounds destabilising the membranes and thus destroying the transport mechanism. Further evidence of the association of polyamine analogues with cell membranes is found in the reported displacement of spermine from intracellular binding sites in *Phytophthora infestans* by E-TED (Havis *et al.* 1996b).

Thus, a prudent area for future work would be the examination of the effect of the compounds on fungal cellular membranes. Ion leakage is a known symptom of membrane damage and conductivity measurements would be a useful aid to determining whether or not membranes were damaged by exposure to the inhibitors. A complicating factor would arise from efflux of amines from the fungus as a compensatory measure to any fluctuations in the polyamine intracellular pool.

Inhibition of spermidine synthase did not occur in fungus exposed to any of the compounds: CHA, norspermidine, N^{1} - and N^{8} -acetylspermidine and the novel spermidine analogues. Only CHA caused an increase in putrescine in *P. avenae*. It should be noted that other perturbations in polyamine metabolism may have occurred following treatment with CHA or the other compounds and yet may have been masked by the many compensatory mechanisms available to cells to respond to fluctuations in the free polyamine pool. However, any undetected changes clearly have little effect on the antifungal or fungicidal activity of the compounds. Thus, alternative modes of action should be sought.

Perturbation of hypusine biosynthesis was shown to be possible in fungal tissue, providing the deoxyhypusine synthase inhibitors tested act against fungi as they do in cultured animal cells. Such inhibition of hypusine synthesis clearly had antifungal effects. The possibility that CHA and the spermidine analogues act by inhibiting hypusine biosynthesis warrants further work.

Many more antiproliferative mechanisms have been suggested as possible functions of polyamine biosynthetic inhibitors and polyamine analogues in various organisms and are thus worthy of consideration for future work in fungal systems. A number of bis(benzyl)polyamine analogues were reported to be potent inhibitors of growth of malaria causing parasites (Bitonti *et al.* 1989). Such analogues were thought to act by binding to DNA and causing disruption of macromolecular synthesis (Bitonti *et al.* 1989). Such a mechanism was originally hypothesised by Porter & Sufrin (1986) when they suggested analogues to be more effective than irreversible enzyme inhibitors.

「「「「「「「」」」」」「「「」」」」」」」」」

Depletion of spermidine has been suggested to be associated with the reduction of the trypanosome spermidine-containing enzyme co-factor, trypanothione, which is obligatory for glutathione reductase activity (Fairlamb *et al.* 1985). Porter & Sufrin (1986) hypothesised that analogues may act by inhibiting or regulating polyamine biosynthetic enzyme activity. It is feasible that other enzymatic processes may involve polyamines and that CHA or spermidine analogues may exert their effects on such alternative cellular functions. Indeed, hypusine biosynthesis is one such example of another cellular process involving spermidine. Zarb & Walters (1994a, c) suggested that polyamine biosynthesis inhibitors may be involved in amino acid synthesis in *Paxillus involutus* and *Laccaria proxima*.

Inhibitors of polyamine biosynthesis were implicated in aberrant methylation in trypanosomes (Byers *et al.* 1991). Indeed, exposure of fungus to E-BED inhibited DNA methylation in *P. avenae* (Walters 1997). Many authors have implicated polyamines and methylation with differentiation of fungi, suggesting that polyamines are able to control expression of genes by interfering with the methylation process (Cano *et al.* 1988; Ruiz-Herrera 1994). An inhibitor of polyamine biosynthesis has also been reported to inhibit DNA methylation (Cano *et al.* 1988; Cano-Canchola *et al.* 1992), thus, clearly, this may be an area worthy of future consideration with regards to determining the mode of action of CHA and the spermidine analogues.

Another cellular process in which polyamines are implicated is protein phosphorylation. Phosphorylation is necessary for the activation of some enzymes and polyamines have been implicated in such regulation of enzyme activity. Differing effects are apparent with polyamines stimulating some enzymes (Velutthambi & Poovaiah 1984), yet depressing activity of others (Bothma & Dubery 1991). Little work has examined fungal protein phosphorylation with regards to the effects of polyamine biosynthetic enzyme inhibitors or polyamine analogues, thus, further work in this area may be prudent.

In conclusion, this study has shown that CHA and norspermidine possess the ability to control plant pathogenic infections on host plants under glasshouse conditions and in the field. Furthermore, pathogens grown on and in artificial media were controlled by CHA, norspermidine, N^1 - and N^8 -acetylspermidine and a number of novel spermidine analogues. Spermidine synthase activity in fungal tissue did not appear to be affected by exposure of the fungus to the compounds. An increase in putrescine was caused by exposure of *P. avenae* to CHA, the cause of which is unknown. It is unlikely that this increase accounted for the observed antifungal activity of CHA. Membrane integrity was drastically affected in fungus exposed to CHA or norspermidine, disrupting the methionine transport system. However, it is not clear whether this effect is primarily involved in the antifungal activity of the compounds or if it is a symptom of another mechanism. Spermidine is involved in another cellular process, hypusine synthesis. *P. avenae* growth has been shown likely to be inhibited by the prevention of hypusine synthesis. The biosynthesis of hypusine is clearly worthy of further investigation in fungal tissue treated with CHA or norspermidine.

Although fungicidal effects of the compounds may differ from antifungal effects, clearly, study of the antifungal mode(s) of action of CHA and the spermidine analogues may be useful in the search for new fungicides with novel modes of action. In spite of plentiful food production in some areas of the world, agricultural productivity must increase to meet rising populations. Crop diseases will continue to reduce potential productivity and

fungicides will remain essential as tools to control such disease. However, fungicide resistance and environmental and toxicological considerations mean that new active ingredients with novel modes of action are required. The antifungal and fungicidal activity exhibited by CHA and spermidine analogues may be useful in the continuing search for novel fungicidal agents.

ŝų.

ŝ

「日本の日本の」という

Bibliography

10. BIBLIOGRAPHY

ABBRUZZESE, A., PARK, M. H., BENINATI, S. & FOLK, J. E. (1989). Inhibition of deoxyhypusine hydroxylase by polyamines and by a deoxyhypusine peptide. Biochimica ct Biophysica Acta 997, 248-255.

ADAMS, D. O., BATES, D. J., ADAMS, D. F. & FRANKE, K. E. (1992). The effect of agmatine and other precursors on the accumulation of putrescine in grape leaves. American Journal of Enology and Viticulture 43, 239-243.

ADAMS, D. O., FRANKE, K. E. & CHRISTENSEN, L. P. (1990). Elevated putrescine levels in grapevine leaves that display symptoms of potassium deficiency. American Journal of Enology and Viticulture 41, 121-125.

AGRAWAL, S. B., AGRAWAL, M., LEE, E. H., KRAMER, G. F. & PILLAI, P. (1992). Changes in polyamine and glutathione contents of a green alga, <u>Chlorogonium</u> <u>elongatum</u> (Dang) France exposed to mercury. Environmental and Experimental Botany 32, 145-151.

AGRIOS, G. N. (1988). Plant Pathology, 3rd edition, Academic Press Ltd., London.

ALHONEN-HONGISTO, L. & JÄNNE, J. (1980). Polyamine depletion induces enhanced synthesis and accumulation of cadaverine in cultured Erlich ascites carcinoma cells. Biochemical and Biophysical Research Communications 93, 1005-1013.

ALHONEN-HONGISTO, L., SEPPÄNEN, P. & JÄNNE, J. (1980). Intracellular putrescine and spermidine deprivation induces increased uptake of the natural polyamines and methylglyoxal bis(guanylhydrazone). Biochemical Journal 192, 941-945.

ALTAMURA, M. M., TORRIGIANI, P., CAPITANI, F., SCARAMAGLI, S. & BAGNI, N. (1991). <u>De novo</u> root formation in tohacco thin layers is affected by inhibition of polyamine biosynthesis. Journal of Experimental Botany **42**, 1575-1582.

ALTMAN, A. & LEVIN, N. (1993). Interactions of polyamines and nitrogen nutrition in plants. Physiologia Plantarum 89, 653-658.

ANON. (1997). *Final view on new sprays.* Crops 15, No. 17 (20 September 1997), Reed Farmers Publishing Group, Surrey, 34-35.

APELBAUM, A., BURGOON, A. C., ANDERSON, J. D., LIEBERMAN, M., BENARIE, R. & MATTOO, A. K. (1981). Polyamines inhibit biosynthesis of ethylene in higher plant tissue and fruit protoplasts. Plant Physiology 68, 453-456.

ARTAMONOVA, E. Y., ZAVALOVA, L. L., KHOMUTOV, R. M. & KHOMUTOV, A. R. (1986). Irreversible inhibition of S-adenosylmethionine decarboxylase by hydroxylamine containing analogues of decarboxylated S-adenosylmethionine. Bioorganicheskaya Khimiya 12, 206-212.

AYRES, P. G. (1984). The interaction between environmental stress injury and biotic disease physiology. Annual Review of Phytopathology 22, 53-75.

AZIZ, S. M., GILLESPIE, M. N., CROOKS, P. A., TOFIQ, S. F., TSUBOI, C. P., OLSON, J. W. & GOSLAND, M. P. (1996). *The potential of a novel polyamine transport inhibitor in cancer chemotherapy*. Journal of Pharmacology and Experimental Therapeutics **278**, 185-192.

BACCHI, C. J., BRUN, R., CROFT, S. L., ALICEA, K. & BUHLER, Y. (1996). In vivo trypanocidal activities of new S-adenosylmethionine decarboxylase inhibitors. Antimicrobial Agents and Chemotherapy 40, 1448-1453.

BACCHI, C. J., GAROFALO, J., MOCKENHAUPT, D., McCANN, P. P., DIEKEMA, K. A., PEGG, A. E., NATHAN, H. C., MULLANEY, E. A., CHUNOSOFF, L., SJOERDSMA, A. & HUTNER, S. H. (1983). In vivo effects of DL- α difluoromethylornithine on the metabolism and morphology of <u>Trypanosoma brucei brucei</u>. Molecular and Biochemical Parisitology 7, 209-225.

BACCHI, C. J., NATHAN, H. C., CLARKSON, A. B., BIENEN, E. J., BITONTI, A. J., McCANN, P. P. & SJOERDSMA, A. (1987). Effects of the ornithine decarboxylase inhibitors DL-α-difluoromethylornithine and α-monofluoromethyldehydroornithine methyl ester alone and in combination with suramin against <u>Trypanosoma brucei brucei</u> central nervous system models. American Journal of Tropical Medicine 36, 46-52.

BACCHI, C. J., NATHAN, H. C., HUTNER, S. H., McCANN, P. P. & SJOERDSMA, A. (1980). Polyamine metabolism - A potential therapeutic target in trypanosomes. Science 210, 332-334.

BACCHI, C. J., VERGARA, C., GAROFALO, J., LIPSCHIK, G. Y. & HUTNER, S.
H. (1979). Synthesis and content of polyamines in bloodstream <u>Trypanosoma brucei</u>. Journal of Protozoology 26, 484-488.

BAGNI, N. & PISTOCCHI, R. (1985). *Putrescine uptake in <u>Saintpaulia</u> petals*. Plant Physiology 77, 398-402.

BARKER, J. H. A., SMITH, T. A. & OWEN, W. J. (1993). Inhibition of polyamine metabolism in <u>Phytophthora</u> species. Mycological Research 97, 1358-1362.

BARTIG, D., LEMKEMEIER, K., FRANK, J., LOTTSPEICH, F. & KLINK, F. (1992). The archaebacterial hypusine containing protein - Structural features suggest common ancestry with eukaryotic translation initiation factor 5A. European Journal of Biochemistry 204, 751-758.

BARTIG, D. & KLINK, F. (1992). Determination of the unusual amino acid hypusine at the lower picomole level by derivatization with 4-dimethylaminoazobenzen-4'-sulfonyl chloride and reversed-phase high-performance or medium-pressure liquid chromatography. Journal of Chromatography 606, 43-48.

BARTIG, D., SCHÜMANN, H. & KLINK, F. (1990). The unique post-translational modification leading to deoxyhypusine or hypusine is a general feature of the archaebacterial kingdom. Systematic and Applied Microbiology 13, 112-116.

BARTHOLEYNS, J., MAMONT, P. & CASARA, P. (1984). Antitumor properties of (2R,5R)-6-heptyne-2,5-diamine, a new potent enzyme activated irreversible inhibitor of ornithine decarboxylase, in rodents. Cancer Research 44, 4972-4977.

BATCHELOR, K. W., SMITH, R. A. & WATSON, N. S. (1986). Dicyclohexylamine is not an inhibitor of spermidine synthase. Biochemical Journal 233, 307-308.

BENINATI, S., ABBRUZZESE, A. & FOLK, J. E. (1990). High-performance liquid chromatographic method for determination of hypusine and deoxyhypusine. Analytical Biochemistry 184, 16-20.

BEPPU, T., SHIRAHATA, A. & SAMEJIMA, K. (1996). Determination of covalently bound hypusine and deoxyhypusine to protein using submilligram of protein samples by *HPLC*. Biological and Pharmaceutical Bulletin 19, 1-5.

BEY, P., DANZIN, C. & JUNG, M. (1987). Inhibition of basic amino acid decarboxylases involved in polyamine biosynthesis. In "Inhibition of Polyamine Metabolism: Biological Significance and Basis for New Therapies", (eds., McCann, P. P., Pegg, A. E. & Sjoerdsma, A.), Academic Press, Orlando, 1-31.

BEY, P., GERHART, F., VAN DORSSELAER, V. & DANZIN, C. (1983). α -(fluoromethyl)dehydroornithine and α -(fluoromethyl)dehydroputrescine analogs as irreversible inhibitors of ornithine decarboxylase. Journal of Medicinal Chemistry 26, 1551-1556.

BHARTI & RAJAM, M. V. (1996). Inhibition of polyamine biosynthesis and growth in plant pathogenic fungi in vitro. Mycopathologia 133, 95-103.

BHARTI & RAJAM, M. V. (1995). Effects of the polyamine biosynthesis inhibitor difluoromethylornithine on growth, polyamine levels, chromosome behaviour and polygenic traits in wheat (<u>Triticum aestivum L.</u>). Annals of Botany 76, 297-301.

BIASI, R., COSTA, G. & BAGNI, N. (1991). Polyamine metabolism as related to fruit set and growth. Plant Physiology and Biochemistry 29, 497-506.

BIONDI, S., BAGNI, N. & SANSOVINI, A. (1986). Dicyclohexylamine uptake and effects on polyamine content in cultured cotyledons of radiata pine. Physiologia Plantarum 66, 41-45.

BIONDI, S., POLGROSSI, I. & BAGNI, N. (1993). Effect of polyamine biosynthesis inhibitors on mycelial growth and concentrations of polyamines in <u>Ophiostoma ulmi</u> (Buism.) Nannf. New Phytologist **123**, 415-419.

BIRECKA, H., GARRAWAY, M. O., BAUMANN, R. J. & McCANN, P. P. (1986). Inhibition of ornithine decarboxylase and growth of the fungus <u>Helminthosporium maydis</u>. Plant Physiology 80, 798-800.

BITONTI, A. J., BACCHI, C. J., McCANN, P. P. & SJOERDSMA, A. (1986a). Uptake of α-difiuoromethylornithine by <u>Trypanosoma brucei brucei</u>. Biochemical Pharmacology 35, 351-354.

BITONTI, A. J., BACCHI, C. J., McCANN, P. P. & SJOERDSMA, A. (1985). Catalytic irreversible inhibition of <u>Trypanosoma brucei brucei</u> ornithine decarboxylase by substrate and product analogs and their effects on murine Trypanosomiasis. Biochemical Pharmacology 34, 1773-1777.

BITONTI, A. J., BYERS, T. L., BUSH, T. L., CASARA, P. J., BACCHI, C. J., CLARKSON, A. B., McCANN, P. P. & SJOERDSMA, A. (1990). Cure of <u>Trypanosoma</u> <u>brucei brucei and Trypanosoma brucei rhodensiense</u> infections in mice with an irreversible inhibitor of S-adenosylmethionine decarboxylase. Antimicrobial Agents and Chemotherapy 34, 1485-1490.

BITONTI, A. J., CASARA, P. J., McCANN, P. P. & BEY, P. (1987a). Catalytic irreversible inhibition of bacterial and plant arginine decarboxylase activities by novel substrate and product analogs. Biochemical Journal 242, 69-74.

BITONTI, A. J., DUMONT, J. A., BUSH, T. L., EDWARDS, M. L., STEMERICK, D. M., McCANN, P. P. & SJOERDSMA, A. (1989). Bis(benzyl)polyamine analogs inhibit the growth of chloroquine-resistant human malaria parasites (*Plasmodium falciparum*) in <u>vitro</u> and in combination with a-difluoromethylornithine cure murine malaria. Proceedings of the National Academy of Sciences of the United States of America 86, 651-655.

BITONTI, A. J., DUMONT, J. A. & McCANN, P. P. (1986b). Characterization of <u>Trypanosoma brucei brucei</u> S-adenosyl-L-methionine decarboxylase and its inhibition by Berenil, pentamidine and methylglyoxal bis(guanylhydrazone). Biochemical Journal 237, 685-689.

BITONTI, A. J., McCANN, P. P. & SJOERDSMA, A. (1987b). <u>Plasmodium falciparum</u> and <u>Plasmodium bergei</u> - Effects of ornithine decarboxylase inhibitors on erythrocytic schizogony. Experimental Parasitology 64, 237-243.

BITONTI, A. J., McCANN, P. P. & SJOERDSMA, A. (1982). Restriction of bacterial growth by inhibition of polyamine biosynthesis by using monofluoromethylornithine, difluoromethylarginine and dicyclohexylammonium sulphate. Biochemical Journal 208, 435-441.

BITONTI, A. J., KELLY, S. E. & McCANN, P. P. (1984). Characterization of spermidine synthase from <u>Trypanosoma bruce</u>i <u>bruce</u>i. Molecular and Biochemical Parasitology 13, 21-28.

BLÄTTLER, R., OCHSENBEIN, A. & BOSCHETTI, A. (1992). Polyamines and cytoplasmic preparations enhance light-driven protein synthesis in isolated chloroplasts of <u>Chlamydomonas reinhardtii</u>. Plant Physiology and Biochemistry **30**, 743-752.

なるとなるのなるの

The state of the s

BOLTON, N. J. E. & SMITH, J. M. (1988). Strategies to combat fungicide resistance in harley powdery mildew. Proceedings Brighton Crop Protection Conference - Pests and Diseases, 367-372.

BOTHA, M. L. & WHITEHEAD, C. S. (1992). The effect of polyamines on ethylene synthesis during normal and pollination-induced senesence of <u>Petunia hybrida L. flowers</u>. Planta **188**, 478-483.

BOTHMA, C. & DUBERY, I. A. (1991). The effect of polyamines on protein kinase activities of wheat (*Triticum aestivum L.*) anthers. Plant Growth Regulation 10, 363-375.

BOYLE, S. M., SRIRANGANATHAN, N. & CORDES, D. (1988). Susceptibility of <u>Microsporum</u> and <u>Trichophyton</u> species to suicide inhibitors of polyamine biosynthesis. Journal of Medical and Veterinary Mycology 26, 227-235.

BYERS, T. L., BUSH, T. L., McCANN, P. P. & BITONTI, A. J. (1991). Antitrypanosomal effects of polyamine biosynthesis inhibitors correlate with increases in <u>Trypanosoma brucei brucei</u> S-adenosyl-L-methionine. Biochemical Journal 274, 527-533.

BYERS, T. L., GANEM, B. & PEGG, A. E. (1992). Cytostasis induced in L1210 murine leukaemia cells by the S-adenosyl-L-methionine decarboxylase inhibitor 5'-{[(Z)-4-amino-2-butenyl[methylamino}-5'-deoxyadenosine may be due to hypusine depletion. Biochemical Journal 287, 717-724.

BYERS, T. L., KAMEJI, R., RANNELS, D. E. & PEGG, A. E. (1987). Multiple pathways for uptake of paraquat, methylglyoxal (bis)guanylhydrazone, and polyamines. American Journal of Physiology 252, C663-C669.

BYERS, T. L., LAKANEN, J. R., COWARD, J. K. & PEGG, A. E. (1994). The role of hypusine depletion in cytostasis induced by S-adenosyl-L-methionine decarboxylase inhibition - New evidence provided by 1-methylspermidine and 1,12-dimethylspermine. Biochemical Journal **303**, 363-368.

BYERS, T. L. & PEGG, A. E. (1989). Properties and physiological functions of the polyamine transport system. American Journal of Physiology 257, C545-C553.

CALVO-MENDEZ, C., MARTINEZ-PACHECO, M. & RUIZ-HERRERA, J. (1987). Regulation of ornithine decarboxylase in <u>Mucor bacilliformis</u> and <u>Mucor rouxi</u>. Experimental Mycology 11, 270-277.

CANELLAKIS, E. S., VICEPS-MADORE, D., KYRIAKIDIS, D. A. & HELLER, J. S. (1979). *Regulation and function of ornithine decarboxylase and of the polyamines*. Current Topics in Cell Regulation 15, 155-202.

CANO, C., HERRERA-ESTRELLA, L. & RUIZ-HERRERA, J. (1988). DNA methylation and polyamines in regulation of development of the fungus <u>Mucor rouxii</u>. Journal of Bacteriology 170, 5946-5948.

CANO-CANCHOLA, C., SOSO, L., FONZI, W., SYPHERD, P. & RUIZ-HERRERA, J. (1992). Developmental regulation of CUP gene expression through DNA methylation in <u>Mucor</u> spp. Journal of Bacteriology 174, 362-366.

CARLILE, W. R. (1995). *Control of Crop Diseases*, 2nd edition, Cambridge University Press, Cambridge.

CASERO, R. A., BERGERON, R. J. & PORTER, C. W. (1984). Treatment with αdifluoromethylornithine plus a spermidine analog leads to spermidine depletion and growth inhibition in cultured L1210 leukemia cells. Journal of Cellular Physiology 121, 476-482.

CATALDI, A. A. & ALGRANATI, I. D. (1986). A probable new pathway for the biosynthesis of putrescine in <u>Escherichia coli</u>. Biochemical Journal 234, 617-622.

CHEN, K. Y. (1983). An 18,000-dalton protein metabolically labeled by polyamines in various mammalian cell lines. Biochimica et Biophysica Acta 756, 395-402.

CHEN, K. Y. & DOU, Q. P. (1988). NAD⁺ stimulated the spermidine dependent hypusine formation on the 18,000 dalton protein in cytosolic lysates derived from NB-15 mouse neuro-blastoma cells. FEBS Letters 229, 325-328.

CHEN, K. Y. & YANG, Y. C. (1988). Spermidine labels a 21,000-dalton protein in Neurospora crasssa both in vivo and in vitro. FASEB Journal 2, A1563.

CHEN, S. L., CHEN, C. T. & KAO, C. H. (1991). Polyamines promote the biosynthesis of ethylene in detached rice leaves. Plant and Cell Physiology 32, 813-817.

CHEN, Z. P., YAN, Y. P., DING, Q. J., KNAPP, S., POTENZA, J. A., SCHUGAR, H. J. & CHEN, K. Y. (1996). Effects of inhibitors of deoxyhypusine synthase on the differentiation of mouse neuroblastoma and erythroleukemia cells. Cancer Letters 105, 233-239.

CHIEN, T. C., CHENG, C. L. & CHING, H. K. (1991). Senescence of rice leaves XXXI. Changes of chlorophyll, protein, and polyamine contents and ethylene production during senescence of a chlorophyll-deficient mutant. Journal of Plant Growth Regulation 10, 201-205.

CLARKSON, A. B., BACCHI, C. J., MELLOW, G. H., NATHAN, H. C. & McCANN, P. P. (1983). Efficacy of combinations of difluoromethylornithine and bleomycin on a mouse model of central nervous system African trypanosomiasis. Proceedings of the National Academy of Sciences of the United States of America - Biological Sciences 80, 5729-5733.

CLARKSON, A. B., BIENEN, E. J., BACCHI, C. J., McCANN, P. P., NATHAN, H. C., HUTNER, S. H. & SJOERDSMA, A. (1984). New drug combination for experimental late stage African trypanosomiasis - DL-α-difluoromethylornithine (DFMO) with suramin. American Journal of Tropical Medicine and Hygiene 33, 1073-1077.

COGHLAN, S. E. & WALTERS, D. R. (1992). Effects of α -difluoromethylornithine on photosynthetic oxygen evolution, respiration and polyamine concentrations in detached barley leaves. Pesticide Science 34, 75-79.

COGHLAN, S. E. & WALTERS, D. R. (1990). Polyamine metabolism in 'green-islands' on powdery mildew-infected barley leaves - Possible interactions with senescence. New Phytologist 116, 417-424.

COHEN, A. S., POPOVIC, R. B. & ZALIK, S. (1979). Effects of polyamines on chlorophyll and protein content, photochemical activity and chloroplast ultrastructure of barley leaf discs during senescence. Plant Physiology 64, 717-720.

COHEN, E., ARAD, S.-M., HEIMER, Y. M. & MIZRAHI, Y. (1982). Participation of ornithine decarboxylase in early stages of tomato fruit development. Plant Physiology 70, 540-543.

COHEN, S. S. (1979). Comparative biochemistry and drug design for infectious disease. Science **205**, 964-971.

COOPER, H. L., PARK, M. H. & FOLK, J. E. (1982). Posttranslational formation of hypusine in a single major protein occurs generally in growing cells and is associated with activation of lymphocyte growth. Cell 29, 791-797.

COOPER, H. L., PARK, M. H., FOLK, J. E., SAFER, B. & BRAVERMAN, R. (1983). Identification of the hypusine-containing protein Hy⁺ as translation initiation factor eIF-4D. Proceedings of the National Academy of Sciences of the United States of America -Biological Sciences 80, 1854-1857

COOPER, T. G. (1982). *Transport in <u>Saccharomyces cerevisiae</u>*. In "The Molecular Biology of the Yeast *Saccharomyces*: Metabolism and Gene Expression", (eds., Strathern, J. N., Jones, E. W. & Broach, J. R.), Cold Spring Harbor Laboratory, New York, 399-461.

COWARD, J. K. & PEGG, A. E. (1987). Specific multisubstrate adduct inhibitors of aminopropyltransferases and their effect on polyamine biosynthesis in cultured cells. Advances in Enzyme Regulation 26, 107-113.

CRISOSTO, C. H., LOMBARD, P. B., RICHARDSON, D. G. & TETLEY, R. (1992). Putrescine extends effective pollination period in 'Comice' pear (<u>Pyrus communis</u> L.) irrespective of post-anthesis ethylene levels. Scientia Horticulturae 49, 211-221.

CSONGA, R., ETTMAYER, P., AUER, M., ECKERSKORN, C., EDER, J. & KLIER, H. (1996). Evaluation of the metal ion requirement of the human deoxyhypusine hydroxylase from HeLa cells using a novel enzyme assay. FEBS Letters 380, 209-214.

DAMAYANTHI (1990). Effect of polyamine precursors and α-difluoromethylornithine on the course of <u>Trypanosoma evansi</u> infection in experimental albino mice. Current Science **59**, 935-937. DANIN, M., UPFOLD, S. J., LEVIN, N., NADEL, B. L., ALTMAN, A. & VAN STADEN, J. (1993). Polyamines and cytokinins in celery embryogenic cell cultures. Plant Growth Regulation 12, 245-254.

DATTA, N., HARDISON, L. K. & ROUX, S. J. (1986). Polyamine stimulation of protein phosphorylation in isolated pea nuclei. Plant Physiology 82, 681-684.

DATTA, N., SCHELL, M. B. & ROUX, S. J. (1987). Spermine stimulation of a nuclear NII kinase from pea plumules and its role in the phosphorylation of a nuclear polypeptide. Plant Physiology **84**, 1397-1401.

DAVE, C. & CABALLERI, L. (1973). Studies on the uptake of methylglyoxal bis(guanylhydrazone) (CH₃-G) and spermidine (SPd) in mouse leukemia L1210 sensitive and resistant to CH₃-G. Federation Proceedings - Federation of American Socieites for Experimental Biology 32, 763.

DAVIES, P. J. (1987). The plant hormones - their nature, occurrence and functions. In "Plant Hormones and their Role in Plant Growth and Development", (ed., Davies, P. J.), Martinus Nijhoff, Boston, 1-23.

DAVIS, R. II. (1990). Management of polyamine pools and the regulation of ornithine decarboxylase. Journal of Cellular Biochemistry 44, 199-205.

DAVIS, R. H., KRASNER, G. N., DIGANGI, J.J. & RISTOW, J. L. (1985). Distinct roles of putrescine and spermidine in the regulation of ornithine decarboxylase in <u>Neurospora crassa</u>. Proceedings of the National Academy of Sciences of the United States of America 82, 4105-4109.

DAVIS, R. H., MORRIS, D. R. & COFFINO, P. (1992). Sequestered end products and enzyme regulation - the case of ornithine decarboxylase. Microbiological Reviews 56, 280-290.

DAVIS, R. H. & RISTOW, J. L. (1991). *Polyamine toxicity in <u>Neurospora</u> <u>crassa</u> - the protective role of the vacuole. Archives of Biochemistry and Biophysics 285, 306-311.*

DAVIS, R. H. & RISTOW, J. L. (1989). Uptake, intracellular binding, and excretion of polyamines during growth of <u>Neurospora crassa</u>. Archives of Biochemistry and Biophysics **271**, 315-322.

224

DAVIS, R. H. & RISTOW, J. L. (1988). Polyamine transport in <u>Neurospora</u> crassa. Archives of Biochemistry and Biophysics 267, 479-489.

DEAK, T. (1978). On the existence of H^+ - symport in yeasts - A comparative study. Archives of Microbiology 116, 205-211.

DEWAARD, M. A., GEORGOPOULOS, S. G., HOLLOMON, D. W., ISHII, H., LEROUX, P., RAGSDALE, N. N. & SCHWINN, F. J. (1993). *Chemical control of plant diseases - problems and prospects*. Annual Review of Phytopathology 31, 403-421.

DE ROUGEMONT, G. M. (1989). A field guide to the crops of Britain and Europe, Collins, London.

DITOMASO, J. M., HART, J. J. & KOCHIAN, L. V. (1992). Transport kinetics and metabolism of exogenously applied putrescine in roots of intact maize seedlings. Plant Physiology 98, 611-620.

DITOMASO, J. M., SHAFF, J. E. & KOCHIAN, L. V. (1989). Putrescine-induced wounding and its effects on membrane integrity and ion-transport processes in roots of intact corn seedlings. Plant Physiology 90, 988-995.

DION, A. S. & COHEN, S. S. (1972). Polyamines in the synthesis of bacteriophage deoxyribonucleic acid. II Requirement for polyamines in T4 infection of a polyamine auxotroph. Journal of Virology 9, 423-430.

DOU, Q. P. & CHEN, K. Y. (1990). Characterization and reconstitution of a cell free system for NAD ⁺ dependent deoxyhypusine formation on the 18,000-dalton eIF-4D precursor. Biochimica et Biophysica Acta 1036, 128-137.

DOUGLAS, K. A., ZORMEIER, M. M., MARCOLINA, L. M. & WOSTER, P. M. (1991). Restricted rotation analogs of decarboxylated S-adenosylmethionine as inhibitors of polyamine biosynthesis. Bioorganic and Medicinal Chemistry Letters 1, 267-270.

DUMBROFF, E. (1991). *Mechanisms of polyamine action during plant development*. In "Lecture Course on Polyamines as Modulators of Plant Development", (eds., Galston, A. W. & Tiburcio, A. F.), **257**, Fundoción Juan, Madrid, 62-66.

EDDY, A. A. (1980). Some aspects of amino acid transport in yeast. In "Microorganisms and Nitrogen Sources", (ed., Payne, J. W.), Wiley, Chichester, 35-62.

EGEA-CORTINES, M., COHEN, E., ARAD, S.-M., BAGNI, N. & MIZRAHI, Y. (1993). Polyamine levels in pollinated and auxin-induced fruit of tomato (*Lycopersicon* esculentum) during development. Physiologia Plantarum 87, 14-20.

ELAD, Y. (1991). An inhibitor of polyamine biosynthesis - difluormethylornithine - and the polyamine spermidine for the control of gray mold (*Botrytis cinerea*). Phytoparasitica 19, 201-209.

EL GHACHTOULI, N., PAYNOT, M., MARTIN-TANGUY, J., MORANDI, D. & GIANINAZZI, S. (1996). Effect of polyamines and polyamine biosynthesis inhibitors on spore germination and hyphal growth of <u>Glomus mosseae</u>. Mycological Research 100, 597-600.

EL HADRAMI, I. & D'AUZAC, J. (1992). Effects of polyamine biosynthetic inhibitors on somatic embryogenesis and cellular polyamines in <u>Hevea brasiliensis</u>. Journal of Plant Physiology 140, 33-36.

「「「「「「「「「「」」」」」

ELO, H., MUTIKAINEN, I., ALHONEN-HONGISTO, L., LAINE, R., JÄNNE, J. & LUMME, P. (1986). Biochemical properties and crystal structure of ethylmethylglyoxal bis(guanylhydrazone) sulfate - an extremely powerful novel inhibitor of adenosylmethionine decarboxylase. Zeitschrift für Naturforschung - Section C - Biosciences 41, 851-855.

ERWIN, B. G. & PEGG, A. E. (1982). Uptake of α-difluoromethylornithine by mouse fibroblasts. Biochemical Pharmacology **31**, 2820-2823.

EVANS, P. T. & MALMBERG, R. L. (1989). Do polyamines have roles in plant development?. Annual Review of Plant Physiology and Plant Molecular Biology 40, 235-269.

FAIRLAMB, A. H., BLACKBURN, P., ULRICH, P., CHAIT, B. T. & CERAMI, A. (1985). Trypanothione - A novel bis(glutathionyl)spermidine cofactor for glutathione reductase in trypanosomatids. Science 227, 1485-1487.

FEIRER, R. P., WANN, S. R. & EINSPAHR, D. W. (1985). The effects of spermidine synthesis inhibitors on in vitro plant development. Plant Growth Regulation 3, 319-327.

FELIX, H. & HARR, J. (1989). Influence of inhibitors of polyamine biosynthesis on polyamine levels and growth of plants. Zeitschrift für Naturforschung C - A Journal of Biosciences 44, 55-58.

FEUERSTEIN, B. G., DEEN, D. F. & MARTON, L. J. (1985). Effects of dicyclohexylamine sulfate, a spermidine synthase inhibitor, in 9L rat brain tumor cells. Cancer Research 45, 4950-4954.

FLORES, H. E., PROTACIO, C. M. & SIGNS, M. W. (1989). Primary and secondary metabolism of polyamines in plants. In "Plant Nitrogen Metabolism", (eds., Poulton, J. E., Romeo, J. T. & Conn, E. E.), Plenum Press, New York, 329-393.

FONG, W. F., HELLER, J. S. & CANELLAKIS, E. S. (1976). The appearance of an ornithine decarboxylase inhibitory protein upon the addition of putrescine to cell cultures. Biochimica et Biophysica Acta 428, 456-465.

FOSTER, S. A. & WALTERS, D. R. (1993). Fungicidal activity of the putrescine analogue, keto-putrescine. Pesticide Science 37, 267-272.

FOSTER, S. A. & WALTERS, D. R. (1992a). Effects of two polyamine biosynthesis inhibitors on infection of wheat by <u>Puccinia graminis tritici</u>. Tests of Agrochemicals and Cultivars 13, 32-33. (Annals of Applied Biology 120 supplement).

FOSTER, S. A. & WALTERS, D. R. (1992b). Polyamine concentrations and activities of ornithine and arginine decarboxylase in wheat infected with the stem rust fungus. Journal of Plant Physiology 140, 134-136.

FOSTER, S. A. & WALTERS, D. R. (1991). Polyamine concentrations and arginine decarboxylase activity in wheat exposed to osmotic stress. Physiologia Plantarum 82, 185-190.

FOSTER, S. A. & WALTERS, D. R. (1990). The effects of polyamine biosynthesis inhibitors on mycelial growth, enzyme activity and polyamine levels in the oat-infecting fungus <u>Pyrenophora avenae</u>. Journal of General Microbiology **136**, 233-239.

FRIEDMAN, R., ALTMAN, A. & LEVIN, N. (1989). The effect of salt stress on polyamine biosynthesis and content of mung bean plants and in halophytes. Physiologia Plantarum 76, 295-302.

GALLARDO, M., BUENO, M., ANGOSTO, T., GALLARDO, E. & MATILLA, A. J. (1992). Free polyamines in <u>Cicer arietinum</u> seeds during the onset of germination. Phytochemistry 31, 2283-2287.

GALSTON, A. W. (1983). Polyamines as modulators of plant development. BioScience 33, 382-388.

GALSTON, A. W. & KAUR-SAWHNEY, R. (1987). Polyamines as endogenous growth substances. In "Plant Hormones and their Role in Plant Growth and Development", (ed., Davies, P. J.), Martinus Nijhoff, Boston, 280-295.

GARCÍA, J. I., NICOLÁS, G. & VALLE, T. (1991). Effect of difluoromethylornithine on growth, cell size and germination of <u>Ceratocystis ulmi</u> spores. Plant Science 77, 131-136.

GERNER, E. W., MAMONT, P. S., BERNHARDT, A. & SIAT, M. (1986). Posttranslational modification of the protein-synthesis initiation factor eIF-4D by spermidine in rat hepatoma cells. Biochemical Journal 239, 379-386.

GIFFIN, B. F., McCANN, P. P., BITONTI, A. J. & BACCIII, C. J. (1986). Polyamine depletion following exposure to DL- α -difluoromethylornithine both <u>in vivo</u> and <u>in vitro</u> initiates morphological alterations and mitochondrial activation in a monomorphic strain of <u>Trypanosoma brucei brucei</u>. Journal of Protozoology **33**, 238-243.

GRIFFEN, D. H. (1994) Fungal Physiology, 2nd edition, Wiley-Liss, Inc., New York.

GOLDEN, J. A., SJOERDSMA, A. & SANTI, D. V. (1984). <u>Pneumocystis carnii</u> pneumonia treated with a-difluoromethylornithine - A prospective study among patients with the Acquired Immunodeficiency Syndrome. Western Journal of Medicine 141, 613-623.

GORDON, E. D., MORA, R., MEREDITH, S. C., LEE, C. & LINDQUIST, S. L. (1987). Eukaryotic initiation factor 4D, the hypusine containing protein is conserved among eukaryotes. Journal of Biological Chemistry 262, 16585-16589.

GREENBERG, M. L. & COHEN, S. S. (1985). Dicyclohexylamine-induced shift of biosynthesis from spermidine to spermine in plant protoplasts. Plant Physiology **78**, 568-575.

228
GROVE, J., FOZARD, J. R. & MAMONT, P. S. (1981). Assay of α difluoromethylornithine in body fluids and tissues by automatic amino acid analysis. Journal of Chromatography 223, 409-416.

GRUHN, C. M. & BOYLE, S. M. (1991). Biochemical and morphological effects of polyamine biosynthesis inhibitors on <u>Trichophyton</u> and <u>Microsporum</u>. Journal of Medical and Veterinary Mycology **29**, 63-72.

GUARINO, L. A. & COHEN, S. S. (1979). Mechanism of toxicity of putrescine in <u>Anacystic nidulans</u>. Proceedings of the National Academy of Sciences of the United States of America 76, 3660-3664.

HAMAGUCIII, K., IWAKIRI, T., IMAMURA, K., FURIHATA, K., SETO, H. & OTAKE, N. (1987). Juglorin, a new spermidine synthase inhibitor. Journal of Antibiotics 40, 717-719.

HANAUSKE-ABEL, H. M., PARK, M. H., HANAUSKE, A. R., POPOWICZ, A. M., LALANDE, M. & FOLK, J. E. (1994). Inhibition of the G1-S transition of the cell cycle by inhibitors of deoxyhypusine hydroxylation. Biochimica et Biophysica Acta - Molecular Cell Research 1221, 115-124.

HARKESS, R. L., LYONS, R. E. & KUSHAD, M. M. (1992). Floral morphogenesis in <u>Rudbeckia hirta</u> in relation to polyamine concentration. Physiologia Plantarum 86, 575-582.

HARTMANN, T., SANDER, H., ADOLPH, R. & TOPPEL, G. (1988). Metabolic links between the biosynthesis of pyrrolizidine alkaloids and polyamines in root cultures of <u>Senecio vulgaris</u>. Planta 175, 82-90.

HASSALL, K. A. (1990). The Biochemistry and Uses of Pesticides. 2nd edition, Macmillan Press Ltd., Hampshire.

HAVIS, N. D. & WALTERS, D. R. (1992a). Evaluation of difluoromethylornithine against powdery mildew on spring barley. Tests of Agrochemicals and Cultivars 13, 16-17. (Annals of Applied Biology 120 supplement).

HAVIS, N. D. & WALTERS, D. R. (1992b). In vitro effects of the polyamine biosynthesis inhibitor, α -difluoromethylornithine, on ornithine decarboxylase activities from three plant pathogenic fungi. Letters in Applied Microbiology 14, 244-246.

HAVIS, N. D., WALTERS, D. R., COOK, F. M. & ROBINS, D. J. (1997). Synthesis, fungicidal activity, and effects on fungal polyamine metabolism of novel cyclic diamines. Journal of Agricultural and Food Chemistry 45, 2341-2344.

HAVIS, N. D., WALTERS, D. R., FOSTER, S. A., MARTIN, W. P., COOK, F. M. & ROBINS, D. J. (1994a). Fungicidal activity of the synthetic putrescine analogue, (E)-1,4diaminobut-2-ene, and derivatives. Pesticide Science 41, 61-69.

HAVIS, N. D., WALTERS, D. R., MARTIN, W. P., COOK, F. M. & ROBINS, D. J. (1996a). *Synthesis and fungicidal activity of alicyclic diamines*. Journal of Agricultural and Food Chemistry 44, 2835-2838.

HAVIS, N. D., WALTERS, D. R., MARTIN, W. P., COOK, F. M. & ROBINS, D. J. (1994b). Evaluation of the putrescine analogue (E)-1,4-diaminobut-2-ene against powdery mildew on spring barley. Tests of Agrochemicals and Cultivars 15, 14-15. (Annals of Applied Biology 124 supplement).

HAVIS, N. D., WALTERS, D. R., MARTIN, W. P., COOK, F. M. & ROBINS, D. J. (1994c). Fungicidal activity of three putrescine analogues. Pesticide Science 41, 71-76,

HAVIS, N. D., WALTERS, D. R. & ROBINS, D. J. (1996b). The fungicidal diamine (E)-N,N,N',N'-tetraethyl-1,4-diaminobut-2-ene increases spermine concentration in the late blight fungus <u>Phytophthora infestans</u>. Letters in Applied Microbiology **25**, 229-232.

HAWTHORNE, T. R. & AUSTIN, J. K. (1996). Synergism of the polyamine analog N^{I} , N^{II} -bisethylnorspermine with cis-diaminedichloroplatinum (II) against murine neoplastic cell lines <u>in vitro</u> and <u>in vivo</u>. Cancer Letters 99, 99-107.

HAYWOOD, G. W. & LARGE, P. J. (1986). 4-acetamidobutyrate deacetylase in the yeast <u>Candida boindinii</u> grown on putrescine or spermidine as sole nitrogen source and its prohable role in polyamine catabolism. Journal of General Microbiology 132, 7-14.

HEREDIA, C. F., SOLS, A. & DE LA FUENTE, G. (1968). Specificity of the constitutive hexose transport in yeast. European Journal of Biochemistry 5, 321-329.

HERSHEY, J. W. B. (1991). Translational control in mammalian cells. Annual Review of Biochemistry 60, 717-755.

HIBASAMI, H., TANAKA, M., NAGAI, J. & IKEDA, T. (1980). Dicyclohexylamine, a potent inhibitor of spermidine synthase in mammalian cells. FEBS Letters 116, 99-101.

HOLM, I., PERSSON, L., STJERNBORG, L., THORSSON, L. & HEBY. O. (1989). Feedback control of ornithine decarboxylase activity expression by polyamines. Biochemical Journal 258, 343-350.

HÖLTTÄ, E., POHJANPELTO, P. & JÄNNE, J. (1979). The association of the early antiproliferative action of methylglyoxal bis(guanylhydrazone) from polyamine depletion. FEBS Letters 97, 9-14.

HUNTER, D. R. & SEGAL, I. H. (1971). Acidic and basic amino acid transport systems of <u>Penicillium chrysogenum</u>. Archives of Biochemistry and Biophysics 144, 168-183.

IGARASHI, K., SAKAMOTO, I., GOTO N., KASHIWAGI, K., HONMA, R. & HIROSE, S. (1982). Interaction between polyamines and nucleic acids or phospholipids. Archives of Biochemistry and Biophysics 219, 438-443.

IMAOKA, N. & NAKAJIMA, T. (1973). N^C-(4-amino-2-hydroxybutyl)-2-6diaminohexanoic acid, in tissue proteins of mammals. Biochimica et Biophysica Acta 320, 97-103.

INDERLIED, C. B., CIHLAR, R. L. & SYPHERD, P. S. (1980). Regulation of ornithine decarboxylase during morphogenesis of <u>Mucor racemosus</u>. Journal of Bacteriology 141, 699-706.

INOUE, H., KATO, Y., TAKIGAWA, M. ADACHI, K. & TAKEDA, Y. (1975). Effect of DL-α-hydrazino-δ-aminovaleric acid, an inhibitor of ornithine decarboxylase, on polyamine metabolism in isoproterenol-stimulated mouse parotid glands. Journal of Biochemistry **77** 879-893.

ISAAC, S. & JENNINGS, D. H. (1995). *Microbial Culture*. Bios Scientific Publishing, Oxford.

ISRAEL, M., ROSENFELD, J. S. & MODEST, E. J. (1964). Analogs of spermidine and spermine. I Synthesis of polymethylene polyamines by reduction of cyanoalkylated alkylenediamines. Journal of Medicinal Chemistry 7, 710-716.

ITO, H., HIBASAMI, H., KESHIRO, S., NAGAI, J. & HIDAKA, H. (1982). Antitumour effect of dicyclohexylamine sulfate, a potent inhibitor of spermidine synthase against P388 leukemia. Cancor Letters 15, 229-235.

JAKUS, J., WOLFF, E. C., PARK, M. H. & FOLK, J. E. (1993). Features of the spermidine-binding site of deoxyhypusine synthase as derived from inhibition studies. Journal of Biological Chemistry 268, 13151-13159.

JÄNNE, J., ALHONEN-HONGISTO, L., NIKULA, P. & ELO, H. (1985). Sadenosylmethionine decarboxylase as a target of chemotherapy. Advances in Enzyme Regulation 24, 125-139.

JENKINS, R. O. (1992). Uptake of nutrients. In "Biosynthesis and the Integration of Cell Metabolism", (ed., Weston, G. D.), Butterworth-Heinemann Ltd., Oxford, 11-40.

JENKYN, J. F. & BAINBRIDGE, A. (1979). *Cereals*. In "The Powdery Mildews", (ed., Spencer, D. M.), Academic Press, London, 284-341.

JOHNSON, G. (1995). *Learning the lessons of a low disease year*. Crops, **13**, No. 20 (11 November 1995), Reed Farmers Publishing Group, Surrey, 10-11.

JONES, D. G. & CLIFFORD, B. C. (1978). Cereal diseases, their pathology and control, BASF United Kingdom Ltd., Suffolk.

JONES, O. T. G. (1963). The accumulation of amino acids by fungi, with particular reference to the plant parasitic fungus <u>Botrytis fabae</u>. Journal of Experimental Botany 14, 339-411.

JORSTAD, C. M., HARADA, J. J. & MORRIS, D. R. (1980). Structural specificity of the spermidine requirment of an <u>Escherichia coli</u> auxotroph. Journal Bacteriology 141, 456-463.

JUPE, E. R., MAGILL, J. M. & MAGILL, C. W. (1986). Stage-specific DNA methylation in a fungal plant pathogen. Journal of Bacteriology 165, 420-423.

KALLIO, A., McCANN, P. P. & BEY, P. (1982). *DL-α-monofluoromethylputrescine is a* potent irreversible inhibitor of <u>Escherichia</u> coli ornithine decarboxylase. Biochemical Journal 204, 771-775.

KALLIO, A., McCANN, P. P. & BEY, P. (1981). DL-α-(difluoromethyl)arginine - A potent enzyme activated irreversible inhibitor of bacterial arginine decarboxylases. Biochemistry 20, 3163-3166.

KALLIO, A. & McCANN, P. P. (1981). Difluoromethylornithine irreversibly inactivates ornithine decarboxylase of <u>Pseudomonas aeruginosa</u>, but does not inhibit the enzymes of <u>Escherichia coli</u>. Biochemical Journal 200, 69-75.

KAMEJI, T., MURAKAMI, Y. & HAYASIII, S. (1979). Effect of diaminobutene and diaminopropane on diet-stimulated polyamine synthesis and cell proliferation in rat liver. Journal of Biochemistry 86, 191-197.

KAUR, K., EMMETT, K., McCANN, P. P., SJOERDSMA, A. & ULLMAN, B. (1986). Effects of DL-α-difluoromethylornithine on <u>Leishmania donovani</u> promastigotes. Journal of Protozoology 33, 518-521.

KAUR-SAWHNEY, R., FLORES, H. E. & GALSTON, A. W. (1980). Polyamineinduced DNA synthesis and mitosis in oat leaf protoplasts. Plant Physiology 65, 368-371.

KEMPER, W. M., BERRY, K. W. & MERRICK, W. C. (1976). Purification of initiation factors M28-α and M28-β. Journal of Biological Chemistry 251, 5551-5557.

KHAN, A. J. & MINOCHA, S. C. (1991). Polyamines and somatic embryogenesis in carrot. II. The effects of cyclohexylammonium phosphate. Journal of Plant Physiology 137, 446-452.

KHAN, A. J. & MINOCHA, S. C. (1989a). Polyamine biosynthetic enzymes and the effect of their inhibition on the growth of some phytopathogenic fungi. Plant and Cell Physiology **30**, 655-663.

KHAN, A. J. & MINOCHA, S. C. (1989b). Biosynthetic arginine decarboxylase in phytopathogenic fungi. Life Sciences 44, 1215-1222.

KHAN, N. A., QUEMENER, V., SEILER, N. & MOULINOUX, J.-P. (1990). Mechanism of spermidine uptake in cultured mammalian cells and its inhibition by some polyamine analogues. Pathobiology 58, 172-178.

KHOMUTOV, R. M., HYVÖNEN, T., KARVONEN, E., KAUPPINEN, L., PAALANEN, T., PAULIN, L., ELORANTA, T., PAJULA, R. L., ANDERSSON, L. C. & PÖSÖ, H. (1985). 1-aminooxy-3-aminopropane, a new and potent inhibitor of polyamine biosynthesis that inhibits ornithine decarboxylase, adenosylmethionine decarboxylase and spermidine synthase. Biochemical and Biophysical Research Communications 130, 596-602.

KIERSZENBAUM, F., WIRTH, J. J., McCANN, P. P. & SJOERDSMA, A. (1987). Impairment of macrophage function by inhibitors of ornithine decarboxylase activity. Infection and Immunity 55, 2461-2464.

KLEIN, H., PRIEBE, A. & JAGER, H. J. (1979). Putrescine and spermidine in peas -Effect of nitrogen source and potassium supply. Physiologia Plantarum 45, 497-499.

KRAMER, D. L., KHOMUTOV, R. M., BUKIN, Y. V., KHOMUTOV, A. R. & PORTER, C. W. (1989). Cellular characterization of a new irreversible inhibitor of Sadenosylmethionine decarboxylase and its use in determining the relative abilities of individual polyamines to sustain growth and viability of L1210 cells. Biochemical Journal 259, 325-331.

KRAMER, G. F., WANG, C. Y. & CONWAY, W. S. (1991). Inhibition of softening by polyamine application in 'Golden Delicious' and 'McIntosh' apples. Journal of the American Society for Horticultural Science 116, 813-817.

LARGE, P. J. (1992). Enzymes and pathways of polyamine breakdown in microorganisms. FEMS Microbiology Reviews 88, 249-262.

LAST, F. T. & HAMLEY, R. E. (1956). A local-lesion technique for measuring the infectivity of conidia of <u>Botrytis fabae</u>. Annals of Applied Biology 44, 410-418.

LAW, D. M., DAVIES, P. J. & MUTSCHLER, M. A. (1991). Polyamine-induced prolongation of storage in tomato fruits. Plant Growth Regulation 10, 283-290.

LEE, T. M. & CHU, C. (1992). Ethylene-induced polyamine accumulation in rice (<u>Oryza</u> sativa L.) coleoptiles. Plant Physiology 100, 238-245.

LEE, Y. B., PARK, M. H. & FOLK, J. E. (1995). Diamine and triamine analogs and derivatives as inhibitors of deoxyhypusine synthase - Synthesis and biological activity. Journal of Medicinal Chemistry 38, 3053-3061.

LESHEM, B., KAUR-SAWHNEY, R. & GALSTON, A. W. (1991). Changes in polyamine levels during organ induction in cultured melon cotyledons. Journal of Plant Physiology **138**, 757-759.

LIBBY, P. R. (1978). Calf liver nuclear N-acetyltransferases. Journal of Biological Chemistry 253, 233-237.

LIN, P.P-C. & KEY, J. L. (1980). Histone kinase from soybean hypocotyls. Plant Physiology 66, 360-367.

LOWRY, O. H., ROSEBROUGH, N. J., FARR, A. L. & RANDALL, R. J. (1951). *Protein measurement with the folin phenol reagent.* Journal of Biological Chemistry 193, 265-272.

LUK, G. D., ABELOFF, M. D., GRIFFIN, C. A. & BAYLIN, S. B. (1983). Successful treatment with DL-α-difluoromethylornithine in established human small cell variant lung carcinoma implants in athymic mice. Cancer Research 43, 4239-4243.

MACHATSCKE, S., KAMROWSKI, C., MOERSCHBACHER, B. M. & REISENER, H-J. (1990). Polyamine levels in stem rust infected wheat leaves and effects of αdifluoromethylornithine on fungal infection. Physiological and Molecular Plant Pathology 36, 451-459.

MAGILL, J. M. & MAGILL, C. W. (1989). DNA methylation in fungi. Developmental Genetics 10, 63-69.

MAMONT, P. S., DANZIN, C., KOLB, M., GERHART, F., BEY, P. & SJOERDSMA, A. (1986). Marked and prolonged inhibition of mammalian ornithine decarboxylase in vitro by esters of (E)-2-(fluoromethyl)dehydroornithine. Biochemical Pharmacology 35, 159-165. MAMONT, P. S., DANZIN, C., WAGNER, J., SLAT, M., JODER-OHENBUSCH, A.-M. & CLAVERIE, N. (1982). Accumulation of decarboxylated S-adenosylmethionine in mammalian cells as a consequence of the inhibition of putrescine synthesis. European Journal of Biochemistry 123, 499-504.

MAMONT, P. S., DUCHESNE, M. C., GROVE, J. & BEY, P. (1978). Anti-proliferative properties of DL- α -difluoromethylornithine in cultured cells. A consequence of the reversible inhibition of ornithine decarboxylase. Biochemical and Biophysical Research Communications 81, 58-66.

MAMONT, P. S., SIAT, M., JODER-OHLENBUSCH, A-M., BERHARDT, A. & CASARA, P. (1984). Effects of (2R.5R)-6-heptyne-2,5-diamine, a potent inhibitor of Lornithine decarboxylase, on rat hepatoma cells cultured in vitro. European Journal of Biochemistry 142, 457-463.

MANDELS, G. R. (1965). *Kinetics of Fungal Growth*. In "The Fungi - Volume 1 - The Fungal Cell", (eds., Ainsworth, G. C. & Sussman, A. S.), Academic Press, London, 599-612.

MARTON, L. J., LEVIN, V. A., HERVATIN, S. J., KOCH-WESER, J., McCANN, P. P. & SJOERDSMA, A. (1981). Potentiation of the anti-tumor therapeutic effects of 1,3bis(2-chloroethyl)-1-nitrosourea by α -difluoromethylornithine, an ornithine decarboxylase inhibitor. Cancer Research 41, 4426-4431.

MARTON, L. J. & MORRIS, D. R. (1987). Molecular and cellular functions of the polyamines. In "Inhibition of Polyamine Biosynthesis: Biological Significance and Basis for New Therapics", (cds., McCann, P. P., Pegg, A. E. & Sjoerdsma, A.), Academic Press, New York, 79-105.

MASSE, J., LABERCHE, J. C. & JEANTY, G. (1988). The effects of nor-spermidine and dicyclohexylamine on in <u>vitro</u> potato development and free polyamines content. Plant Growth Regulation 7, 249-260.

MASSE, J., LABERCHE, J. C. & JEANTY, G. (1985). Effect of exogenous polyamines on growth of corn seedlings (Zea mays L.). Comptes Rendus de l'Academie des Sciences Serie III - Sciences de la Vie. 301, 27-32.

MATEJKO, C. & DAHLHELM, H. (1991). Polyamine synthesis and its relation to dormancy in <u>Allium cepa L</u>. Biochemie und Physiologic der Pflanzen 187, 217-226.

MATTILA, T., HONKANEN-BUZALSKI, T. & PÖSÖ, H. (1984). Reversible inhibition of bacterial growth after specific inhibition of spermidine synthase by dicyclohexylamine. Biochemical Journal 223, 823-830.

McCANN, P. P., BACCHI, C. J., CLARKSON, A. B., BEY, P., SJOERDSMA, A., SCHECTER, P. J., WALTZER, P. D. & BARLOW, J. L. R. (1986). Inhibition of polyamine biosynthesis by α -difluoromethylornithine in African trypanosomes and <u>Pneumocystis carnii</u> as a basis of chemotherapy: biochemical and clinical aspects. American Journal of Tropical Medicine and Hygiene 35, 1153-1156.

McCANN, P. P., BACCHI, C. J., HANSON, W. L., CAIN, G. D., NATHAN, H. C., HUTNER S. H. & SJOERDSMA, A. (1981). Effect on parasitic protozoa of adifluoromethylornithine - an inhibitor of ornithine decarboxylase. In "Advances in Polyamine Research - Volume 3", (eds., Caldarea, C. M., Zappia, V. & Bachrach, U.), Raven Press, New York, 97-110.

McCLOSKEY, D. E., YANG, J. N., WOSTER, P. M., DAVIDSON, N. & CASERO, R. A. (1996). Polyamine analog induction of programmed cell death in human lung tumor cells. Clinical Cancer Research 2, 441-446.

McDERMOTT, J. C. D. & JENNINGS, D. H. (1976). The relationship between the uptake of glucose and 3-0-methyl glucose and soluble carbohydrate and polysaccharide in the fungus <u>Dendryphiella salina</u>. Journal of General Microbiology **97**, 193-209.

MENGOLI, M., BAGNI, N. & BIONDI, S. (1989). Effect of α-difluoromethylornithine on carrot cell cultures. Journal of Plant Physiology **129**, 479-485.

MENNUCCI, L., ROJAS, S. & CAMARGO, F. P. (1975). Polyamines and ornithine decarboxylase activity during growth and differentiation in <u>Blastocladiella emersonii</u>. Biochimica et Biophysica Acta **404**, 249-256.

MERALI, S. & CLARKSON, A. B. (1996). Polyamine content of <u>Pneumocystis carinii</u> and response to ornithine decarboxylase inhibitor DL-α-difluoromethylornithine. Antimicrobial Agents and Chemotherapy **40**, 973-978. METCALF, B. W., BEY, P., DANZIN, C., JUNG, M. J., CASARA, J. & VEVERT, J. P. (1978). Catalytic irreversible inhibition of mammalian ornithine decarboxylase (EC 4. 1. 1. 17) by substrate and product analogues. Journal of the American Chemical Society 100, 2551-2553.

MEYSKENS, F. L., KINGSLEY, E. M., GLATTKE, T., LOESCHER, L. & BOOTH, A. (1986). A phase II study of α -difluoromethylornithine (DFMO) for the treatment of metastatic melanoma. Investigational New Drugs 4, 257-262.

MINOCHA, S. C. & KHAN, A. J. (1991). Effects of combinations of polyamine biosynthetic inhibitors on cellular polyamines in carrot cell cultures. Journal of Plant Physiology 137, 507-510.

MINOCHA, R., MINOCHA, S. C., KOMAMINE, A. & SHORTLE, W. C. (1991). Regulation of DNA synthesis and cell division by polyamines in <u>Catharanthus roseus</u> suspension cultures. Plant Cell Reports 10, 126-130.

MITCHELL, J. L. A., MAHAN, D. W., McCANN, P. P. & QASBA, P. (1985). Dicyclohexylamine effects on HTC polyamine content and ornithine decarboxylase activity. Biochimica et Biophysica Acta 840, 309-316.

MUKIIOPADIIYAY, R., KAPOOR, P. & MADHUBALA, R. (1996). Antileishmanial effect of a potent S-adenosylmethionine decarboxylase inhibitor – CGP 40215A. Pharmacological Research 33, 67-70.

MURAKAMI, Y., MATSUFUJI, S., MIYAZAKI, Y. & HAYASHI, S. (1992a). Destabilization of ornithine decarboxylase by transfected antizyme gene expression in hepatoma tissue culture cells. Journal of Biological Chemistry 267, 13138-13141.

MURAKAMI, Y., TANAKA, K., MATSUFUJI, S., MIYAZAKI, Y. & HAYASHI, S. (1992b). Antizyme, a protein induced by polyamines, accelerates the degradation of ornithine decarboxylase in Chinese-hamster ovary-cell extracts. Biochemical Journal 283, 661-664.

MURPHEY, R. J. & GERNER, E. W. (1987). Hypusine formation in protein by a twostep process in cell lysates. Journal of Biological Chemistry 262, 15033-15036. MUSSELL, H., OSMELOSKI, J., WETTLAUFER, S. H. & WEINSTEIN, L. (1987). Suppression of Verticillium wilt of tomato by difluoromethylornithine, a suicidal inhibitor of polyamine biosynthesis. Plant Disease 71, 313-316.

NAIK, B. I., & SRIVASTAVA, S. K. (1978). Effect of polyamines on tissue permeability. Biochemistry 17, 1885-1887.

NAIR, K. R. S. & ELLINBOE, A. H. (1962). A method of controlled inoculations with conidiospores of <u>Ervsiphe graminis</u> var. <u>tritici</u>. Phytopathology 52, 714.

NAKAJIMA, T., MATSUBAYASHI, T., KAKIMOTO, Y. & SANO, I. (1971). Distribution of hypusine, N⁶(4-amino-2-hydroxybutyl) 2,6-diaminohexanoic acid, in mammalian organs. Biochimica et Biophysica Acta 252, 92-97.

NATHAN, H. C., BACCHI, C. J., HUTNER, S. H., RESCIGNO, D., McCANN, P. P. & SJOERDSMA, A. (1981). Antagonism by polyamines of the curative effects of αdifluoromethylornithine on <u>Trypanosoma brucei brucei</u> infections. Biochemical Pharmacology 30, 3010-3013.

NIKULA, P., ALHONEN-HONGISTO, L., SEPPANEN, P. & JÄNNE, J. (1984). Inhibition of long-chain fatty acid oxidation by methylglyoxal bis(guanylhydrazone). Biochemical and Biophysical Research Communications 120, 9-14.

NOVAK, S., D'AMORE, T. & STEWART, G. G. (1990). 2-Deoxy-D-glucose resistant yeast with altered sugar transport activity. FEBS Letters 269, 202-204.

OKADA, H. & HALVORSON, H. O. (1964). Uptake of o-thioethyl-D-glucopyranoside by Saccharomyces cerevisiae. II General characteristics of an active transport system. Biochimica et Biophysica Acta 82, 547-555.

OKUYAMA, T. & KUBAYASHI, Y. (1961). Determination of diamine oxidase activity by liquid scintillation counting. Archives of Biochemistry and Biophysics **95**, 242-251.

OREDSSON, S., ANEHUS, S. & HEBY, O. (1980). Irreversible inhibition of the early increase in ornithine decarboxylase activity following growth stimulation is required to block Ehrlich ascites tumour cell proliferation in culture. Biochemical and Biophysical Research Communications 94, 151-158.

OREDSSON, S. M., DEEN, D. F. & MARTON, L. J. (1982). Decreased cyto-toxicity of cis-diaminedichloroplatinum (II) by α-difluoromethylornithine depletion of polyamines in 9L rat brain tumor cells <u>in vitro</u>. Cancer Research 42, 1296-1299.

OUTSIDERS GUIDE, (1995). The Outsiders Guide to Crop Production, Outsiders Guide, London.

PALL, M. L. & KELLY, K. A. (1971). Specificity of transinhibition of amino acid transport in <u>Neurospora crassa</u>. Biochemical and Biophysical Research Communications 42, 940-947.

PALMER, T. (1991). Understanding enzymes. 3rd edition., Ellis Horwood Ltd., Chichester.

PARK, K. Y. & LEE, S. H. (1994). Effects of ethylene and auxin on polyamine levels in suspension-cultured tobacco cells. Physiologia Plantarum 90, 382-390.

PARK, M. H. (1989). The essential role of hypusine in eukaryotic translation initiation factor 4D (eIF-4D) - Purification of eIF-4D and its precursors and comparison of their activities. Journal of Biological Chemistry 264, 18531-18535.

PARK, M. H. (1987). Regulation of biosynthesis of hypusine in Chinese hamster ovary cells - Evidence for eIF-4D precursor polypeptides. Journal of Biological Chemistry 262, 12730-12734.

PARK, M. H., COOPER, H. L. & FOLK, J. E. (1982). The biosynthesis of protein bound hypusine $[N^{\text{E}}-(4-amino-2-hydroxybutyl)]$ ysine] - Lysine as the amino acid precursor and the intermediate role of deoxyhypusine $[N^{\text{E}}-(4-amino-butyl)]$ ysine]. Journal of Biological Chemistry **257**, 7217-7222.

PARK, M. H., COOPER, H. L. & FOLK, J. E. (1981). Identification of hypusine, an unusual amino acid, in a protein from human lymphocytes and of spermidine as its biosynthetic precursor. Proceedings of the National Academy of Sciences of the United States of America - Biological Sciences 78, 2869-2873.

PARK, M. H. & WOLFF, E. C. (1988). Cell free synthesis of deoxyhypusine - Separation of protein substrate and enzyme and identification of 1,3-diaminopropane as a product of spermidine cleavage. Journal of Biological Chemistry 263, 15264-15269.

PARK, M. H., WOLFF, E. C. & FOLK, J. E. (1993a) Hypusine – Its posttranslational formation in eukaryotic initiation factor 5A and its potential role in cellular regulation. Biofactors 4, 95-104.

PARK, M. H., WOLFF, E. C. & FOLK, J. E. (1993b). Is hypusine essential for eukaryotic cell proliferation. Trends in Biochemical Sciences 18, 475-479.

PARK, M. H., WOLFF, E. C., LEE, Y. B. & FOLK, J. E. (1994). Antiproliferative effects of inhibitors of deoxyhypusine synthase. Journal of Biological Chemistry 269, 27827-27832.

PARK, M. H., WOLFF, E. C., SMIT-McBRIDE, Z., HERSHEY, J. W. B. & FOLK, J.
E. (1991). Comparison of the activities of variant forms of eIF-4D. Journal of Biological Chemistry 266, 7988-7994.

PARRY, D. (1990). Plant Pathology in Agriculture, Cambridge University Press, Cambridge.

PATTERSON, T. F. & ANDRIOLE, V. T. (1989). *Current concepts in cryptococcosis*. European Journal of Clinical Microbiology and Infectious Diseases **8**, 457-465.

PAULIN, L., VEHMAANPERÄ, J., NYKANEN, I. & PÖSÖ, H. (1983). *GTP* insensitive ornithine decarboxylase in acetobacteria able to synthesize spermine. Biochemical and Biophysical Research Communications 114, 779-784.

PAULUS, T. J., CRAMER, C. L. & DAVIS, R. H. (1983). Compartmentation of spermidine in <u>Neurospora crassa</u>. Journal of Biological Chemistry 258, 8608-8612.

PAULUS, T. J., KIYONO, P. & DAVIS, R. H. (1982). Polyamine-deficient <u>Neurospora</u> <u>crassa</u> mutants and synthesis of cadaverine. Journal of Bacteriology 152, 291-297.

PAZ, M. A., TORRELIO, B. M. & GALLOP, P. M. (1984). Hydralazine inhibition of the post-translational hydroxylation of deoxyhypusine, a polyamine derived amino acid. Biochemical Pharmacology 33, 779-785.

PEGG, A. E. (1988). Polyamine metabolism and its importance in neoplastic growth as a target for chemotherapy. Cancer Research 48, 759-774.

PEGG, **A. E.** (1987). The use of inhibitors to study the biochemistry and molecular biology of polyamine biosynthesis and uptake. In "Inhibition of Polyamine Metabolism: Biological Significance and Basis for New Therapies", (eds., McCann, P. P., Pegg, A. E. & Sjoerdsma, A.), Academic Press, Orlando, 107-119.

PEGG, A. E. (1986). Recent advances in the biochemistry of polyamines in eukaryotes. Biochemical Journal 234, 249-262.

PEGG, A. E. (1984). *S*-adenosylmethionine decarboxylase - A brief review. Cell Biochemistry and Function 2, 11-15.

PEGG, A. E., BITONTI, A. J., McCANN, P. P. & COWARD, J. K. (1983). Inhibition of bacterial aminopropyltransferases by S-adenosyl-1,8-diamino-3-thiooctane and by dicyclohexylamine. FEBS Letters 155, 192-196.

PEGG, A. E., COWARD, J. K., TALEKAR, R. R. & SECRIST, J. A. (1986). Effects of certain 5'-substituted adenosines on polyamine synthesis - Selective inhibitors of spermine synthese. Biochemistry 25, 4091-4097.

PEGG, A. E. & JACOBS, G. (1983). Comparison of inhibitors of S-adenosylmethionine decarboxylase from different species. Biochemical Journal 213, 495-502.

PEGG, A. E., JONES, D. B. & SECRIST, J. A. (1988a). Effect of inhibitors of Sadenosylmethionine decarboxylase on polyamine content and growth of L1210 cells. Biochemistry 27, 1408-1415.

PEGG, A. E., KAMEJI, T., SHIRAHATA, A., STANLEY, B., MADHUBALA, R. & PAJUNEN, A. (1988b). Regulation of mammalian S-adenosylmethionine decarboxylase. Advances in Enzyme Regulation 27, 43-55.

PEGG, A. E. & McCANN, P. P. (1982). *Polyamine metabolism and function*. American Journal of Physiology **234**, C212-C221.

PEGG, A. E. & McGILL, S. (1979). Decarboxylation of ornithine and lysine in rat tissues. Biochimica et Biophysica Acta 586, 416-427.

PEGG, A. E., PÖSÖ, H., SHUTTLEWORTH, K. & BENNETT, R. A. (1982a). Effect of inhibition of polyamine synthesis on the content of decarboxylated S-adenosylmethionine. Biochemical Journal 202, 519-526.

PEGG, A. E., TANG, K-C. & COWARD, J. K. (1982b). Effect of S-adenosyl-1,8diamino-3-thiooctane. Biochemistry 21, 5082-5089.

PEGG, A. E. & WILLIAMS-ASHMAN, H. G. (1987). Pharmacological interference with enzymes of polyamine biosynthesis and of 5'-methylthioadenosine metabolism. "Inhibition of Polyamine Metabolism: Biological Significance and Basis for New Therapies", (eds., McCann, P. P., Pegg, A. E. & Sjoerdsma, A.), Academic Press, Orlando, 33-48.

PERSSON, L., KHOMUTOV, A. R. & KHOMUTOV, R. M. (1989). Feedback regulation of S-adenosylmethionine decarboxylase synthesis. Biochemical Journal 257, 929-931.

PETERS, W. (1985). The problem of drug resistance in malaria. Parasitology **90**, 705-715.

PFALLER, M. A., GERARDEN, T. & RILEY, J. (1987). Growth inhibition of pathogenic yeast isolates by α -difluoromethylornithine – An inhibition of ornithine decarboxylase. Mycopathologia **98**, 3-8.

PFALLER, M. A., RILEY, J. & GERARDEN, T. (1990). Polyamine depletion and growth inhibition of <u>Cryptococcus</u> neoformans by α-difluoromethylornithine and cyclohexylamine. Mycopathologia **112**, 27-32.

PFALLER, M. A., RILEY, J. & GERARDEN, T. (1988). Polyamine depletion and growth inhibition in <u>Candida</u> <u>albicans</u> and <u>Candida</u> <u>tropicalis</u> by αdifluoromethylornithine and cyclohexylamine. Journal of Medical and Veterinary Mycology **26**, 119-126.

PFOSSER, M., MENGL, M., KONIGSHOFER, H. & KANDELER, R. (1992). Time course of polyamine levels during starvation of <u>Medicago varia</u> cells and its correlation with cell cycle data. Journal of Plant Physiology 140, 334-338.

PHILLIPS, M. A., COFFINO, P. & WANG, C. C., (1987). Cloning and sequencing of the ornithine decarboxylase gene from <u>Trypanosoma brucei</u> - Implications for enzyme turnover and selective difluoromethylornithine inhibition. Journal of Biological Chemistry 262, 8721-8727.

PHILLIPS, M. A. & WANG, C. C. (1987). A <u>Trypanosoma brucei</u> mutant resistant to αdifluoromethylornithine. Molecular and Biochemical Parasitology 22, 9-17.

PISTOCCHI, R., BAGNI, N. & CREUS, J. A. (1987). Polyamine uptake in carrot cell cultures. Plant Physiology 84, 374-380.

POLYA, G. M. & MICUCCI, V. (1985). Interaction of wheat germ Ca²⁺ dependent protein kinase with calmodulin antagonists and polyamines. Plant Physiology 79, 968-972.

PONAPPA, T., LAWSON, D. R. & MILLER, A. R. (1992). Inhibition of ornithine decarboxylase activity reduces polyamine levels and growth of <u>Agrobacterium tumefaciens</u>. Current Microbiology **2**4, 269-273.

PORTER, C. W. & BERGERON, R. J. (1983). Spermidine requirement for cell proliferation in eukaryotic cells - Structural specificity and quantitation. Science **219**, 1083-1085.

PORTER, C. W., BERGERON, R. J. & STOLOWICH, N. J. (1982). Biological properties of N^4 -spermidine derivatives and their potential in anticancer chemotherapy. Cancer Research 42, 4072-4078.

PORTER, C. W., CAVANAUGH, P. F., STOLOWICH, N., GANIS, B., KELLY, E. & BERGERON, R. J. (1985). Biological properties of N^4 - and N^1, N^8 -spermidine derivatives in cultured L1210 leukemia cells. Cancer Research 45, 2050-2057.

PORTER, C. W. & JÄNNE, J. (1987). *Modulation of antineoplastic drug action by inhibitors of polyamine biosynthesis.* In "Inhibition of Polyamine Metabolism: Biological Significance and Basis for New Therapies", (eds., McCann, P. P., Pegg, A. E. & Sjoerdsma, A.), Academic Press, Orlando, 203-248.

PORTER, C. W., MILLER, J. & BERGERON, R. J. (1984). Aliphatic chain length specificity of the polyamine transport system in ascities L1210 leukemia cells. Cancer Research 44, 126-128.

PORTER, C. W. & SUFRIN, J. R. (1986). Interference with polyamine biosynthesis and/or function by analogs of polyamines or methionine as a potential anticancer chemotherapeutic strategy. Anticancer Research 6, 525-542.

PÖSÖ, H., PAULIN, L. & BRANDER, E. (1983). Specific inhibition of spermidine synthase from mycobacteria by ethambutol. Lancet 2, 1418.

PRAKASH, N. J., BOWLIN, T. L., DAVIS, G. F., SUNKARA, P. S. & SJOERDSMA, A. (1988). Antitumour activity of norspermidine, a structural homolog of the natural polyamine spermidine. Anticancer Research **8**, 563-568.

PRAKASH, N. J., SCHECTER, P. J., GROVE, J. & KOCH-WESER, J. (1978). Effect of α-difluoromethylornithine, an enzyme activated irreversible inhibitor of ornithine decarboxylase, on L1210 leukemia in mice. Cancer Research **38**, 3059-3062.

PRIEBE, A., KLEIN, H. & JAGER, H. J. (1978). Role of polyamines on SO₂-polluted pea plants. Journal of Experimental Botany 29, 1045-1050.

RAJAM, M. V. (1991). Insecticidal activity of inhibitors of polyamine biosynthesis on <u>Spodoptera litura</u> F. larvae. Indian Journal of Experimental Biology **29**, 881-882.

RAJAM, M. V. (1988). Restriction of pollen germination and tube growth in lily pollen by inhibitors of polyamine biosynthesis metabolism. Plant Science 59, 53-56.

RAJAM, M. V. & GALSTON, A. W. (1985). The effects of some polyamine biosynthetic inhibitors on growth and morphology of phytopathogenic fungi. Plant Cell Physiology **26**, 683-692.

RAJAM, M. V., WEINSTEIN, L. H. & GALSTON, A. W. (1991). Effect of polyamine biosynthesis inhibitors on polyamine levels in bean seedlings. Current Science **60**, 178-179.

RAJAM, M. V., WEINSTEIN, L. H. & GALSTON, A. W. (1989). Inhibition of uredospore germination and germ tube growth by inhibitors of polyamine metabolism in <u>Uromyces phaseoli</u> L. Plant Cell Physiology 30, 37-41.

RAJAM, M. V., WEINSTEIN, L. H. & GALSTON, A. W. (1986). Kinetic studies on the control of the bean rust fungus (<u>Uromyces phaseoli</u> L.) by an inhibitor of polyamine biosynthesis. Plant Physiology 82, 485-487.

RAJAM, M. V., WEINSTEIN, L. II. & GALSTON, A. W. (1985). Prevention of a plant disease by specific inhibition of fungal polyamine biosynthesis. Proceedings of the National Academy of Sciences of the United States of America **82**, 6874-6878.

RAZIN, A., CEDAR, H. & RIGGS, D. (1984). DNA methylation, biochemistry and biological significance. Springer-Verlag, New York.

REGENASS, U., METT, H., STANEK, J., MUELLER, M., KRAMER, D. & PORTER, C. W. (1994). CGP 48664, a new S-adenosylmethionine decarboxylase inhibitor with broad spectrum antiproliferative and antitumor activity. Cancer Research 54, 3210-3217.

REINHOLD, L. & KAPLAN, A. (1984). *Membrane transport of sugars and amino acids.* Annual Review of Plant Physiology **35**, 45-83.

REITZ, M., WALTERS, D. R. & MOERSCHBACHER, B. (1995a). Germination and appressorial formation by uredospores of <u>Uromyces viciae-fabae</u> exposed to inhibitors of polyamine biosynthesis. European Journal of Plant Pathology 101, 573-578.

REITZ, M., WALTERS, D.R., MOERSCHBACHER, B. & ROBINS, D. J. (1995b). The effects of two synthetic putrescine analogues on germination and appressorium formation by uredospores of <u>Uromyces viciae-fabae</u> on artificial membranes. Letters in Applied Microbiology **21**, 285-287.

RELYEA, N. & RANDO, R. R. (1975). Potent inhibition of ornithine decarboxylase by βunsaturated substrate analogs. Biochemical and Biophysical Research Communications 67, 392-402.

RICHARDS, F. J. & COLEMAN, R. G. (1952). Occurrence of putrescine in potassiumdeficient barley. Nature 170, 460.

RUHL, M., HIMMELSPACH, M., BAHR, G. M., HAMMERSCHMID, F., JAKSCHE, H., WOLFF, B., ASCHAUER, H., FARRINGTON, G. K., PROBST, H., BEVEC, D. & HAUBER, J. (1993). Eukaryotic initiation factor SA is a cellular target of the Human Immunodeficiency Virus Type 1 Rev activation domain mediating trans-activation. Journal of Cell Biology **123**, 1309-1320.

RUIZ-HERRERA, J. (1994). *Polyamines, DNA methylation, and fungal differentiation*. Critical Reviews in Microbiology **20**, 143-150.

RUIZ-HERRERA, J., RUIZ, A. & LOPEZ-ROMERO, E. (1983). Isolation and biochemical analysis of <u>Mucor bacilliformis</u> monomorphic mutants. Journal of Bacteriology 156, 264-272.

RUIZ-HERRERA, J., RUIZ-MEDRANO, R. & DOMINGUEZ, A. (1995). Selective inhibition of cytosine-DNA methylases by polyamines. FEBS Letters **357**, 192-196.

SAMEJIMA, K., KAWASE, M., SAKAMOTO, S., OKADA, M. & ENDO, Y. (1976). A sensitive fluorometric method for the determination of aliphatic diamines and polyamines in biological materials by high-speed liquid chromatography. Analytical Biochemistry 76, 392-406.

SANDHOLZER, U., CENEA-INTEMANN, M., NOEGEL, A. A. & LOTTSPEICH, F. (1989). *c-DNA and derived amino-acid sequence of the hypusine containing protein from Dictyostelium discoideum*. FEBS Letters **246**, 94-100.

SANO, A., MIYAKE, M. & KAKIMOTO, Y. (1984). A rapid and sensitive method for the determination of hypusine in proteins and its distribution and developmental changes. Biochimica et Biophysica Acta 800, 135-139.

SCARBOROUGH, G. A. (1970). Sugar transport in <u>Neurospora</u> crassa. Journal of Biological Chemistry 245, 1694-1698.

SCHECTER, P. J., BARLOW, J. L. R. & SJOERDSMA, A. (1987). Clinical aspects of inhibition of ornithine decarboxylase with emphasis on therapeutic trials of effornithine (DFMO) in cancer and protozoan diseases. In "Inhibition of Polyamine Metabolism: Biological Significance and Basis for New Therapics", (eds., McCann, P. P., Pegg, A. E. & Sjoerdsma, A.), Academic Press, Orlando, 345-364.

SCHNIER, J., SCHWELBERGER, H. G., SMIT-McBRIDE, Z., KANG, H. A. & HERSHEY, J. W. B. (1991). Translation initiation factor 5A and its hypusine

modification are essential for cell viability in the yeast <u>Saccharomyces</u> <u>cerevisiae</u>. Molecular and Cellular Biology 11, 3105-3114.

SCHÜMANN, H. & KLINK, F. (1989). Archaebacterial protein contains hypusine, a unique amino acid characteristic for eukaryotic translation initiation factor 4D. Systematic and Applied Microbiology 11, 103-107.

SCHWELBERGER, H. G., KANG, H. A. & HERSHEY, J. W. B. (1993). Translation initiation factor elF-5A expressed from either of two yeast genes or from human cDNA - functional identity under aerobic and anaerobic conditions. Journal of Biological Chemistry 268, 14018-14025.

SEASTON, A., INKSON, C. & EDDY, A. A. (1973). The absorption of protons with specific amino acids and carbohydrates by yeast. Biochemical Journal 134, 1031-1043.

SECRIST, J. A. (1987). New substrate analogs as inhibitors of S-adenosylmethionine decarboxylase. Nucleosides and Nucleotides 6, 73-84.

SEILER, N. (1988). Potential roles of polyamine interconversion in the mammalian organism. In "Progress in Polyamine Research: Novel Biochemical, Pharmacological and Clinical Aspects", (eds., Zappia, V. & Pegg, A. E.), Plenum Press, New York, 127-145.

SEILER, N. & KNÖDGEN, B. (1985). Determination of polyamines and related compounds by reversed-phase high-performance liquid chromatography - Improved separation systems. Journal of Chromatography 339, 45-57.

SEPPÄNEN, P. (1981). Some properties of the polyamine deprivation inducible uptake system for methylglyoxal bis(guanylhydrazone) in tumor cells. Acta Chemica Scandinavica Series B - Organic Chemistry and Biochemistry. **35**, 731-736.

SEPPÄNEN, P., ALHONEN-HONGISTO, L. & JÄNNE, J. (1981). Polyamine deprivation induced enhanced uptake of methylglyoxal bis(guanylhydrazone) by tumor cells. Biochimica et Biophysica Acta 674, 169-177.

SERAFINI-FRACASSINI, D. N., BAGNI, N., CIONINI, P. G. & BENNICI, A. (1980). Polyamines and nucleic acids during the first cell cycle of <u>Helianthus tuberosus</u> tissue after the dormancy break. Planta 148, 332-337. SHAPIRA, R., ALTMAN, A., HENIS, Y. & CHET, I. (1989). Polyamines and ornithine decarboxylase activity during growth and differentiation in <u>Sclerotium rolfsii</u>. Journal of General Microbiology 135, 1361-1367.

SHEARD, J. & FARRAR, J. F. (1987). Transport of sugar in <u>Phytophthora palmivora</u> (Butl.) Butl. New Phytologist 105, 265-272.

SHI, X. P., YIN, K. C., AHERN, A., DAVIS, J., STERN, A. M., WAXMAN, L. (1996). Effects of N^1 -guanyl-1,7-diaminoheptane - an inhibitor of deoxyhypusine synthase, on the growth of tumorigenic cell lines in culture. Biochimica et Biophysica Acta 1310, 119-126.

SHIBA, T., AKIYAMA, H., UMEDA, I., OKADA, S. & WAKAMIYA, T. (1972). Synthesis and stereochemistry of hypusine, a new amino acid in bovine brain. Bulletin of the Chemical Society of Japan 55, 899-903.

SHIBA, T., MIZOTE, H., KANEKO, T., NAKAJIMA, T., KAKIMOTO, Y. & SANO, I. (1971). Hypusine, a new amino acid occurring in bovine brain. Isolation and structural determination. Biochimica et Biophysica Acta 244, 523-531.

SHIRAHATA, A., MOROHOSHI, T. & SAMEJIMA, K. (1988). Trans-4methylcyclohexylamine, a potent new inhibitor of spermidine synthase. Chemical and Pharmaceutical Bulletin 36, 3220-3222. SHIRAHATA, A. & PEGG, A. E. (1985). Regulation of S-adenosylmethionine decarboxylase activity in rat liver and prostate. Journal of Biological Chemistry 260, 9583-9588.

SIIMES, M., SEPPÄNEN, P., ALHONEN-HONGISTO, L. & JÄNNE, L. (1981). Synergistic action of two polyamine antimetabolites leads to a rapid therapeutic response in childhood leukemia. International Journal of Cancer 28, 567-570.

SINDHU, R. K. & COHEN, S. S. (1984). Propylamine transferases in Chinese cabbage leaves. Plant Physiology 74, 645-649.

SINGH, P. P., SINGH, S., BASRA, A. S., BEDI, P. S. (1989). Inhibition of mycelial growth and sporidial formation in the wheat pathogen <u>Neovossia indica</u> by a polyamine biosynthetic inhibitor. Phytoparasitica 17, 323-326.

SJOERDSMA, A. & SCHECHTER, P. J. (1984). Chemotherapeutic implications of polyamine biosynthesis inhibition. Clinical Pharmacology and Therapeutics 35, 287-300.

SLOCUM, R. D. & GALSTON, A. W. (1987). Inhibition of polyamine biosynthesis in plants and plant pathogenic fungi. In "Inhibition of Polyamine Metabolism: Biological Significance and Basis for New Therapies", (eds., McCann, P. P., Pegg, A. E. & Sjoerdsma, A.), Academic Press, Orlando, 305-316.

SLOCUM, R. D. & GALSTON, A. W. (1985). In vivo inhibition of polyamine biosynthesis and growth in tobacco ovary tissues. Plant and Cell Physiology 26, 1519-1526.

SMITH, T. A. (1985). Polyamines. Annual Review of Plant Physiology 36, 117-143.

SMITH, T. A. (1982). The physiology of the polyamines and related compounds. Endeavour 31, 22-28.

SMITH, T. A. (1973). Amine levels in mineral deficient <u>Hordeum vulgare</u> leaves. Phytochemistry 12, 2093-2100.

SMITH, T. A. (1971). Occurrence, metabolism and functions of amines in plants. Biological Reviews 46, 201-212.

SMITH, T. A., BARKER, J. H. A. & OWEN, W. J. (1992). Insensitivity of <u>Septoria</u> tritici and <u>Ustilago maydis</u> to inhibitors of ornithine decarboxylase. Mycological Research 96, 395-400.

SMITH, T. A., BARKER, J. H. A. & JUNG, M. (1990a). Effect of enzyme-activated inhibitors on ornithine decarboxylase and growth of <u>Botrytis cinerea</u>. Phytochemistry 29, 1759-1762.

SMITH, T. A., BARKER, J. H. A. & JUNG, M. (1990b). Growth inhibition of <u>Botrytis</u> <u>cinerea</u> by compounds interfering with polyamine metabolism. Journal of General Microbiology 136, 985-992.

SMITH, T. A. & SINCLAIR, C. (1967). *The effect of acid feeding on amine formation in barley*. Annals of Botany **31**, 103-111.

SMIT-McBRIDE, Z., SCHNIER, J., KAUFMAN, R. J. & HERSHEY, J. W. B. (1989). Protein synthesis initiation factor eIF-4D - Functional comparison of native and unhypusinated forms of the protein. Journal of Biological Chemistry 264, 18527-18530

SPATHUS, D. H., PATEMAN, J. A. & CLUTTERBUCK, A. J. (1982). Polyamine transport in <u>Aspergillus nidulans</u>. Journal of General Microbiology 128, 557-563.

SPERANZA, A. & BAGNI, N. (1977). Putrescine biosynthesis in <u>Agrobacterium</u> <u>tumefaciens</u> and in normal and crown gall tissues of <u>Scorzonera hispanica</u> L. Zeitschrift fuer Pflanzenphysiologie 81, 226-233. あたいのでは、あまいにあい、 ないろう

SRIVASTAVA, S. K. & SMITH, T. A. (1982). The effect of some oligoamines and guanidines on membrane permeability in higher plants. Phytochemistry 21, 997-1008.

STAPLEY, J. H. & GAYNER, F. C. H. (1969). World Crop Protection - Vol 1 - Pests and Diseases, Iliffe Books Ltd., London.

STEVENS, L. (1970). Biochemical role of naturally occurring polyamines in nucelic acid synthesis. Biological Reviews 45, 1-27.

STEVENS, L., McKINNON, I. M. & WINTHER, M. (1977). Effects of 1,4diaminobutanone on polyamine synthesis in <u>Aspergillus nidulans</u>. FEBS Letters 75, 180-182.

STEVENS, L., McKINNON, I. M. & WINTHER, M. (1976). Polyamine and ornithine metabolism during the germination of conidia of <u>Aspergillus nidulans</u>. Biochemical Journal **158**, 235-241.

STEVENS, L. & WINTHER, M. D. (1979). Spermine, spermidine and putrescine in fungal development. Advances in Microbial Physiology 19, 63-148.

SUNKARA, P. S., BAYLIN, S. B. & LUK, G. D. (1987). Inhibitors of polyamine biosynthesis - Cellular and in vivo effects on tumor proliferation. In "Inhibition of Polyamine Metabolism: Biological Significance and Basis for New Therapies", (eds., McCann, P. P., Pegg, A. E. & Sjoerdsma, A.), Academic Press, Orlando, 121-140.

SUNKARA, P. S., FOWLER, S. K., NISHIOKA, K. & RAO, P. N. (1980). Inhibition of polyamine biosynthesis by α -difluoromethylornithine potentiates the cylotoxic effects of

arabinosylcytosine in HeLa cells. Biochemical and Biophysical Research Communications 95, 423-430.

SUNKARA, P. S., PRAKASH, N. J., CHANG, C. C. & SJOERDSMA, A. (1983a). Cyto-toxicity of methylglyoxal bis(guanylhydrazone) in combination with α difluoromethylornithine against HeLa cells and mouse L1210 leukemia. Journal of the National Cancer Institute 70, 505-509.

SUNKARA, P. S., PRAKASH, N. J., MAYER, G. D. & SJOERDSMA, A. (1983b). Tumor suppression with a combination of α -difluoromethylornithine and interferon. Science 219, 851-853.

SUNKARA, P. S., ZWOLSHEN, J. H., PRAKASH, N. J. & BOWLIN, T. J. (1988). Mechanism of antitumour activity of norspermidine, a structural homologue of spermidine. In "Progress in Polyamine Research: Novel Biochemical, Pharmacological and Clinical Aspects", (eds., Zappia, V. & Pegg, A. E.), Plenum Press, New York, 707-716.

TABOR, C. W. & TABOR, II. (1985). Polyamines in microorganisms. Microbiological Reviews 49, 81-99.

TABOR, C. W. & TABOR, H. (1984). *Polyamines*. Annual Review Biochemistry 53, 749-790.

TABOR, H. & TABOR, C. W. (1964). Spermidine, spermine and related amines. Pharmacological Reviews 16, 245-300.

TAELMAN, H., SCHECHTER, P. J., MARCELIS, L., SONNET, J., KAZYUMBA, G., DASNOY, J., HAEGELE, K. D., SJOERDSMA, A. & WERY, M. (1987). Difluoromethylornithine, an effective new treatment of Gambian trypanosomiasis. The American Journal of Medicine 82, 607-614.

TALPAZ, M., PLAGER, C., QUESADA, J., BENJAMIN, R., KANTARJIAN, H. & GUTTERMAN, J. (1986). Difluoromethylornithine and leukocyte interferon - a phase I study in cancer patients. European Journal of Cancer and Clinical Oncology 22, 685-689.

TIBURCIO, A. F., KAUR-SAWHNEY, R. & GALSTON, A. W. (1987). Effect of polyamine biosynthetic inhibitors on alkaloids and organogenesis in tobacco callus cultures. Plant Cell Tissue and Organ Culture 9, 111-120.

TOMS, A. M. & DAHL, M. H. (1976). Pests and Diseases of Fruit and Vegetables in the Garden, Blandford Press, Dorset.

TORRELIO, B. M., PAZ, M. A. & GALLOP, P. M. (1984). Cellular proliferation and hypusine synthesis. Experimental Cell Research 154, 454-463.

TORRIGIANI, P., SERAFINI-FRACASSINI, D. & BAGNI, N. (1987). Polyamine biosynthesis and effect of dicyclohexylamine during the cell cycle of <u>Helianthus tuberosus</u> tuber. Plant Physiology **84**, 148-152.

TRIONE, E. J., STOCKWELL, V. O. & AUSTIN, H. A. (1988). The effects of polyamines on the growth and development of the wheat bunt fungi. Botanical Gazette 149, 173-178.

TURNER, L. B. & STEWART, G. R. (1986). The effect of water stress upon polyamine levels in barley (*Hordeum yulgare L.*) leaves. Journal of Experimental Botany **37**, 170-177.

UPFOLD, S. J. & VAN STADEN, J. (1991). Polyamines and carnation flower senescence - Endogenous levels and the effect of applied polyamines on senescence. Plant Growth Regulation 10, 355-362.

VELUTHAMBI, K. & POOVAIAH, B. W. (1984). Polyamine stimulated phosphorylation of proteins from corn (<u>Zea mays</u> L.) coleoptiles. Biochemical and Biophysical Research Communications 122, 1374-1380.

WALE, S. J., OXLEY, S. J. P. & BAIN, R. A. (1993). Effective spring barley disease control using low dose fungicide mixtures. Proceedings of the Crop Protection in Northern Britain Conference, 127-132.

WALTERS, D. R. (1997). The putrescine analogue (E)-1,4-diaminobut-2-ene reduces DNA methylation in the plant pathogenic fungus <u>Pyrenophora avenae</u>. FEMS Microbiology Letters 154, 215-218.

WALTERS, D. R. (1995). Inhibition of polyamine biosynthesis in fungi. Mycological Research 99, 129-139.

WALTERS, D. R. (1986). The effects of a polyamine biosynthesis inhibitor on infection of *Vicia faba L. by the rust fungus*, *Uromyces viciae-fabae (Pers.) Schroet*. New Phytologist 104, 613-619.

WALTERS, D. R., COWLEY, T. & McPHERSON, A. (1995a). Effects of the trypanocidal agents Berenil and pentamidine on growth, enzyme activites, and polyamine concentrations in the rice blast pathogen <u>Pyricularia oryzae</u> and on powdery mildew infection of barley seedling. Pesticide Biochemistry and Physiology 53, 147-151.

WALTERS, D. R., COWLEY, T., McPHERSON, A., MARSHALL, G. & McROBERTS, N. (1996). Sugar transport in the light leaf spot pathogen <u>Pyrenopeziza</u> brassicae. FEMS Microbiology Letters 143, 285-289.

WALTERS, D. R., KEENAN, J. P., COWLEY, T., MCPHERSON, A. & HAVIS, N. D. (1995b). Inhibition of polyamine biosynthesis in <u>Phytophthora infestans</u> and <u>Pythium</u> <u>ultimum</u>. Plant Pathology 44, 80-85.

WALTERS, D. R. & KINGHAM, G. (1990). Uptake and translocation of α difluoromethylornithine, a polyamine biosynthesis inhibitor, by barley seedlings - effects on mildew infection. New Phytologist 114, 659-665.

WALTERS, D.R. & McPHERSON, A. (1997). Effect of the synthetic putrescine analogue (E)-1,4-diaminobut-2-ene on uptake of methionine, polyamines and sugars in *Pyrenophora avenae*. Pesticide Biochemistry and Physiology 57, 63-67.

WALTERS, D. R., McPHERSON, A., MARSHALL, G. & McROBERTS, N. (1997). Transport of glutamate and glutamine in the light leaf spot pathogen <u>Pyrenopeziza</u> <u>brassicae</u>. Journal of Phytopathology 145, 393-395.

WALTERS, D. R., WILSON, P. W. F. & SHUTTLETON, M. A. (1985). Relative changes in levels of polyamines and activities of their biosynthetic enzymes in barley infected with the powdery mildew fungus, <u>Erysiphe graminis</u> D. C. ex Merat f. sp. <u>hordei</u> Marchal. New Phytologist 101, 695-703.

WALTERS, D. R. & WYLIE, M. A. (1986). Polyamines in discrete regions of barley leaves infected with the powdery mildew fungus, <u>Erysiphe graminis</u>. Physiologia Plantarum 67, 630-633.

WATSON, P. A., HANAUSKE-ABEL, H. H., FLINT, A. & LALANDE, M. (1991). Mimosine reversibly arrests cell cycle progression at the G1-S phase border. Cytometry 12, 242-246.

WEINSTEIN, L. H., OSMELOSKI, J. F., WETTLAUFER, S. H. & GALSTON, A. W. (1987). Protection of wheat against leaf and stem rust and powdery mildew diseases by inhibition of polyamine metabolism. Plant Science 51, 311-316.

WEST, H. M. & WALTERS, D. R. (1991a). Uptake of the polyamine biosynthesis inhibitor α-difluoromethylornithine by the plant pathogenic fungus <u>Fusarium culmorum</u>. Mycological Research 95, 621-623.

WEST, H. M. & WALTERS, D. R. (1991b). Polyamine uptake by the plant pathogenic fungus, *Fusarium culmorum*. Mycological Research 95, 715-719.

WEST, H. M. & WALTERS, D. R. (1989). Effects of polyamine biosynthesis inhibitors on growth of <u>Pyrenophora teres</u>, <u>Gaeumannomyces graminis</u>, <u>Fusarium culmorum</u> and <u>Septoria nodorum in vitro</u>. Mycological Research 92, 453-457.

WEST, H. M. & WALTERS, D. R. (1988). The effects of polyamine biosynthesis inhibitors on infection of <u>Hordeum vulgare</u> L. by <u>Erysiphe graminis</u> f. sp. <u>hordei</u> Marchal. New Phytologist **110**, 193-200.

WHITAKER, A. (1976). Amino acid transport into fungi - An essay. Transactions of the British Mycological Society 67, 365-376.

WILLIAMS-ASHMAN, H. G. & PEGG, A. E. (1981). Aminopropyl group transfers in polyamine biosynthesis. In "Polyamines in Biology and Medicine", (cds., Morris, D. R. & Marton, L. J.), Marcel Dekker, Inc., New York, 43-73.

WOLFF, E. C., JAKUS, J., PARK, M. H. & FOLK, J. E. (1992). Inhibitors of deoxyhypusine synthase as potential regulators of hypusine biosynthesis in eIF-5A. FASEB Journal 6, A39.

YAKUBA, M. A., BASSO, B., KIERSZENBAUM, F. (1992). DL-αdifluoromethylarginine inhibits intracellular <u>Trypanosoma cruzi</u> multiplication by affecting cell division but not tryomastigote-amastigote transformation. Journal of Parasitology 78, 414-419. **YAMANOHA**, B & COHEN, S. S. (1985). S-adenosylmethionine decarboxylase and spermidine synthase from Chinese cabbage. Plant Physiology 78, 784-790.

YAMANOHA, B., SAMEJIMA, K., NAKAJIMA, T. & YASUHARA, T. (1984). Differences between homogeneous spermidine synthases isolated from rat and pig liver. Journal of Biochemistry 96, 1273-1281.

YANG, Y. C., CHEN, K. Y., SEYFZADEH, M. & DAVIS, R. H. (1990). Deoxyhypusine hypusine formation on a 21000-dalton cellular protein in a <u>Neurospora crassa</u> mutant <u>in</u> vivo and <u>in vitro</u>. Biochimica et Biophysica Acta 1033, 133-138.

YARLETT, N. & BACCHI, C. J. (1994). Parasitic polyamine metabolism - targets for chemotherapy. Biochemical Society Transactions **22**, 875-879.

ZARB, J. (1995). *Polyamine metabolism in mycorrhizal fungi*. Ph.D. Thesis, University of Glasgow.

ZARB, J. & WALTERS, D. R. (1994a). The effect of the polyamine biosynthesis inhibitor DFMO on the ectomycorrhizal fungus <u>Paxillus involutus</u>. Letters in Applied Microbiology 18, 5-7.

ZARB, J. & WALTERS, D. R. (1994b). The formation of cadaverine, aminopropylcadaverine and N,N bis(3-aminopropyl)cadaverine in mycorrhizal and phytopathogenic fungi. Letters in Applied Microbiology 19, 277-280.

ZARB, J. & WALTERS, D. R. (1994c). The effects of polyamine biosynthesis inhibitors on growth, enzyme activities and polyamine concentrations in the mycorrhizal fungus Laccaria proxima. New Phytologist 126, 99-104.

ZARB, J. & WALTERS, D. R. (1993). The effect of ornithine decarboxylase inhibition on growth, enzyme activities, and polyamine concentrations in <u>Crinipellis perniciosa</u>. Pesticide Biochemistry and Physiology 47, 44-50.

Appendices

ć

11. APPENDICES

「「「「「「「「」」」」

A STATES

Treatment	Mean percentage of leaf area affected by disease	
Control	11.5 ± 1.16	
0.5 mM pre CHA	$4.5 \pm 0.80 ***$	
1.0 mM pre CHA	3.3 ± 0.77 ***	
2.0 mM pre CHA	$4.8 \pm 0.70 $ ***	
0.5 mM pre norspermidine	4.5 ± 1.20 ***	
1.0 mM pre norspermidine	5.5±1.14 ***	
2.0 mM pre norspermidine	4.1 ± 0.44 ***	
0.5 mM post CHA	7.5 ± 1.25 *	
1.0 mM post CHA	8.1 ± 1.54 ns	
2.0 mM post CHA	6.7 ± 1.64 *	
0.5 mM post norspermidine	4.7 ± 1.06 ***	
1.0 mM post norspermidine	6.1 ± 0.76 ***	
2.0 mM post norspermidine	4.9 ± 0.66 ***	

Appendix 1. Effect of CHA and norspermidine against rust on broad bean

Note: Significant differences from control are shown as: * $p \le 0.05$; ** $p \le 0.01$; *** $p \le 0.001$

ns no significant difference

Treatment	Mean percentage of leaf area affected by disease	
Control	22.9 ± 1.49	
0.5 mM pre CHA	17.2 ± 2.37 ns	
1.0 mM pre CHA	13.7 ± 1.56 ***	
2.0 mM pre CHA	15.4 ± 0.95 ***	
0.5 mM pre norspermidine	14.8 ± 2.27 **	
1.0 mM pre norspermidine	16.1 ± 0.93 ***	
2.0 mM pre norspermidine	$16.6 \pm 1.71 **$	
0.5 mM post CHA	20.4 ± 2.06 ns	
1.0 mM post CHA	21.4 ± 1.85 ns	
2.0 mM post CHA	16.9 ± 0.77 ***	
0.5 mM post norspermidine	21.8 ± 1.24 ns	
1.0 mM post norspermidine	18.9 ± 1.61 ns	
2.0 mM post norspermidine	18.6 ± 1.33 *	

Appendix 2. Effect of CHA and norspermidine against chocolate spot on broad bean

l

Note: Significant differences from control are shown as: * $p \le 0.05$; ** $p \le 0.01$; *** $p \le 0.001$

ns no significant difference

Treatment	Mean percentage of leaf area affected by disease	
Control	58 ± 10.2	
0.5 mM pre CHA	11 ± 1.9 *	
1.0 mM pre CHA	6±1,9 **	
2.0 mM pre CHA	7±0.8 **	
0.5 mM pre norspermidine	8 ± 2.0 **	
1.0 mM pre norspermidine	9 ± 3.3 **	
2.0 mM pre norspermidine	7 ± 2.6 **	
0.5 mM post CHA	22 ± 9.4 *	
1.0 mM post CHA	15 ± 2.2 *	
2.0 mM post CHA	9±3.1 **	
0.5 mM post norspermidine	13 ± 3.8 **	
1.0 mM post norspermidine	12 ± 4.5 **	
2.0 mM post norspermidine	7±1.4 **	

Appendix 3. Effect of CHA and norspermidine against powdery mildew on barley

a 812 - 1 AM

and the second

Note: Significant differences from control are shown as: * $p \le 0.05$; ** $p \le 0.01$

Treatment	Mean radial mycelial growth (mm) Days of growth			
	3 days	6 days	8 days	
Control	12.0 ± 0.26	28.7 ± 0.42	36.2 ± 0.79	
0.05 mM CHA	12.2 ± 0.30	29.5 ± 0.44	37.5 ± 0.31	
0.1 mM CHA	12.0 ± 0.59	$\textbf{28.8} \pm \textbf{0.32}$	37.9 ± 0.26	
0.5 mM CHA	11.8 ± 0.64	26.4 ± 0.99	37.0 ± 0.47	
1 mM CHA	12.2 ± 0.59	26.1 ± 0.93 *	35.4 ± 1.00	
2 mM CHA	3.7 ± 0.22 ***	12.2 ± 1.49 ***	16.2 ± 2.04 ***	
0.05 mM norspermidine	12.6 ± 0.31	$\textbf{28.6} \pm \textbf{0.47}$	37.6 ± 0.36	
0.1 mM norspermidine	$\textbf{2.8} \pm \textbf{0.31}$	28.3 ± 0.62	38.0 ± 0.36	
0.5 mM norspermidine	10.2 ± 0.45 **	25.3 ± 0.68 ***	3.8 ± 0.68 *	
1 mM norspermidine	7.5 ± 0.68 ***	15.8 ± 1.59 ***	21.6 ± 2.39 ***	
2 mM norspermidine	4.4 ± 0.15 ***	9.8 ± 0.27 ***	13.3 ± 0.70 ***	

家庭に開業

Appendix 4. Effect of varying concentrations of CHA and norspermidine on mycelial growth of *Pyrenophora avenae* on solid media

Note: Significant differences from control are shown as: * $p \le 0.05$; ** $p \le 0.01$; *** $p \le 0.001$

Treatment	Mean radial mycelial growth (mm) Days of growth			
—	7 days	14 days	21 days	
Control	12.7 ± 0.14	27.2 ± 0.37	35.2 ± 0.63	
1 mM CHA	11.8 ± 0.11 ***	25.9 ± 0.19 **	35.1 ± 0.40	
2 mM CHA	10.7 ± 0.14 ***	22.8 ± 0.30 ***	30.9 ± 0.47 ***	
3 mM CHA	7.8±0.34 ***	18.4 ± 0.58 ***	27.4 ± 0.64 ***	
4 mM CHA	7.2 ± 0.18 ***	19.4 ± 0.15 ***	28.9 ± 0.36 ***	
5 mM CHA	3.6 ± 0.23 ***	14.4 ± 0.36 ***	22.6 ± 0.66 ***	
1 mM norspermidine	10.2 ± 0.11 ***	23.8 ± 0.18 ***	33.2 ± 0.37 *	
2 mM norspermidine	8.0 ± 0.21 ***	$20.4 \pm 0.26 ***$	29.1 ± 0.42 ***	
3 mM norspermidine	4.6 ± 0.15 ***	$15.8 \pm 0.24^{***}$	$22.9 \pm 0.60 ***$	
4 mM norspermidine	3.2 ± 0.27 ***	14.1 ± 0.95 ***	23.9 ± 1.12 ***	
5 mM norspermidine	0.6 ± 0.15 ***	8.8 ± 1,10 ***	17.4 ± 1.39 ***	

Appendix 5. Effect of varying concentrations of CHA and norspermidine on mycelial growth of *Pyricularia oryzae* on solid media

Note: Significant differences from control are shown as: * $p \le 0.05$; ** $p \le 0.01$; *** $p \le 0.001$

Regression equation is: Fluorescence = 9.75 + 1.35 (Concentration), p < 0.001



Regression equation is: Fluorescence = 9.11 + 0.55 (Concentration), p < 0.001


Treatment	Mean radial mycelial growth (mm) Days of growth			
	3 days	6 days	8 days	
Control	11.8 ± 0.68	30.9 ± 0.58	37,2 ± 0.34	
0.05 mM C59	10.9 ± 0.31	29.6 ± 0.96	35.4 ± 0.75	
0.1 mM C59	10.8 ± 0.32	30.7 ± 0.54	36.9 ± 0.15	
0.5 mM C59	12.3 ± 0.38	30.8 ± 0.37	36.9 ± 0.19	
1.0 mM C59	12.4 ± 0.23	31.6 ± 0.26	$\textbf{36.8} \pm \textbf{0.18}$	
2.0 mM C59	12.6 ± 0.19	30.8 ± 0.21	36.8 ± 0.18	
0.05 mM C73	11.8 ± 0.24	30.8 ± 0.39	36.7 ± 0.22	
0.1 mM C73	11.5 ± 0.26	31.0 ± 0.52	35.8±0.18 **	
0.5 mM C73	11.8 ± 0.18	30.9 ± 0.23	37.4 ± 0.19	
1.0 mM C73	10.9 ± 0.31	29.2 ± 0.66	34.8 ± 0.27 ***	
2.0 mM C73	9.9 ± 0.42 *	28.3 ± 0.80 *	34.1 ± 0.57 ***	
0.05 mM LAS 2/11	12.4 ± 0.19	30.8 ± 0.59	35.9±0.31 ***	
0.1 mM LAS 2/11	11.0 ± 0.30	30.8 ± 0.47	36.5 ± 0.29	
0.5 mM LAS 2/11	11.2 ± 0.60	29.5 ± 0.93	35.4 ± 0.48 **	
1.0 mM LAS 2/11	10.4 ± 0.38	28.7 ± 0.29 **	36.0 ± 0.24 *	
2.0 mM LAS 2/11	10.1 ± 0.54	28.6 ± 0.38 **	33.9 ± 0.31 ***	
0.05 mM LAS 2/12	11.5 ± 0.34	30.1 ± 0.38	36.0 ± 0.21 *	
0.1 mM LAS 2/12	11.2 ± 0.22	30.1 ± 0.40	37.1 ± 0.23	
0.5 mM LAS 2/12	9.8 ± 0.41 *	29.8 ± 0.34	36.8 ± 0.17	
1.0 mM LAS 2/12	10.1 ± 0.15 *	28.6±0.31 **	34.8 ± 0.58 **	
2.0 mM LAS 2/12	5.3 ± 0.14 ***	13.9 ± 0.60 ***	19.6 ± 1.46 ***	

Appendix 8a. Effect of varying concentrations of novel spermidine analogues of mycelial growth of *Pyrenophora avenae* on solid media.

Note: Significant differences from control are shown as: * $p \le 0.05$; ** $p \le 0.01$; *** $p \le 0.001$

Treatment	Mean radial mycelial growth (mm) Days of growth			
	Control	11.7 ± 0.38	31.0 ± 0.78	38.4±0.51
0.05 mM LAS 2/26	12.2 ± 0.28	32.0 ± 0.35	$\textbf{37.8} \pm \textbf{0.35}$	
0.1 mM LAS 2/26	9.9 ± 0.38 **	28.2 ± 0.98 *	$\textbf{37.7} \pm \textbf{0.26}$	
0.5 mM LAS 2/26	9.2 ± 0.60 **	25.9 ± 1.12 **	36.2 ± 0.22 **	
1.0 mM LAS 2/26	10.7 ± 0.36	29.0 ± 1.08	37.1 ± 0.26 *	
2.0 mM LAS 2/26	10.7 ± 0.37	28.8 ± 0.22 *	37.6 ± 0.29	
0.05 mM LAS 2/27	12.2 ± 0.28	31.6 ± 0.74	39.1 ± 0.34	
0.1 mM LAS 2/27	8.7 ± 0.70 **	26.0 ± 1.22 **	34.9 ± 0.69 ***	
0.5 mM LAS 2/27	11.9 ± 0.31	31.5 ± 0.31	$\textbf{37.9} \pm \textbf{0.15}$	
1.0 mM LAS 2/27	9.2 ± 0.85 *	27.7 ± 1.37	36.9 ± 0.82	
2.0 mM LAS 2/27	10.6 ± 0.31 *	30.9 ± 0.36	38.0 ± 0.21	
0.05 mM LAS 2/28	10.7 ± 0.56 *	30.2 ± 0.25	37.6 ± 0.23	
0.1 mM LAS 2/28	12.2 ± 0.15	31.7 ± 0.41	38.4 ± 0.18	
0.5 mM LAS 2/28	11.0 ± 0.37	30.0 ± 0.35	38.2 ± 0.35	
1.0 mM LAS 2/28	10.8 ± 0.22	30.0 ± 0.29	$\textbf{37.3} \pm \textbf{0.17}$	
2.0 mM LAS 2/28	5.6±0.38 ***	17.1 ± 1.20 ***	25.2 ± 1.73 ***	
0.05 mM LAS 3/8	10.8 ± 0.28	31.8 ± 0.32	38.5 ± 0.61	
0.1 mM LAS 3/8	11.7 ± 0.33	31.8 ± 0.43	38.7 ± 0.41	
0.5 mM LAS 3/8	11.6 ± 0.38	31.5 ± 0.44	39.2 ± 0.25	
1.0 mM LAS 3/8	11.2 ± 0.21	30.2 ± 0.24	38.8 ± 0.39	
2.0 mM LAS 3/8	9.6 ± 0.40 ***	27.8 ± 0.74 **	36.8 ± 0.59	

Appendix 8b. Effect of varying concentrations of novel spermidine analogues of mycelial growth of *Pyrenophora avenae* on solid media.

「日本になる」の言語を読みた

Contraction of the local sector

Ŀ

Note: Significant differences from control are shown as: * $p \le 0.05$; ** $p \le 0.01$; *** $p \le 0.001$



266